CWI Tracts Managing Editors

A.M.H. Gerards (CWI, Amsterdam) M. Hazewinkel (CWI, Amsterdam) J.W. Klop (CWI, Amsterdam) N.M. Temme (CWI, Amsterdam)

Executive Editor

M. Bakker (CWI Amsterdam, e-mail: Miente.Bakker@cwi.nl)

Editorial Board

W. Albers (Enschede)
K.R. Apt (Amsterdam)
M.S. Keane (Amsterdam)
P.W.H. Lemmens (Utrecht)
J.K. Lenstra (Eindhoven)
M. van der Put (Groningen)
A.J. van der Schaft (Enschede)
J.M. Schumacher (Tilburg)
H.J. Sips (Delft, Amsterdam)
M.N. Spijker (Leiden)
H.C. Tijms (Amsterdam)

CWI

P.O. Box 94079, 1090 GB Amsterdam, The Netherlands Telephone + 31 - 20 592 9333 Telefax + 31 - 20 592 4199 WWW page http://www.cwi.nl/publications_bibl/

CWI is the nationally funded Dutch institute for research in Mathematics and Computer Science.

Multivariable orthogonal polynomials and quantum Grassmanniams

- ·

J.V. Stokman

.

2000 Mathematics Subject Classification:

- 33D52 Basic orthogonal polynomials and functions associated with root systems (Macdonald polynomials, etc.)
- 33D80 Connections with quantum groups, Chevalley groups, p-adic groups, Hecke algebras, and related topics

ISBN 90 6196 506 3

NUGI-code: 811

Copyright ©2001, Stichting Centrum voor Wiskunde en Informatica, Amsterdam Printed in the Netherlands

Contents

.

Editorial Preface Foreword	vii viii
Chapter 1. General introduction	1
1.1. Introduction	1
1.2. q -Analogues of Euler's beta integral	1
1.3. Plücker coordinates on the Grassmann manifold	4
1.4. Overview of the remaining chapters	9
1.5. Notations and conventions	15
Chapter 2. A residue calculus for Koornwinder polynomials	17
2.1. Introduction	17
2.2. One variable Askey-Wilson and q -Racah polynomials	18
2.3. Koornwinder polynomials for generic parameters values	21
2.4. Residue calculus for the orthogonality measure $d\nu$	29
2.5. Multivariable q -Racah polynomials	35
2.6. Koornwinder polynomials with positive orthogonality measure	40
Chapter 3. Limit transitions for multivariable orthogonal polynomials	47
3.1. Introduction	47
3.2. One variable big and little q -Jacobi polynomials	48
3.3. Limit transitions to little q-Jacobi polynomials	52
3.4. Limit transitions to big q -Jacobi polynomials	58
3.5. The special case $t = q^k$ with $k \in \mathbb{N}$	64
3.6. Limit transitions: an algebraic approach	67
3.7. Limit of the orthogonality measure (little q-Jacobi case)	76
3.8. Limit of the orthogonality measure (big q -Jacobi case)	85
Chapter 4. Harmonic analysis on quantum Grassmannians	95
4.1. Introduction	95
4.2. The classical complex Grassmannian	96
4.3. Preliminaries on the quantum unitary group	100
4.4. Spherical corepresentations	108
4.5. A one parameter family of quantum Grassmannians	112

4.6. Zonal (σ, τ) -spherical functions	122
4.7. Limit transitions on quantum Grassmannians	125
Chapter 5. Quantum Plücker coordinates	133
5.1. Introduction	133
5.2. The contraction map	135
5.3. Algebraic generators of $A_q(U/K)$	139
5.4. Algebras generated by quantum Plücker coordinates	142
5.5. Algebraic properties of anti-holomorphic Plücker coordinates	145
5.6. The algebraic structure of $A_q(U/K_0)$ and $A_q(U/K)$	148
5.7. Some remarks on the subalgebra of $bi-A_q(K)$ -fixed elements	152
Chapter 6. Quantized flag manifolds and irreducible *-representations	155
6.1. Introduction	155
6.2. Bruhat-Poisson brackets on flag manifolds	158
6.3. Preliminaries on the quantized function algebra $\mathbb{C}_q[U]$	162
6.4. Quantized function algebras on generalized flag manifolds	167
6.5. Restriction of irreducible *-representations to $\mathbb{C}_{q}[U/K]$	173
6.6. Irreducible *-representations of A_S	178
Bibliography	185

EDITORIAL PREFACE

Editorial Preface

This CWI Tract contains the PhD dissertation of Dr. Jasper V. Stokman, as it has been defended at the University of Amsterdam on June 11, 1998. The research of this thesis has been performed within the framework of the Dutch Research School 'Thomas Stieltjes Institute for Mathematics'. The Stieltjes Institute has awarded Dr. Stokman's thesis with a prize, as being the best 1998 thesis written in this Research School. We congratulate Dr. Stokman with this prize, and appreciate that we have obtained his permission to include his thesis into the CWI Tract Series.

The editors of CWI Tracts

vii

Foreword

The intimate relation between representation theory and the theory of special functions is a continuing source of new and beautiful results in both fields of mathematics. An example of the interaction between representation theory and special functions is the development of a general theory on hypergeometric functions associated with root systems by Heckman and Opdam, which was motivated by their interpretation (for very special parameter values) as zonal spherical functions on Riemannian symmetric spaces. Another more recent example is the representation theoretic interpretation of Macdonald polynomials associated with root systems. The Macdonald polynomials were related to representation theory of affine Hecke algebras by Cherednik and to harmonic analysis on certain quantizations of homogeneous spaces by Noumi and Sugitani. The last example is an illustration of the remarkable fact that the "q" in q-special function theory is essentially the same q as in quantum groups, if the "right" q-deformations are chosen.

One of the interesting new aspects of the harmonic analysis on quantized homogeneous spaces is the implicit role played by the Poisson structure on the underlying spaces. The Poisson structure is built in the quantization, so inequivalent Poisson structures give rise to different quantizations. Hence the harmonic analysis on quantizations of homogeneous spaces depends on the choice of Poisson structures on the underlying spaces. A nice illustration of this phenomenon is the interpretation of several families of orthogonal polynomials on different quantizations of the 2-sphere by Koornwinder, Mimachi, Noumi and others.

Poisson structures also play an important role in the representation theory of the quantized homogeneous spaces themselves. The origin of this observation lies in the orbit method of Kostant, Kirillov and Souriau in which irreducible unitary representations of Lie groups are related to coadjoint orbits of the corresponding Lie algebras. Coadjoint orbits are symplectic submanifolds for the Kostant-Kirillov Poisson bracket. The quantum orbit method deals with the problem of relating representation theory of arbitrary quantized Poisson algebras to the Poisson geometry of the underlying spaces. A beautiful example of the quantum orbit method is Soibelman's classification of the irreducible *-representations of the quantized function algebras on compact simple Lie groups. Many properties of the irreducible *-representations were shown to be closely related to geometric properties of the underlying Poisson-Lie groups.

In the present CWI Tract the above mentioned ideas are developed further in several different directions. It is entirely based on my dissertation which I have completed in June, 1998 at the KdV institute for Mathematics, University of Amsterdam under the supervision of Prof. Tom H. Koornwinder. The main text of my dissertation is reproduced here without major changes; I have corrected some misprints, and updated the references.

Acknowledgments: I thank my former thesis-advisor Tom H. Koornwinder for his advise and guidance during the four years that I have worked on my dissertation. I am financially supported by a fellowship from the Royal Netherlands Academy of Arts and Sciences (KNAW).

viii

CHAPTER 1

General introduction

1.1. Introduction

The present Chapter contains a general introduction to the different topics of the Tract. I aim to clarify some of the main ideas and techniques by considering certain simplified examples in detail.

In Chapter 2 and 3 of the Tract certain families of multivariable orthogonal polynomials are studied. A main tool in the analysis is the development of a residue calculus for a specific multidimensional contour integral. In Section 1.2 we illustrate some of these techniques for certain q-analogues of Euler's beta integral.

The remaining chapters of the Tract rely on representation theoretic methods, which are mostly of algebraic nature. In Chapter 5 and Chapter 6 intensive use is made of generalizations of Plücker coordinates. As an illustration, I give in Section 1.3 the construction of Plücker coordinates on complex Grassmannians. I only use here some well-known notions from algebraic geometry and algebraic groups, which can for instance be found in [112].

In Section 1.4 I give a detailed description of the contents of the Tract while refering to the simplified examples given in Section 1.2 and Section 1.3. The notations and conventions which are used throughout the Tract, are listed in Section 1.5.

1.2. q-Analogues of Euler's beta integral

A well-known identity in special function theory is Euler's beta integral,

(1.2.1)
$$\int_0^1 z^{\alpha} (1-z)^{\beta} dz = \frac{\Gamma(\alpha+1)\Gamma(\beta+1)}{\Gamma(\alpha+\beta+2)}, \quad (\operatorname{Re}(\alpha), \operatorname{Re}(\beta) > -1)$$

where $\Gamma(z)$ is the Gamma-function

$$\Gamma(z) := \int_0^\infty t^{z-1} e^{-t} dt, \quad (\operatorname{Re}(z) > 0).$$

In this section a discrete and a continuous q-deformation of (1.2.1) is considered, where q is a fixed real number in the open interval (0, 1). An identity (depending on q) is called a q-deformation or a q-analogue of Euler's beta integral when Euler's beta integral (1.2.1) can be recovered by taking the limit $q \uparrow 1$ in a suitable manner. The q-analogue is called discrete (respectively continuous) if it is an evaluation of a possibly infinite sum (respectively contour integral).

1. GENERAL INTRODUCTION

The theory of q-special functions has its origin in the work of Euler, Gauss, Jacobi and in particular Heine, who derived several fundamental properties of a q-analogue of Gauss's hypergeometric series. Nowadays q-deformations have been found for many identities in classical special function theory. In particular, several q-analogues of Euler's beta integral (1.2.1) have been found. The discrete q-analogue of Euler's beta integral which are considered in this section can be most conveniently expressed in terms of Jackson's q-integral,

(1.2.2)
$$\int_{c}^{d} f(z)d_{q}z := \int_{0}^{d} f(z)d_{q}z - \int_{0}^{c} f(z)d_{q}z,$$
$$\int_{0}^{c} f(z)d_{q}z := (1-q)\sum_{i=0}^{\infty} f(cq^{i})cq^{i},$$

which are considered here for functions f such that both sums are absolutely convergent. Observe that the q-integral of a continuous function f over the interval [c, d] tends to the usual Riemann integral of f over [c, d] when $q \uparrow 1$. The discrete q-beta integral is given by

(1.2.3)
$$\int_0^1 \frac{(qz;q)_{\infty}}{(q^{\beta+1}z;q)_{\infty}} z^{\alpha} d_q z = \frac{\Gamma_q(\alpha+1)\Gamma_q(\beta+1)}{\Gamma_q(2+\alpha+\beta)}$$

 $(\alpha,\beta>-1),$ where $\left(a;q\right)_k$ $(k\in\mathbb{Z}_+\cup\{\infty\})$ is the q -shifted factorial,

$$\left(a;q\right)_{k}:=\prod_{i=0}^{k-1}(1-aq^{i}),\quad\left(a;q
ight)_{\infty}:=\lim_{k\to\infty}\left(a;q
ight)_{k}$$

and $\Gamma_q(z)$ is the q-Gamma function

(1.2.4)
$$\Gamma_q(z) := \frac{(q;q)_{\infty}}{(q^z;q)_{\infty}} (1-q)^{1-z}, \quad z \notin -\mathbb{Z}_+,$$

where \mathbb{Z}_+ is the set of positive integers. Observe that $z \in -\mathbb{Z}_+$ should be excluded in the definition of $\Gamma_q(z)$ since $(q^z; q)_{\infty} = 0$ for $z \in -\mathbb{Z}_+$. The discrete *q*-beta integral (1.2.3) tends formally to Euler's beta integral (1.2.1) in the limit $q \uparrow 1$ since the *q*-Gamma function $\Gamma_q(z)$ tends to the Gamma function $\Gamma(z)$ when $q \uparrow 1$, and

$$\lim_{q\uparrow 1} \frac{\left(qz;q\right)_{\infty}}{\left(q^{\beta+1}z;q\right)_{\infty}} = (1-z)^{\beta}.$$

A very general continuous q-analogue of the beta integral which depends, besides q, on four additional parameters $\underline{t} = (t_0, t_1, t_2, t_3)$, is the Askey-Wilson integral [7]

(1.2.5)
$$\frac{\frac{1}{2\pi i} \int_{z \in T} \frac{(z^2, z^{-2}; q)_{\infty}}{(t_0 z, t_0 z^{-1}, t_1 z, t_1 z^{-1}, t_2 z, t_2 z^{-1}, t_3 z, t_3 z^{-1}; q)_{\infty}} \frac{dz}{z}}{\left(q, t_0 t_1, t_0 t_2, t_0 t_3, t_1 t_2, t_1 t_3, t_2 t_3; q\right)_{\infty}}$$

where the parameters t_i are generic complex with moduli < 1 and with T the unit circle in the complex plane. Here the shorthand notation $(a_1, \ldots, a_r; q)_k := (a_1; q)_k \ldots (a_r; q)_k$ for products of q-shifted factorials is used. Although this is not clear at first sight, the beta integral (1.2.1) is a limit case of the Askey-Wilson integral (1.2.5). This can be shown by making the change of variables $x = z + z^{-1}$, substituting $t_0 = q^{\alpha + \frac{1}{2}}$, $t_1 = -q^{\beta + \frac{1}{2}}$, $t_2 = q^{\frac{1}{2}}$ and $t_3 = -q^{\frac{1}{2}}$ and taking the limit $q \uparrow 1$ in (1.2.5). Then, we obtain

$$\int_{-1}^{1} (1-x)^{\alpha} (1+x)^{\beta} dx = 2^{\alpha+\beta+1} \frac{\Gamma(\alpha+1)\Gamma(\beta+1)}{\Gamma(\alpha+\beta+2)}$$

which is equivalent with (1.2.1) (cf. [30, Section 6.1]).

It turns out that the discrete q-beta integral (1.2.3) can be considered as a limit case of the Askey-Wilson integral (1.2.5). This is illustrated here for special parameter values. The q-beta integral (1.2.3) for $\alpha = -1/2$ and $\beta = \infty$ (i.e. $q^{\beta} = 0$) is equivalent to the summation formula

(1.2.6)
$$\sum_{k=0}^{\infty} \frac{q^{k/2}}{(q;q)_k} = \frac{1}{(q^{\frac{1}{2}};q)_{\infty}}$$

which is a special case of the q-binomial theorem

$$\sum_{k=0}^{\infty} \frac{\left(a;q\right)_k}{\left(q;q\right)_k} z^k = \frac{\left(az;q\right)_{\infty}}{\left(z;q\right)_{\infty}}, \quad (|z|<1).$$

The summation formula (1.2.6) can be derived from the Askey-Wilson integral (1.2.5) by analytic continuation, residue computation, and a limit transition as follows. Specializing the parameters by

$$(t_0, t_1, t_2, t_3) = (\varepsilon^{-1} q^{\frac{1}{2}}, -1, 0, -q^{\frac{1}{2}})$$

in (1.2.5) and using that $(z^2;q)_{\infty} = (z, -z, q^{\frac{1}{2}}z, -q^{\frac{1}{2}}z;q)_{\infty}$, we obtain

(1.2.7)
$$\frac{1}{2\pi i} \int_{z \in T} w_c(z;\varepsilon) dz = \frac{2}{\left(q, -\varepsilon^{-1}q^{\frac{1}{2}}, -\varepsilon^{-1}q, q^{\frac{1}{2}}; q\right)_{\infty}}$$

for $\varepsilon > q^{-\frac{1}{2}}$ with weight function $w_c(z;\varepsilon)$ given by

$$w_c(z;\varepsilon) := rac{\left(z, q^{rac{1}{2}} z, z^{-1}, q^{rac{1}{2}} z^{-1}; q
ight)_{\infty}}{z \left(\varepsilon^{-1} q^{rac{1}{2}} z, \varepsilon^{-1} q^{rac{1}{2}} z^{-1}; q
ight)_{\infty}}.$$

Now $w_c(.;\varepsilon)$ for generic $\varepsilon > 0$ has two infinite sequences of simple poles, namely the converging sequence of poles $\{\varepsilon^{-1}q^{\frac{1}{2}+k}\}_{k\in\mathbb{Z}_+}$ respectively the diverging sequence of poles $\{\varepsilon q^{-\frac{1}{2}-k}\}_{k\in\mathbb{Z}_+}$. By analytic continuation, (1.2.7) is valid for arbitrary $\varepsilon > 0$ if the integration contour T is replaced by a rectifiable Jordan curve T_{ε} for which the converging sequence of poles is contained in its interior and the diverging sequence of poles is contained in its exterior. Let $w_d(k;\varepsilon)$ ($k \in \mathbb{Z}_+$) be the residue of $w_c(z;\varepsilon)$ at $z = \varepsilon^{-1}q^{\frac{1}{2}+k}$.

1. GENERAL INTRODUCTION

By an easy computation, the residues $w_d(k;\varepsilon)$ for generic ε are computed explicitly as

$$w_d(k;\varepsilon) = \left(\frac{\left(\varepsilon^{-2}q;q\right)_k}{\left(\varepsilon^{-1}q^{\frac{1}{2}},\varepsilon^{-1}q;q\right)_k}\right)\frac{1}{\left(-\varepsilon^{-1}q^{\frac{1}{2}},-\varepsilon^{-1}q;q\right)_{\infty}\left(q;q\right)_{\infty}\left(q^{-k};q\right)_k}$$

for $k \in \mathbb{Z}_+$. Observe that the residue of $w_c(z;\varepsilon)$ at $z = \varepsilon q^{-\frac{1}{2}-k}$ equals $-w_d(k;\varepsilon)$. For generic $\varepsilon > 0$, the contour T_{ε} in $\int_{T_{\varepsilon}} w_c(z;\varepsilon) dz$ can now be pulled back to the unit circle T while picking up residues. Then, it follows for generic $\varepsilon > 0$ by Cauchy's theorem that

(1.2.8)
$$\frac{\frac{1}{2\pi i} \int_{z \in T} w_c(z;\varepsilon) dz + 2 \sum_{k \in \mathbb{Z}_+:\varepsilon < q^{\frac{1}{2}+k}} w_d(k;\varepsilon)}{= \frac{2}{(q, -\varepsilon^{-1}q^{\frac{1}{2}}, -\varepsilon^{-1}q, q^{\frac{1}{2}};q)_{\infty}}}$$

4

Observe that $w_d(k;\varepsilon)$ and the right hand side of (1.2.8) have $\left(-\varepsilon^{-1}q^{\frac{1}{2}}, -\varepsilon^{-1}q;q\right)_{\infty}$ as a common factor in their denominators. Now multiply the left and right hand side of (1.2.8) with this common factor and take the limit $\varepsilon \downarrow 0$. By Lebesgue's dominated convergence theorem, the continuous part in the left hand side of (1.2.8) disappears in this limit, while the finite sum in the left hand side of (1.2.8) tends to the infinite sum

$$2\sum_{k\in\mathbb{Z}_{+}}\frac{q^{k/2}}{\left(q;q\right)_{k}\left(q;q\right)_{\infty}}$$

It follows that the remaining summation formula is equivalent to (1.2.6). This type of computation can be done for general $\alpha, \beta > -1$ (see [122] and Chapter 3), which then shows that the discrete q-beta integral (1.2.3) is a limit case of the continuous q-beta integral (1.2.5).

1.3. Plücker coordinates on the Grassmann manifold

Let l and n be non-zero positive integers such that $l \leq [n/2]$ and let $Y_{l,n}$ be the set of l dimensional subspaces of $V := \mathbb{C}^n$. In the first part of this section we associate to $Y_{l,n}$ an algebra of functions, the so-called homogeneous coordinate ring of $Y_{l,n}$. In order to define the homogeneous coordinate ring of $Y_{l,n}$, we first need to give a different description of $Y_{l,n}$.

The tensor algebra T(V) of the vector space V is the linear space

$$T(V) = \mathbb{C} \oplus V \oplus V^{\otimes 2} \oplus V^{\otimes 3} \oplus \dots$$

with multiplication defined by $t.t' := t \otimes t'$ $(t, t' \in T(V))$. The direct sum decomposition $T(V) = \bigoplus_{m \in \mathbb{Z}_+} V^{\otimes m}$ is a grading with respect to this multiplication.

The exterior algebra $\Lambda(V)$ of the vector space V is by definition the quotient of T(V) with the two-sided ideal \mathcal{J} generated by $v \otimes v$ ($v \in V$). The product in $\Lambda(V)$ is called the wedge product, and is denoted by \wedge .

Let $\{v_i\}_{i=1}^n$ be the canonical basis of V and let $P_m(n)$ be the collection of subsets of $\{1, \ldots, n\}$ of cardinality m. The elements

$$v_J := v_{j_1} \wedge v_{j_2} \wedge \ldots \wedge v_{j_m} \in \Lambda(V)$$

with $J = \{j_1 < \ldots < j_m\} \in P_m(n)$ and $0 \le m \le n$ form a linear basis for $\Lambda(V)$. Here we have used the convention that $v_{\emptyset} := 1$. The direct sum decomposition

$$\Lambda(V) = \bigoplus_{m=0}^{n} \Lambda^{m}(V),$$

with $\Lambda^m(V)$ the span of the basis elements v_J ($J \in P_m(n)$) is a grading of $\Lambda(V)$ with respect to the wedge product.

Let $X := \mathbb{P}(\Lambda^l(V))$ be the projective space associated with $\Lambda^l(V)$, i.e. X is the collection of one dimensional subspaces of $\Lambda^l(V)$. The natural projection $\pi : \Lambda^l(V) \setminus \{0\} \to X$ sends $0 \neq w \in \Lambda^l(V)$ to the one dimensional subspace [w] containing w. Now $Y_{l,n}$ can be embedded in X via the so-called Plücker embedding. The Plücker embedding is defined by

$$Y_{l,n} \to X, \quad W = \operatorname{span}\{w_1, \ldots, w_l\} \mapsto [w_1 \wedge \ldots \wedge w_l].$$

It is easily seen that the Plücker embedding is well-defined and injective.

It turns out that there is a nice description of the image of $Y_{l,n}$ under the Plücker embedding. To give this description of the image, we first need to introduce a group action on the projective space X.

Let $G := GL(n, \mathbb{C})$ be the group of n by n invertible matrices over \mathbb{C} . It is wellknown that G has the structure of an irreducible affine variety, i.e. G is a connected linear algebraic group. The group G acts on V by the usual matrix multiplication

(1.3.1)
$$g.v_j := \sum_i \xi_j^i(g)v_i, \quad g \in G,$$

where ξ_j^i is the coordinate function on G defined by $\xi_j^i(g) = g_{ij}$ if $g = (g_{ij})_{i,j} \in G$.

The left G-action (1.3.1) on V extends to a G-action on $\Lambda(V)$ as follows. The tensor algebra T(V) has a unique left G-module algebra structure such that its action on V coincides with the action (1.3.1). In particular, this means that $g.(v \otimes w) = g.v \otimes g.w$ for $g \in G$ and $v, w \in T(V)$. It follows that the two-sided ideal \mathcal{J} is stable under the G-action, i.e. $g.\mathcal{J} \subset \mathcal{J}$ for all $g \in G$. Consequently, the exterior algebra $\Lambda(V)$ inherets a G-module algebra structure from the G-action on the tensor algebra T(V).

Explicitly, the action of G on the linear basis $\{v_I | I \in P_m(n)\}$ of $\Lambda^m(V)$ is given by

(1.3.2)
$$g.v_J = \sum_{I \in P_m(n)} \xi^I_J(g) v_I, \quad J \in P_m(n),$$

where $\xi_J^I(g)$ is the determinant of the *m* by *m* submatrix of *g* obtained from *g* by deleting all rows with row index $I^c := \{1, \ldots, n\} \setminus I$ and by deleting all columns with column

1. GENERAL INTRODUCTION

index J^c . Formula (1.3.2) for the action on $\Lambda^m(V)$ is a direct consequence of the well-known expansion formula for determinants

(1.3.3)
$$\xi_J^I = \sum_{\sigma \in \mathfrak{S}_m} (-1)^{l(\sigma)} \xi_{j_1}^{i_{\sigma(1)}} \xi_{j_2}^{i_{\sigma(2)}} \dots \xi_{j_m}^{i_{\sigma(m)}},$$

where $I = \{i_1 < \ldots < i_m\}, J = \{j_1 < \ldots < j_m\} \in P_m(n), \mathfrak{S}_m$ is the permutation group of $\{1, \ldots, m\}$ and

$$l(\sigma) := \#\{1 \le i < j \le m \,|\, \sigma(i) > \sigma(j)\}$$

is the so-called length function on \mathfrak{S}_m . In particular, it follows from (1.3.2) that the graded pieces $\Lambda^m(V)$ are invariant subspaces for the *G*-action. Observe that

(1.3.4)
$$g.[w] := [g.w], \quad (g \in G, w \in \Lambda^m(V)),$$

is a well-defined left G-action on X. Set

$$x := v_{\{n-l+1,\dots,n\}} \in \Lambda^l(V),$$

then it follows by elementary linear algebra that the *G*-orbit $G.[x] := \{g.[x] | g \in G\} \subset X$ is exactly the image of $Y_{l,n}$ under the Plücker embedding. From now on, we identify $Y_{l,n}$ with its image under the Plücker embedding, i.e. we identify $Y_{l,n}$ with the *G*-orbit G.[x].

Observe that via the map $G \to G.[x]$, $g \mapsto g.[x]$, the G-orbit $Y_{l,n} = G.[x]$ can be identified with the coset space $G/G_x := \{gG_x \mid g \in G\}$, where

$$G_x := \{g \in G \mid g.[x] = [x]\}$$

is the so-called isotropy subgroup of [x]. It is not difficult to verify that

(1.3.5)
$$G_x = \{g = (g_{ij})_{i,j} \in G \mid g_{ij} = 0 \text{ for } 1 \le i \le n-l, \ n-l+1 \le j \le n\}.$$

In particular, the lower triangular matrices are contained in the isotropy subgroup G_x . This fact implies that G_x is a so-called parabolic subgroup. A parabolic subgroup P of G has the important property that the corresponding quotient G/P is a projective G-variety, i.e. the quotient G/P can be realized as the zero set of a finite collection of homogeneous polynomials. So $Y_{l,n} = G[x]$ becomes a (n - l)l dimensional irreducible projective variety, the so-called (complex) projective Grassmann manifold.

The homogeneous coordinate ring A of $Y_{l,n} \subset \mathbb{P}(\Lambda^l(V))$ is defined as

$$A := \mathbb{C}[\Lambda^l(V)]/\Im,$$

where $\mathbb{C}[\Lambda^{l}(V)]$ is the coordinate ring of $\Lambda^{l}(V)$ and \mathfrak{I} is the ideal generated by the homogeneous polynomials $f \in \mathbb{C}[\Lambda^{l}(V)]$ which vanish on $Y_{l,n}$. Equivalently, A is the coordinate ring of the affine cone $Y_{l,n}^{a} := \pi^{-1}(Y_{l,n}) \cup \{0\} \subset \Lambda^{l}(V)$ over $Y_{l,n}$, where $\pi : \Lambda^{l}(V) \setminus \{0\} \to X$ is the natural projection. The algebra A can be realized as a subalgebra of the coordinate ring $\mathbb{C}[G]$ of G as follows. Consider the morphism $\phi : G \to Y_{l,n}^{a}$ defined by $\phi(g) := g.x$. The dual mapping ϕ^* is by definition the algebra homomorphism

$$\phi^* : A \to \mathbb{C}[G], \quad \phi^*(f) := f \circ \phi.$$

The dual map ϕ^* is injective since $\overline{\phi(G)} = Y_{l,n}^a$, where $\overline{\phi(G)}$ is the closure of $\phi(G)$ in $Y_{l,n}^a$. Hence ϕ^* embeds A into the coordinate ring $\mathbb{C}[G]$ of G. We can give now an explicit set of algebraic generators for the image of ϕ^* by computing the image under ϕ^* of the coordinate functions on $\Lambda^l(V)$. We take here the coordinate functions $\eta_I \in \mathbb{C}[\Lambda^l(V)]$ $(I \in P_l(n))$ with respect to the linear basis $\{v_I \mid I \in P_l(n)\}$ of $\Lambda^l(V)$, i.e. $\eta_I : \Lambda^l(V) \to \mathbb{C}$ is the linear mapping which maps the basis elements v_J $(J \neq I)$ to 0 and which maps v_I to 1. Then,

$$(\phi^*(\eta_I))(g) = \eta_I(g.x) = t_I(g), \quad (I \in P_l(n)),$$

where the t_I ($I \in P_l(n)$) are the so-called *Plücker coordinates*

(1.3.6)
$$t_I := \xi_{\{n-l+1,\dots,n\}}^I, \quad (I \in P_l(n))$$

In other words, the algebra A may be identified with the subalgebra $\mathbb{C}[t_I | I \in P_l(n)]$ of $\mathbb{C}[G]$ generated by the Plücker coordinates t_I $(I \in P_l(n))$.

A set of homogeneous algebraic relations between the Plücker coordinates t_I are called a set of defining relations if its pre-image under ϕ^* generates \Im as an ideal. We describe here one well-known set of defining relations, namely the set of Plücker relations. To give these quadratic relations it is convenient to define for arbitrary i_1, \ldots, i_l between 1 and n,

$$t_{i_1,\dots,i_l} := \sum_{\sigma \in \mathfrak{S}_l} (-1)^{l(\sigma)} \xi_{n-l+1}^{i_{\sigma(1)}} \xi_{n-l+2}^{i_{\sigma(2)}} \cdots \xi_n^{i_{\sigma(l)}}$$

Observe that t_{i_1,\ldots,i_l} is anti-symmetric in the *l* indices i_1,\ldots,i_l and that its definition coincides with the definition (1.3.6) of the Plücker coordinate $t_{\{i_1,\ldots,i_l\}}$ if $i_1 < \ldots < i_l$. In particular, $t_{i_1,\ldots,i_l} = 0$ when two indices i_p, i_r $(p \neq r)$ coincide. The Plücker relations are now given by the quadratic relations

(1.3.7)
$$\sum_{i=1}^{l+1} (-1)^i t_{k_1,\dots,\hat{k}_i,\dots,k_{l+1}} t_{k_i,j_1,\dots,j_{l-1}} = 0$$

for subsets $K = \{k_1 < ... < k_{l+1}\} \in P_{l+1}(n)$ and $J = \{j_1 < ... < j_{l-1}\} \in P_{l-1}(n)$, where \hat{k} means that the element k should be omitted. There is a rich combinatorial structure related to the defining relations of the Grassmann manifold, for which we refer to Towber [128].

The so-called Levi subgroup L corresponding to the maximal parabolic subgroup G_x consists of the n by n invertible matrices of the form

(1.3.8)
$$\begin{pmatrix} B & 0\\ 0 & C \end{pmatrix}, \quad B \in GL(n-l,\mathbb{C}), \ C \in GL(l,\mathbb{C}).$$

The algebra of right L-invariant polynomial functions on G,

(1.3.9)
$$\mathbb{C}[G/L] := \{ f \in \mathbb{C}[G] \mid f(gp) = f(g), \ \forall g \in G, \forall p \in L \},\$$

turns out to be closely related to the homogeneous coordinate ring A of the projective Grassmann manifold $Y_{l,n}$. Since the precise connection between A and $\mathbb{C}[G/L]$ in a

1. GENERAL INTRODUCTION

more general setting plays a fundamental role in the last two chapters of the Tract, we explain it here in more detail for the complex Grassmannian.

For the precise connection between A and $\mathbb{C}[G/L]$ we have to consider a specific G-orbit within $\Lambda^l(V) \otimes \Lambda^l(V^*)$, where V^* is the dual module of V. First, we define the dual module of V and we give the dual construction of the Grassmannian manifold $Y_{l,n}$ by replacing the role of the G-module V by its dual. Detailed discussions are omitted here since the constructions and techniques are the same as before.

The G-action on the linear dual $V^* := \{f : V \to \mathbb{C} \mid f \text{ linear } \}$ of V is given by

$$(g.f)(v) := f(g^{-1}.v), \quad (f \in V^*, v \in V, g \in G).$$

Let $\{v_i^*\}_i$ be the dual basis with respect to $\{v_i\}_i$. A linear basis of the *m*th graded part of the exterior algebra $\Lambda(V^*)$ is given by the elements

$$v_I^* := v_{i_1}^* \wedge v_{i_2}^* \wedge \ldots \wedge v_{i_m}^*,$$

where $I = \{i_1 < ... < i_m\} \in P_m(n)$. Set

$$\overline{\xi}_J^I(g):=\xi_I^J(g^{-1}) \quad (g\in G \text{ and } I,J\in P_m(n)),$$

then the $\overline{\xi}_J^I$ occur as matrix coefficients of the induced *G*-action on $\Lambda^m(V^*)$,

$$g.v_J^* = \sum_{I \in P_m(n)} \overline{\xi}_J^I(g) v_I^*, \quad (J \in P_m(n)).$$

Now consider the action of G on the projective space $X^* := \mathbb{P}(\Lambda^m(V^*))$ and let $Y_{l,n}^*$ be the G-orbit of $[x^*] \in X^*$, where

$$x^* := v^*_{\{n-l+1,\dots,n\}} \in \Lambda^l(V^*).$$

Then $Y_{l,n}^*$ is an irreducible projective variety of dimension l(n - l) with homogeneous coordinate ring isomorphic to the subalgebra

$$A^* := \mathbb{C}[t_I^* \mid I \in P_l(n)],$$

where

(1.3.10)
$$t_I^* := \overline{\xi}_{\{n-l+1,\dots,n\}}^I, \quad (I \in P_l(n))$$

are the dual Plücker coordinates.

The map $*: \xi_j^i \mapsto \overline{\xi}_j^i, *: \det^{-1} \mapsto \det$ extends uniquely to a anti-linear involution * of $\mathbb{C}[G]$, which maps A bijectively onto A^* . In fact, * maps the Plücker coordinate t_I to the dual Plücker coordinate t_I^* for all $I \in P_l(n)$. Hence the algebraic structure of A^* is determined by the algebraic structure of A.

A so-called real form U of G is associated with the involution * on $\mathbb{C}[G]$ by

$$U := \{g \in G \mid f(g) = f^*(g) \ \forall f \in \mathbb{C}[G]\}.$$

Here U is the group consisting of the n by n unitary matrices.

Observe that the isotropy subgroup $G^*_x := \{g \in G \, | \, g.[x^*] = [x^*]\}$ of $[x^*]$ is given by

$$G_x^* := \{ g = (g_{ij})_{i,j} \in G \, | \, g_{ij} = 0 \text{ for } n - l + 1 \le i \le n, \ 1 \le j \le n - l \}$$

which is a parabolic subgroup of G since it contains the subgroup of upper triangular matrices. Observe furthermore that the Levi subgroup L is equal to the intersection $G_x \cap G_x^*$.

Now we can relate the homogeneous coordinate rings A and A^* with the subalgebra $\mathbb{C}[G/L]$ (1.3.9) as follows. Consider the diagonal action of G on $\Lambda^l(V) \otimes \Lambda^l(V^*)$, which is defined by $g.(v \otimes w) := g.v \otimes g.w$ for $g \in G$, $v \in \Lambda^l(V)$ and $w \in \Lambda^l(V^*)$. The diagonal action descends to a well-defined action on $\mathbb{P}(\Lambda^l(V) \otimes \Lambda^l(V^*))$, cf. (1.3.4). Let B be the homogeneous coordinate ring of the closure of the orbit G.[y], where y is given by

(1.3.11)
$$y := v_{\{n-l+1,\dots,n\}} \otimes v_{\{n-l+1,\dots,n\}}^* \in \Lambda^l(V) \otimes \Lambda^l(V^*).$$

The algebra B can be identified with the subalgebra of $\mathbb{C}[G]$ generated by the elements $t_I t_J^*(I, J \in P_l(n))$ via the dual ψ^* of the morphism $\psi : g \mapsto g.y$. On the other hand, L stabilizes y, hence the image of ψ^* is contained in $\mathbb{C}[G/L]$. It can be shown that $\mathbb{C}[G/L]$ is in fact generated by the elements $t_I t_J^*(I, J \in P_l(n))$, i.e. any right L-invariant polynomial function on G is a polynomial in the elements $t_I t_J^*(I, J \in P_l(n))$. This result can be informally restated as follows: any right L-invariant polynomial function on G is a solution of f is a polynomial and an anti-holomorphic polynomial in the Plücker coordinates t_I ($I \in P_l(n)$). From this viewpoint we have thus obtained a factorization of the algebra $\mathbb{C}[G/L]$ in terms of algebras of holomorphic and anti-holomorphic polynomials.

1.4. Overview of the remaining chapters

1.4.1. Multivariable orthogonal polynomials. The study of orthogonal polynomials related to multivariable beta type integrals started in the 1970's with the work of James and Constantine [45], Vretare [131], [132] and Koornwinder and Sprinkhuizen-Kuyper [61], [62] [69], [113]. In the late 1980's Heckman and Opdam [36], [37], [38] associated to each irreducible root system certain multivariable analogues of the Jacobi polynomials. Many important properties of the polynomials were derived, such as orthogonality relations, quadratic norm evaluations and the existence of a "large enough" system of differential equations for which the Jacobi polynomials are joint eigenfunctions; the so-called hypergeometric differential equations.

Macdonald [82] introduced q-deformations of the Heckman-Opdam polynomials and proved orthogonality relations with respect to multivariable continuous q-beta type integrals. Cherednik's affine Hecke-algebraic approach [12], [13], [14], [15] has led to a good understanding of the basic properties of the Macdonald polynomials.

The Macdonald polynomials associated to the non-reduced root system BC of rank 1 form a two parameter subfamily of the four parameter family of Askey-Wilson polynomials. The Askey-Wilson polynomials are the orthogonal polynomials with respect to the continuous q-beta integral (1.2.5). Koornwinder [65] extended the definition of the BC type Macdonald polynomials to a five parameter family of orthogonal polynomials, which for rank 1 reduce to the full four parameter family of Askey-Wilson polynomials. Similar results as mentioned above for the Heckman-Opdam polynomials and the Macdonald polynomials have been proved for the Koornwinder polynomials by work of Koornwinder [65], van Diejen [17], [18], Noumi [91], Macdonald [86] and Sahi [106].

The Askey-Wilson polynomials are on top of the so-called Askey tableau. The Askey tableau is a hierarchy of families of basic hypergeometric orthogonal polynomials which are joint eigenfunctions of a second order q-difference operator. Certain families can be obtained from other families by limit transitions or specializations of the parameters, which induces the hierarchy structure between the families in the Askey tableau. The example treated in Section 1.2 is directly related to the hierarchy structure between the Askey-Wilson polynomials and the orthogonal polynomials associated to the discrete q-beta integral (1.2.3), which are called the little q-bacobi polynomials. More explicitly, in Section 1.2 it is mentioned that the discrete q-beta integral (1.2.3) is a limit case of the Askey-Wilson integral (1.2.5). The corresponding limit on the level of orthogonal polynomials gives the hierarchy structure between Askey-Wilson polynomials and the little q-Jacobi polynomials.

In Chapter 2 and Chapter 3 multivariable analogues of three families of orthogonal polynomials in the Askey tableau are introduced, namely the q-Racah polynomials and the big and little q-Jacobi polynomials. It is also proved that limit transitions between these three families and the Koornwinder polynomials exist, which can be seen as a multivariable generalization of the hierarchy structure in the Askey tableau. Furthermore, full orthogonality of the polynomials is established with respect to multivariable q-beta type integrals. Their quadratic norms are computed and it is shown that for each family there is a second order q-difference operator which is diagonalized by the orthogonality relations with respect to multivariable big and little q-Jacobi polynomials satisfy orthogonality relations with respect to multivariable discrete q-beta integrals which have been introduced in 1980 by Askey [5], and have been studied intensively thereafter in several papers, e.g. [5], [33], [54], [4], [29], [127].

An important tool for obtaining the above mentioned results on the multivariable Askey tableau is the development of a residue calculus for Gustafson's multidimensional analogue of the continuous q-beta integral (1.2.5), which is presented in Chapter 2. Discrete multivariable q-analogues of the beta integral can now be obtained from Gustafson's integral by suitable limit transitions. This was shown for a special one variable example in Section 1.2. Several properties of the limit cases of the Koornwinder polynomials are proved by taking the limits in the corresponding results for the Koornwinder polynomials.

The results of Chapter 2 and Chapter 3 have appeared in several papers. The multivariable big and little q-Jacobi polynomials were introduced and studied in [116]. Several limit transitions between multivariable orthogonal polynomials were studied in an algebraic manner, cf. Section 3.6, in the joint paper [121] with Koornwinder. The multivariable q-Racah polynomials were introduced and studied in the joint paper [20] with van Diejen. In the papers [117], [118] the Koornwinder polynomials for special parameter values are studied with respect to partially discrete orthogonality measures. The general residue calculus for Koornwinder polynomials has appeared in [119]. In Chapter 2 and Chapter 3 we have mainly followed the approach of [119]. Finally, the theory in Chapter 2 and Chapter 3 in the one variable case has appeared in the joint paper [122] with Koornwinder.

1.4.2. Multivariable orthogonal polynomials and quantum Grassmannians. In the mid 1980's Drinfeld [27] and Jimbo [47] quantized the universal enveloping algebra $U(\mathfrak{g})$ of a simple Lie algebra \mathfrak{g} . They obtained the "standard" quantized universal enveloping algebra $U_h(\mathfrak{g})$, which is a Hopf-algebra quantization of $U(\mathfrak{g})$ endowed with the co-Poisson structure induced by the standard solution r of the modified classical Yang-Baxter equation. The defining relations of $U_h(\mathfrak{g})$ are given by the so-called quantized Serre relations. In particular, the definition of $U_h(\mathfrak{g})$ depends on a particular choice of a Cartan subalgebra $\mathfrak{h} \subset \mathfrak{g}$ and of a choice of simple roots $\Delta = \{\alpha_1, \ldots, \alpha_r\}$ for the root system associated with $(\mathfrak{g}, \mathfrak{h})$.

The Hopf-algebra dual $\mathbb{C}_h[G]$ of $U_h(\mathfrak{g})$ is a quantization of the function algebra of regular functions on the connected, simply connected, simple Lie group G with Lie algebra \mathfrak{g} . The associated Poisson-Lie group structure on G in the semi-classical limit is given by the so-called Sklyanin-bracket on G associated with r. This bracket is also known as the Bruhat-Poisson bracket on G.

In Chapter 4,5 and 6 of the Tract an important object of study is a rational form of $\mathbb{C}_h[G]$, which is denoted here by $\mathbb{C}_q[G]$. Here q is assumed to be specialized to a value in the open interval (0, 1). For several simple Lie groups G, there is an explicit realization of $\mathbb{C}_q[G]$ in terms of generators and relations. For the purpose of this section, we give here the construction of $\mathbb{C}_q[G]$ for the reductive Lie group $G = GL(n, \mathbb{C})$ in terms of generators and relations. The quantized function algebra $\mathbb{C}_q[SL(n, \mathbb{C})]$ is then an Hopf subgroup of $\mathbb{C}_q[GL(n, \mathbb{C})]$, which can formally be obtained from $\mathbb{C}_q[GL(n, \mathbb{C})]$ by setting the quantum determinant equal to 1. The algebra $\mathbb{C}_q[GL(n, \mathbb{C})]$ is generated by the elements t_{ij} $(1 \le i, j \le n)$ and \det_q^{-1} , subject to the relations

$$t_{ki}t_{kj} = qt_{kj}t_{ki}, \quad t_{ik}t_{jk} = qt_{jk}t_{ik} \quad (i < j),$$

$$t_{il}t_{kj} = t_{kj}t_{il}, \quad t_{ij}t_{kl} - t_{kl}t_{ij} = (q - q^{-1})t_{il}t_{kj} \quad (i < k, j < l),$$

and det_a^{-1} is defined as the inverse of the quantum determinant

$$\det_q := \sum_{\sigma \in \mathfrak{S}_n} (-q)^{l(\sigma)} t_{\sigma(1)1} t_{\sigma(2)2} \dots t_{\sigma(n)n}.$$

Observe the striking similarity with the coordinate ring $\mathbb{C}[G]$ of $G = GL(n, \mathbb{C})$ as considered in Section 1.3. In particular, it is immediate from the above description of $\mathbb{C}_q[G]$ that the coordinate ring $\mathbb{C}[G]$ is formally reobtained by taking the limit $q \uparrow 1$.

The function algebra $\mathbb{C}_q[G/L]$ of the regularly embedded quantum Grassmannian is defined as the "standard" quantization of the algebra of functions $\mathbb{C}[G/L]$, where G =

1. GENERAL INTRODUCTION

 $GL(n, \mathbb{C}), L = GL(n - l, \mathbb{C}) \times GL(l, \mathbb{C})$ and $\mathbb{C}[G/L]$ is the subalgebra of right Linvariant functions in $\mathbb{C}[G]$, cf. (1.3.9). The quantized function algebra $\mathbb{C}_q[G/L]$ can be alternatively described as the subalgebra of right $\mathbb{C}_q[L]$ -invariant elements in $\mathbb{C}_q[G]$, where $\mathbb{C}_q[L]$ is the obvious quantum analogue of the subgroup $L = GL(n - l, \mathbb{C}) \times$ $GL(l, \mathbb{C})$, cf. (1.3.8). This quantization of the Grassmann manifold is called regularly embedded because the quantum subgroup $\mathbb{C}_q[L]$ contains the quantized diagonal matrices. This property turns out to have vast implications for the associated harmonic analysis. For instance, it implies that the associated zonal spherical functions satisfy orthogonality relations with respect to completely discrete orthogonality measures.

In Chapter 4 it is shown that the zonal spherical functions on the regularly embedded quantum Grassmannian can in fact be identified with multivariable big and little q-Jacobi polynomials. The strategy for obtaining this result is as follows. Noumi, Dijhuizen and Sugitani [92] introduced a one parameter family of quantum complex Grassmannians and they identified the associated zonal spherical functions with a subfamily of the Koornwinder polynomials. The regularly embedded quantum Grassmannians can be formally reobtained from the one parameter family of quantum Grassmannians by sending the parameter to infinity. In Chapter 4 it is shown that such limits on the level of quantum Grassmannians correspond on the level of zonal spherical functions with the limits from Koornwinder polynomials to multivariable big and little q-Jacobi polynomials as studied in Chapter 2. The results of Chapter 1 and Chapter 2 therefore play an essential role in the study of the limit transitions on the one parameter family of quantum Grassmannians.

The rank 1 case of these results was obtained by Koornwinder [64], [66] for 2-spheres and by Dijkhuizen and Noumi [24]. Chapter 4 is based on the joint paper [25] with Dijkhuizen. The results in Section 4.5 and Section 4.6 have appeared before in the paper [92] of Noumi, Dijkhuizen and Sugitani without proofs.

1.4.3. Quantum Plücker coordinates and their generalizations. The concept of Plücker coordinates on Grassmannians can be generalized to arbitrary flag manifolds. The general definition is based on the characterization of the Plücker coordinates as matrix coefficients of a finite dimensional (irreducible) representation, cf. (1.3.2), (1.3.6).

Flag manifolds are defined as the homogeneous spaces of the form G/P, where G is a connected, simply connected, simple Lie group and P is a parabolic subgroup of G. In this section the parabolic subgroup P is assumed to be standard with respect to the fixed Cartan subalgebra \mathfrak{h} and with respect to the fixed simple roots Δ . The standard parabolic subgroups are naturally parametrized by subsets of Δ .

On the other hand, irreducible finite dimensional representations of G are parametrized by an integral cone $\bigoplus_{\alpha \in \Delta} \mathbb{Z}_+ \varpi_\alpha$, the so-called integral cone of dominant integral weights. The ϖ_α ($\alpha \in \Delta$) are called the fundamental dominant weights and form a type of dual basis with respect to the simple roots Δ . For a parabolic subgroup P of Gcorresponding to a particular subset $I \subset \Delta$, the (generalized) Plücker coordinates can be defined as certain matrix coefficients of the finite dimensional irreducible representations of G corresponding to the fundamental weights ϖ_α ($\alpha \in I$). Dual (generalized) Plücker coordinates are then defined as certain matrix coefficients of the corresponding

dual representations. Quantum analogues of the generalized Plücker coordinates and of the generalized dual Plücker coordinates can be defined in a similar manner by using the finite dimensional corepresentation theory of $\mathbb{C}_q[G]$.

If L is the Levi component of the parabolic subgroup P, then the "standard" quantized function algebra $\mathbb{C}_q[G/L]$ can be defined as the subalgebra of right $\mathbb{C}_q[L]$ -invariant elements of the quantized function algebra $\mathbb{C}_q[G]$, where $\mathbb{C}_q[L]$ is the quantum subgroup associated with the Levi component L. Then, products of (generalized) quantum Plücker coordinates and dual (generalized) quantum Plücker coordinates lie in the quantized function algebra $\mathbb{C}_q[G/L]$. From an informal point of view, this means that $\mathbb{C}_q[G/L]$ contains the quantized algebra of zero-weighted complex-valued polynomial functions on G/L.

One of the questions which is addressed in Chapter 5 and Chapter 6 is the following: Is $\mathbb{C}_q[G/L]$ generated as algebra by the products of the quantum Plücker coordinates and the dual quantum Plücker coordinates? If the answer is affirmative, then the quantized function algebra $\mathbb{C}_q[G/L]$ is called factorizable. Affirmative answers to this factorization problem is given for an interesting class of flag manifolds. In fact it turns out that the answer is affirmative for *all* flag manifolds (unpublished result of the author).

In Chapter 5 the factorization problem is proved for the regularly embedded quantum Grassmannian. Furthermore, the defining relations between the quantum Plücker coordinates and the dual quantum Plücker coordinates are derived in case of the complex Grassmannian. To arrive at this result, it is important to understand the algebraic structure of the algebra A_q , which is by definition the algebra generated by the quantum Plücker coordinates. The algebra A_q is the quantized homogeneous coordinate ring of the Grassmann manifold $Y_{l,n}$, cf. Section 1.3. The algebraic structure of A_q is well understood due to the work of Taft and Towber [**126**] and of Noumi, Yamada and Mimachi [**96**]. In [**126**] it was shown that the defining relations of the quantum Plücker coordinates are given by Young symmetry relations, or equivalently by q-Garnir relations. Both of these relations contain the quantum analogues of the Plücker relations (1.3.7) as special cases. The results of Chapter 5 are based on a handwritten manuscript of the author.

In Chapter 6 the factorization problem is considered for arbitrary flag manifolds. The factorization is proved for an interesting class of flag manifolds, which contains in particular the irreducible compact Hermitean symmetric spaces. This part is based on the joint paper [123] with Dijkhuizen.

1.4.4. Quantum orbit method for flag manifolds. In Section 1.3 of this chapter an involution on the coordinate ring $\mathbb{C}[G]$ of $G = GL(n, \mathbb{C})$ was introduced. This involution corresponds on the level of groups with choosing the *n* by *n* unitary matrices as real form of *G*. The subalgebra $\mathbb{C}[G/L]$ (1.3.9) of right $L = GL(n-l, \mathbb{C}) \times GL(l, \mathbb{C})$ invariant polynomial functions on $G = GL(n, \mathbb{C})$ is stable under this involution. In fact, the involution maps the Plücker coordinates to the dual Plücker coordinates. There is a natural anti-linear anti-involution * of the standard quantized function algebra $\mathbb{C}_q[G]$ which, in the semi-classical limit, reduces to choosing the *n* by *n* unitary matrices as real form of $G = GL(n, \mathbb{C})$. Again, $\mathbb{C}_q[G/L]$ is *-stable and the anti-involution maps the quantum Plücker coordinates to the dual quantum Plücker coordinates.

1. GENERAL INTRODUCTION

For a given connected, simply connected, simple Lie group G with Lie algebra \mathfrak{g} and Cartan subalgebra \mathfrak{h} there is a standard way to construct a compact real form U of G using a special root basis of \mathfrak{g} , cf. [40, Theorem 6.3]. The compact real form U has the property that for any standard parabolic subgroup $P, K = P \cap U$ is a compact real form of the Levi component L of P. Furthermore, G/P is isomorphic to U/K as a real manifold.

There exists an anti-linear anti-involution * on the quantized function algebra $\mathbb{C}_q[G]$ which corresponds in the semi-classical limit to the above mentioned choice of compact real form U. To emphasize that $\mathbb{C}_q[G]$ is considered with this *-structure, it is customary to write $\mathbb{C}_q[U]$ instead of $\mathbb{C}_q[G]$. The subalgebra $\mathbb{C}_q[G/L]$ with L the Levi component of a standard parabolic subgroup P is stable under the *-involution. Furthermore, the *-involution maps quantum Plücker coordinates to dual quantum Plücker coordinates. In Chapter 6, the algebra $\mathbb{C}_q[G/L]$ is considered with the above mentioned *-structure. To emphasize the choice of *-structure, we write $\mathbb{C}_q[U/K]$ ($K := P \cap U$) for the algebra $\mathbb{C}_q[G/L]$ with this particular choice of *-structure.

In Chapter 6 the quantum orbit method is developed for flag manifolds U/K by relating the irreducible *-representations of the *-algebra $\mathbb{C}_q[U/K]$ to the geometry of the underlying Poisson structure on U/K. Since $\mathbb{C}_q[U/K]$ is a *-subalgebra of $\mathbb{C}_q[U]$, the Poisson structure of U/K in the semi-classical limit is the induced Bruhat-Poisson structure of U, i.e. it is the unique Poisson structure on U/K such that the natural projection $U \to U/K$ preserves the Poisson structures.

The Poisson geometry of U and U/K with respect to the Bruhat-Poisson bracket was studied by Soibelman [110], respectively Lu and Weinstein [80]. It was shown that the symplectic foliation of U is a refinement of the Bruhat decomposition of U and that the symplectic foliation of U/K coincides with the Schubert cell decomposition of U/K. Every symplectic leaf of U is mapped surjectively onto a Schubert cell of U/K under the natural projection $U \to U/K$.

The irreducible *-representations of the standard quantized function algebra of the group of n by n unitary matrices were classified independently by Koelink [60] and Soibelman [109]. In [110], Soibelman classified the irreducible *-representations of $\mathbb{C}_q[U]$ for an arbitrary compact simple Lie group U. Soibelman showed that the irreducible *-representations of $\mathbb{C}_q[U]$ are naturally parametrized by the symplectic leaves of the Bruhat-Poisson structure on U.

The natural embedding $\mathbb{C}_q[U/K] \hookrightarrow \mathbb{C}_q[U]$ can be interpreted as the quantized dual of the natural projection $U \to U/K$. In Chapter 6 it is shown there is a close connection between the properties of the natural projection $U \to U/K$ and the properties of its quantized dual. For instance, it is shown that an irreducible *-representation π of $\mathbb{C}_q[U]$ remains irreducible as *-representation of $\mathbb{C}_q[U/K]$ if and only if the symplectic leaf corresponding to π is mapped isomorphically onto its image under the natural projection $U \to U/K$. Since for each Schubert cell Y of U/K there exist symplectic leaves of U which are mapped isomorphically onto Y, we thus obtain in a natural way an irreducible *-representation of $\mathbb{C}_q[U/K]$ for every symplectic leaf Y of U/K. Irreducible *-representations corresponding to different Schubert cells turn out to be inequivalent.

A *-representation of a factorizable quantized function algebras $\mathbb{C}_q[U/K]$ is completely determined by its action on the products of the quantum Plücker coordinates and the dual quantum Plücker coordinates. This fact is used to give a complete classification of the irreducible *-representations of a factorizable quantized function algebra $\mathbb{C}_q[U/K]$. For factorizable quantized function algebras $\mathbb{C}_q[U/K]$ it is shown that the irreducible *-representations of $\mathbb{C}_q[U/K]$ obtained by restriction of irreducible *-representations of $\mathbb{C}_q[U/K]$ obtained by restriction of irreducible *-representations of $\mathbb{C}_q[U/K]$. In particular, the complete classification of the irreducible *-representations is obtained for the quantized function algebras of the irreducible *-representations is metric spaces.

These and other connections between the Poisson geometry of flag manifolds and the irreducible *-representations of the quantized function algebras of flag manifolds are explored in full detail in Chapter 6.

Chapter 6 is based on the joint paper [123] with Dijkhuizen.

1.5. Notations and conventions

The following notations and conventions are used throughout the Tract.

- $\mathbb{N} := \{1, 2, 3, \dots\}$ and $\mathbb{Z}_+ = \{0, 1, 2, \dots\}.$
- Sums over empty index sets are equal to 0, products over empty index sets are equal to 1.
- q is a fixed real number in the open interval (0, 1), unless explicitly stated otherwise.

-
$$[k, l] := \{k, k + 1, \dots, l - 1, l\}$$
 for integers k, l with $k < l$.

- An associative algebra is by definition an associative algebra with unit.
- The q-shifted factorial is defined by

(1.5.1)
$$(a;q)_b := \frac{(a;q)_\infty}{(aq^b;q)_\infty}, \quad (a;q)_\infty := \prod_{j=0}^\infty (1-q^j a),$$

provided that $aq^b \notin \{q^{-k}\}_{k \in \mathbb{Z}_+}$. For $b = k \in \mathbb{Z}_+$, this reduces to $(a;q)_k = \prod_{i=0}^{k-1} (1 - aq^i)$, which is well defined for all $a \in \mathbb{C}$. The notation

$$(a_1,\ldots,a_m;q)_b := \prod_{j=1}^m (a_j;q)_b$$

is used for products of q-shifted factorials.

CHAPTER 2

A residue calculus for Koornwinder polynomials

2.1. Introduction

Koornwinder [**65**] introduced a five parameter multivariable generalization of the four parameter family of Askey-Wilson polynomials [**7**] and showed that the polynomials satisfy orthogonality relations with respect to the absolutely continuous measure of Gustafson's multidimensional Askey-Wilson integral [**31**]

(2.1.1)
$$\frac{1}{(2\pi i)^{n}} \iint_{z \in T^{n}} \prod_{\substack{1 \le k < l \le n \\ \varepsilon_{k}, \varepsilon_{l} \in \{\pm 1\}}} \frac{\left(z_{k}^{\varepsilon_{k}} z_{l}^{\varepsilon_{l}}; q\right)_{\infty}}{\left(t z_{k}^{\varepsilon_{k}} z_{l}^{\varepsilon_{l}}; q\right)_{\infty}} \prod_{i=1}^{n} \frac{\left(z_{i}^{2}, z_{i}^{-2}; q\right)_{\infty}}{\prod_{j=0}^{3} \left(t_{j} z_{i}, t_{j} z_{i}^{-1}; q\right)_{\infty}} \frac{dz_{i}}{z_{i}}$$
$$= 2^{n} n! \prod_{i=1}^{n} \frac{\left(t, t^{2n-i-1} t_{0} t_{1} t_{2} t_{3}; q\right)_{\infty}}{\left(q, t^{n-i+1}; q\right)_{\infty} \prod_{0 \le k < l \le 3} \left(t^{n-i} t_{k} t_{l}; q\right)_{\infty}},$$

where T^n is the *n*-torus in \mathbb{C}^n and $|t_i| < 1$, $t \in (0, 1)$. This five parameter family of multivariable orthogonal polynomials is nowadays known as the family of Koornwinder polynomials. Certain three parameter subfamilies of the Koornwinder polynomials coincide with the families of BC type Macdonald polynomials, which were previously introduced and studied by Macdonald [86]. Van Diejen [16] has shown that the Koornwinder polynomials contain all families of Macdonald polynomials associated with the classical root systems as special cases or, in the case of root system A, as the highest homogeneous part.

Several important properties of the Koornwinder polynomials were derived in the past years. We mention here two properties which will play an important role in this chapter. Koornwinder [65] introduced an explicit second order q-difference operator which simultaneously diagonalizes the Koornwinder polynomials. Van Diejen [17] evaluated the quadratic norms for a four parameter family of the Koornwinder polynomials using so-called Pieri formulas. By recent results of Sahi [106], van Diejen's quadratic norm evaluations are valid for the full five parameter family of Koornwinder polynomials, see also [120].

In [65], [17] the orthogonality relations and norm evaluations with respect to the measure associated with (2.1.1) were derived for parameters (\underline{t}, t) with $|t_i| \leq 1$. In Section 2.3 we extend these results to the case that the four parameters \underline{t} are generically complex. Intuitively, the results for \underline{t} generically complex follow from the results for $|t_i| \leq 1$ by analytic continuation, after replacing the torus T^n in the orthogonality measure by a suitable Cartesian product C^n of a Jordan curve C. Informally rephrased, the results in Section 2.3 follow from a multidimensional analytic version of Cauchy's Theorem for the measure associated with (2.1.1), analytic in the sense that no poles of the measure are picked up. The technicalities are a little bit more involved and will be discussed in Section 2.3.

In Section 2.4 a multidimensional meromorphic version of Cauchy's Theorem is developed for the measure associated with (2.1.1), so that certain poles of the measure can be picked up by shifting the multidimensional contour. The completely discrete weights which arise from this residue calculus contain a common factor which has a pole when the parameters satisfy a type of truncation condition. By dividing out this common factor, the partly continuous weights are zero for parameters satisfying the truncation condition, and all that remains is a discrete, finite orthogonality measure. The Koornwinder polynomials with parameters satisfying this truncation condition are multivariable analogues of the well-known family of one variable q-Racah polynomials. These multivariable analogues of the q-Racah polynomials are studied in Section 2.5.

In Section 2.6 orthogonality relations for the Koornwinder polynomials are derived with respect to positive, partly discrete orthogonality measures. The results in Section 2.6 are important for the understanding of limit transitions from Koornwinder polynomials to multivariable analogues of big and little q-Jacobi polynomials, which will be the main subject of the next chapter.

This chapter is started with a short review on the theory of one variable Askey-Wilson polynomials and one variable q-Racah polynomials.

2.2. One variable Askey-Wilson and q-Racah polynomials

Following the notations in [30] we define the basic hypergeometric series $_r\phi_s$ by

$$(2.2.1) \ _{r}\phi_{s}\left(\begin{array}{c}a_{1},\ldots,a_{r}\\b_{1},\ldots,b_{s}\end{array};q,z\right)=\sum_{m=0}^{\infty}\frac{(a_{1},\ldots,a_{r};q)_{m}}{(b_{1},\ldots,b_{s},q;q)_{m}}\left((-1)^{m}q^{\binom{m}{2}}\right)^{1+s-r}z^{m}$$

A set of orthogonal polynomials is called of basic hypergeometric type if the orthogonal polynomials can be expressed in terms of the basic hypergeometric series. Askey and Wilson [7] introduced a very general family of basic hypergeometric orthogonal polynomials depending on four parameters $\underline{t} = (t_0, t_1, t_2, t_3)$ which is nowadays known as the family of Askey-Wilson polynomials. In terms of the basic hypergeometric series (2.2.1) they are given by

$$P_n(z;\underline{t}) := \frac{(t_0t_1, t_0t_2, t_0t_3; q)_n}{t_0^n (t_0t_1t_2t_3q^{n-1}; q)_n} {}_4\phi_3 \begin{pmatrix} q^{-n}, q^{n-1}t_0t_1t_2t_3, t_0z, t_0z^{-1} \\ t_0t_1, t_0t_2, t_0t_3 \end{pmatrix}$$

for $n \in \mathbb{Z}_+$. Observe that $P_n(z)$ is a monic polynomial in $z + z^{-1}$ of degree n.

The orthogonality relations and norm evaluations for the monic Askey-Wilson polynomials can be stated as follows.

THEOREM 2.2.1. ([7, Theorem 2.3]) Assume that pairwise products of t_0 , t_1 , t_2 , t_3 as a multiset (so both t_0^2 and t_0t_1 are considered among the products) do not belong to

the set $\{q^{-j}\}_{j \in \mathbb{Z}_+}$. Then the monic Askey-Wilson polynomials satisfy the orthogonality relations

$$\frac{1}{2\pi i} \int_{z \in C} \left(P_m P_n \right)(z; \underline{t}) w_c(z; \underline{t}) \frac{dz}{z} = \delta_{m,n} \mathcal{N}(n; \underline{t})$$

with weight function

(2.2.3)
$$w_c(x;\underline{t}) := \frac{(x^2, x^{-2}; q)_{\infty}}{(t_0 x, t_0 x^{-1}, t_1 x, t_1 x^{-1}, t_2 x, t_2 x^{-1}, t_3 x, t_3 x^{-1}; q)_{\infty}}.$$

Here C is a positively oriented, continuous differentiable Jordan curve containing 0 and the four sequences $\{t_i q^j\}_{j \in \mathbb{Z}_+}$ $(i \in [0,3])$ and seperating them from $\{t_i^{-1}q^{-j}\}_{j \in \mathbb{Z}_+}$ $(i \in [0,3])$. The quadratic norms $\mathcal{N}(n)$ of the monic Askey-Wilson polynomials are explicitly given by

$$\mathcal{N}(n;\underline{t}) = \frac{2(q^{2n-1}t_0t_1t_2t_3, q^{2n}t_0t_1t_2t_3; q)_{\infty}}{(q^{n+1}, q^{n-1}t_0t_1t_2t_3, q^nt_0t_1, q^nt_0t_2, q^nt_0t_3, q^nt_1t_2, q^nt_1t_3, q^nt_2t_3; q)_{\infty}}$$

The theorem implies in particular that the monic Askey-Wilson polynomials $P_n(z; \underline{t})$ are symmetric with respect to permutations of the four variables \underline{t} (which cannot be immediately read off from the explicit expressions (2.2.2)). For the proof of the orthogonality relations and norm evaluations, Askey and Wilson [7] used the q-Pfaff-Saalschütz sum [7, (1.29)], [30, (II.12), p. 237] and the explicit evaluation of the integral over the weight function,

(2.2.4)
$$\frac{1}{2\pi i} \int_{z \in C} w_c(z;\underline{t}) \frac{dz}{z} = \frac{2(t_0 t_1 t_2 t_3; q)_{\infty}}{(q, t_0 t_1, t_0 t_2, t_0 t_3, t_1 t_2, t_1 t_3, t_2 t_3; q)_{\infty}}$$

(cf. [7, Theorem 2.1]). The integral (2.2.4) is a q-analogue of the classical beta integral and its evaluation is proved in [7] by summing up four sequences of residues by a summation formula of a very-well poised $_6\phi_5$ series [7, (2.2)], [30, (II.20), p.238] and subsequently summing the four remaining terms with the help of an elliptic function identity. More elementary proofs of (2.2.4) were obtained, for instance, in [103], [43] and [67].

A partly discrete orthogonality measure can be obtained by pulling C over some of the poles of w_c . The poles of w_c are simple for generic parameters \underline{t} and are given by the eight sequences $\{t_i q^j\}_{j \in \mathbb{Z}_+}, \{t_i^{-1} q^{-j}\}_{j \in \mathbb{Z}_+}$ $(i \in [0, 3])$. We write

(2.2.5)
$$w_d(eq^i; e) = w_d(eq^i; e; f, g, h) := \operatorname{res}_{z=eq^i}\left(\frac{w_c(z; \underline{t})}{z}\right)$$

for the residues, where f, g, h are such that $\{e, f, g, h\} = \{t_0, t_1, t_2, t_3\}$ (counted with multiplicity). When w_c has a simple pole in eq^i , then

(2.2.6)
$$\operatorname{res}_{z=e^{-1}q^{-i}}\left(\frac{w_c(z;\underline{t})}{z}\right) = -w_d(eq^i; e; f, g, h)$$

2. A RESIDUE CALCULUS FOR KOORNWINDER POLYNOMIALS

by the invariance of $w_c(z)$ under the inversion $z \mapsto z^{-1}$, and we have the explicit formula

(2.2.7)
$$w_{d}(eq^{i}; e; f, g, h) := \frac{(e^{-2}; q)_{\infty}}{(q, ef, f/e, eg, g/e, eh, h/e; q)_{\infty}} \cdot \frac{(e^{2}, ef, eg, eh; q)_{i}}{(q, qe/f, qe/g, qe/h; q)_{i}} \frac{(1 - e^{2}q^{2i})}{(1 - e^{2})} \left(\frac{q}{efgh}\right)^{i}$$

(see [7, Theorem 2.4] or [30, (7.5.22)] to avoid a small misprint).

The monic q-Racah polynomials depend (apart from q) on three continuous parameter and one discrete parameter $N \in \mathbb{N}$. They are the finite family

$$\{P_n(.;\underline{t}_N;q) \mid n \in [0,N]\}$$

of Askey-Wilson polynomials for the special parameters $\underline{t}_N := (t_0, t_1, t_2, t_0^{-1}q^{-N})$ where $N \in \mathbb{N}$. Since the parameters \underline{t}_N do no longer satisfy the assumptions of Theorem 2.2.1, it is convenient to think of the q-Racah polynomials as limit cases of the monic Askey-Wilson polynomials where the limit is given by sending t_3 to $t_0^{-1}q^{-N}$.

The orthogonality relations and norm evaluations for the monic q-Racah polynomials can be stated as follows.

THEOREM 2.2.2. ([6, Section 2]) Let $N \in \mathbb{N}$. For generic parameters t_0, t_1, t_2 we have the orthogonality relations

$$\sum_{i=0}^{N} (P_m P_n)(t_0 q^i; \underline{t}_N) \Delta^{qR}(t_0 q^i; \underline{t}_N) = \delta_{m,n} \mathcal{N}^{qR}(n; \underline{t}_N)$$

for $m, n \in \{0, ..., N\}$, with

$$\Delta^{qR}(t_0q^i;\underline{t}) := \frac{(1 - t_0^2q^{2i})(t_0^2, t_0t_1, t_0t_2, t_0t_3; q)_i}{(t_0t_1t_2t_3q^{-1})^i(1 - t_0^2)(q, qt_1^{-1}t_0, qt_2^{-1}t_0, qt_3^{-1}t_0; q)_i}$$

The quadratic norms of the monic q-Racah polynomials are explicitly given by

$$\mathcal{N}^{qR}(n;\underline{t}) := \frac{\left(q, t_0 t_1, t_0 t_2, t_0 t_3, t_1 t_2, t_1 t_3, t_2 t_3; q\right)_n}{\left(q^{n-1} t_0 t_1 t_2 t_3; q\right)_n \left(t_0 t_1 t_2 t_3; q\right)_{2n}} \frac{\left(t_1 / t_0, t_2 / t_0, t_3 / t_0, t_0 t_1 t_2 t_3; q\right)_{\infty}}{\left(t_1 t_2, t_1 t_3, t_2 t_3, t_0^{-2}; q\right)_{\infty}}$$

Askey and Wilson [6] obtained the orthogonality relations and norm evaluations for the *q*-Racah polynomials from a summation formula for very well poised terminating $_6\phi_5$ series [6, (2.3)], [30, (II.21), p.238] and the *q*-Pfaff-Saalschütz sum [6, (2.5)], [30, (II.12), p.237]. In particular, they obtained the summation formula

(2.2.8)
$$\sum_{i=0}^{N} \Delta^{qR}(t_0 q^i; \underline{t}_N) = \frac{\left(q t_0^2, q/t_1 t_2; q\right)_N}{\left(q t_0/t_1, q t_0/t_2; q\right)_N}$$

using a summation formula for very well poised terminating $_6\phi_5$ series [6, (2.3)], [30, (II.21), p.238].

It turn out that the orthogonality relations and norm evaluations for the q-Racah polynomials can be obtained by taking suitable limits in the orthogonality relations and norm

evaluations for the Askey-Wilson polynomials. The main trick here is to rescale a suitable partly discrete orthogonality measure for the Askey-Wilson polynomials such that certain common poles of the discrete weights become zeros for the continuous part of the orthogonality measure. These zeros cause the vanishing of the continuous part of the orthogonality measure in the limit $\underline{t} \rightarrow \underline{t}_N$ (see [122] for the details). In Section 2.5 this approach is used to generalize Theorem 2.2.2 to the multivariable setting.

2.3. Koornwinder polynomials for generic parameters values

In this section the multivariable analogue of Theorem 2.2.1 is given. We first need to introduce some notations and conventions which will be used throughout this chapter. The number $n \in \mathbb{N}$ denotes the number of independent variables of the multivariable polynomials under consideration. The *n* variables are denoted by $z = (z_1, \ldots, z_n)$. A function h(z) for which a definition is given with respect to the *n* variables $z = (z_1, \ldots, z_n)$, should be read with n = m if it follows from the context that $z \in \mathbb{C}^m$. If h(z) appears in formulas with $z \in \mathbb{C}^m$ and m = 0, then h(z) should be read as 1.

Let $\mathfrak{S} = \mathfrak{S}_n$ be the group of permutations of the set [1, n] and $\mathcal{W} = \mathfrak{S} \ltimes \{\pm 1\}^n$ the Weyl group of type BC_n . \mathcal{W} acts by permutations and inversions on the algebra $A := \mathbb{C}[z_1^{\pm 1}, \ldots, z_n^{\pm 1}]$. Denote $A^{\mathcal{W}}$ for the subalgebra of \mathcal{W} -invariant functions in A. A basis for $A^{\mathcal{W}}$ is given by the monomials $\{m_{\lambda} \mid \lambda \in \Lambda\}$, where $\Lambda := \{\lambda \in \mathbb{Z}_+^n \mid \lambda_1 \geq \ldots \geq \lambda_n\}$, and

$$m_\lambda(z) := \sum_{\mu \in \mathcal{W}\lambda} z^\mu$$

with $z^{\mu} = z_1^{\mu_1} \dots z_n^{\mu_n}$. The \mathcal{W} -orbit of $\lambda \in \Lambda \subset \mathbb{Z}^n$ is with respect to the natural action of \mathcal{W} on \mathbb{Z}^n .

Set $\underline{t} = (t_0, t_1, t_2, t_3)$ for the four tuple of parameters t_0, t_1, t_2, t_3 . We assume in this section that $\underline{t} \in V$, where $V \subset \mathbb{C}^4$ is the following parameter domain.

DEFINITION 2.3.1. Let V be the set of parameters $\underline{t} \in (\mathbb{C}^*)^4$ for which

 $#\{arg(t_i), arg(t_i^{-1}) \mid i \in [0,3]\} = 8$

and for which $t_0t_1t_2t_3 \notin \mathbb{R}_{\geq 1}$. Here $arg(u) \in [0, 2\pi)$ is the argument of $u \in \mathbb{C}^*$ and $\mathbb{R}_{>1} := \{r \in \mathbb{R} \mid r \geq 1\}.$

The measures which will be considered in this chapter will have their support on certain deformed *n*-tori $C^n \subset \mathbb{C}^n$, where $C \subset \mathbb{C}$ are the following type of deformations of the unit circle T.

DEFINITION 2.3.2. A continuous rectifiable Jordan curve $C = \phi_C([0,1]) \subset \mathbb{C}$ is called a deformed circle if C has a parametrization ϕ_C of the form

(2.3.1) $\phi_C(x) = r_C(x)e^{2\pi i x} \ (x \in [0,1]), \quad r_C : [0,1] \to (0,\infty)$

and if C is invariant under inversion, i.e. $C^{-1} := \{z^{-1} | z \in C\} = C$. For $\underline{t} \in V$, a deformed circle C is called a \underline{t} -contour if the four parameters t_0, t_1, t_2, t_3 are in the interior of C.

2. A RESIDUE CALCULUS FOR KOORNWINDER POLYNOMIALS

For a deformed circle $C = \phi_C([0, 1])$ the radial function r_C satisfies $r_C(1 - x) = (r_C(x))^{-1}$ since $C = C^{-1}$. Since a deformed circle C is by definition a closed contour, we furthermore have that $r_C(0) = r_C(1/2) = r_C(1) = 1$. Observe that the unit circle T is a deformed circle with $r_T \equiv 1$. If $\underline{t} \in V$ and C is a \underline{t} -contour, then the radial function r_C satisfies the extra conditions

$$r_C(\alpha_i^+) > |t_i|, \quad i \in [0,3],$$

where $\alpha_i = \alpha_i(\underline{t})$ is defined by

(2.3.2)
$$\alpha_i^{\pm} := \frac{\arg(t_i^{\pm 1})}{2\pi}, \quad i \in [0,3].$$

In particular, the unit circle T is a <u>t</u>-contour if $|t_i| < 1$ for all $i \in [0,3]$. Observe furthermore that $\alpha_i^{\pm} \neq 0, 1/2$ and that $\alpha_i^{-} = 1 - \alpha_i^{+}$ for all i.

The convention will be used that a deformed circle C is counterclockwise oriented (i.e. has the orientation induced from the parametrization ϕ_C) when we integrate over C.

Let $t \in (0, 1)$, $\underline{t} \in V$ and let C be a deformed circle such that $t_i q^j \notin C$ for $i \in [0, 3]$ and $j \in \mathbb{Z}_+$. Let $d\nu(z; \underline{t}; t)$ be the measure on C^n given by

(2.3.3)
$$d\nu(z;\underline{t};t) := \Delta(z;\underline{t};t) \frac{dz}{z}$$

for $z \in C^n$ with $\frac{dz}{z} := \frac{dz_1}{z_1} \dots \frac{dz_n}{z_n}$ and with weight function $\Delta(z; \underline{t}; t)$ given by

(2.3.4)
$$\Delta(z;\underline{t};t) = \left(\prod_{j=1}^{n} w_c(z_j;\underline{t})\right) \delta(z;t),$$

with $w_c(x;\underline{t})$ (2.2.3) the weight function for the continuous part of the orthogonality measure of the Askey-Wilson polynomials, and $\delta(z;t)$ given by

(2.3.5)
$$\delta(z;t) := \prod_{1 \le i < j \le n} \left(z_i z_j, z_i^{-1} z_j, z_i z_j^{-1}, z_i^{-1} z_j^{-1}; q \right)_{\tau}, \quad t = q^{\tau}.$$

The interaction factor $\delta(z; t)$ is only present in the multivariable setting (i.e. when n > 1). In particular the measure is independent of the deformation parameter t when n = 1, and coincides with the orthogonality measure of the Askey-Wilson polynomials as given in Theorem 2.2.1.

The measure $d\nu(.;\underline{t};t)$ on C^n is well defined, since the poles of $\Delta(.;\underline{t};t)$ do not intersect with the integration domain C^n . Indeed, the poles of the weight function $\Delta(z;\underline{t};t)$ lie on hyperplanes

(2.3.6)
$$z_i = t_m q^j$$
 or $z_i = t_m^{-1} q^{-j}$

with $m \in [0,3], j \in \mathbb{Z}_+$ and $i \in [1,n]$ (the poles coming from $w_c(z_i)$) and on hypersurfaces

with $1 \le k \ne l \le n$, $j \in \mathbb{Z}_+$ and $\varepsilon_k, \varepsilon_l \in \{-1, 1\}$ (the poles coming from $\delta(z; \underline{t})$). We have $z \notin C^n$ for a pole z of the form (2.3.6) since $C^{-1} = C$ and $t_i q^j \notin C$ for $i \in [0, 3]$ and $j \in \mathbb{Z}_+$. Similarly it follows from the definition of a deformed circle C that $z \notin C^n$ for a pole z of the form (2.3.7).

In this section we will study the orthogonal polynomials related to the complex measure $(C^n, d\nu(.; \underline{t}; t))$ where C is an arbitrary \underline{t} -contour. We first show that the measure $d\nu(.; \underline{t}; t)$ is independent of the \underline{t} -contour C when integrating against \mathcal{W} -invariant Laurent polynomials. In order to obtain this result, we consider specific subsets of $(\mathbb{C}^*)^n$ on which the interaction factor $\delta(.; t)$ is analytic.

Let C and C be deformed circles, with parametrization given by $\phi_C(x) = r_C(x)e^{2\pi ix}$ respectively $\phi_{\mathfrak{C}}(x) = r_{\mathfrak{C}}(x)e^{2\pi ix}$. Let $A^+(C, \mathfrak{C})$ be the open subset

(2.3.8)
$$A^+(C,\mathfrak{C}) := \{ x \in [0,1] \mid r_C(x) > r_\mathfrak{C}(x) \} \subset (0,1).$$

Set

(2.3.9)
$$\Omega(C, \mathfrak{C}) := \Omega^+(C, \mathfrak{C}) \cup \Omega^+(\mathfrak{C}, C) \cup C$$

where $\Omega^+(C, \mathfrak{C})$ is given by

(2.3.10)
$$\Omega^+(C,\mathfrak{C}) := \bigcup_{x \in A^+(C,\mathfrak{C})} \{ y(x) e^{2\pi i x} \, | \, r_C(x) \ge y(x) \ge r_{\mathfrak{C}}(x) \}.$$

The following properties of $\Omega(C, \mathfrak{C}) \subset \mathbb{C}^*$ follow easily from the definitions: (*i*) $\Omega(C, \mathfrak{C}) = \Omega(\mathfrak{C}, C)$.

We will use, in view of (i), the notation $\Omega = \Omega(C, \mathfrak{C})$ when it is clear from the context which pair of contours C, \mathfrak{C} is meant.

(*ii*) $\Omega^{-1} = \Omega$.

(*iii*) The contour C can de deformed homotopically to \mathfrak{C} within Ω .

We call $\Omega \subset \mathbb{C}^*$ the domain associated with the pair (C, \mathfrak{C}) . We write $\mathcal{O}_{\mathcal{W}}(\Omega^n)$ for the ring of \mathcal{W} -invariant functions f which are analytic on Ω^n . We have now the following crucial lemma.

LEMMA 2.3.3. Let $t \in (0,1)$ and let C, \mathfrak{C} be deformed circles satisfying the condition $t(r_C(x)) < r_{\mathfrak{C}}(x)$ for all $x \in A^+(C, \mathfrak{C})$. Then $\delta(.; t) \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$.

PROOF. Let C, \mathfrak{C} be deformed circles such that $t(r_C(x)) < r_{\mathfrak{C}}(x)$ for $x \in A^+(C, \mathfrak{C})$. Let $z \in (\mathbb{C}^*)^n$ such that $z_k^{\varepsilon_k} z_l^{\varepsilon_l} = t^{-1}q^{-j}$ for some $j \in \mathbb{Z}_+$, some $k \neq l$ and some $\varepsilon_k, \varepsilon_l \in \{\pm 1\}$. Write $\beta_k := \arg(z_k)/2\pi$ and $\beta_l := \arg(z_l)/2\pi$. For the proof of the lemma it suffices to show that either $z_k \notin \Omega$ or $z_l \notin \Omega$.

As an example, we check that either $z_k \notin \Omega$ or $z_l \notin \Omega$ when $\beta_k \in A^+(C, \mathfrak{C})$ and $z_k z_l = t^{-1}q^{-j}$ for some $j \in \mathbb{Z}_+$. Then $\beta_k = 1 - \beta_l$ and $\beta_l \in A^+(\mathfrak{C}, C)$, so in particular $r_C(\beta_k) = r_C(\beta_l)^{-1}$, $r_{\mathfrak{C}}(\beta_k) = r_{\mathfrak{C}}(\beta_l)^{-1}$. Suppose that $z_l \in \Omega$, then

$$|z_k| = t^{-1}q^{-j}|z_l^{-1}| \ge q^{-j}t^{-1}r_{\mathfrak{C}}(\beta_k) > q^{-j}r_C(\beta_k) \ge r_C(\beta_k),$$

hence $z_k \notin \Omega$. All the other cases are checked similarly.

LEMMA 2.3.4. Let $\underline{t} \in V$, $t \in (0, 1)$ and $f \in A^{\mathcal{W}}$. Then

(2.3.11)
$$\iint_{z \in C^n} f(z) d\nu(z; \underline{t}; t)$$

is independent of the choice of \underline{t} -contour C.

PROOF. With the shorthand notation $N_f(C)$ for the integral (2.3.11), we have to show that $N_f(C) = N_f(\mathfrak{C})$ for arbitrary pairs of <u>t</u>-contours (C, \mathfrak{C}) .

Let \mathcal{L} be the collection of pairs of \underline{t} -contours (C, \mathfrak{C}) for which $A^+(C, \mathfrak{C})$ is a finite disjoint union of open intervals and for which $t(r_C(x)) < r_{\mathfrak{C}}(x)$ for all $x \in A^+(C, \mathfrak{C})$. Fix a pair $(C, \mathfrak{C}) \in \mathcal{L}$ and let Ω be the associated domain. Since the four parameters t_0, t_1, t_2, t_3 are in the interior of C and \mathfrak{C} we have $w_c(.; \underline{t}) \in \mathcal{O}_{\{\pm 1\}}(\Omega)$, and by Lemma 2.3.3 we have $\delta(.; t) \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$. So Cauchy's Theorem implies that $N_f(C) = N_f(\mathfrak{C})$.

Suppose now that (C, \mathfrak{C}) is an arbitrary pair of \underline{t} -contours. Then there exists a finite sequence of \underline{t} -contours C_0, C_1, \ldots, C_s such that $C_0 = \mathfrak{C}, C_s = C$ and such that $(C_i, C_{i-1}) \in \mathcal{L}$ for all $i \in [1, s]$. It follows that $N_f(C) = N_f(\mathfrak{C})$. \Box

Define for parameters $\underline{t} \in V$ and $t \in (0, 1)$ a symmetric bilinear form $(., .)_{\underline{t}, t}$ on $A^{\mathcal{W}}$ by

(2.3.12)
$$(f,g)_{\underline{t},t} := \frac{1}{(2\pi i)^n} \iint_{z \in C^n} f(z)g(z)d\nu(z;\underline{t};t), \quad f,g \in A^{\mathcal{W}}$$

where C is a <u>t</u>-contour. The bilinear form (2.3.12) is independent of the choice of <u>t</u>contour C by Lemma 2.3.4. An important tool for studying orthogonal polynomials with respect to the bilinear form $(.,.)_{\underline{t},t}$ is an explicit second order q-difference operator $D = D_{\underline{t},t}$ which preserves the algebra $A^{\mathcal{W}}$ and which is symmetric with respect to $(.,.)_{\underline{t},t}$. The second order q-difference operator D was introduced by Koornwinder [65]. It is explicitly given by

(2.3.13)
$$D := \sum_{j=1}^{n} \left(\phi_j^+(z) (T_j^+ - \mathrm{Id}) + \phi_j^-(z) (T_j^- - \mathrm{Id}) \right),$$

where T_j^{\pm} is the $q^{\pm 1}$ -shift in the *j*th coordinate,

$$(T_j^{\pm}f)(z) := f(z_1, \ldots, z_{j-1}, q^{\pm 1}z_j, z_{j+1}, \ldots, z_n),$$

and the functions $\phi_j^+(z;\underline{t};t)$ and $\phi_j^-(z;\underline{t};t)$ are given by

$$\phi_j^+(z;\underline{t};t) := \frac{\prod_{i=0}^3 (1-t_i z_j)}{(1-z_j^2)(1-q z_j^2)} \prod_{l \neq j} \frac{(1-t z_l z_j) \left(1-t z_l^{-1} z_j\right)}{(1-z_l z_j) \left(1-z_l^{-1} z_j\right)},$$
$$\phi_j^-(z;\underline{t};t) := \phi_j^+(z^{-1};\underline{t};t),$$

2.3. KOORNWINDER POLYNOMIALS FOR GENERIC PARAMETERS VALUES

where $z^{-1} := (z_1^{-1}, \dots, z_n^{-1})$. The *BC* type dominance order on Λ is defined by

(2.3.14)
$$\mu \le \lambda \quad \Leftrightarrow \quad \sum_{j=1}^{i} \mu_j \le \sum_{j=1}^{i} \lambda_j \quad (i \in [1, n])$$

for $\lambda, \mu \in \Lambda$.

REMARK 2.3.5. Choose for the root system $R = R^+ \cup (-R^+)$ of type BC_n , the positive roots R^+ by

$$R^{+} = \{e_i\}_{i=1}^n \cup \{e_i \pm e_j\}_{1 \le i < j \le n} \cup \{2e_i\}_{i=1}^n,$$

with $\{e_i\}_{i=1}^n$ the standard orthonormal basis for \mathbb{R}^n , then Λ coincides with the set of dominant weights, and $\lambda > \mu$ for $\lambda, \mu \in \Lambda$ iff $\lambda - \mu$ is a sum of positive roots.

Koornwinder proved the following triangularity property of D (see [65, Lemma 5.2] and the remark after [121, Proposition 4.1]).

PROPOSITION 2.3.6. Let $\lambda \in \Lambda$. For arbitrary $\underline{t} \in \mathbb{C}^4$ and $t \in \mathbb{C}$ we have

$$(2.3.15) Dm_{\lambda} = \sum_{\mu \le \lambda} E_{\lambda,\mu} m_{\mu}$$

with $E_{\lambda,\mu} = E_{\lambda,\mu}(\underline{t};t) \in \mathbb{C}$ depending polynomially on the parameters \underline{t} and t. The leading term $E_{\lambda,\lambda}(\underline{t};t)$ will be denoted by $E_{\lambda}(\underline{t};t)$ and is given by

(2.3.16)
$$E_{\lambda}(\underline{t};t) := \sum_{j=1}^{n} \left(q^{-1} t_0 t_1 t_2 t_3 t^{2n-j-1} (q^{\lambda_j} - 1) + t^{j-1} (q^{-\lambda_j} - 1) \right)$$

In particular, D preserves the algebra $A^{\mathcal{W}}$. An other important property of D is the symmetry of D with respect to the bilinear form (.,.), i.e.

$$(2.3.17) \qquad \qquad \left(D_{\underline{t},t}f,g\right)_{\underline{t},t} = \left(f,D_{\underline{t},t}g\right)_{\underline{t},t}, \quad f,g \in A^{\mathcal{W}}$$

for parameters $\underline{t} \in V$ and $t \in (0, 1)$. Koornwinder [65, Lemma 5.3] proved (2.3.17) for parameters \underline{t} with $|t_i| < 1$ (then the unit circle T can be chosen as \underline{t} -contour). By Proposition 2.3.6, (2.3.17) follows for $\underline{t} \in V$ by analytic continuation.

Define now explicit expressions $\mathcal{N}(\lambda; \underline{t}; t)$ for $\lambda \in \Lambda$ by

(2.3.18)
$$\mathcal{N}(\lambda;\underline{t};t) := 2^n n! \mathcal{N}^+(\lambda;\underline{t};t) \mathcal{N}^-(\lambda;\underline{t};t)$$

where $\mathcal{N}^+(\lambda) := \mathcal{N}^+(\lambda; \underline{t}; t)$ is given by (2.3.19)

$$\mathcal{N}^{+}(\lambda) := \prod_{i=1}^{n} \frac{\left(q^{2\lambda_{i}-1}t^{2(n-i)}t_{0}t_{1}t_{2}t_{3};q\right)_{\infty}}{\left(q^{\lambda_{i}-1}t^{n-i}t_{0}t_{1}t_{2}t_{3},q^{\lambda_{i}}t^{n-i}t_{0}t_{1},q^{\lambda_{i}}t^{n-i}t_{0}t_{2},q^{\lambda_{i}}t^{n-i}t_{0}t_{3};q\right)_{\infty}} \\ \cdot \prod_{1 \le j < k \le n} \frac{\left(q^{\lambda_{j}+\lambda_{k}-1}t^{2n-j-k}t_{0}t_{1}t_{2}t_{3},q^{\lambda_{j}-\lambda_{k}}t^{k-j};q\right)_{\infty}}{\left(q^{\lambda_{j}+\lambda_{k}-1}t^{2n-j-k+1}t_{0}t_{1}t_{2}t_{3},q^{\lambda_{j}-\lambda_{k}}t^{k-j+1};q\right)_{\infty}},$$

2. A RESIDUE CALCULUS FOR KOORNWINDER POLYNOMIALS

and $\mathcal{N}^-(\lambda):=\mathcal{N}^-(\lambda;\underline{t};t)$ is given by

(2.3.20)
$$\mathcal{N}^{-}(\lambda) := \prod_{i=1}^{n} \frac{\left(q^{2\lambda_{i}}t^{2(n-i)}t_{0}t_{1}t_{2}t_{3};q\right)_{\infty}}{\left(q^{\lambda_{i}+1}t^{n-i}, q^{\lambda_{i}}t^{n-i}t_{1}t_{2}, q^{\lambda_{i}}t^{n-i}t_{1}t_{3}, q^{\lambda_{i}}t^{n-i}t_{2}t_{3};q\right)_{\infty}} \\ \cdot \prod_{1 \le j < k \le n} \frac{\left(q^{\lambda_{j}+\lambda_{k}}t^{2n-j-k}t_{0}t_{1}t_{2}t_{3}, q^{\lambda_{j}-\lambda_{k}+1}t^{k-j};q\right)_{\infty}}{\left(q^{\lambda_{j}+\lambda_{k}}t^{2n-j-k-1}t_{0}t_{1}t_{2}t_{3}, q^{\lambda_{j}-\lambda_{k}+1}t^{k-j-1};q\right)_{\infty}}$$

The following theorem extends the results of Koornwinder [65] (the orthogonality relations for the Koornwinder polynomials) and van Diejen in [17], Sahi [106] (the quadratic norm evaluations for the Koornwinder polynomials) to parameters $\underline{t} \in V$ (in [65], [17] and [106] these results were obtained for a parameter domain such that $|t_i| \leq 1$ for all *i*). See also [120].

THEOREM 2.3.7. For parameters $(\underline{t}, t) \in V \times (0, 1)$ there exists a unique linear basis $\{P_{\lambda}(:, \underline{t}; t)\}_{\lambda \in \Lambda}$ of $A^{\mathcal{W}}$ such that

(2.3.21)
$$P_{\lambda}(:;\underline{t};t) = m_{\lambda} + \sum_{\mu < \lambda} c_{\lambda,\mu}(\underline{t},t)m_{\mu}, \quad c_{\lambda,\mu}(\underline{t},t) \in \mathbb{C}$$

(2.3.22)
$$\left(P_{\lambda}(:;\underline{t};t), P_{\mu}(:;\underline{t};t) \right)_{t,t} = 0 \quad \text{if } \mu \neq \lambda$$

Furthermore, $P_{\lambda}(.;\underline{t};t)$ is an eigenfunction of $D_{\underline{t},t}$ with eigenvalue $E_{\lambda}(\underline{t};t)$ and we have the explicit evaluation formula

(2.3.23)
$$(P_{\lambda}(:;\underline{t};t), P_{\lambda}(:;\underline{t};t))_{t \ t} = \mathcal{N}(\lambda;\underline{t};t)$$

for the quadratic norms of the polynomials P_{λ} .

DEFINITION 2.3.8. The W-invariant Laurent polynomial $P_{\lambda}(.;\underline{t};t)$ is called the monic Koornwinder polynomial of degree $\lambda \in \Lambda$.

We end this section with a sketch of the proof for Theorem 2.3.7 using the techniques of Koornwinder [65] and van Diejen [17]. For more details, we refer the reader to these two papers.

Fix arbitrary $0 \neq \nu \in \Lambda$. It is sufficient to prove the existence and uniqueness of a set of \mathcal{W} -invariant Laurent polynomials $\{P_{\lambda}(.;\underline{t};t)\}_{\lambda \leq \nu}$ satisfying (2.3.21) and (2.3.22) for $\lambda, \mu \leq \nu$ and to prove the remaining assertions of the theorem for the polynomials $\{P_{\lambda}(.;\underline{t};t)\}_{\lambda < \nu}$.

We first define the polynomials $\{P_{\lambda}(.;\underline{t};t)\}_{\lambda \leq \nu}$ for a dense parameter domain $U_{\nu} \subset V \times (0,1)$. The subset U_{ν} is by definition the set of parameters $(\underline{t},t) \in V \times (0,1)$ such that $E_{\mu}(\underline{t};t) \neq E_{\lambda}(\underline{t};t)$ for all $\lambda, \mu \leq \nu, \lambda \neq \mu$. Observe that $U_{\nu} \subset V \times (0,1)$ is open and dense since the eigenvalues $\{E_{\lambda}(\underline{t};t)\}_{\lambda \in \Lambda}$ are mutually different as polynomials in the parameters \underline{t}, t .

The polynomials $P_{\lambda}(.;\underline{t};t) \in A^{\mathcal{W}}$ for $(\underline{t},t) \in U_{\nu}$ and $\lambda \leq \nu$ are defined by

(2.3.24)
$$P_{\lambda}(.;\underline{t};t) := \left(\prod_{\mu < \lambda} \frac{D_{\underline{t},t} - E_{\mu}(\underline{t};t)}{E_{\lambda}(\underline{t};t) - E_{\mu}(\underline{t};t)}\right) m_{\lambda}$$

(cf. [82], [121]). It is an easy consequence of the triangularity of D (cf. Proposition 2.3.6) and of the Cayley-Hamilton Theorem that $P_{\lambda}(.;\underline{t};t)$ (2.3.24) is the unique function of the form (2.3.21) which is an eigenfunction of $D_{\underline{t},t}$ with eigenvalue $E_{\lambda}(\underline{t};t)$. The polynomials $\{P_{\lambda}(.;\underline{t};t)\}_{\lambda \leq \nu}$ (2.3.24) satisfy the orthogonality relations (2.3.22) for parameter values $(\underline{t},t) \in U_{\nu}$ since D is symmetric with respect to (.,.) and $E_{\mu}(\underline{t};t) \neq E_{\mu'}(\underline{t};t)$ for $\mu, \mu' \leq \lambda, \mu \neq \mu'$.

Next we establish the quadratic norm evaluations (2.3.23) for $\{P_{\lambda}(.;\underline{t};t)\}_{\lambda \leq \nu}$ with $(\underline{t},t) \in U_{\nu}$. In the special case $\lambda = 0$, (2.3.23) reduces to

$$(1,1)_{\underline{t},t} = \mathcal{N}(0;\underline{t};t)$$

(2.3.25)
$$= 2^{n} n! \prod_{i=1}^{n} \frac{\left(t, t^{2n-i-1} t_0 t_1 t_2 t_3; q\right)_{\infty}}{\left(q, t^{n-i+1}; q\right)_{\infty} \prod_{0 \le j < k \le 3} \left(t^{n-i} t_j t_k; q\right)_{\infty}}$$

which was proved by Gustafson [31] for parameters $t \in (0, 1)$ and $\underline{t} \in \mathbb{C}^4$ with $|t_i| < 1$ (since then the torus T can be chosen as \underline{t} -contour). The second equality follows by a straightforward computation (see also [83]). By analytic continuation, (2.3.25) is valid for parameters $t \in (0, 1)$ and $\underline{t} \in V$.

For general λ , van Diejen [17] proved explicit Pieri formulas for the renormalized Koornwinder polynomials

(2.3.26)
$$p_{\lambda}(:;\underline{t};t) := c(\lambda;\underline{t};t)P_{\lambda}(:;\underline{t};t), \quad c(\lambda;\underline{t};t) := \frac{\mathcal{N}^{+}(0)}{\mathcal{N}^{+}(\lambda)}\prod_{j=1}^{n}(t_{0}t^{n-j})^{\lambda_{j}}.$$

The renormalization constant $c(\lambda; \underline{t}; t)$ is a rational expression in the parameters \underline{t}, t . The Pieri formulas give explicit expressions for the coefficients $d_{\lambda}^{(r)}(\mu; \underline{t}; t)$ in the expansions

(2.3.27)
$$E_r(z;\underline{t};t)p_{\lambda}(z;\underline{t};t) = \sum_{\mu \le \lambda + (1^n)} d_{\lambda}^{(r)}(\mu;\underline{t};t)p_{\mu}(z;\underline{t};t), \quad (r \in [1,n])$$

where $\{E_r(z;\underline{t};t)\}_{r=1}^n$ are explicit algebraic generators of the algebra of \mathcal{W} -invariant Laurent polynomials $A^{\mathcal{W}}$ and where $(1^n) := (1, \ldots, 1) \in \Lambda$ is the *n*th fundamental weight (see [17] for the explicit formulas, or [20, Appendix B] where the notations are closer to the ones used in this Chapter).

In [17] the proof of the Pieri formulas was derived for a four parameter subfamily [17]. It turns out that Cherednik's affine Hecke-algebraic approach to the study of basic hypergeometric orthogonal polynomials related to root systems (cf. [12], [13], [14]) can in fact be worked out for the complete five parameter family of Koornwinder polynomials. This was worked out in detail by Noumi [90], Sahi [106], see also [120]. The results of Sahi [106] imply that van Diejen's Pieri formulas are valid for the full five parameter family of Koornwinder polynomials. In particular, (2.3.27) may be viewed as an identity

in the algebra of W-invariant Laurent polynomials over the quotient field $\mathbb{C}(\underline{t}, t)$, where \underline{t}, t are considered as indeterminates.

The Pieri formulas and the orthogonality relations for the renormalized Koornwinder polynomials with real parameters t_i and $|t_i| < 1$ allowed van Diejen (cf. [17, Theorem 4]) to reduce the norm computation for arbitrary λ to the case $\lambda = 0 \in \Lambda$. Gustafson's evaluation (2.3.25) then completes the evaluation for general λ . By taking a dense subset of the parameter domain U_{ν} if necessary, exactly the same reduction can be done for the norm evaluations of the polynomials $\{p_{\lambda}(.;\underline{t};t)\}_{\lambda \leq \nu}$ for parameters $(\underline{t},t) \in U_{\nu}$. The extension of Gustafson's result (2.3.25) then completes the proof of (2.3.23) for $\{P_{\lambda}(.;\underline{t};t)\}_{\lambda \leq \nu}$ and $(\underline{t},t) \in U_{\nu}$.

The polynomials $\{P_{\lambda}(.;\underline{t};t)\}_{\lambda \leq \nu}$ with parameter values $(\underline{t},t) \in U_{\nu}$ are uniquely characterized by (2.3.21) and (2.3.22) for $\lambda, \mu \leq \nu$, since their quadratic norms (2.3.23) are non-zero. Indeed, the functions

$$(2.3.28) \qquad (\underline{t}, t) \mapsto \mathcal{N}^{\pm}(\lambda; \underline{t}; t) : \quad V \times (0, 1) \to \mathbb{C}$$

are well defined, continuous functions which do not have zeros on the domain $V \times (0, 1)$. This is immediately clear except for $\mathcal{N}^+(\lambda)$ with $\lambda \in \Lambda$ and $\lambda_n = 0$. But then the expression for $\mathcal{N}^+(\lambda)$ can be simplified, similarly as the simplification of the expression for $\mathcal{N}(0)$ in (2.3.25), from which it follows that $\mathcal{N}^+(\lambda; \underline{t}; t)$ is a well defined, continuous function of $(\underline{t}, t) \in V \times (0, 1)$ without zeros.

The proof of the theorem can now be finished by extending these results to parameter values $(\underline{t}, t) \in V \times (0, 1)$ using a continuity argument, as follows. The Koornwinder polynomial P_{λ} satisfies the following Gram-Schmidt formula,

(2.3.29)
$$P_{\lambda}(z;\underline{t};t) = m_{\lambda}(z) - \sum_{\mu < \lambda} \frac{\left(m_{\lambda}, P_{\mu}(.;\underline{t};t)\right)_{\underline{t},t}}{\mathcal{N}(\mu;\underline{t};t)} P_{\mu}(z;\underline{t};t)$$

for $(\underline{t}, t) \in U_{\nu}$ and $\lambda \leq \nu$. By induction, it follows from (2.3.29) that the coefficients $c_{\lambda,\mu}: U_{\nu} \to \mathbb{C}$ in (2.3.21) uniquely extend to continuous functions $c_{\lambda,\mu}: V \times (0,1) \to \mathbb{C}$ for all $\mu < \lambda \leq \nu$. Hence existence and uniqueness of $\{P_{\lambda}(.;\underline{t};t)\}_{\lambda \leq \nu}$ as well as the other assertions follow now by continuity for all $(\underline{t}, t) \in V \times (0, 1)$. This completes the proof of the theorem.

REMARK 2.3.9. For n = 1 the polynomials $\{P_{\lambda}(z; \underline{t}) \mid \lambda \in \mathbb{Z}_+\}$ are independent of t and are the monic Askey-Wilson polynomials (2.2.2) as defined in Section 2.2. Theorem 2.3.7 reduces to the orthogonality relation and quadratic norm evaluation stated in Theorem 2.2.1. The renormalized Askey-Wilson polynomial $p_{\lambda}(z; \underline{t})$ (2.3.26) is then exactly the $_4\phi_3$ part of (2.2.1).

REMARK 2.3.10. The renormalization constant $c(\lambda; \underline{t}; t)$ (2.3.26) is easily seen to be regular and non-zero at $(\underline{t}, t) \in V \times (0, 1)$ for all $\lambda \in \Lambda$. Hence the renormalized Koornwinder polynomials $\{p_{\lambda}(z; \underline{t}; t)\}_{\lambda \in \Lambda}$ form an orthogonal basis of $A^{\mathcal{W}}$ with respect to the bilinear form $(., .)_{t,t}$ for all parameter values $(\underline{t}, t) \in V \times (0, 1)$.
REMARK 2.3.11. Several elementary properties of the Koornwinder polynomials can be deduced using the fact that for generic parameters, $P_{\lambda}(.;\underline{t};t)$ is the unique function of the form (2.3.21) which is an eigenfunction of $D_{\underline{t},t}$ with eigenvalue $E_{\lambda}(\underline{t};t)$. For instance, it follows that the Koornwinder polynomial $P_{\lambda}(.;\underline{t};t)$ is symmetric in the four parameters \underline{t} , and that $P_{\lambda}(z;-\underline{t};t) = (-1)^{|\lambda|}P_{\lambda}(-z;\underline{t};t)$ where $|\lambda| := \sum_{i=1}^{n} \lambda_i$, $-z := (-z_1, \ldots, -z_n)$ and, similarly, $-\underline{t} = (-t_0, -t_1, -t_2, -t_3)$.

2.4. Residue calculus for the orthogonality measure $d\nu$

In this section a residue calculus is developed for integrals of the form

(2.4.1)
$$\frac{1}{(2\pi i)^n} \iint_{z \in C^n} f(z) d\nu(z) = \frac{1}{(2\pi i)^n} \iint_{z \in C^n} f(z) \Delta(z) \frac{dz}{z}, \quad f \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$$

when C^n is shifted to \mathfrak{C}^n , where $\Omega \subset \mathbb{C}^*$ is the domain associated with the pair (C, \mathfrak{C}) and (C, \mathfrak{C}) is a so-called (n, t_0) -residue pair, which is defined as follows.

DEFINITION 2.4.1. Let $\underline{t} = (t_0, t_1, t_2, t_3) \in V$ (V as given in Definition 2.3.1). A pair of contours (C, \mathfrak{C}) is called a (n, t_0) -residue pair if C and \mathfrak{C} are deformed circles satisfying the following three properties.

(i) The subset $A^+(C, \mathfrak{C})$ (2.3.8) is an open interval for which $\alpha_0^+ \in A^+(C, \mathfrak{C})$ but $\alpha_i^{\pm} \notin A^+(C, \mathfrak{C})$ (i = 1, 2, 3);

(ii) $t_i q^r \notin C \cup \mathfrak{C}$ for $r \in \mathbb{Z}_+$ and $i \in [0,3]$; (iii) $t_0 t^p q^r \notin \mathfrak{C}$ for $p \in [-1, n-1]$ and $r \in \mathbb{Z}$.

The poles which are picked up when deforming C^n to \mathfrak{C}^n in (2.4.1) for a (n, t_0) residue pair (C, \mathfrak{C}) will only depend on q, t and t_0 . We therefore fix in this section $t \in$ (0,1) and $t_1, t_2, t_3 \in \mathbb{C}^*$ such that $\#\{\alpha_i^+, \alpha_i^- | i = 1, 2, 3\} = 6$ (α_i^{\pm} given by (2.3.2)) and simplify the notations by omitting the dependance on these parameters. For instance, we will write $w_c(x; t_0)$ instead of $w_c(x; \underline{t}), w_d(x; t_0)$ instead of $w_d(x; t_0; t_1, t_2, t_3)$, etc.

Observe that due to the symmetry of $\Delta(z; \underline{t}; t)$ in the four parameters t_0, t_1, t_2 and t_3 , all the results on the residue calculus for (n, t_0) -residue pairs can be reformulated for (n, t_i) -residue pairs with $i \in [0, 3]$ arbitrary by relabeling the parameters \underline{t} .

The measure which is obtained after deforming the contour C^n in (2.4.1) to \mathbb{C}^n for a (n, t_0) -residue pair (C, \mathbb{C}) is at first site rather complicated. We will therefore first give the result for the special case $t = q^k$ ($k \in \mathbb{N}$), in which case the answer as well as the proof is much simpler. The rather dramatic simplification of the proof is a consequence of the fact that the interaction factor $\delta(.; q^k)$ is analytic on $(\mathbb{C}^*)^n$, which is certainly not the case for general t. Despite this simplification, the answer has the same form as in the general case.

LEMMA 2.4.2. Let $\underline{t} \in V$, $t = q^k$ with $k \in \mathbb{N}$ and let (C, \mathfrak{C}) be an (n, t_0) -residue pair. Let M be the smallest positive integer such that $|t_0q^M| < r_C(\alpha_0^+)$ and let N be the largest positive integer such that $|t_0q^N| > r_{\mathfrak{C}}(\alpha_0^+)$. Then

(2.4.2)
$$\frac{1}{(2\pi i)^n} \iint_{z \in C^n} f(z) d\nu(z) = \sum_{r=0}^n \frac{2^r \binom{n}{r}}{(2\pi i)^{n-r}} \sum_{\substack{z_i \in \{t_0 q^l\}_{l=M}^N \\ i=1,\dots,r}} \iint_{\substack{(z_{r+1},\dots,z_n) \in \mathfrak{C}^{n-r} \\ i=1,\dots,r}} f(z)$$
$$\cdot \delta(z;q^k) \prod_{i=1}^r w_d(z_i;t_0) \prod_{j=r+1}^n w_c(z_j) \frac{dz_j}{z_j}$$

for $f \in A^{\mathcal{W}}$.

PROOF. Write $I_n(f)$ for the left hand side of (2.4.2) and $\tilde{I}_n(f)$ for the right hand side of (2.4.2). We prove $I_n(f) = \tilde{I}_n(f)$ for $f \in A^{\mathcal{W}}$ by induction on the number of variables n. The equality $I_1(f) = \tilde{I}_1(f)$ for $f \in \mathbb{C}[z+z^{-1}]$ is well-known (see [7], or Section 2.2). For n > 1, we obtain from Cauchy's Theorem and the fact that $\delta(.; q^k) \in \mathcal{O}_{\mathcal{W}}((\mathbb{C}^*)^n)$,

(2.4.3)

$$I_n(f) = \frac{1}{2\pi i} \int_{z_1 \in \mathfrak{C}} w_c(z_1) I_{n-1}(f_{z_1}) \frac{dz_1}{z_1} + 2 \sum_{z_1 = \{t_0 q^i\}_{i=M}^N} w_d(z_1; t_0) I_{n-1}(f_{z_1})$$

for $f \in A^{\mathcal{W}}$, where $f_x(w)$ $(x \in \mathbb{C}^*)$ is the \mathcal{W} -invariant Laurent polynomial in n-1 variables $w = (w_1, \ldots, w_{n-1})$ given by

$$f_x(w) := f(x,w) \prod_{i=1}^{n-1} \left(xw_i, xw_i^{-1}, x^{-1}w_i, x^{-1}w_i^{-1}; q \right)_k.$$

By the induction hypotheses we may replace $I_{n-1}(f_x)$ by $\tilde{I}_{n-1}(f_x)$ in (2.4.3). Then the equality $I_n(f) = \tilde{I}_n(f)$ follows from the \mathcal{W} -invariance of f and $\delta(.;q^k)$, and from the formula

$$\frac{1}{2\pi i} \frac{2^r \binom{n-1}{r}}{(2\pi i)^{n-1-r}} + 2\frac{2^{r-1} \binom{n-1}{r-1}}{(2\pi i)^{n-r}} = \frac{2^r \binom{n}{r}}{(2\pi i)^{n-r}} \quad (r \in [0,n]),$$

where we use the convention that $\binom{m}{r} = 0$ if r < 0 and if r > m.

Since $\delta(z;q^k) = 0$ if $z_i = q^l z_j$ for some $i \neq j$ and some $l \in [0, k-1]$ and using S-symmetry properties of the weights in the right hand side of (2.4.2), it follows that the discrete parts are actually supported on points of the form

$$(t_0q^{\lambda_1}, t_0tq^{\lambda_2}, \ldots, t_0t^{r-1}q^{\lambda_r})$$

with $\lambda_1 \leq \lambda_2 \leq \ldots \leq \lambda_r$ positive integers and $t = q^k$. This turn out to be also the case for arbitrary $t \in (0, 1)$, as will be shown in a moment. We first need to introduce some notations. Set

(2.4.4)
$$P(r) := \{\lambda \in \mathbb{Z}_+^r \mid \lambda_1 \le \lambda_2 \le \ldots \le \lambda_r\}$$

and set $\lambda_0 := 0$ for arbitrary $\lambda \in P(r)$. We write $\rho_i = \rho_i(t_0; t) := t_0 t^{i-1}$ for $i \in \mathbb{Z}$ and, for $\lambda \in P(r)$,

(2.4.5)
$$\rho q^{\lambda} := (\rho_1 q^{\lambda_1}, \rho_2 q^{\lambda_2}, \dots, \rho_r q^{\lambda_r}) = (t_0 q^{\lambda_1}, t_0 t q^{\lambda_2}, \dots, t_0 t^{r-1} q^{\lambda_r}).$$

Define $D(r) = D(r; C, \mathfrak{C}; t_0)$ for $r \in [1, n]$ by

(2.4.6)
$$D(r) := \{ \rho q^{\lambda} \mid \lambda \in P(r) \text{ and } r_C(\alpha_0^+) > |\rho_i q^{\lambda_i}| > r_{\mathfrak{C}}(\alpha_0^+) \quad (i \in [1, r]) \}.$$

Observe that for $\omega \in D(r)$ we have $\omega_i \in int(\Omega)$ for all *i*, where $int(\Omega)$ is the interior of Ω . For $\rho q^{\lambda} \in D(r)$, we set

(2.4.7)
$$\Delta^{(d)}(\rho q^{\lambda}; t_0) := \left(\prod_{j=1}^r w_d(\rho_j q^{\lambda_j}; \rho_j q^{\lambda_{j-1}})\right) \delta_d(\rho q^{\lambda})$$

where w_d is given by (2.2.5), (2.2.7), and with interaction factor

$$(2.4.8) \quad \delta_d(\rho q^{\lambda}) := \prod_{1 \le k < l \le r} \frac{\left(\rho_k^{-1} \rho_l q^{\lambda_l - \lambda_k}, \rho_k^{-1} \rho_l^{-1} q^{-\lambda_k - \lambda_l}; q\right)_{\tau}}{\left(\rho_k \rho_l q^{\lambda_{k-1} + \lambda_l}, \rho_k \rho_l^{-1} q^{\lambda_{k-1} - \lambda_l}; q\right)_{\lambda_k - \lambda_{k-1}}} \quad (t = q^{\tau}).$$

Observe that the weight function $w_c(x; \rho_j q^{\lambda_{j-1}})/x$ has a simple pole at $x = \rho_j q^{\lambda_j}$ since $\lambda_j - \lambda_{j-1} \in \mathbb{Z}_+$ and $\underline{t} \in V$; hence the factors $w_d(\rho_j q^{\lambda_j}; \rho_j q^{\lambda_{j-1}})$ in (2.4.7) are non-zero. The discrete parts of the measure which will appear by deforming C^n to \mathfrak{C}^n in (2.4.1) will involve the weights $\Delta^{(d)}$. Observe furthermore that for r = 1 and $\rho q^{\lambda} = t_0 q^i \in D(1)$, we have $\Delta^{(d)}(t_0q^i; t_0) = w_d(t_0q^i; t_0)$. For $\rho q^{\lambda} \in D(r)$ and $z \in \mathfrak{C}^{n-r}$, set

(2.4.9)
$$d\nu_r(\rho q^{\lambda}, z; t_0) = \Delta_r(\rho q^{\lambda}, z; t_0) \frac{dz}{z}$$

with weight function $\Delta_r(\rho q^{\lambda}, z; t_0)$ given by

(2.4.10)
$$\Delta_r(\rho q^{\lambda}, z; t_0) := \Delta^{(d)} \left(\rho q^{\lambda}; t_0\right) \Delta(z; t_0) \delta_c \left(\rho q^{\lambda}; z\right),$$

where $\Delta(z; t_0)$ is the weight function (2.3.4) defined with respect to the variables $z = (z_1, \ldots, z_{n-r})$ and where $\delta_c(\rho q^{\lambda}; z)$ is an interaction factor given by

$$(2.4.11) \quad \delta_c(\rho q^{\lambda}; z) := \prod_{\substack{1 \le k \le r \\ 1 \le l \le n-r}} \left(\rho_k q^{\lambda_k} z_l, \rho_k q^{\lambda_k} z_l^{-1}, \rho_k^{-1} q^{-\lambda_k} z_l, \rho_k^{-1} q^{-\lambda_k} z_l^{-1}; q \right)_{\tau}$$

with $t = q^{\tau}$. In particular, $\Delta_n(\rho q^{\lambda}; t_0) = \Delta^{(d)}(\rho q^{\lambda}; t_0)$ for $\rho q^{\lambda} \in D(n)$. The measure $d\nu_r(\rho q^{\lambda}, z; t_0)$ is well defined on $D(r) \times \mathfrak{C}^{n-r}$ since the denominator of $\Delta_r(\rho q^{\lambda}, z; t_0)$ is non-zero by properties (ii) and (iii) of the (n, t_0) -residue pair (C, \mathfrak{C}) (Definition 2.4.1). We call $d\nu_r$ the *r*th measure associated with the (n, t_0) -residue pair (C, \mathfrak{C}) .

2. A RESIDUE CALCULUS FOR KOORNWINDER POLYNOMIALS

PROPOSITION 2.4.3. Let (C, \mathfrak{C}) be a (n, t_0) -residue pair and let $\Omega = \Omega(C, \mathfrak{C})$ be the associated domain. Let $d\nu_r$ be the rth measure associated with (C, \mathfrak{C}) , then

(2.4.12)
$$\frac{\frac{1}{(2\pi i)^n} \iint\limits_{z \in C^n} f(z) d\nu(z) = \frac{1}{(2\pi i)^n} \iint\limits_{z \in \mathfrak{C}^n} f(z) d\nu(z) + \sum_{r=1}^n \frac{2^r (n-r+1)_r}{(2\pi i)^{n-r}} \sum_{\omega \in D(r)} \iint\limits_{z \in \mathfrak{C}^{n-r}} f(\omega, z) d\nu_r(\omega, z)$$

for $f \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$, where $(u)_r := \prod_{i=0}^{r-1} (u+i)$ is the shifted factorial.

REMARK 2.4.4. It is easy to show that Proposition 2.4.3 reduces to the statement in Lemma 2.4.2 for $t = q^k$ ($k \in \mathbb{N}$) using the fact that $\delta(z; q^k) = 0$ if $z_i = q^l z_j$ for some $i \neq j$ and some $l \in [0, k - 1]$ and the fact that

$$w_d(x;e_iq^l) = (e_ix, e_ix^{-1};q)_l w_d(x;e_i), \quad (x = e_iq^{l+m}, m \in \mathbb{Z}_+).$$

In the next lemma the proof of Proposition 2.4.3 is given for (n, t_0) -residue pairs (C, \mathfrak{C}) such that the interaction factor $\delta(.; t)$ is analytic on $(\Omega(C, \mathfrak{C}))^n$.

LEMMA 2.4.5. Suppose that (C, \mathfrak{C}) is a (n, t_0) -residue pair such that

(2.4.13)
$$t(r_C(x)) < r_{\mathfrak{C}}(x), \qquad \forall x \in A^+(C, \mathfrak{C})$$

Then (2,4.12) *is valid.*

PROOF. Fix a (n, t_0) -residue pairs (C, \mathfrak{C}) satisfying (2.4.13). We will prove by induction on $l \in [0, n]$ that

(2.4.14)
$$\frac{1}{(2\pi i)^n} \iint_{z \in C^l \times \mathfrak{C}^{n-l}} f(z)\Delta(z;t_0) \frac{dz}{z} = \frac{1}{(2\pi i)^n} \iint_{z \in \mathfrak{C}^n} f(z)\Delta(z;t_0) \frac{dz}{z} + \frac{2l}{(2\pi i)^{n-1}} \sum_{i \in I} \iint_{z \in \mathfrak{C}^{n-1}} f(t_0q^i,z)\Delta_1(t_0q^i,z;t_0) \frac{dz}{z}$$

for $f \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$, where

(2.4.15)
$$I = \{ i \in \mathbb{Z}_+ | r_C(\alpha_0^+) > | t_0 q^i | > r_{\mathfrak{C}}(\alpha_0^+) \}.$$

Then (2.4.12) is the special case l = n in (2.4.14), since $D(r) = \emptyset$ for r > 1 by (2.4.13). For l = 0, (2.4.14) is trivial. Let $l \in [1, n]$. Since $\delta(.; t) \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$ by Lemma 2.3.3, we can shift C to \mathfrak{C} for the first variable z_1 in the left hand side of (2.4.14) and we obtain by (2.2.5), by the \mathcal{W} -invariance of $\Delta(z)$ and by Cauchy's Theorem

$$(2.4.16) \quad \frac{1}{(2\pi i)^n} \iint_{z \in C^l \times \mathfrak{C}^{n-l}} f(z)\Delta(z;t_0) \frac{dz}{z} = \frac{1}{(2\pi i)^n} \iint_{z \in C^{l-1} \times \mathfrak{C}^{n-l+1}} f(z)\Delta(z;t_0) \frac{dz}{z} + \frac{2}{(2\pi i)^{n-1}} \sum_{i \in I} \iint_{z \in C^{l-1} \times \mathfrak{C}^{n-l}} f(t_0q^i,z)\Delta_1(t_0q^i,z;t_0) \frac{dz}{z}$$

for $f \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$. Here we have used that the residue at $z_1 = t_0^{-1}q^{-i}$ $(i \in I)$ of the function $\Delta(z_1, z'; t_0)/z_1$ is equal to $-\Delta_1(t_0q^i, z'; t_0)$ in view of (2.2.6). The weight function $\Delta_1(t_0q^i, z; t_0)$ in (2.4.16) can be rewritten as

(2.4.17)
$$\Delta_1(t_0q^i, z; t_0) = h(t_0q^i, z; t_0)\Delta(z; tt_0q^i)$$

with

$$(2.4.18) heta(t_0q^i,z;t_0) = w_d(t_0q^i;t_0) \prod_{s=1}^{n-1} \frac{\left(t_0^{-1}q^{-i}z_s,t_0^{-1}q^{-i}z_s^{-1};q\right)_{\tau}}{\left(t_0z_s,t_0z_s^{-1};q\right)_i} \quad (t=q^{\tau}).$$

Formula (2.4.17) follows by interchanging the factor $(t_0z_s, t_0z_s^{-1}; q)_{\infty}$ in the denominator of the weight $w_c(z_s; t_0)$ with the factor $(tt_0q^iz_s, tt_0q^iz_s^{-1}; q)_{\infty}$ in the denominator of the interaction factor $\delta_c(t_0q^i; z)$ for $s \in [1, n-1]$. We have $\Delta(.; tt_0q^i) \in \mathcal{O}_{\mathcal{W}}(\Omega^{n-1})$ for $i \in I$ by (2.4.13) and Lemma 2.3.3. We claim that $h(t_0q^i, .; t_0) \in \mathcal{O}_{\mathcal{W}}(\Omega^{n-1})$ for $i \in I$. Indeed, it is sufficient to check that the map

(2.4.19)
$$x \mapsto \frac{\left(t_0^{-1}q^{-i}x, t_0^{-1}q^{-i}x^{-1}; q\right)_{\infty}}{\left(t_0x, t_0x^{-1}; q\right)_i \left(tt_0^{-1}q^{-i}x, tt_0^{-1}q^{-i}x^{-1}; q\right)_{\infty}}$$

is analytic on Ω when $i \in I$. The zeros of the factor $(t_0x, t_0x^{-1}; q)_i$ in the denominator are compensated by zeros in the numerator. Next, we check that $(tt_0^{-1}q^{-i}x; q)_{\infty}$ is nonzero for $x \in \Omega$ and $i \in I$. Now $(tt_0^{-1}q^{-i}x; q)_{\infty} = 0$ iff $x = t^{-1}q^{i-m}t_0$ for some $m \in \mathbb{Z}_+$. In particular, we must have $\arg(x) = \arg(t_0) = 2\pi\alpha_0^+$. Since $i \in I$, we have for $m \in \mathbb{Z}_+$,

$$|t^{-1}q^{i-m}t_0| > t^{-1}r_{\mathfrak{C}}(\alpha_0^+)q^{-m} \ge t^{-1}r_{\mathfrak{C}}(\alpha_0^+) > r_C(\alpha_0^+)$$

where the last inequality is obtained from the extra condition (2.4.13). Since $x \in \Omega$ with $\arg(x) = 2\pi\alpha_0^+$ implies that $r_{\mathfrak{C}}(\alpha_0^+) \leq |x| \leq r_C(\alpha_0^+)$, it follows that $(tt_0^{-1}q^{-i}x;q)_{\infty} \neq 0$ for $x \in \Omega$ and $i \in I$. Since $\Omega^{-1} = \Omega$, we then also have $(tt_0^{-1}q^{-i}x^{-1};q)_{\infty} \neq 0$ for $x \in \Omega$ and $i \in I$. Thus the map given by (2.4.19) is analytic on Ω if $i \in I$. In particular,

$$f(t_0q^i,.)\Delta_1(t_0q^i,.;t_0) \in \mathcal{O}_{\mathcal{W}}(\Omega^{n-1})$$

for $i \in I$, so we obtain by Cauchy's Theorem and (2.4.16),

(2.4.20)
$$\frac{1}{(2\pi i)^n} \iint_{z \in C^l \times \mathfrak{C}^{n-l}} f(z)\Delta(z;t_0) \frac{dz}{z} = \frac{1}{(2\pi i)^n} \iint_{z \in C^{l-1} \times \mathfrak{C}^{n-l+1}} f(z)\Delta(z;t_0) \frac{dz}{z} + \frac{2}{(2\pi i)^{n-1}} \sum_{i \in I} \iint_{z \in \mathfrak{C}^{n-1}} f(t_0q^i,z)\Delta_1(t_0q^i,z;t_0) \frac{dz}{z}$$

for $f \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$. Then (2.4.14) follows by applying the induction hypotheses on the integral over $C^{l-1} \times \mathfrak{C}^{n-l+1}$ in (2.4.20).

Lemma 2.4.5 can be used to prove Proposition 2.4.3 inductively. The following definition will be used to formulate the induction hypotheses.

2. A RESIDUE CALCULUS FOR KOORNWINDER POLYNOMIALS

DEFINITION 2.4.6. Let (C, \mathfrak{C}) be a (n, t_0) -residue pair and let $A^+(C, \mathfrak{C})$ (2.3.8) be the corresponding open interval. A sequence of closed contours (C_0, \ldots, C_s) is called a (n, t_0) -resolution for (C, \mathfrak{C}) if the contours C_l are deformed circles satisfying the following four conditions (we write r_l for the (radial) functions r_{C_l} in the parametrization ϕ_{C_l} of C_l):

(i) $C_0 = \mathfrak{C}$ and $C_s = C$; (ii) $r_l(x) = r_{\mathfrak{C}}(x) = r_C(x)$ for $x \notin A^+(C, \mathfrak{C}) \cup A^+(\mathfrak{C}, C)$ and $l \in [0, s]$; (iii) $t(r_{l+1}(x)) < r_l(x) < r_{l+1}(x)$ for $x \in A^+(C, \mathfrak{C})$ and $l \in [0, s-1]$; (iv) $t_0 t^p q^r \notin C_l$ for $p \in [-1, n-1]$, $r \in \mathbb{Z}$ and $l \in [1, s-1]$. We call s the length of the resolution.

Observe that there exists a (n, t_0) -resolution for every (n, t_0) -residue pair (C, \mathfrak{C}) . If (C_0, \ldots, C_s) is a (n, t_0) -resolution for a (n, t_0) -residue pair (C, \mathfrak{C}) , then (C_l, C_{l-1}) is a (n, t_0) -residue pair satisfying the extra condition (2.4.13) used to prove Lemma 2.4.5 ($l \in [1, s]$). Proposition 2.4.3 can now be proved by induction on the length of the resolution.

PROOF OF PROPOSITION 2.4.3. Suppose that for all $n \in \mathbb{N}$ and all $t_0 \in \mathbb{C}^*$ with $\underline{t} = (t_0, t_1, t_2, t_3) \in V$, Proposition 2.4.3 has been proved for (n, t_0) -residue pairs which have a (n, t_0) -resolution of length $\leq s - 1$, where $s \geq 2$.

Fix arbitrary $n \in \mathbb{N}$ and $t_0 \in \mathbb{C}^*$ such that $\underline{t} = (t_0, t_1, t_2, t_3) \in V$. The induction step is clear for n = 1, so we may assume that n > 1. Let (C, \mathfrak{C}) be a (n, t_0) -residue pair with a (n, t_0) -resolution (C_0, \ldots, C_s) of length s. It suffices to prove (2.4.12) for the (n, t_0) -residue pair (C, \mathfrak{C}) . We write $\Omega^{(l)}$ and $\Omega_{(l)}$ for the domains associated with the (n, t_0) -residue pairs (C_l, \mathcal{C}_{l-1}) and (C_l, \mathfrak{C}) respectively $(l \in [1, s])$. Note that $\Omega_{(1)} \subset$ $\Omega_{(2)} \subset \ldots \subset \Omega_{(s)} = \Omega$ where Ω is the domain associated with the (n, t_0) -residue pair (C, \mathfrak{C}) . By (2.4.14) and (2.4.17), we have

$$+ \frac{1}{(2\pi i)^n} \iint_{z \in C^n} f(z) \Delta(z; t_0) \frac{dz}{z} = \frac{1}{(2\pi i)^n} \iint_{z \in (C_{s-1})^n} f(z) \Delta(z; t_0) \frac{dz}{z} + \frac{2n}{(2\pi i)^{n-1}} \sum_{i \in I_s} \iint_{z \in (C_{s-1})^{n-1}} f_i(z) \Delta(z; tt_0 q^i) \frac{dz}{z}$$

(2.4.21)

for $f \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$, where

$$(2.4.22) I_s := \{i \in \mathbb{Z}_+ | r_s(\alpha_0^+) > | t_0 q^i | > r_{s-1}(\alpha_0^+) \}$$

and $f_i(z) := f(t_0q^i, z)h(t_0q^i, z; t_0)$ with h given by (2.4.18). We will apply the induction hypotheses on all the terms in the right hand side of (2.4.21). For the integral over $(C_{s-1})^n$ note that (C_0, \ldots, C_{s-1}) is a (n, t_0) -resolution of length s - 1 for the

 (n, t_0) -residue pair (C_{s-1}, \mathfrak{C}) . Hence, by the induction hypotheses,

(2.4.23)
$$\frac{1}{(2\pi i)^n} \iint_{z \in (C_{s-1})^n} f(z)\Delta(z;t_0)\frac{dz}{z} = \frac{1}{(2\pi i)^n} \iint_{z \in \mathfrak{C}^n} f(z)\Delta(z;t_0)\frac{dz}{z} + \sum_{r=1}^n \frac{2^r (n-r+1)_r}{(2\pi i)^{n-r}} \sum_{\omega \in D(r;C_{s-1},\mathfrak{C};t_0)} \iint_{z \in \mathfrak{C}^{n-r}} f(\omega,z)\Delta_r(\omega,z;t_0)\frac{dz}{z}$$

for all $f \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$.

Now fix an $i \in I_s$. In the proof of Lemma 2.4.5 it was shown that $h(t_0q^i, .; t_0) \in \mathcal{O}_{\mathcal{W}}((\Omega^{(s)})^{n-1})$. In fact it follows from the proof that $h(t_0q^i, .; t_0) \in \mathcal{O}_{\mathcal{W}}(\Omega^{n-1})$. In particular we have $f_i \in \mathcal{O}_{\mathcal{W}}((\Omega_{(s-1)})^{n-1})$ for $f \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$. Observe furthermore that $(tt_0q^i, t_1, t_2, t_3) \in V$ since $\arg(tt_0q^i) = \arg(t_0)$, and that the sequence (C_0, \ldots, C_{s-1}) is a $(n-1, tt_0q^i)$ -resolution of length s-1 for the $(n-1, tt_0q^i)$ -residue pair (C_{s-1}, \mathfrak{C}) . So the induction hypotheses can be applied to all the terms in the second line of (2.4.21), and we obtain

$$\frac{2n}{(2\pi i)^{n-1}} \iint_{z \in (C_{s-1})^{n-1}} f_i(z)\Delta(z;tt_0q^i)\frac{dz}{z} = \frac{2n}{(2\pi i)^{n-1}} \iint_{z \in \mathfrak{C}^{n-1}} f_i(z)\Delta(z;tt_0q^i)\frac{dz}{z} + \sum_{r=2}^n \frac{2^r (n-r+1)_r}{(2\pi i)^{n-r}} \sum_{\omega \in D(r-1;C_{s-1},\mathfrak{C};tt_0q^i)} \iint_{z \in \mathfrak{C}^{n-r}} f_i(\omega,z)\Delta_{r-1}(\omega,z;tt_0q^i)\frac{dz}{z}$$

for $f \in \mathcal{O}_{\mathcal{W}}(\Omega^n)$ and $i \in I_s$.

Substitution of (2.4.23) and (2.4.24) in the right hand side of (2.4.21) completes the proof of (2.4.12), since

$$\begin{split} D(1; C, \mathfrak{C}; t_0) &= D(1; C_{s-1}, \mathfrak{C}; t_0) \cup \{t_0 q^i\}_{i \in I_s}, \\ D(r; C, \mathfrak{C}; t_0) &= D(r; C_{s-1}, \mathfrak{C}; t_0) \cup \bigcup_{i \in I_s} \left\{ (t_0 q^i, \omega) \,|\, \omega \in D(r-1; C_{s-1}, \mathfrak{C}; tt_0 q^i) \right\} \end{split}$$

disjoint unions $(r \in [2, n])$ and

(2.4.25)
$$\begin{aligned} f_i(z)\Delta(z;tt_0q^i) &= f(t_0q^i,z)\Delta_1(t_0q^i,z;t_0),\\ f_i(\omega,z)\Delta_{r-1}(\omega,z;tt_0q^i) &= f(t_0q^i,\omega,z)\Delta_r(t_0q^i,\omega,z;t_0) \end{aligned}$$

for $i \in I_s$, $r \in [2, n]$ and $\omega \in D(r - 1; C_{s-1}, \mathfrak{C}; tt_0q^i)$.

2.5. Multivariable q-Racah polynomials

In this section the Koornwinder polynomials are studied for parameter values (\underline{t}, t) satisfying a particular truncation condition. By applying the residue calculus (cf. Section 2.4) to the results in Theorem 2.3.7 it will be shown that the Koornwinder polynomials for these parameter values are orthogonal with respect to a finite, discrete measure and the corresponding quadratic norms will be computed. The orthogonality relations which

will be obtained reduce to the orthogonality relations of the q-Racah polynomials in the one variable setting (n = 1) (cf. Theorem 2.2.2).

For $\lambda \in P(r)$ set

(2.5.1)
$$\Delta^{qR}(\rho q^{\lambda};\underline{t};t) := \prod_{i=1}^{r} \left(\frac{(q\rho_{i}^{2};q)_{2\lambda_{i}}}{(\rho_{i}^{2};q)_{2\lambda_{i}}(q^{-1}t_{0}t_{1}t_{2}t_{3}t^{2i-2})^{\lambda_{i}}} \prod_{j=0}^{3} \frac{(t_{j}\rho_{i};q)_{\lambda_{i}}}{(qt_{j}^{-1}\rho_{i};q)_{\lambda_{i}}} \right) \\ \cdot \prod_{1 \leq k < l \leq r} \frac{(q\rho_{k}\rho_{l}, t\rho_{k}\rho_{l};q)_{\lambda_{k}+\lambda_{l}}(q\rho_{k}^{-1}\rho_{l}, t\rho_{k}^{-1}\rho_{l};q)_{\lambda_{l}-\lambda_{k}}}{(t^{-1}q\rho_{k}\rho_{l}, \rho_{k}\rho_{l};q)_{\lambda_{k}+\lambda_{l}}(t^{-1}q\rho_{k}^{-1}\rho_{l}, \rho_{k}^{-1}\rho_{l};q)_{\lambda_{l}-\lambda_{k}}}$$

where $\rho_i := t_0 t^{i-1}$ and set for $r \in \mathbb{Z}_+$,

(2.5.2)

$$K_{r}(\underline{t};t) := \prod_{i=1}^{r} \frac{(\rho_{i}^{-2};q)_{\infty}}{(q,\rho_{i}t_{1},\rho_{i}^{-1}t_{1},\rho_{i}t_{2},\rho_{i}^{-1}t_{2},\rho_{i}t_{3},\rho_{i}^{-1}t_{3};q)_{\infty}} \\ = \prod_{1 \le k < l \le r}^{r} (\rho_{k}^{-1}\rho_{l},\rho_{k}^{-1}\rho_{l}^{-1};q)_{\tau} \\ = \prod_{i=1}^{r} \frac{(\rho_{i}^{-2},t,t_{0}^{-2}t^{2-i-r};q)_{\infty}}{(q,\rho_{i}t_{1},\rho_{i}^{-1}t_{1},\rho_{i}t_{2},\rho_{i}^{-1}t_{2},\rho_{i}t_{3},\rho_{i}^{-1}t_{3},t^{i},t_{0}^{-2}t^{2-2i};q)_{\infty}}$$

where $t = q^{\tau}$. The discrete weights $\Delta^{(d)}(\rho q^{\lambda}; \underline{t}; t)$ (2.4.7) can now be rewritten as follows.

PROPOSITION 2.5.1. For $\lambda \in P(r)$ we have

$$\Delta^{(d)}(\rho q^{\lambda};\underline{t};t) = K_r(\underline{t};t)\Delta^{qR}(\rho q^{\lambda};\underline{t};t),$$

where $\rho_i = t_0 t^{i-1}$.

PROOF. We rewrite every factor $(aq^m;q)_b$ in the explicit expression of the discrete weight $\Delta^{(d)}(\rho q^{\lambda};\underline{t};t)$ in which m only depends on λ as a quotient of infinite products using (1.5.1). Then replace the factors of the form $(cq^m;q)_{\infty}$ by $(c;q)_{\infty}(c;q)_m^{-1}$ if $m \in \mathbb{Z}_+$, respectively by $(c;q)_{\infty}(-c)^{-m}q^{-\binom{1-m}{2}}(qc^{-1};q)_{-m}$ if $m \in -\mathbb{N}$. For the case $m \in -\mathbb{N}$, we used here the formula

(2.5.3)
$$(q^{1-l}x;q)_l = (-x)^l q^{-\binom{l}{2}} (x^{-1};q)_l, \quad (l \in \mathbb{N}).$$

Using this method we obtain for $j \in [1, r]$,

$$(2.5.4) \qquad \prod_{i=1}^{j-1} \frac{1}{(\rho_{i}\rho_{j}q^{\lambda_{i-1}+\lambda_{j}}, \rho_{i}\rho_{j}^{-1}q^{\lambda_{i-1}-\lambda_{j}}; q)_{\lambda_{i}-\lambda_{i-1}}} \\ = (-1)^{\lambda_{j-1}} t^{(j-1)\lambda_{j}}q^{\lambda_{j-1}-(\lambda_{j}-\lambda_{j-1})+(\lambda_{j})} \frac{(q;q)_{\lambda_{j}-\lambda_{j-1}}(t_{0}\rho_{j};q)_{\lambda_{j}}}{(\rho_{j}^{2};q)_{\lambda_{j}+\lambda_{j-1}}(qt_{0}^{-1}\rho_{j};q)_{\lambda_{j}}} \\ \cdot \prod_{i=1}^{j-1} \frac{(t\rho_{i}\rho_{j};q)_{\lambda_{i}+\lambda_{j}}(q\rho_{i}^{-1}\rho_{j};q)_{\lambda_{j}-\lambda_{i}}}{(\rho_{i}\rho_{j};q)_{\lambda_{i}+\lambda_{j}}(qt^{-1}\rho_{i}^{-1}\rho_{j};q)_{\lambda_{j}-\lambda_{i}}} t^{\lambda_{i}-\lambda_{j}}.$$

Using (2.5.4) and applying the same method to the explicit expression for the weight w_d (2.2.7) gives

$$(2.5.5) \qquad w_{d}(\rho_{j}q^{\lambda_{j}};\rho_{j}q^{\lambda_{j-1}})\prod_{i=1}^{j-1}\frac{1}{(\rho_{i}\rho_{j}q^{\lambda_{i-1}+\lambda_{j}},\rho_{i}\rho_{j}^{-1}q^{\lambda_{i-1}-\lambda_{j}};q)_{\lambda_{i}-\lambda_{i-1}}} \\ = \frac{(\rho_{j}^{-2};q)_{\infty}}{(q,\rho_{j}t_{1},\rho_{j}^{-1}t_{1},\rho_{j}t_{2},\rho_{j}^{-1}t_{2},\rho_{j}t_{3},\rho_{j}^{-1}t_{3};q)_{\infty}} \\ \cdot \frac{(q\rho_{j}^{2};q)_{2\lambda_{j}}}{(\rho_{j}^{2};q)_{2\lambda_{j}}(q^{-1}t_{0}t_{1}t_{2}t_{3})^{\lambda_{j}}}\prod_{k=0}^{3}\frac{(t_{k}\rho_{j};q)_{\lambda_{j}}}{(qt_{k}^{-1}\rho_{j};q)_{\lambda_{j}}} \\ \cdot \prod_{i=1}^{j-1}\frac{(t\rho_{i}\rho_{j};q)_{\lambda_{i}+\lambda_{j}}(qt_{k}^{-1}\rho_{i}^{-1}\rho_{j};q)_{\lambda_{j}-\lambda_{i}}}{(\rho_{i}\rho_{j};q)_{\lambda_{i}+\lambda_{j}}(qt_{k}^{-1}\rho_{i}^{-1}\rho_{j};q)_{\lambda_{j}-\lambda_{i}}}t^{\lambda_{i}-\lambda_{j}}}$$

for $j \in [1, r]$. Now again applying the above mentioned method, gives

(2.5.6)
$$\prod_{i=1}^{j-1} \left(\rho_i^{-1} \rho_j q^{\lambda_j - \lambda_i}, \rho_i^{-1} \rho_j^{-1} q^{-\lambda_i - \lambda_j}; q \right)_{\tau} \\ = \prod_{i=1}^{j-1} \left(\rho_i^{-1} \rho_j, \rho_i^{-1} \rho_j^{-1}; q \right)_{\tau} \frac{\left(t \rho_i^{-1} \rho_j; q \right)_{\lambda_j - \lambda_i} \left(q \rho_i \rho_j; q \right)_{\lambda_i + \lambda_j}}{\left(\rho_i^{-1} \rho_j; q \right)_{\lambda_j - \lambda_i} \left(q t^{-1} \rho_i \rho_j; q \right)_{\lambda_i + \lambda_j}} t^{-\lambda_i - \lambda_j}$$

for $j \in [1, r]$, where $t = q^{\tau}$. Now the proposition follows by multiplying (2.5.5) and (2.5.6) and taking the product over $j \in [1, r]$.

In the remainder of the section we fix a $N \in \mathbb{N}$. In the next theorem the orthogonality relations for the Koornwinder polynomials are given when the parameters (\underline{t}, t) satisfy the truncation condition $t^{n-1}t_0t_3 = q^{-N}$. The theorem will be formulated with the parameters considered as indeterminates. Set $F := \mathbb{C}(\underline{t}, t)$, $\mathbb{F} := \mathbb{C}(t_0, t_1, t_2, t)$ and $\underline{t}_N := (t_0, t_1, t_2, t^{1-n}t_0^{-1}q^{-N})$. Let $A_F^{\mathcal{W}}$ respectively $A_F^{\mathcal{W}}$ be the algebra of \mathcal{W} -invariant Laurent polynomials over the field F respectively \mathbb{F} . Define the *Koornwinder polynomial* $P_{\lambda}(:;\underline{t};t) \in A_F^{\mathcal{W}}$ of degree λ over the field F by

(2.5.7)
$$P_{\lambda}(.;\underline{t};t) := \left(\prod_{\mu < \lambda} \frac{D_{\underline{t},t} - E_{\mu}(\underline{t},t)}{E_{\lambda}(\underline{t},t) - E_{\mu}(\underline{t};t)}\right) m_{\lambda}$$

Observe that $P_{\lambda}(.;\underline{t}_{N};t) \in A_{\mathbb{F}}^{\mathcal{W}}$ is well defined since the eigenvalues $\{E(\lambda;\underline{t}_{N};t)\}_{\lambda\in\Lambda}$ (2.3.16) are mutually different as elements in $\mathbb{C}[t_{0}, t_{1}, t_{2}, t]$. The Koornwinder polynomial $P_{\lambda}(.;\underline{t};t) \in A^{\mathcal{W}}$ of degree λ as defined in Theorem 2.3.7 (cf. Definition 2.3.8) can be reobtained from (2.5.7) by specializing the parameters (\underline{t}, t) to values in $V \times (0, 1)$.

DEFINITION 2.5.2. $\{P_{\lambda}(.;\underline{t}_{N};t)\}_{\lambda \in \Lambda_{N}} \subset A_{\mathbb{F}}^{\mathcal{W}}$ with $\Lambda_{N} := \{\lambda \in \Lambda \mid \lambda_{1} \leq N\}$ are called the multivariable (BC type) q-Racah polynomials.

Let $\lambda \in P(n)$, then the weight $\Delta^{qR}(\rho q^{\lambda}; \underline{t}_N; t) \in \mathbb{F}$ is well defined $(\Delta^{qR} \text{ given by} (2.5.1))$ and it is non-zero if and only if $\lambda_n \leq N$ due to the factor $(\rho_n t_3; q)_{\lambda_n}$ in the numerator of $\Delta^{qR}(\rho q^{\lambda}; \underline{t}; t)$. So the bilinear form

(2.5.8)
$$\langle f,g \rangle_{qR,\underline{t}_N,t} := \sum_{\lambda \in P(n)} f(\rho q^{\lambda}) g(\rho q^{\lambda}) \Delta^{qR}(\rho q^{\lambda};\underline{t}_N;t), \quad f,g \in A_{\mathbb{F}}^{\mathcal{W}},$$

takes its values in the field \mathbb{F} . Let $\mathcal{N}^{qR}(\lambda; \underline{t}; t)$ for $\lambda \in \Lambda$ be given by

(2.5.9)
$$\mathcal{N}^{qR}(\lambda;\underline{t};t) := \frac{\mathcal{N}(\lambda;\underline{t};t)}{K_n(t;t)2^n n!}$$

where $\mathcal{N}(\lambda)$ (2.3.18) is the expression for the quadratic norms of the Koornwinder polynomial P_{λ} . Substitution of the explicit expressions for $\mathcal{N}(\lambda)$ and K_n in (2.5.9) yields that $\mathcal{N}^{qR}(\lambda; \underline{t}_N; t) \in \mathbb{F}$ and that $\mathcal{N}^{qR}(\lambda; \underline{t}_N; t)$ is non-zero if and only if $\lambda \in \Lambda_N$.

THEOREM 2.5.3. Let $N \in \mathbb{N}$. The q-Racah polynomials $P_{\lambda}(.;\underline{t}_N;t)$ $(\lambda \in \Lambda_N)$ are orthogonal with respect to $\langle ., . \rangle_{qR,\underline{t}_N,t}$ and the quadratic norms are given by

(2.5.10)
$$\langle P_{\lambda}(:;\underline{t}_{N};t), P_{\lambda}(:;\underline{t}_{N};t) \rangle_{qR,\underline{t}_{N},t} = \mathcal{N}^{qR}(\lambda;\underline{t}_{N};t) \qquad (\lambda \in \Lambda_{N}).$$

PROOF. Let $\tilde{V} \subset (\mathbb{C}^*)^4$ be the set of parameters $\underline{t} \in (\mathbb{C}^*)^4$ for which $t_0 t_1 t_2 t_3 \in \mathbb{C} \setminus \mathbb{R}$. Note that there exists an open dense subset $I_N \subset (0, 1)$ such that $E_{\lambda}(\underline{t}; t) \neq E_{\mu}(\underline{t}; t)$ for all $\underline{t} \in \tilde{V}$, $t \in I_N$ and all $\lambda, \mu \in \Lambda_N$ with $\lambda \neq \mu$.

for all $\underline{t} \in \tilde{V}$, $t \in I_N$ and all $\lambda, \mu \in \Lambda_N$ with $\lambda \neq \mu$. Fix $t_0, t_1, t_2 \in \mathbb{C}^*$ such that $\#\{\arg(t_i), \arg(t_i^{-1}) \mid i = 0, 1, 2\} = 6$ and $t \in I_N$. Then $\underline{t}_N \in \tilde{V}$ and there exists a sequence $\{t_{3,i}\}_{i \in \mathbb{Z}_+} \subset \mathbb{C}^*$ converging to $t^{1-n}t_0^{-1}q^{-N}$ such that $\underline{t}_i := (t_0, t_1, t_2, t_{3,i}) \in V \cap \tilde{V}$ for all i (V given in Definition 2.3.1). By considering a subsequence if necessary, we may assume that there exist (n, t_0) -residue pairs (C_i, \mathfrak{C}) where C_i is a \underline{t}_i -contour and where \mathfrak{C} is a deformed circle such that the sequences $\{t_1q^j, t_2q^j, t_{3,i}q^j\}_{j\in\mathbb{Z}_+}$ are in the interior of \mathfrak{C} for all i and such that $t^{n-1}t_0q^N$ is in the exterior of \mathfrak{C} . Then it follows from Theorem 2.3.7, Proposition 2.4.3 and Proposition

(2.5.11)
$$\frac{\mathcal{N}(\lambda;\underline{t}_{i};t)}{K_{n}(\underline{t}_{i};t)}\delta_{\lambda,\mu} = \frac{1}{(2\pi i)^{n}} \iint_{z\in\mathfrak{C}^{n}} (P_{\lambda}P_{\mu})(z;\underline{t}_{i};t) \frac{\Delta(z;\underline{t}_{i};t)}{K_{n}(\underline{t}_{i};t)} \frac{dz}{z} + \sum_{r=1}^{n} \frac{2^{r}(n-r+1)_{r}}{(2\pi i)^{n-r}} \sum_{\omega\in D(r)} \iint_{z\in\mathfrak{C}^{n-r}} (P_{\lambda}P_{\mu})(\omega,z;\underline{t}_{i};t) \frac{\Delta_{r}(\omega,z;\underline{t}_{i};t)}{K_{n}(\underline{t}_{i};t)} \frac{dz}{z}$$

where $\delta_{\lambda,\mu}$ is the Kronecker-delta and $D(r) = D(r; C_i, \mathfrak{C}; t_0; t)$ (2.4.6) (which is independent of *i*). By (2.4.10) and Proposition 2.5.1 we have

(2.5.12)
$$\Delta_r(\omega, z; \underline{t}_i; t) = K_r(\underline{t}_i; t) \Delta^{qR}(\omega; \underline{t}_i; t) \Delta(z; \underline{t}_i; t) \delta_c(\omega; z).$$

After substitution of (2.5.12) in the right hand side of (2.5.11) for all r, it follows from the bounded convergence Theorem that the limit $i \to \infty$ may be pulled through the integrals in the right hand side of (2.5.11). Only the completely discrete part survives the limit $i \to \infty$ in the equality (2.5.11) since

$$\lim_{i \to \infty} \frac{K_r(\underline{t}_i; t)}{K_n(\underline{t}_i; t)} = 0, \qquad 0 \le r < n$$

by the factor $(\rho_n t_3; q)_{\infty}$ in the denominator of $K_n(\underline{t}; t)$. The theorem follows now for the specialized parameter values t_0, t_1, t_3, t from the fact that

$$\{\rho q^{\lambda} \mid \lambda \in P(n), \lambda_n \leq N\} \subset D(n)$$

and the fact that $\Delta^{qR}(\rho q^{\lambda}; \underline{t}_N; t) = 0$ for $\lambda \in P(n)$ with $\lambda_n > N$. It is now clear that the theorem also holds over the field \mathbb{F} . \Box

The constant term identity can be simplified as follows.

COROLLARY 2.5.4. For $N \in \mathbb{N}$ we have the summation formula

(2.5.13)
$$\langle 1,1\rangle_{qR,\underline{t}_N,t} = \prod_{i=1}^n \frac{\left(qt_0^2 t^{2n-i-1}, qt_1^{-1} t_2^{-1} t^{i-n}; q\right)_N}{\left(qt_0 t_1^{-1} t^{n-i}, qt_0 t_2^{-1} t^{n-i}; q\right)_N}$$

PROOF. First note that by (2.3.25), (2.5.2) and (2.5.9) we have the explicit formula

$$(2.5.14) \quad \mathcal{N}^{qR}(0;\underline{t};t) = \prod_{i=1}^{n} \frac{\left(t_0 t_1 t_2 t_3 t^{2n-i-1}, t_0^{-1} t_1 t^{1-i}, t_0^{-1} t_2 t^{1-i}, t_0^{-1} t_3 t^{1-i}; q\right)_{\infty}}{\left(t_0^{-2} t^{1+i-2n}, t_1 t_2 t^{i-1}, t_1 t_3 t^{i-1}, t_2 t_3 t^{i-1}; q\right)_{\infty}}.$$

Then (2.5.13) follows by substitution of $t_3 = t_0^{-1}t^{1-n}q^{-N}$ in (2.5.14) and by applying formula (2.5.3) repeatedly (see also [18, Section 2.3]).

The second order q-difference operator $D_{\underline{t}_N,t}$ (2.3.13) diagonalizes the q-Racah polynomials $\{P_{\lambda}(.;\underline{t}_N;t)\}_{\lambda \in \Lambda_N}$. By Theorem 2.5.3 we conclude that $D_{\underline{t}_N,t}$ is symmetric with respect to $\langle ., . \rangle_{qR,\underline{t}_N,t}$. In [20] the symmetry of $D_{\underline{t}_N,t}$ was proved by direct calculations and the orthogonality relations for the multivariable q-Racah polynomials were proved using the symmetry of $D_{\underline{t}_N,t}$. Furthermore, in [20] the quadratic norms of the

2. A RESIDUE CALCULUS FOR KOORNWINDER POLYNOMIALS

q-Racah polynomials were expressed in terms of the quadratic norm of the unit polynomial by studying Pieri formulas for the q-Racah polynomials. The constant term identity (2.5.13) was recently proved by van Diejen [18, Theorem 3] by truncating a multivariable analogue of Roger's $_6\phi_5$ -series [18, Theorem 2], which in turn is closely related to an Aomoto-Ito type sum (cf. [4], [44]) for the non-reduced root system BC_n . The proofs of the summation formulas in [18] are based on a multiple $_6\psi_6$ summation formula of Gustafson.

In the one variable case it is known that the Askey-Wilson integral can be rewritten as an infinite sum of residues for some parameter region by shifting the contour over four infinite sequences of poles (see [7, Theorem 2.1]). More generally one can ask the question whether a completely discrete orthogonality measure for the Koornwinder polynomials can be obtained by pulling the <u>t</u>-contours over certain infinite sequence of poles in the orthogonality relations of the Koornwinder polynomials (Theorem 2.3.7).

Strong indications in that direction can be found in Gustafson's paper [32] and the recent paper of Tarasov and Varchenko [127] where contours in multidimensional integrals are shifted over infinite sequences of poles in order to arrive at (purely discrete) multidimensional Jackson integrals. Another strong indication is the fact that the Macdonald polynomials are orthogonal with respect to Aomoto-Ito type (cf. [4], [44]) weight functions (see Cherednik [15]). Since the B, C and D type Macdonald polynomials can be obtained from the Koornwinder polynomials by suitable specialization of the parameters we thus have orthogonality relations for these subfamilies of the Koornwinder polynomials with respect to infinite discrete measures (and the corresponding discrete weights are directly related to (2.5.1), see [18]).

We will not consider here the above mentioned questions. In the next chapter we will look instead at the implications of the residue calculus for certain limit cases of the Koornwinder polynomials (multivariable big and little q-Jacobi polynomials). In order to study these limit cases we first need to consider the Koornwinder polynomials for yet another parameter domain. This will be the subject of the next section.

2.6. Koornwinder polynomials with positive orthogonality measure

In this section the Koornwinder polynomials are considered for parameters \underline{t} in the following parameter domain.

DEFINITION 2.6.1. Let V_K be the set of parameters $\underline{t} = (t_0, t_1, t_2, t_3)$ which satisfy the following conditions:

(1) The parameters t_0, t_1, t_2, t_3 are real, or if complex, then they appear in conjugate pairs.

(2) $t_k t_l \notin \mathbb{R}_{\geq 1}$ for all $0 \leq k < l \leq 3$.

Observe that parameters $\underline{t} \in V_K$ satisfy the following properties:

(A) $t_i \in \mathbb{R}$ if $|t_i| \ge 1$;

(B) There are at most two parameters with modulus ≥ 1 . If there are two, then one is positive and the other is negative.

It will be shown that the multivariable Koornwinder polynomials are orthogonal with respect to a positive, partly discrete orthogonality measure for $t \in (0, 1)$ and $\underline{t} \in V_K$ by shifting the contour C^n in the integral

(2.6.1)
$$\frac{1}{(2\pi i)^n} \iint_{z \in C^n} P_{\lambda}(z) P_{\mu}(z) \Delta(z) \frac{dz}{z}$$

to the *n*-torus T^n for a specific parameter domain $V_0 \subset V$ (here *C* is a <u>*t*</u>-contour (Definition 2.3.2) and *V* is the parameter domain given in Definition 2.3.1). Then a partly discrete orthogonality measure will be obtained which turns out to be well defined and positive for parameter values $\underline{t} \in V_K$. Orthogonality relations for parameter values $\underline{t} \in V_K$ with respect to this positive, partly discrete orthogonality measure can then be derived by suitable continuity arguments.

The parameter domain V_0 is defined as follows.

DEFINITION 2.6.2. Let V_0 be the set of parameters $\underline{t} \in V$ for which (i) at most two parameters have modulus > 1; (ii) $t_i t^j q^p \notin T$ for $i \in [0,3]$, $j \in [-1, n-1]$ and $p \in \mathbb{Z}$.

Fix $t \in (0, 1)$, $\underline{t} \in V_0$ and $0 \le i \ne j \le 3$ such that $|t_k| < 1$ for $k \ne i, j$. Write $\rho_p^{(i)} := t^{p-1}t_i$ respectively $\rho_p^{(j)} := t^{p-1}t_j$ for $p \in \mathbb{Z}$. Define for $r \in \mathbb{N}$ a finite discrete set $D_i(r) = D_i(r; \underline{t}; t) \subset \mathbb{C}^r$ by

$$(2.6.2) D_i(r) := \{ \rho^{(i)} q^{\mu} \mid \mu \in P(r), \ |\rho_r^{(i)} q^{\mu_r}| > 1 \}$$

and similarly for $D_i(r)$, where P(r) is given by (2.4.4) and

$$\rho^{(i)}q^{\mu} = (\rho_1^{(i)}q^{\mu_1}, \dots, \rho_r^{(i)}q^{\mu_r})$$

for $\mu \in P(r)$. Observe that $D_i(r) = \emptyset$ if $|t_i| < 1$. Furthermore, write $F(r) = F(r; \underline{t}; t) \subset \mathbb{C}^r$ for the disjoint union

(2.6.3)
$$F(r) := \bigcup_{\substack{l+m=r\\l,m\in\mathbb{Z}_+}} D_i(l) \times D_j(m) \qquad (r \in [1,n]).$$

Here the convention is used that $D_i(l) \times D_j(m) = \emptyset$ if l > 0 and $D_i(l) = \emptyset$ or if m > 0and $D_j(m) = \emptyset$, and that $D_i(0) \times D_j(m) = D_j(m)$, $D_i(l) \times D_j(0) = D_i(l)$. Let $\omega \in F(r), z \in T^{n-r}$ and set

(2.6.4)
$$d\nu_r^K(\omega, z; \underline{t}; t) := \Delta_r^K(\omega, z; \underline{t}; t) \frac{dz}{z}$$

with weight function $\Delta_r^K(\omega, z)$ for $\omega = (\vartheta, \zeta) \in D_i(l) \times D_j(m)$ given by

(2.6.5)
$$\Delta_r^K(\vartheta,\zeta,z;\underline{t};t) := \Delta^{(d)}(\vartheta;t_i)\Delta^{(d)}(\zeta;t_j)\Delta(z;\underline{t};t)\delta_c(\vartheta;\zeta,z)\delta_c(\zeta;z)$$

where $\Delta^{(d)}$ is given by (2.4.7) and δ_c is given by (2.4.11). For the special case l = 0 respectively m = 0, (2.6.5) simplifies to

(2.6.6)
$$\Delta_r^K(\zeta, z; \underline{t}; t) = \Delta^{(d)}(\zeta; t_j) \Delta(z; \underline{t}; t) \delta_c(\zeta; z) = \Delta_r(\zeta, z; t_j)$$

2. A RESIDUE CALCULUS FOR KOORNWINDER POLYNOMIALS

respectively

(2.6.7)
$$\Delta_r^K(\vartheta, z; \underline{t}; t) = \Delta^{(d)}(\vartheta; t_i) \Delta(z; \underline{t}; t) \delta_c(\vartheta; z) = \Delta_r(\vartheta, z; t_i)$$

where Δ_r is given by (2.4.10). The following lemma is an easy consequence of the residue calculus developed in Section 2.4.

LEMMA 2.6.3. Let $t \in (0, 1)$ and $\underline{t} \in V_0$. Let C be a \underline{t} -contour and $f \in A^{\mathcal{W}}$. Then,

(2.6.8)
$$\frac{\frac{1}{(2\pi i)^n} \iint\limits_{z \in C^n} f(z) d\nu(z) = \frac{1}{(2\pi i)^n} \iint\limits_{z \in T^n} f(z) d\nu(z) + \sum_{r=1}^n \frac{2^r (n-r+1)_r}{(2\pi i)^{n-r}} \sum_{\omega \in F(r)} \iint\limits_{z \in T^{n-r}} f(\omega, z) d\nu_r^K(\omega, z).$$

PROOF. If $|t_k| < 1$ for all k then we only have the completely continuous measure $d\nu$ on T^n in the right hand side of (2.6.8) since $F(r) = \emptyset$. Since T is a <u>t</u>-contour in this case, the lemma follows from Lemma 2.3.4.

Suppose that at most one parameter has modulus > 1. By the symmetry of $d\nu(z; \underline{t}; t)$ in the four parameters \underline{t} , we may assume that $|t_0| > 1$. By Lemma 2.3.4, we may assume that the \underline{t} -contour C satisfies the additional conditions that $A^+ := \{x \in [0,1] | r_C(x) > 1\}$ is an open interval and that $\alpha_0^+ \in A^+$ but $\alpha_i^{\pm} \notin A^+$ for i = 1, 2, 3 (here r_C is as in Definition 2.3.2, and α_i^+ is given by (2.3.2)). Then (C,T) is a (n, t_0) -residue pair since $\underline{t} \in V_0$ (Definition 2.6.2), and $D_0(l) = D(l; C, T; t_0)$ (2.4.6) since t_0 is in the interior of C. The lemma is then a direct consequence of Proposition 2.4.3 and (2.6.7).

Suppose now that two parameters have moduli > 1. Without loss of generality, we may assume that $|t_0| > 1$ and $|t_1| > 1$, and that the <u>t</u>-contour C satisfies the additional condition that

$$\{x \in [0,1] \mid r_C(x) > 1\} = A_0^+ \cup A_1^+$$

disjoint union, with the A_i^+ open intervals such that $\alpha_i^+ \in A_i^+$ and $\alpha_j^{\pm} \notin A_i^+$ for $j \neq i$ and i = 0, 1. Let $C' := \phi_{C'}([0, 1])$ be the deformed circle with parametrization $\phi_{C'}(x) = r_{C'}(x)e^{2\pi ix}$ given by

$$r_{C'}(x) := r_C(x) \ (x \notin A_0^+ \cup A_0^-), \quad r_{C'}(x) := 1 \ (x \in A_0^+ \cup A_0^-),$$

where $A_0^- := (1 - \beta, 1 - \alpha)$ when $A_0^+ = (\alpha, \beta)$. Then (C, C') is a (n, t_0) -residue pair, (C', T) is a (n, t_1) -residue pair and $D_0(l) = D(l; C, C'; t_0)$ respectively $D_1(m) = D(m; C', T; t_1)$ since t_0 and t_1 are in the interior of C. Write Ω' for the domain associated with (C', T), then $\delta_c(\vartheta; .) \in \mathcal{O}_{\mathcal{W}}((\Omega')^{n-l})$ for $\vartheta \in D_0(l)$, hence the lemma follows by applying Proposition 2.4.3 first to the (n, t_0) -residue pair (C, C'), and then to the (n, t_1) residue pair (C', T).

For $t = q^k$ with $k \in \mathbb{N}$, formula (2.6.8) can be rewritten as

(2.6.9)
$$\frac{\frac{1}{(2\pi i)^n} \iint\limits_{z \in C^n} f(z) d\nu(z) = \sum_{r=0}^n \frac{2^r \binom{n}{r}}{(2\pi i)^{n-r}} \sum_{\substack{e_1, \dots, e_r \\ i=1, \dots, r}} \sum_{\substack{z_i \in \{e_i q^l\}_{l=0}^{Ne_i} \\ i=1, \dots, r}} \int\limits_{\substack{(z_{r+1}, \dots, z_n) \in T^{n-r}}} f(z) \delta(z; q^k) \prod_{i=1}^r w_d(z_i; e_i) \prod_{j=r+1}^n w_c(z_j) \frac{dz_j}{z_j}}{\sum_{i=1}^{n-r}}$$

for $f \in A^{\mathcal{W}}$, where the second sum is over $e_i \in \{t_j \mid j \in [0,3], |t_j| > 1\}$ and N_{e_i} is the largest positive integer such that $|e_i q^{N_{e_i}}| > 1$. Formula (2.6.9) can also be proved directly by induction on n using similar arguments as in the proof of Lemma 2.4.2.

Define now bilinear forms $\langle ., . \rangle_{r,t,t}$ on $A^{\mathcal{W}}$ for $r \in [0, n], \underline{t} \in V_0$ and $t \in (0, 1)$ by

$$\begin{split} \langle f,g\rangle_0 &:= \iint_{z \in T^n} f(z)g(z)d\nu(z), \\ \langle f,g\rangle_r &:= \sum_{\omega \in F(r)} \iint_{z \in T^{n-r}} f(\omega,z)g(\omega,z)d\nu_r^K(\omega,z), \quad r \in [1,n] \end{split}$$

for $f, g \in A^{\mathcal{W}}$ and set

(2.6.10)

(2.6.11)
$$\langle f,g\rangle_{\underline{t},t} := \sum_{r=0}^{n} \frac{2^r (n-r+1)_r}{(2\pi i)^{n-r}} \langle f,g\rangle_{r,\underline{t},t}, \quad f,g \in A^{\mathcal{W}}.$$

In the following lemma the symmetric bilinear form $\langle ., . \rangle_{\underline{t},t}$ is considered for parameter values $(\underline{t},t) \in V_K \times (0,1)$.

LEMMA 2.6.4. Let $t \in (0, 1)$ and $\underline{t} \in V_K$. (i) The bilinear form $\langle ., . \rangle_{\underline{t}, t}$ is well defined; (ii) The weight function $\Delta(z; \underline{t}; t)$ respectively $\Delta_r^K(\omega, z; \underline{t}; t)$ is positive for $z \in T^n$ respectively $(\omega, z) \in F(r) \times T^{n-r}$ $(r \in [1, n])$.

PROOF. The discrete weights w_d (2.2.7) appearing as factors of the weight function $\Delta_r^K(\omega, z)$ for r > 0 are well defined and strictly positive. Indeed if $t_0t_1t_2t_3 = 0$, then the factors $(t_iq/t_j;q)_k t_j^k$ in the denominator of w_d should be read as $\prod_{l=0}^{k-1} (t_j - t_iq^{l+1})$. The factor $\delta(z;t) = |\delta_+(z;t)|^2$ is also well defined and positive for $z \in T^{n-r}$.

Without loss of generality we may assume that $|t_2|, |t_3| < 1$. Fix $\omega = (\vartheta, \zeta) \in F(r)$ with $\vartheta \in D_0(l)$ and $\zeta \in D_1(m)$ (r = l + m). The factor $\delta_d(\vartheta)$ respectively $\delta_d(\zeta)$ (2.4.8) appearing in the discrete weights $\Delta^{(d)}(\vartheta; t_0)$ respectively $\Delta^{(d)}(\zeta; t_1)$ when l > 0respectively m > 0 is well defined and strict positive. Indeed, if $\vartheta \in D_0(l)$ and l > 0, then $|t_0| > 1$, hence $t_0 \in \mathbb{R}$. Then $\delta_d(\vartheta) > 0$ follows easily from the definition of the set $D_0(l)$ (2.6.2).

It remains to show that $h(z) := (\prod_{l=1}^{n-r} w_c(z_l; \underline{t})) \delta_c(\vartheta; \zeta, z) \delta_c(\zeta; z)$ is well defined and positive for $z \in T^{n-r}$. We check the case that both t_0 and t_1 have moduli ≥ 1 , and

2. A RESIDUE CALCULUS FOR KOORNWINDER POLYNOMIALS

that t_0 is positive real and t_1 negative real (see property (**B**) for parameters $\underline{t} \in V_K$). The case that at most one parameter has modulus ≥ 1 will then also be clear.

Rewrite the factor $(x^2, x^{-2}; q)_{\infty}$ appearing in the numerator of $w_c(x; \underline{t})$ as

$$(x^2, x^{-2}; q)_{\infty} = (x, -x, x^{-1}, -x^{-1}; q)_{\infty} (qx^2, qx^{-2}; q^2)_{\infty}$$

then it is sufficient to check that the factors of the form

$$\begin{split} h_0(x) &:= \frac{\left(x, x^{-1}; q\right)_{\infty}}{\left(t_0 x, t_0 x^{-1}; q\right)_{\infty}} \prod_{k=1}^l \left(\vartheta_k x, \vartheta_k x^{-1}, \vartheta_k^{-1} x, \vartheta_k^{-1} x^{-1}; q\right)_{\tau}, \\ h_1(x) &:= \frac{\left(-x, -x^{-1}; q\right)_{\infty}}{\left(t_1 x, t_1 x^{-1}; q\right)_{\infty}} \prod_{k=1}^m \left(\zeta_k x, \zeta_k x^{-1}, \zeta_k^{-1} x, \zeta_k^{-1} x^{-1}; q\right)_{\tau} \end{split}$$

 $(l, m \in [0, n - 1])$ are well defined and positive for $x \in T$ (here $t = q^{\tau}$). Indeed, the remaining factors of $w_c(x; \underline{t})$ are easily seen to be well defined and positive since $|t_2|, |t_3| < 1$ and t_2, t_3 are both real or are a conjugate pair, while the remaining factors

$$\prod_{\varepsilon_i,\varepsilon_j=\pm 1} \left(\vartheta_i^{\varepsilon_i} \zeta_j^{\varepsilon_j}; q\right)_{\tau} \quad (i \in [1, l], j \in [1, m])$$

of $\delta_c(\vartheta; \zeta, z)$ are well defined and positive since t_0 is positive real and t_1 is negative real. Now let $\lambda \in P(l)$ such that $\vartheta = \rho^{(0)}q^{\lambda} \in D_0(l)$, then $h_0(x) = |h_0^+(x)|^2$ for $x \in T$ with h_0^+ given by

$$\begin{aligned} h_0^+(x) &:= \quad \frac{(x;q)_\infty}{(t_0x;q)_\infty} \prod_{k=1}^l \left(\vartheta_k x, \vartheta_k^{-1} x;q\right)_\tau \\ &= \quad \frac{(x;q)_\infty}{(t\vartheta_l x;q)_\infty} \prod_{k=1}^l \frac{\left(\vartheta_k^{-1} x;q\right)_\tau}{(t\vartheta_{k-1} x;q)_{\lambda_k-\lambda_{k-1}}} \end{aligned}$$

where $\vartheta_0 := t^{-1}t_0$ and $\lambda_0 := 0$. It follows that $h_0^+(x)$ is well defined for $x \in T$, since the possible zero at x = 1 of the factor $(t\vartheta_l x; q)_{\infty}$ in the denominator can be compensated by the zero at x = 1 of the factor $(x; q)_{\infty}$. Similarly, one deals with $h_1(x)$. \Box

We write $A_{\mathbb{R}}^{\mathcal{W}}$ for the \mathbb{R} -algebra of \mathcal{W} -invariant Laurent polynomials in the variables z_1, \ldots, z_n . The following corollary is a direct consequence of Lemma 2.6.4.

COROLLARY 2.6.5. Let $\underline{t} \in V_K$ and $t \in (0, 1)$. Then the restriction of the bilinear form $\langle ., . \rangle_{\underline{t},t}$ to $A_{\mathbb{R}}^{\mathcal{W}} \times A_{\mathbb{R}}^{\mathcal{W}}$ maps into \mathbb{R} and is positive definite.

PROOF. The monomials m_{λ} ($\lambda \in \Lambda$) are real-valued on $F(r) \times T^{n-r}$ since $F(r) \subset \mathbb{R}^r$ by property (A) for parameters in V_K (Definition 2.6.1), so the assertion follows from Lemma 2.6.4(ii).

The following theorem defines the Koornwinder polynomials for parameters $\underline{t} \in V_K$ and $t \in (0,1)$ as a special choice of orthogonal basis for $A_{\mathbb{R}}^{\mathcal{W}}$ with respect to the *positive definite* bilinear form $\langle ., . \rangle_{\underline{t},t} : A_{\mathbb{R}}^{\mathcal{W}} \times A_{\mathbb{R}}^{\mathcal{W}} \to \mathbb{R}$.

2.6. KOORNWINDER POLYNOMIALS WITH POSITIVE ORTHOGONALITY MEASURE

45

THEOREM 2.6.6. For parameters $(\underline{t},t) \in V_K \times (0,1)$ there exists a unique basis $\{P_{\lambda}(.;\underline{t};t)\}_{\lambda \in \Lambda}$ of $A_{\mathbb{R}}^{\mathcal{W}}$ such that

(i) $P_{\lambda}(:;\underline{t};t) = \overline{m}_{\lambda} + \sum_{\mu < \lambda} c_{\lambda,\mu}(\underline{t};t)m_{\mu}$, some $c_{\lambda,\mu}(\underline{t};t) \in \mathbb{R}$;

(ii) $\langle P_{\lambda}(:;\underline{t};t), P_{\mu}(:;\underline{t};t) \rangle_{\underline{t},t} = 0 \text{ if } \mu \neq \lambda.$

Furthermore, $P_{\lambda}(.;\underline{t};t)$ is an eigenfunction of $D_{\underline{t},t}$ with eigenvalue $E_{\lambda}(\underline{t};t)$ and we have the explicit evaluation formula

$$\langle P_{\lambda}(:;\underline{t};t), P_{\lambda}(:;\underline{t};t) \rangle_{\underline{t},t} = \mathcal{N}(\lambda;\underline{t};t), \quad \lambda \in \Lambda$$

for the quadratic norms of the polynomials P_{λ} .

PROOF. Fix $t \in (0,1)$ and $\underline{t} \in V_K$. Since $\langle ., . \rangle_{\underline{t},t}$ is positive definite on $A_{\mathbb{R}}^{\mathcal{W}}$, there exists for $\lambda \in \Lambda$ a unique \mathcal{W} -invariant Laurent polynomial $P_{\lambda}(.;\underline{t};t) \in A_{\mathbb{R}}^{\mathcal{W}}$ satisfying (i) and the conditions $\langle P_{\lambda}(.;\underline{t};t), m_{\mu} \rangle_{\underline{t};t} = 0$ for all $\mu < \lambda$. Furthermore,

$$(2.6.12) \quad P_{\lambda}(z;\underline{t};t) = m_{\lambda}(z) - \sum_{\mu < \lambda} \frac{\langle m_{\lambda}, P_{\mu}(.;\underline{t};t) \rangle_{\underline{t},t}}{\langle P_{\mu}(.;\underline{t};t), P_{\mu}(.;\underline{t};t) \rangle_{\underline{t},t}} P_{\mu}(z;\underline{t};t), \quad \lambda \in \Lambda.$$

The polynomials $P_{\lambda}(z; \underline{t}; t) = m_{\lambda}(z) + \sum_{\mu < \lambda} c_{\lambda,\mu}(\underline{t}; t)m_{\mu}(z)$ as defined in Theorem 2.3.7 also satisfy the formula (2.6.12) when $\underline{t} \in V_0$, in view of Lemma 2.6.3. Fix $\underline{t} \in V_K \setminus V_K^-$, where V_K^- is the set of parameters $\underline{t} \in V_K$ such that $t_i = \pm t^{-m}q^{-s}$ for some $i \in [0,3], m \in [0,n-1]$ and $s \in \mathbb{Z}_+$. Let $\{\underline{t}_k\}_{k \in \mathbb{Z}_+}$ be a sequence in V_0 converging to \underline{t} . Then, by the bounded convergence Theorem,

(2.6.13)
$$\lim_{k \to \infty} \langle f, g \rangle_{\underline{t}_k, t} = \langle f, g \rangle_{\underline{t}, t}, \quad \forall f, g \in A^{\mathcal{W}}.$$

Indeed, by assuming $\underline{t} \notin V_K^-$, we have $F(r; \underline{\tau}; t) = F(r; \underline{t}; t)$ for $\underline{\tau}$ in an open neighbourhood of \underline{t} ($r \in [1, n]$), and no zeros in the denominator of the expression for $\Delta_r^K(\omega, .; \underline{t}; t)$ ($\omega \in F(r), r \in [0, n - 1]$) occur which need to be compensated by zeros in the numerator (see the proof of Lemma 2.6.4). Hence the bounded convergence Theorem may be applied at once.

By induction on λ we then obtain from (2.6.12) and (2.6.13) that

(2.6.14)
$$\lim_{t \to \infty} c_{\lambda,\mu}(\underline{t}_k; t) = c_{\lambda,\mu}(\underline{t}; t), \quad \mu < \lambda,$$

where $c_{\lambda,\mu}$ are the expansion coefficients of P_{λ} with respect to the monomials m_{μ} ($\mu \in \Lambda$). By the residue calculus given in Lemma 2.6.3, Theorem 2.3.7 can be reformulated with respect to the bilinear form $\langle ., . \rangle_{\underline{\tau},t}$ for $\underline{\tau} \in V_0$. The theorem follows then for $\underline{t} \in V_K \setminus V_K^-$ by taking limits in the reformulated results using Proposition 2.3.6 and (2.6.14).

To prove the theorem for $\underline{t} \in V_K^-$, we use again a continuity argument. We treat here one typical example, the general case is derived similarly. Assume that $\underline{t} \in V_K^-$ with $t_0 = t^{-m}q^{-s}$ for some $m \in [0, n-1]$, $s \in \mathbb{Z}_+$ and that $t_i \neq t^{-l}q^{-s'}$ for all $i \in \{1, 2, 3\}$, $l \in [0, n-1]$ and $s' \in \mathbb{Z}_+$. Then there exists an $\varepsilon > 0$ such that $(\tau_0, t_1, t_2, t_3) \in V_K \setminus V_K^$ and $F(r; \tau_0, t_1, t_2, t_3; t) = F(r; \underline{t}; t)$ for all $r \in [1, n]$ and all $\tau_0 \in \mathbb{R}_{>0}$ with $t_0 - \tau_0 < \varepsilon$. We claim that

(2.6.15)
$$\lim_{\tau_0\uparrow t_0} \langle f,g\rangle_{\tau_0,t_1,t_2,t_3,t} = \langle f,g\rangle_{\underline{t},t}, \qquad \forall f,g \in A^{\mathcal{W}}.$$

2. A RESIDUE CALCULUS FOR KOORNWINDER POLYNOMIALS

We use the bounded convergence Theorem. In Lemma 2.6.4 it was shown that zeros in the denominator of the expression for the weight function $\Delta_r^K(\omega, .; \underline{\tau}; t)$ can occur when $\omega \in F(r; \underline{\tau}; t)$ and $\underline{\tau} \in V_K^-$, and that these zeros can be compensated by zeros in the numerator. It follows from the specific form of these compensated zeros and from the fact that the functions

$$h^{\pm}(u,x) := \begin{cases} \frac{(1\pm x)}{(1\pm ux)} & \text{if } u \neq 1\\ 1 & \text{if } u = 1 \end{cases}$$

are bounded on $U \times T$ where $U \subset \mathbb{R}_{>0}$ is some open set containing 1, that the bounded convergence Theorem may be applied in the limit (2.6.15). Now the theorem for the specific parameter values <u>t</u> follows by continuity arguments from (2.6.15).

For parameters $\underline{t} \in V_K$ with $|t_i| \leq 1$ the orthogonality measure is completely continuous (i.e. $\langle ., . \rangle = \langle ., . \rangle_0$) and coincides with Koornwinder's orthogonality measure [**65**]. In particular the orthogonality relations reduce to Koornwinder's orthogonality relations (see [**65**]) and the quadratic norm evaluations reduce to van Diejen's quadratic norm evaluations (see [**17**]) for parameter values $\underline{t} \in V_K$ with $|t_i| \leq 1$.

Theorem 2.6.6 for n = 1 reduces to the orthogonality relations and norm evaluations stated in [7, Theorem 2.5].

For $t = q^k$ ($k \in \mathbb{N}$) the bilinear form $\langle ., . \rangle_{t,q^k}$ (2.6.11) can be rewritten as

$$\langle f,g \rangle = \sum_{r=0}^{n} \frac{2^{r} \binom{n}{r}}{(2\pi i)^{n-r}} \sum_{\substack{e_{1},\dots,e_{r}\\i=1,\dots,r}} \sum_{\substack{z_{i} \in \{e_{i}q^{l}\}_{l=0}^{N_{e_{i}}}} \iint_{\substack{(z_{r+1},\dots,z_{n}) \in T^{n-r}\\i=1,\dots,r}} \int_{f(z)g(z)\delta(z;q^{k})} \prod_{i=1}^{r} w_{d}(z_{i};e_{i}) \prod_{j=r+1}^{n} w_{c}(z_{j}) \frac{dz_{j}}{z_{j}}$$

for $f, g \in A^{\mathcal{W}}$. This is the form of the partly discrete orthogonality measure for the Koornwinder polynomials with discrete deformation parameter $t = q^k$ which was studied earlier in [117] and [118].

46

(2.6.1)

CHAPTER 3

Limit transitions for multivariable orthogonal polynomials

3.1. Introduction

The one variable big and little q-Jacobi polynomials depend apart from q on (essentially) three and two parameters, respectively. The associated orthogonality measures are completely discrete and have infinitely many discrete mass points. The orthogonality measures can be expressed most conveniently in terms of Jackson integrals. The one variable big and little q-Jacobi polynomials are q-analogues of the classical Jacobi polynomials in the sense that when q tends to 1, the big and little q-Jacobi polynomials tend, up to a possible translation and dilation of the variable, to the classical Jacobi polynomials.

The families of one variable big and little q-Jacobi polynomials are members of the Askey tableau. The Askey tableau consists of families of (basic hypergeometric) orthogonal polynomials which are joint eigenfunctions of a second order q-difference operator. Some families can be obtained from others by limit transitions or by specializations of parameters. This induces the hierarchy structure between the families. From this point of view, the four parameter family of Askey-Wilson polynomials is on top of the hierarchy and the families of big and little q-Jacobi polynomials are directly below the Askey-Wilson polynomials. Suitable limit transitions are known from the Askey-Wilson polynomials to the big and little q-Jacobi polynomials (cf. [66]).

In this chapter the limit transitions from Askey-Wilson polynomials to big and little q-Jacobi polynomials are generalized to the multivariable setting. It is shown that in these limits, the positive, partly continuous orthogonality measure for the Koornwinder polynomials (cf. Section 2.6) tend to certain completely discrete measures. These discrete measures can be expressed most conveniently in terms of multidimensional Jackson integrals. They reduce to the orthogonality measures for the little and big q-Jacobi polynomials in the one variable setting. These discrete measures are used to define a four parameter family of multivariable big q-Jacobi polynomials and a three parameter family of multivariable little q-Jacobi polynomials. Orthogonality relations and quadratic norm evaluations for the multivariable big and little q-Jacobi polynomials then follow by taking the limits in the orthogonality relations and the quadratic norm evaluations for the Koornwinder polynomials (cf. previous chapter).

The constant term identities for the multivariable little and big q-Jacobi polynomials reduce to well-known q-analogues of the Selberg integral. The constant term identity

for the multivariable little *q*-Jacobi polynomials (Corollary 3.3.6) is known as the Askey-Habsieger-Kadell formula (see [**5**],[**33**],[**54**]) and was proved in full generality by Aomoto [**4**]. The constant term identity for the multivariable big *q*-Jacobi polynomials with one of the parameters discrete (Corollary 3.5.3) was conjectured by Askey [**5**] and proved by Evans [**29**]. The constant term identity in the general form (Corollary 3.4.8) is equivalent to Tarasov's and Varchenko's constant term identity [**127**, Theorem (E.10)].

In this chapter we develop also a more algebraic method for proving limit transitions between families of multivariable orthogonal polynomials. The method is based on the fact that the multivariable basic hypergeometric orthogonal polynomials under consideration can be characterized for generic parameter values as a special type of eigenfunction for an explicitly known second order q-difference operator. The proof of the limit transitions reduces then to the problem of computing limits of second order q-difference operators and their eigenvalues. Usually this method is less computational and valid for generic parameter values. The disadvantage of this method is that properties of the polynomials, such as orthogonality relations and quadratic norm evaluations, can not be rigorously transported to their limit cases. In this chapter we apply this algebraic method to prove limit transitions from Koornwinder polynomials, multivariable big and little q-Jacobi polynomials to generalized Jacobi polynomials.

This chapter is organized as follows. In Section 3.2 the one variable little and big q-Jacobi polynomials are defined as limit cases of the Askey-Wilson polynomials and their orthogonality relations and quadratic norm evaluations are recalled. In Section 3.3 respectively Section 3.4, the multivariable little respectively big q-Jacobi polynomials are introduced and studied. The proof of the limit from the orthogonality measure of the Koornwinder polynomials to the discrete orthogonality measures of the multivariable little respectively big q-Jacobi polynomials is postponed to Section 3.7 respectively Section 3.8. In Section 3.5 the orthogonality measures of the multivariable big and little q-Jacobi polynomials are considered when one parameter is discrete. Then it is shown that the constant term identities for the multivariable big and little q-Jacobi polynomials reduce to well-known identities which were already studied in the beginning of the 1980's, cf. [5]. In Section 3.6 the algebraic approach for proving limit transitions is developed.

3.2. One variable big and little *q*-Jacobi polynomials

In this section the families of little and big q-Jacobi polynomials are introduced as limit cases of the Askey-Wilson polynomials. For the notations on Askey-Wilson polynomials, we refer the reader to Section 2.2.

The monic little q-Jacobi polynomials $\{P_n^L(.; a, b)\}_{n \in \mathbb{Z}_+}$ can be considered as limit cases of the monic Askey-Wilson polynomials by substituting

(3.2.1)
$$\underline{t}_{L}(\varepsilon) := (\varepsilon^{-1}q^{\frac{1}{2}}, -aq^{\frac{1}{2}}, \varepsilon bq^{\frac{1}{2}}, -q^{\frac{1}{2}})$$

for the four variables of the Askey-Wilson polynomials, rescaling of the z-variable, and taking the limit $\varepsilon \downarrow 0$,

(3.2.2)
$$P_n^L(z;a,b) := \lim_{\varepsilon \downarrow 0} \left(\varepsilon q^{-\frac{1}{2}}\right)^n P_n\left(\varepsilon^{-1} q^{\frac{1}{2}} z; \underline{t}_L(\varepsilon)\right)$$

(3.2.3)
$$= \lim_{\varepsilon \downarrow 0} \left(\varepsilon q^{-\frac{1}{2}} \right)^n P_n \left(\varepsilon^{-1} q^{\frac{1}{2}} z; t_{L,2}(\varepsilon), t_{L,0}(\varepsilon), t_{L,3}(\varepsilon), t_{L,1}(\varepsilon) \right)$$

(3.2.4)
$$= \frac{(qb;q)_n}{(qb)^n (q^{n+1}ab;q)_n} {}_3\phi_2 \begin{pmatrix} q^{-n}, q^{n+1}ab, qbz \\ qb, 0 \end{pmatrix}; q,q \end{pmatrix}$$

(3.2.5)
$$= \frac{(-1)^n q^{\binom{n}{2}} (qa;q)_n}{(q^{n+1}ab;q)_n} _2 \phi_1 \begin{pmatrix} q^{-n}, q^{n+1}ab \\ qa \\ ;q,qz \end{pmatrix}$$

(cf. [66, Proposition 6.3] and take into account that the Askey-Wilson polynomials used in [66] are written as functions of $(z + z^{-1})/2$ and are normalized differently). In fact, an easy calculation yields that the right hand side of (3.2.3) is equal to

(3.2.6)
$$\frac{(qb;q)_n}{(qb)^n (q^{n+1}ab;q)_n} \sum_{m=0}^n \frac{(q^{-n}, q^{n+1}ab;q)_m}{(q, qb;q)_m} (-\varepsilon q^{m+1}b, -\varepsilon q^{m+1}ab;q)_{n-m} \\ \cdot q^m \prod_{i=0}^{m-1} \left((1+\varepsilon^2 b^2 q^{2i+1}) - q^{i+1}b\varepsilon q^{-\frac{1}{2}}h_1(\varepsilon^{-1}q^{\frac{1}{2}}z) \right)$$

with $h_1(z) := z + z^{-1}$, so (3.2.4) follows from the observation that $\lim_{\varepsilon \downarrow 0} (u\varepsilon; q)_m = 1$ and

(3.2.7)
$$\lim_{u \downarrow 0} u h_1(u^{-1}z) = z.$$

A transformation formula for terminating $_2\phi_1$ series [**30**, (III.7), p. 241] yields (3.2.5) and shows that the little *q*-Jacobi polynomials are also defined for b = 0. The little *q*-Jacobi polynomial $P_n^L(z; a, b)$ is a monic polynomial of degree *n* in the variable *z*. So in the limit (3.2.2) we go from a polynomial in $z + z^{-1}$ to a polynomial in *z*. This can be made more transparent as follows. Expand P_n in powers of $z + z^{-1}$,

$$P_n(z;\underline{t}) = \sum_{r=0}^n c_{n,r}(\underline{t})h_r(z) \qquad (c_{n,n} = 1)$$

with $h_r(z) := (h_1(z))^r = (z + z^{-1})^r$. Then (3.2.7) extends to the limit

(3.2.8)
$$\lim_{u \downarrow 0} u^r h_r(u^{-1}z) = z^r \qquad (r \in \mathbb{N})$$

so by (3.2.6) and (3.2.8) we conclude that

$$P_n^L(z; a, b) = \sum_{r=0}^n c_{n,r}^L(a, b) z^r$$

3. LIMIT TRANSITIONS FOR MULTIVARIABLE ORTHOGONAL POLYNOMIALS

with

(3.2.9)
$$\lim_{\varepsilon \downarrow 0} \left(\varepsilon q^{-\frac{1}{2}} \right)^{n-r} c_{n,r}(\underline{t}_L(\varepsilon)) = c_{n,r}^L(a,b).$$

The monic big q-Jacobi polynomials $\{P_n^B(.; a, b, c, d)\}_{n \in \mathbb{Z}_+}$ may be considered as limit cases of the monic Askey-Wilson polynomials by substituting

(3.2.10)
$$\underline{t}_B(\varepsilon) := (\varepsilon^{-1} (qc/d)^{\frac{1}{2}}, -\varepsilon^{-1} (qd/c)^{\frac{1}{2}}, \varepsilon a (qd/c)^{\frac{1}{2}}, -\varepsilon b (qc/d)^{\frac{1}{2}})$$

for the four variables of the Askey-Wilson polynomials, rescaling of the z-variable, and taking the limit $\varepsilon \downarrow 0$:

$$P_{n}^{B}(z; a, b, c, d) := \lim_{\varepsilon \downarrow 0} \left(\varepsilon(cd/q)^{\frac{1}{2}} \right)^{n} P_{n} \left(\varepsilon^{-1}(q/cd)^{\frac{1}{2}} z; \underline{t}_{B}(\varepsilon) \right)$$

(3.2.11)
$$= \lim_{\varepsilon \downarrow 0} \left(\varepsilon(cd/q)^{\frac{1}{2}} \right)^{n} P_{n} \left(\varepsilon^{-1}(q/cd)^{\frac{1}{2}} z; t_{B,2}(\varepsilon), t_{B,0}(\varepsilon), t_{B,1}(\varepsilon), t_{B,3}(\varepsilon) \right)$$

$$= \frac{(qa, -qad/c; q)_{n}}{(q^{n+1}ab; q)_{n}(qa/c)^{n}} _{3} \phi_{2} \left(\begin{array}{c} q^{-n}, q^{n+1}ab, qza/c \\ qa, -qad/c \end{array}; q, q \right)$$

(cf. [66, Proposition 6.1]). Observe that $P_n^B(z; a, b, c, d)$ is a monic polynomial of degree n in the variable z.

Similarly as in the little q-Jacobi case, we have

$$P_n^B(z; a, b, c, d) = \sum_{r=0}^n c_{n,r}^B(a, b, c, d) z^r$$

with

(3.2.12)
$$c_{n,r}^B(a,b,c,d) = \lim_{\varepsilon \downarrow 0} \left(\varepsilon(cd/q)^{\frac{1}{2}} \right)^{n-r} c_{n,r}(\underline{t}_B(\varepsilon)).$$

In the next theorem we give the orthogonality relations and norm evaluations for the monic little q-Jacobi polynomials with parameters $(a, b) \in V_L$, where the parameter domain V_L is defined as follows.

DEFINITION 3.2.1. Let V_L be the set of parameters (a, b) for which $a \in (0, 1/q)$ and $b \in (-\infty, 1/q)$.

Recall the definition of the Jackson *q*-integral and the *q*-Gamma function, which were introduced in Section 1.2.

,

THEOREM 3.2.2. ([1, Theorem 9]) Let
$$(a, b) \in V_L$$
. Then

$$\int_0^1 (P_m^L P_n^L)(z; a, b) v_L(z; a, b) d_q z = \delta_{m,n} \mathcal{N}^L(n; a, b)$$

with

(3.2.13)
$$v_L(z;a,b) := \frac{(qz;q)_{\infty}}{(qbz;q)_{\infty}} z^{\alpha} \qquad (a=q^{\alpha}).$$

The quadratic norms $\mathcal{N}^L(n)$ of the monic little q-Jacobi polynomials are explicitly given by

$$\mathcal{N}^{L}(n;a,b) = \frac{\Gamma_{q}(n+1)\Gamma_{q}(n+1+\alpha)\Gamma_{q}(n+1+\beta)\Gamma_{q}(n+1+\alpha+\beta)}{\Gamma_{q}(2n+1+\alpha+\beta)\Gamma_{q}(2n+2+\alpha+\beta)}q^{(n+\alpha)n},$$

where $b = q^{\beta}$.

Observe that the weights $v_L(z; a, b)$ are positive since $(a, b) \in V_L$. The little q-Jacobi polynomials were first observed by Hahn [33]. A detailed discussion of the orthogonality relations and norm evaluations was given by Andrews and Askey [1]. The orthogonality relations and norm evaluations were derived from the q-binomial formula [1, (3.6)], [30, (II.3), p.236] and the q-Pfaff-Saalschütz formula [1, (3.7)], [30, (II.12), p.237]. The evaluation of the q-Jackson integral over the weight function

(3.2.14)
$$\int_0^1 v_L(z;a,b) d_q z = \frac{\Gamma_q(\alpha+1)\Gamma_q(\beta+1)}{\Gamma_q(2+\alpha+\beta)} \qquad (a=q^{\alpha},b=q^{\beta})$$

is a well-known q-analogue of the beta integral, and is equivalent with the q-binomial formula [30, (II.3), p.236] (see $\S1.2$).

We end this section with the orthogonality relations and norm evaluations for the monic big q-Jacobi polynomials with parameters (a, b, c, d) in the following parameter domain.

DEFINITION 3.2.3. Let V_B be the set of parameters (a, b, c, d) for which c, d > 0and $a \in (-c/dq, 1/q)$, $b \in (-d/cq, 1/q)$ or a = cu, $b = -d\overline{u}$ with $u \in \mathbb{C} \setminus \mathbb{R}$

THEOREM 3.2.4. ([3, Section 3]) Let $(a, b, c, d) \in V_B$. Then

(3.2.15)
$$\int_{-d}^{c} \left(P_{m}^{B} P_{n}^{B} \right)(z; a, b, c, d) v_{B}(z; a, b, c, d) d_{q} z = \delta_{m,n} \mathcal{N}^{B}(n; a, b, c, d),$$

with

(3.2.16)
$$v_B(z; a, b, c, d) := \frac{(qz/c, -qz/d; q)_{\infty}}{(qaz/c, -qbz/d; q)_{\infty}}$$

The quadratic norms $\mathcal{N}^B(n)$ of the monic big q-Jacobi polynomials are explicitly given by

$$\mathcal{N}^{B}(n;a,b,c,d) := \frac{\Gamma_{q}(n+1)\Gamma_{q}(n+1+\alpha)\Gamma_{q}(n+1+\beta)\Gamma_{q}(n+1+\alpha+\beta)}{\Gamma_{q}(2n+1+\alpha+\beta)\Gamma_{q}(2n+2+\alpha+\beta)} \cdot \frac{(cd)^{n+1}q^{\binom{n}{2}}(-c/d,-d/c;q)_{\infty}}{(c+d)\left(-q^{n+1}bc/d,-q^{n+1}ad/c;q\right)_{\infty}}$$

where $a = q^{\alpha}$ and $b = q^{\beta}$.

Observe that the weights $v_B(z; a, b, c, d)$ are positive since $(a, b, c, d) \in V_B$. The big q-Jacobi polynomials were first hinted at by Hahn [33]. A detailed discussion of the orthogonality relations and norm evaluations was given by Andrews and Askey [3].

3. LIMIT TRANSITIONS FOR MULTIVARIABLE ORTHOGONAL POLYNOMIALS

The orthogonality relations and norm evaluations were derived using the q-Vandermonde formula [3, (3.29)], [30, (II.6), p.236] and the evaluation of the q-Jackson integral over the weight function

$$(3.2.17)$$

$$\int_{-d}^{c} v_B(z;a,b,c,d) d_q z = \frac{\Gamma_q(1+\alpha)\Gamma_q(1+\beta)}{\Gamma_q(2+\alpha+\beta)} \frac{\left(-c/d,-d/c;q\right)_{\infty} cd}{\left(-qbc/d,-qad/c;q\right)_{\infty} (c+d)}$$

$$= (1-q)c \frac{\left(q,-d/c,-qc/d,q^2ab;q\right)_{\infty}}{\left(qa,qb,-qbc/d,-qad/c;q\right)_{\infty}}$$

where $a = q^{\alpha}$, $b = q^{\beta}$. The summation formula (3.2.17) is a q-analogue of the beta integral which first appeared in [3, Theorem 1].

It turns out that the orthogonality relations and norm evaluations for the little and big q-Jacobi polynomials can be obtained by taking the limit (3.2.2) respectively (3.2.11) in the orthogonality relations for the Askey-Wilson polynomials. If one chooses the continuous part of the orthogonality measure for the Askey-Wilson polynomials supported on the circle $T = \{z \in \mathbb{C} \mid |z| = 1\}$ using Cauchy's Theorem, one can show by Lebesgue's dominated convergence Theorem that the continuous part of the measure disappears in the limit (3.2.2) respectively (3.2.11). See Section 1.2 for a simplified example of this phenomenon. In case of the little q-Jacobi polynomials, the discrete part of the orthogonality measure blows up to one infinite discrete serie of weights, since there is one parameter in $\underline{t}_L(\varepsilon)$ (3.2.1) which blows up to infinity in absolute value when $\varepsilon \to 0$. In case of the big q-Jacobi polynomials, the discrete part of the orthogonality measure polynomials blows up to two infinite discrete series of weights, since there are two parameters in $\underline{t}_B(\varepsilon)$ (3.2.10) which blows up to infinity in absolute value when $\varepsilon \to 0$. Making this explicit, one obtains rigorous proofs of Theorem 3.2.2 and Theorem 3.2.4 as corollaries of the orthogonality relations and norm evaluations of the Askey-Wilson polynomials (see [122] for details). In Section 3.3 and Section 3.4 this approach is used to generalize Theorem 3.2.2 and Theorem 3.2.4 to the multivariable setting.

3.3. Limit transitions to little *q*-Jacobi polynomials

In this section a multivariable analogue of the limit from Askey-Wilson polynomials to little q-Jacobi polynomials (cf. (3.2.3)) is considered. We will show that in this limit, the positive partly discrete orthogonality measure (Theorem 2.6.6) for the Koornwinder polynomials tends to an infinite discrete measure. The rigorous proof of this fact will be given in Section 3.7. This discrete measure is used to define the multivariable analogues of the little q-Jacobi polynomials. Orthogonality relations and quadratic norm evaluations for the multivariable little q-Jacobi polynomials are then derived by taking limits in the corresponding results for the Koornwinder polynomials.

The orthogonality measure of the multivariable little q-Jacobi polynomials can be expressed most conveniently in terms of certain multidimensional Jackson integrals, which

we will define now first. For a point $\xi \in (\mathbb{C}^*)^n$, the Jackson integral of f over the set (3.3.1) $\langle \xi \rangle_n := \{\xi q^{\nu} \mid \nu \in P(n)\}$

where $\xi q^{\nu} := (\xi_1 q^{\nu_1}, \dots, \xi_n q^{\nu_n})$ and P(n) is given by (2.4.4), is defined by

(3.3.2)
$$\iint_{\langle \xi \rangle_n} f(z) d_q z := (1-q)^n \sum_{\nu \in P(n)} f(\xi q^{\nu}) \prod_{i=1}^n \xi_i q^{\nu_i}$$

provided that the multisum is absolutely convergent. Note that for special points $\xi = (\xi_1, \xi_1 \gamma, \dots, \xi_1 \gamma^{n-1}) \in (\mathbb{C}^*)^n$, the multisum (3.3.2) can be expressed as an iterated Jackson integral (1.2.2) by

(3.3.3)
$$\iint_{\langle \xi \rangle_n} f(z) d_q z = \int_{z_1=0}^{\xi_1} \int_{z_2=0}^{\gamma z_1} \dots \int_{z_n=0}^{\gamma z_{n-1}} f(z) d_q z_n \dots d_q z_1$$

Let $A_{\mathbb{R}}^{\mathfrak{S}}$ be the \mathbb{R} -algebra of \mathfrak{S} -invariant polynomials in the variables z_1, \ldots, z_n . An \mathbb{R} basis for $A_{\mathbb{R}}^{\mathfrak{S}}$ is given by the set of monomials $\{\tilde{m}_{\lambda}\}_{\lambda \in \Lambda}$, where $\tilde{m}_{\lambda}(z) := \sum_{\mu \in \mathfrak{S}^{\lambda}} z^{\mu}$. Define a symmetric bilinear form $\langle ., . \rangle_{L,t}^{a,b}$ on $A_{\mathbb{R}}^{\mathfrak{S}}$ for $t \in (0,1)$ and $(a,b) \in V_L$ (cf. Definition 3.2.1) by

(3.3.4)
$$\langle f,g\rangle_L := \iint_{\langle \rho_L\rangle_n} f(z)g(z)\Delta^L(z)d_qz, \quad f,g \in A^{\mathfrak{S}}_{\mathbb{R}}$$

where $\rho_{L,i} := t^{i-1}$ and where the weight function $\Delta^L(z) = \Delta^L(z; a, b; t)$ is given by

(3.3.5)
$$\Delta^{L}(z) := q^{-2\tau^{2}\binom{n}{3}} t^{-(\alpha+1)\binom{n}{2}} \left(\prod_{i=1}^{n} v_{L}(z_{i})\right) \delta_{qJ}(z), \qquad (a = q^{\alpha}, t = q^{\tau})$$

with v_L (3.2.13) the weight function for the one variable little q-Jacobi polynomials and with interaction factor $\delta_{qJ}(z;t)$ given by

(3.3.6)

$$\delta_{qJ}(z;t) := \prod_{1 \le i < j \le n} |z_i - z_j| |z_i|^{2\tau - 1} (qt^{-1}z_j/z_i;q)_{2\tau - 1}, \qquad (t = q^{\tau}, \tau > 0).$$

It can be shown that the weights $\Delta^L(z)$ in the bilinear form $\langle ., . \rangle_L$ are strict positive for $z \in \langle \rho_L \rangle_n$ and that $\langle f, g \rangle_L$, written out as a multidimensional infinite sum, is absolutely convergent for all $f, g \in A^{\mathfrak{S}}_{\mathbb{R}}$. For a detailed proof of this fact, we refer the reader to [116, Section 6].

DEFINITION 3.3.1. Let $t \in (0,1)$ and $(a,b) \in V_L$. The multivariable little q-Jacobi polynomials $\{P_{\lambda}^L(.;a,b;t)\}_{\lambda \in \Lambda}$ are by definition the unique symmetric polynomials which satisfy

mials which satisfy (a) $P_{\lambda}^{L} = \tilde{m}_{\lambda} + \sum_{\mu < \lambda} c_{\lambda,\mu}^{L} \tilde{m}_{\mu}$ for certain $c_{\lambda,\mu}^{L} = c_{\lambda,\mu}^{L}(a,b;t) \in \mathbb{R};$ (b) $\langle P_{\lambda}^{L}, \tilde{m}_{\mu} \rangle_{L} = 0$ for $\mu < \lambda$.

3. LIMIT TRANSITIONS FOR MULTIVARIABLE ORTHOGONAL POLYNOMIALS

In the following proposition the link is established between Koornwinder polynomials with positive partly discrete orthogonality measure and the little q-Jacobi polynomials. We use in the formulation of the proposition the notation $|\lambda| := \sum_{i=1}^{n} \lambda_i$ for the length of a partition $\lambda \in \Lambda$.

PROPOSITION 3.3.2. Let $t \in (0,1)$ and $(a,b) \in V_L$. There exists a sequence of positive real numbers $\{\varepsilon_k\}_{k \in \mathbb{Z}_+}$ which converges to 0, such that

(3.3.7)
$$\lim_{k \to \infty} \left(\prod_{i=1}^{n} \left(-\varepsilon_k^{-1} q t^{i-1}, -\varepsilon_k^{-1} q a t^{i-1}; q \right)_{\infty} \right) \left(\varepsilon_k q^{-\frac{1}{2}} \right)^{|\lambda| + |\mu|} \langle m_\lambda, m_\mu \rangle_{\underline{t}_L(\varepsilon_k), t}$$
$$= 2^n n! \left(q; q \right)_{\infty}^{-2n} (1-q)^{-n} \langle \tilde{m}_\lambda, \tilde{m}_\mu \rangle_{L, t}^{a, b}$$

for all $\lambda, \mu \in \Lambda$, where $\langle ., . \rangle_{t,t}$ is given by (2.6.11) and \underline{t}_L is given by (3.2.1).

The proof of the proposition will be given in Section 3.7. Observe that $\underline{t}_L(\varepsilon) \in V_K$ for $\varepsilon \in \mathbb{R}_{>0}$ sufficiently small, so $\langle ., . \rangle_{\underline{t}_L(\varepsilon), t}$ is well defined and positive definite for $\varepsilon \in \mathbb{R}_{>0}$ sufficiently small by Lemma 2.6.4 and Corollary 2.6.5.

Proposition 3.3.2 will be used to prove that the multivariable little q-Jacobi polynomials are limit cases of the Koornwinder polynomials and to establish orthogonality relations and norm evaluations for the little q-Jacobi polynomials with respect to the scalar product $\langle ., . \rangle_L$ on $A_{\mathbb{R}}^{\mathfrak{S}}$.

The following definition of limit transitions between \mathfrak{S} -invariant Laurent polynomials will be used. Let f(.; u) $(u \in \mathbb{R}^*)$ and f be \mathfrak{S} -invariant Laurent polynomials in n variables z_1, \ldots, z_n , then we write $\lim_{u\to 0} f(.; u) = f$ if $\lim_{u\to 0} f(z; u) = f(z)$ for all $z \in (\mathbb{C}^*)^n$. Observe that the \mathbb{R} -algebra of \mathfrak{S} -invariant Laurent polynomials has as \mathbb{R} -basis the set of monomials $\{\tilde{m}_{\lambda}(z)\}_{\lambda\in\tilde{\Lambda}}$, where $\tilde{\Lambda} := \{\lambda \in \mathbb{Z}^n \mid \lambda_1 \ge \lambda_2 \ge \ldots \ge \lambda_n\}$ and $\tilde{m}_{\lambda}(z) := \sum_{\mu\in\mathfrak{S}\lambda} z^{\mu}$. If $f(.; u) = \sum_{\lambda\in\tilde{\Lambda}} c_{\lambda}(u)\tilde{m}_{\lambda}$ $(u \in \mathbb{R}^*)$ and $f = \sum_{\lambda\in\tilde{\Lambda}} c_{\lambda}\tilde{m}_{\lambda}$ satisfy the additional condition that $\{\lambda \in \tilde{\Lambda} \mid c_{\lambda}(u) \neq 0\}$ is contained in some finite u-independent subset for |u| sufficiently small, then $\lim_{u\to 0} f(.; u) = f$ iff $\lim_{u\to 0} c_{\lambda}(u) = c_{\lambda}$ for all $\lambda \in \tilde{\Lambda}$. Crucial in the limit from Koornwinder polynomials to little q-Jacobi polynomials is a limit from rescaled monomials $m_{\lambda}(z|u)$ to $\tilde{m}_{\lambda}(z)$, where the rescaled monomial $m_{\lambda}(z|u)$ for $u \in \mathbb{R}^*$ is the \mathfrak{S} -invariant Laurent polynomial given by

(3.3.8)
$$m_{\lambda}(z|u) := u^{|\lambda|} m_{\lambda}(u^{-1}z), \quad \lambda \in \Lambda,$$

where $u^{-1}z := (u^{-1}z_1, \ldots, u^{-1}z_n)$. In terms of the basis $\{\tilde{m}_{\mu}\}_{\mu \in \tilde{\Lambda}}$, we have

$$m_{\lambda}(z|u) = \sum_{\mu \in \tilde{\Lambda} \cap \mathcal{W}\lambda} d_{\lambda,\mu}(u) \tilde{m}_{\mu}(z), \quad \lambda \in \Lambda$$

with $d_{\lambda,\mu}(u)$ homogeneous of degree $|\lambda| - |\mu|$ and $d_{\lambda,\lambda}(u) \equiv 1$. Furthermore, $|\mu| \le |\lambda|$ if $\mu \in \mathcal{W}\lambda$ and $|\lambda| = |\mu|$ iff $\mu \in \mathfrak{S}\lambda$. Hence we obtain the limit transitions

(3.3.9)
$$\lim_{u \to 0} m_{\lambda}(z|u) = \tilde{m}_{\lambda}(z) \qquad (\lambda \in \Lambda).$$

This limit will play a fundamental role for the multivariable generalization of the limit from Askey-Wilson polynomials to little q-Jacobi polynomials.

The quadratic norm of the little q-Jacobi polynomials can be expressed in terms of functions $\mathcal{N}_{qJ}^+(\lambda) = \mathcal{N}_{qJ}^+(\lambda; a, b; t)$ and $\mathcal{N}_{qJ}^-(\lambda) = \mathcal{N}_{qJ}^-(\lambda; a, b; t)$ which are defined by

$$\mathcal{N}_{qJ}^{+}(\lambda) := \prod_{i=1}^{n} \frac{\Gamma_{q}(\lambda_{i}+1+(n-i)\tau+\alpha+\beta)\Gamma_{q}(\lambda_{i}+1+(n-i)\tau+\alpha)}{\Gamma_{q}(2\lambda_{i}+1+2(n-i)\tau+\alpha+\beta)}$$

$$(3.3.10) \qquad \qquad \prod_{1 \leq j < k \leq n} \left(\frac{\Gamma_{q}(\lambda_{j}+\lambda_{k}+1+(2n-j-k+1)\tau+\alpha+\beta)}{\Gamma_{q}(\lambda_{j}+\lambda_{k}+1+(2n-j-k)\tau+\alpha+\beta)} \\ \cdot \frac{\Gamma_{q}(\lambda_{j}-\lambda_{k}+(k-j+1)\tau)}{\Gamma_{q}(\lambda_{j}-\lambda_{k}+(k-j)\tau)} \right),$$

$$\mathcal{N}_{qJ}^{-}(\lambda) := \prod_{i=1}^{n} \frac{\Gamma_{q}(\lambda_{i}+1+(n-i)\tau)\Gamma_{q}(\lambda_{i}+1+(n-i)\tau+\beta)}{\Gamma_{q}(2\lambda_{i}+2+2(n-i)\tau+\alpha+\beta)}$$

$$(3.3.11) \qquad \cdot \prod_{1 \le j < k \le n} \left(\frac{\Gamma_{q}(\lambda_{j}+\lambda_{k}+2+(2n-j-k-1)\tau+\alpha+\beta)}{\Gamma_{q}(\lambda_{j}+\lambda_{k}+2+(2n-j-k)\tau+\alpha+\beta)} \\ \cdot \frac{\Gamma_{q}(\lambda_{j}-\lambda_{k}+1+(k-j-1)\tau)}{\Gamma_{q}(\lambda_{j}-\lambda_{k}+1+(k-j)\tau)} \right),$$

where $a = q^{\alpha}$, $b = q^{\beta}$, $t = q^{\tau}$ and where $\Gamma_q(z)$ is the q-Gamma function given by (1.2.4). For $\lambda \in \Lambda$, $(a, b) \in V_L$ and $t \in (0, 1)$ define $\mathcal{N}^L(\lambda) = \mathcal{N}^L(\lambda; a, b; t)$ by

(3.3.12)
$$\mathcal{N}^{L}(\lambda) := q^{\sum_{i=1}^{n} (\lambda_i + \alpha + 2(n-i)\tau)\lambda_i} \mathcal{N}^{+}_{qJ}(\lambda) \mathcal{N}^{-}_{qJ}(\lambda)$$

Observe that $\mathcal{N}^L(\lambda)$ is well defined and positive. We have now the following multivariable generalization of Theorem 3.2.2 and of the limit transition (3.2.3).

THEOREM 3.3.3. Let $(a, b) \in V_L$ and $t \in (0, 1)$. There exists a sequence of positive real numbers $\{\varepsilon_k\}_{k \in \mathbb{Z}_+}$ which converges to 0, such that

(3.3.13)
$$\lim_{k \to \infty} \left(q^{-\frac{1}{2}} \varepsilon_k \right)^{|\lambda|} P_{\lambda} \left(q^{\frac{1}{2}} \varepsilon_k^{-1} z; \underline{t}_L(\varepsilon_k); t \right) = P_{\lambda}^L(z; a, b; t)$$

for all $\lambda \in \Lambda$. Furthermore, the polynomials $\{P_{\lambda}^{L}\}_{\lambda \in \Lambda}$ are orthogonal with respect to $\langle ., . \rangle_{L}$ and the quadratic norms of the little q-Jacobi polynomials are given by

(3.3.14)
$$\langle P_{\lambda}^{L}, P_{\lambda}^{L} \rangle_{L} = \mathcal{N}^{L}(\lambda), \quad \lambda \in \Lambda.$$

PROOF. We write

(3.3.15)
$$(q^{-\frac{1}{2}}\varepsilon)^{|\lambda|} P_{\lambda}(z; \underline{t}_{L}(\varepsilon); t) = \sum_{\mu \leq \lambda} c_{\lambda,\mu}(\varepsilon) (q^{-\frac{1}{2}}\varepsilon)^{|\mu|} m_{\mu}(z),$$
$$P_{\lambda}^{L}(z; a, b; t) = \sum_{\mu \leq \lambda} c_{\lambda,\mu}^{L} \tilde{m}_{\mu}(z)$$

3. LIMIT TRANSITIONS FOR MULTIVARIABLE ORTHOGONAL POLYNOMIALS

for the expansions of the Koornwinder polynomial and the multivariable little q-Jacobi polynomial in terms of monomials. In particular, we have $c_{\lambda,\lambda}(\varepsilon) = c_{\lambda,\lambda}^L = 1$.

Let \leq be a total order on Λ such that $\mu \leq \lambda$ if $\mu \leq \lambda$. Let $\{\varepsilon_k\}_{k \in \mathbb{Z}_+}$ be a sequence in $\mathbb{R}_{>0}$ converging to 0 such that (3.3.7) is satisfied for all $\lambda, \mu \in \Lambda$. We prove that

(3.3.16)
$$\lim_{k \to \infty} c_{\lambda,\mu}(\varepsilon_k) = c_{\lambda,\mu}^L, \quad \forall \mu \le \lambda,$$

and we prove full orthogonality for the subset $\{P_{\mu}^{L}\}_{\mu \leq \lambda}$ of multivariable little *q*-Jacobi polynomials by induction on $\lambda \in \Lambda$ along \leq . The limit (3.3.13) follows then from (3.3.9), (3.3.15) and (3.3.16), and the quadratic norm evaluations (3.3.14) are then immediate consequences of the quadratic norm evaluations of the Koornwinder polynomials (see Theorem 2.6.6), Proposition 3.3.2, (3.3.16) and the observation that

$$\begin{split} \lim_{\varepsilon \to 0} \left(\prod_{i=1}^n \left(-\varepsilon^{-1} q t^{i-1}, -\varepsilon^{-1} q a t^{i-1}; q \right)_{\infty} \right) (q^{-1/2} \varepsilon)^{2|\lambda|} \mathcal{N}(\lambda; \underline{t}_L(\varepsilon); t) \\ &= 2^n n! (q; q)_{\infty}^{-2n} (1-q)^{-n} \mathcal{N}^L(\lambda; a, b; t). \end{split}$$

So it remains to prove the induction step (the case $\lambda = 0$ being trivial). For $\lambda \neq 0$, note that $\underline{t}_L(\varepsilon) \in V_K$ for $\varepsilon > 0$ sufficiently small, hence by Theorem 2.6.6 we can write

(3.3.17)
$$(q^{-\frac{1}{2}}\varepsilon)^{|\lambda|} P_{\lambda}(z;\underline{t}_{L}(\varepsilon);t) = (q^{-\frac{1}{2}}\varepsilon)^{|\lambda|} m_{\lambda}(z) - \sum_{\nu < \lambda} d_{\lambda,\nu}(\varepsilon) (q^{-\frac{1}{2}}\varepsilon)^{|\nu|} P_{\nu}(z;\underline{t}_{L}(\varepsilon);t)$$

with

$$d_{\lambda,\nu}(\varepsilon) := \frac{(q^{-\frac{1}{2}}\varepsilon)^{|\lambda|+|\nu|} \langle m_{\lambda}, P_{\nu}(.; \underline{t}_{L}(\varepsilon); t) \rangle_{\underline{t}_{L}(\varepsilon), t}}{(q^{-\frac{1}{2}}\varepsilon)^{2|\nu|} \langle P_{\nu}(.; \underline{t}_{L}(\varepsilon); t), P_{\nu}(.; \underline{t}_{L}(\varepsilon); t) \rangle_{\underline{t}_{L}(\varepsilon), t}}$$

for $\varepsilon > 0$ sufficiently small. By the induction hypotheses, we also have

(3.3.18)
$$P_{\lambda}^{L}(z;a,b;t) = \tilde{m}_{\lambda}(z) - \sum_{\nu < \lambda} d_{\lambda,\nu}^{L} P_{\nu}^{L}(z;a,b;t),$$

with

$$d_{\lambda,\nu}^{L} = \frac{\langle \tilde{m}_{\lambda}, P_{\nu}^{L}(.; a, b; t) \rangle_{L,t}^{a,o}}{\langle P_{\nu}^{L}(.; a, b; t), P_{\nu}^{L}(.; a, b; t) \rangle_{L,t}^{a,b}}.$$

It follows that for $\mu < \lambda$,

(3.3.19)
$$c_{\lambda,\mu}(\epsilon) = -\sum_{\mu \le \nu < \lambda} d_{\lambda,\nu}(\epsilon) c_{\nu,\mu}(\epsilon), \qquad c_{\lambda,\mu}^L = -\sum_{\mu \le \nu < \lambda} d_{\lambda,\nu}^L c_{\nu,\mu}^L$$

for $\epsilon \in \mathbb{R}_{>0}$ sufficiently small. Again by the induction hypotheses and by Proposition 3.3.2, we obtain

(3.3.20)
$$\lim_{k \to \infty} d_{\lambda,\nu}(\varepsilon_k) = d_{\lambda,\nu}^L \quad \forall \, \nu < \lambda.$$

So the limits (3.3.16) follow from the induction hypotheses, (3.3.19) and (3.3.20). The orthogonality relations for $\{P_{\mu}^{L}\}_{\mu \leq \lambda}$ now follow by taking limits in the orthogonality relations for the Koornwinder polynomials (see Theorem 2.6.6). This completes the proof of the induction step.

Write $e_r(z) = m_{(1^r)}(z)$ and $\tilde{e}_r(z) = \tilde{m}_{(1^r)}(z)$ for the \mathcal{W} -invariant and \mathfrak{S} -invariant monomials corresponding to the fundamental weights $(1^r) \in \Lambda$ $(r \in [1, n])$. The monomials $\{e_r\}_{r=1}^n$ and $\{\tilde{e}_r\}_{r=1}^n$ are algebraically independent generators of the algebras $A^{\mathcal{W}}$ and $A^{\mathfrak{S}}$ respectively. Let \hat{P}^L_{λ} $(\lambda \in \Lambda)$ be the unique polynomial in n variables $y = (y_1, \ldots, y_n)$ satisfying

$$\hat{P}^L_{\lambda}(\tilde{e}_1(z),\ldots,\tilde{e}_n(z)) = P^L_{\lambda}(z).$$

Similarly, we set \hat{P}_{λ} for the unique polynomial in the *n* variables *y* satisfying

$$\hat{P}_{\lambda}(e_1(z),\ldots,e_n(z))=P_{\lambda}(z).$$

The limit transition (3.3.13) can now be reformulated as follows.

COROLLARY 3.3.4. Let
$$\lambda \in \Lambda$$
, $(a,b) \in V_L$ and $t \in (0,1)$, then

$$\lim_{k \in k} (s_{\varepsilon_k})^{-|\lambda|} \hat{P}_{\lambda}(s_{\varepsilon_k}y_1, \dots, (s_{\varepsilon_k})^n y_n; \underline{t}_L(\varepsilon_k); t) = \hat{P}_{\lambda}^L(y; a, b; t)$$

for certain sequence $\{\varepsilon_k\}_{k\in\mathbb{Z}_+}$ in $\mathbb{R}_{>0}$ converging to zero, where $s_{\varepsilon} := q^{\frac{1}{2}}\varepsilon^{-1}$.

PROOF. The corollary follows from the proof of Theorem 3.3.3 since

$$(s_{\varepsilon})^{-r}e_r(s_{\varepsilon}z) = \tilde{e}_r(z) + \mathcal{O}(\varepsilon)$$

for $r \in [1, n]$.

Let $D_L = D_{L,a,b,t}$ be the second order q-difference operator

(3.3.21)
$$D_L = \sum_{j=1}^n \left(\phi_{L,j}^+(z) (T_j^+ - \operatorname{Id}) + \phi_{L,j}^-(z) (T_j^- - \operatorname{Id}) \right)$$

where $\phi^{\pm}_{L,j}(z)=\phi^{\pm}_{L,j}(z;a,b;t)$ is given by

(3.3.22)

$$\phi_{L,j}^{+}(z) := qt^{n-1}a\left(b - \frac{1}{qz_{j}}\right)\prod_{i \neq j} \frac{z_{i} - tz_{j}}{z_{i} - z_{j}},$$

$$\phi_{L,j}^{-}(z) := \left(1 - \frac{1}{z_{j}}\right)\prod_{i \neq j} \frac{z_{j} - tz_{i}}{z_{j} - z_{i}}.$$

Then we have the following corollary of Theorem 3.3.3.

COROLLARY 3.3.5. Let $(a,b) \in V_L$ and $t \in (0,1)$. Then $P^L_{\lambda}(.;a,b;t)$ is an eigenfunction of $D_{L,a,b,t}$ with eigenvalue

(3.3.23)
$$E_{\lambda}^{qJ}(a,b;t) := \sum_{j=1}^{n} \left(qabt^{2n-j-1}(q^{\lambda_j}-1) + t^{j-1}(q^{-\lambda_j}-1) \right)$$

for all $\lambda \in \Lambda$.

PROOF. Follows from (3.3.13) by taking the limit $k \to \infty$ in the equations

(3.3.24)
$$(q^{-1}\varepsilon_k)^{|\lambda|} ((D-E_\lambda)P_\lambda) (q^{\frac{1}{2}}\varepsilon_k^{-1}z; \underline{t}_L(\varepsilon_k); t) = 0 \quad (\lambda \in \Lambda)$$

where D is given by (2.3.13) and E_{λ} is given by (2.3.16).

In [16] van Diejen introduced n commuting q-difference operators D_i $(i \in [1, n])$ with $D_1 = D$ which simultaneously diagonalize the Koornwinder polynomials. It is possible to explicitly compute the limit (3.3.13) in the corresponding eigenvalue equations, as was done for $D_1 = D$ in Corollary 3.3.5. This yields n explicit, commuting q-difference operators $D_{L,1} = D_L, D_{L,2}, \ldots, D_{L,n}$ which simultaneously diagonalize the little q-Jacobi polynomials. The explicit formulas are omitted here, since we will not be needing them in the remainder of the thesis.

The constant term identity for the little q-Jacobi polynomials can be rewritten as follows.

COROLLARY 3.3.6. For $t \in (0, 1)$ and $(a, b) \in V_L$ we have

(3.3.25)
$$\langle 1,1\rangle_{L,t}^{a,b} = \prod_{j=1}^{n} \frac{\Gamma_q(\alpha+1+(j-1)\tau)\Gamma_q(\beta+1+(j-1)\tau)\Gamma_q(j\tau)}{\Gamma_q(\alpha+\beta+2+(n+j-2)\tau)\Gamma_q(\tau)},$$

where $a = q^{\alpha}, b = q^{\beta}$ and $t = q^{\tau}$.

The constant term identity (3.3.25) has been studied extensively in the past 20 years. It was conjectured by Askey [5] for $t = q^k$, $k \in \mathbb{N}$ and proved in this case independently by Habsieger [33] and Kadell [54] (see Section 3.5 for more details). For arbitrary $t \in (0, 1)$ the first proof appeared in Aomoto's paper [4] (see also [55] and [127] for alternative proofs).

3.4. Limit transitions to big q-Jacobi polynomials

In this section a multivariable analogue of the limit transition from Askey-Wilson polynomials to big q-Jacobi polynomials (cf. (3.2.11)) is considered. By repeating the methods of the previous section, we derive orthogonality relations and quadratic norm evaluations for multivariable analogues of the big q-Jacobi polynomials.

Before we define the orthogonality measure for the multivariable big q-Jacobi polynomials we first need to introduce some more notations. Set

$$\langle \xi, \eta \rangle_n := \bigcup_{j=0}^n \langle \xi \rangle_j \times \langle \eta \rangle_{n-j} \subset \mathbb{C}^n$$

where $\eta, \xi \in (\mathbb{C}^*)^n$ and $\langle \xi \rangle_n$ is defined by (3.3.1). Here we use the convention that $\langle \xi \rangle_n \times \langle \eta \rangle_0 = \langle \xi \rangle_n$ and $\langle \xi \rangle_0 \times \langle \eta \rangle_n = \langle \eta \rangle_n$. Let $\underline{c} = (c_0, \ldots, c_n) \in (\mathbb{C}^*)^{n+1}$, then we

58

define the <u>c</u>-weighted Jackson integral of f over the set $\langle \eta, \xi \rangle_n$ by

(3.4.1)
$$\iint_{\langle \xi,\eta\rangle_n} f(z)d_q^c z := \sum_{j=0}^n (-1)^{n-j} c_j \iint_{z\in\langle\xi\rangle_j} \iint_{w\in\langle\eta\rangle_{n-j}} f(z,w)d_q z d_q w$$
$$= (1-q)^n \sum_{j=0}^n \sum_{\substack{\mu\in P(j)\\\nu\in P(n-j)}} c_j f(\xi q^\mu, \eta q^\nu) \prod_{l=1}^j \xi_l q^{\mu_l} \prod_{m=1}^{n-j} (-\eta_m q^{\nu_m}),$$

where the j = 0 respectively j = n term in (3.4.1) should be read as

$$(-1)^{n} c_{0} \iint_{w \in \langle \eta \rangle_{n}} f(w) d_{q} w = (1-q)^{n} \sum_{\nu \in P(n)} c_{0} f(\eta q^{\nu}) \prod_{m=1}^{n} \left(-\eta_{m} q^{\nu_{m}} \right)$$

respectively

$$c_n \iint_{z \in \langle \xi \rangle_n} f(z) d_q z = (1-q)^n \sum_{\mu \in P(n)} c_n f(\xi q^{\mu}) \prod_{l=1}^n \xi_l q^{\mu_l}.$$

If $\eta = (\eta_1, \eta_1 \gamma', \dots, \eta_1 (\gamma')^{n-1})$ and $\xi = (\xi_1, \xi_1 \gamma, \dots, \xi_1 \gamma^{n-1})$, then the <u>c</u>-weighted Jackson integral over $\langle \eta, \xi \rangle_n$ can be rewritten as an iterated Jackson integral by

$$\iint_{\langle \xi, \eta \rangle_n} f(z) d_q^c z = \sum_{j=0}^n c_j \int_{z_1=0}^{\xi_1} \int_{z_2=0}^{\gamma z_1} \dots \int_{z_j=0}^{\gamma z_{j-1}} \int_{z_{j+1}=\eta_1}^0 \int_{z_{j+2}=\gamma' z_{j+1}}^0 \dots \int_{z_n=\gamma' z_{n-1}}^0 f(z) d_q z_n \dots d_q z_1.$$

Define a symmetric bilinear form $\langle ., . \rangle_{B,t}^{a,b,c,d}$ on $A_{\mathbb{R}}^{\mathfrak{S}}$ for $t \in (0,1)$ and $(a,b,c,d) \in V_B$ (cf. Definition 3.2.3) by

(3.4.2)
$$\langle f,g\rangle_B := \iint_{\langle \rho_B,\sigma_B\rangle_n} f(z)g(z)\Delta^B(z)d_q^{\mathcal{L}_B}z, \quad f,g\in A^{\mathfrak{S}}_{\mathbb{R}},$$

with $\rho_{B,i}:=ct^{i-1},$ $\sigma_{B,i}:=-dt^{i-1}$ and with weight function

(3.4.3)
$$\Delta^B(z) := \left(\prod_{i=1}^n v_B(z_i)\right) \delta_{qJ}(z),$$

where v_B (3.2.16) is the weight function in the orthogonality measure for the one variable big q-Jacobi polynomials and $\delta_{qJ}(z) = \delta_{qJ}(z;t)$ is given by (3.3.6). The weight $\underline{c}_B = \underline{c}_B(c,d;t)$ is of the form $c_{B,j} := c_B d_{B,j}$, with

(3.4.4)
$$d_{B,j} := \prod_{\substack{1 \le k < m \le n \\ k \le j}} \Psi_t(-t^{n-m-k+1}d/c)$$

where $\Psi_t(x)$ is defined by

(3.4.5)
$$\Psi_t(x) := |x|^{2\tau - 1} \frac{\theta(tx)}{\theta(qt^{-1}x)}$$

with $\theta(x)$ the Jacobi theta function

(3.4.6)
$$\theta(x) := \left(q, x, qx^{-1}; q\right)_{\infty}$$

The constant c_B is defined by

(3.4.7)
$$c_B := \frac{(q;q)_{\infty}^n q^{-2\tau^2\binom{n}{3}} d^{-2\tau\binom{n}{2}-n} t^{-\binom{n}{2}}}{\prod_{i=1}^n \theta(-t^{1-i}c/d)}$$

The positive constant c_B is not essential for the definition of $\langle ., . \rangle_B$. We have chosen to take this constant within the definition of $\langle ., . \rangle_B$ because it will simplify formulas and notations later on. The Jacobi theta function satisfies the functional relation

(3.4.8)
$$\theta(q^k x) = (-x^{-1})^k q^{-\binom{k}{2}} \theta(x), \quad k \in \mathbb{Z}_+.$$

This implies that Ψ_t is a quasi constant, i.e. $\Psi_t(qx) = \Psi_t(x)$. In particular, the weight \underline{d}_B (3.4.4) is independent of a, b and quasi constant in the parameters c, d.

The weight \underline{c}_B in the definition of $\langle ., . \rangle_B$ is needed in order to obtain good asymptotic behaviour of the weights occuring in $\langle ., . \rangle_B$. To be more precise, let $j \in [1, n], \lambda \in P(j-1), \mu \in P(n-j)$ and set $\lambda^{(l)} := (\lambda, l) \in P(j)$ for $l \geq \lambda_{j-1}$, respectively $\mu^{(m)} := (\mu, m) \in P(n-j+1)$ for $m \geq \mu_{n-j}$. For j = 1 respectively j = n, this should be read as $\lambda^{(l)} = l \in P(1)$, respectively $\mu^{(m)} = m \in P(1)$. Define

(3.4.9)
$$z^{+}(l;\lambda,\mu) := \left(\rho_{B}q^{\lambda^{(l)}}, \sigma_{B}q^{\mu}\right) \in \langle\rho_{B}\rangle_{j} \times \langle\sigma_{B}\rangle_{n-j} \quad (l \ge \lambda_{j-1}), \\ z^{-}(m;\lambda,\mu) := \left(\rho_{B}q^{\lambda}, \sigma_{B}q^{\mu^{(m)}}\right) \in \langle\rho_{B}\rangle_{j-1} \times \langle\sigma_{B}\rangle_{n-j+1} \quad (m \ge \mu_{n-j}),$$

then we have

$$\lim_{l \to \infty} z^+(l; \lambda, \mu) = (\rho_B q^\lambda, 0, \sigma_B q^\mu), \quad \lim_{m \to \infty} z^-(m; \lambda, \mu) = (\rho_B q^\lambda, \sigma_B q^\mu, 0)$$

(with the obvious conventions when j = 1 respectively j = n). We have now good asymptotic behaviour of the weights in the following sense.

LEMMA 3.4.1. Let $(a, b, c, d) \in V_B$ and $t \in (0, 1)$. Then

(3.4.10)
$$\lim_{l \to \infty} c_{B,j} \Delta^B(z^+(l;\lambda,\mu)) = \lim_{m \to \infty} c_{B,j-1} \Delta^B(z^-(m;\lambda,\mu))$$

for all $\lambda \in P(j-1)$, $\mu \in P(n-j)$ and $j \in [1,n]$. The conditions (3.4.10) for $\lambda \in P(j-1)$, $\mu \in P(n-j)$ and $j \in [1,n]$ characterize the weight \underline{c}_B uniquely up to a multiplicative constant.

PROOF. Let $\lambda \in P(j-1)$ and $\mu \in P(n-j)$. By direct calculation we obtain

$$\lim_{m \to \infty} \Delta^B(z^{-}(m;\lambda,\mu)) = \delta_{qJ} \left(\rho_B q^{\lambda}, \sigma_B q^{\mu} \right) \prod_{i=1}^{j-1} \left(c t^{i-1} q^{\lambda_i} \right)^{2\tau} \prod_{k=1}^{n-j} \left(d t^{k-1} q^{\mu_k} \right)^{2\tau},$$

where $t = q^{\tau}$ and δ_{qJ} is the interaction factor (3.3.6) considered as function of n - 1 variables. Using the equality

$$q^{k(2\tau-1)} (q^{1-\tau-k}x;q)_{2\tau-1} = \frac{(q^{\tau}x^{-1};q)_k}{(q^{1-\tau}x^{-1};q)_k} (q^{1-\tau}x;q)_{2\tau-1}$$

for $k \in \mathbb{Z}_+$ we furthermore have that

$$\lim_{l \to \infty} \Delta^B(z^+(l;\lambda,\mu)) = \delta_{qJ} \left(\rho_B q^\lambda, \sigma_B q^\mu \right) \\ \cdot \prod_{i=1}^{j-1} \left(c t^{i-1} q^{\lambda_i} \right)^{2\tau} \prod_{k=1}^{n-j} \left\{ \left(d t^{k-1} q^{\mu_k} \right)^{2\tau} \Psi_t \left(-t^{j-k} q^{-\mu_k} c/d \right) \right\}.$$

The lemma follows now from the formula

$$\frac{c_{B,j-1}}{c_{B,j}} = \prod_{k=1}^{n-j} \Psi_t(-t^{j-k}c/d)$$

since Ψ_t is a quasi-constant.

Finally, it can be shown that the bilinear form $\langle ., . \rangle_B$ (3.4.2) on $A_{\mathbb{R}}^{\mathfrak{S}}$ is well defined and positive definite. Indeed, for parameters $(a, b, c, d) \in V_B$ and $t \in (0, 1)$ it can be shown that the weights $\Delta^B(z)$ in the bilinear form $\langle ., . \rangle_B$ are strict positive for $z \in \langle \rho_B, \sigma_B \rangle_n$ and that $\langle f, g \rangle_B$, written out as a multidimensional infinite sum over $\langle \sigma_B, \rho_B \rangle_n$, is absolutely convergent for all $f, g \in A_{\mathbb{R}}^{\mathfrak{S}}$. We refer the reader to [116, Section 6] for a detailed proof of this fact.

DEFINITION 3.4.2. Let $t \in (0,1)$ and $(a,b,c,d) \in V_B$. The multivariable big q-Jacobi polynomials $\{P_{\lambda}^B(.;a,b,c,d;t)\}_{\lambda \in \Lambda}$ are by definition the unique symmetric polynomials which satisfy

(a) $P_{\lambda}^{B} = \tilde{m}_{\lambda} + \sum_{\mu < \lambda} c_{\lambda,\mu}^{B} \tilde{m}_{\mu}$ for certain constants $c_{\lambda,\mu}^{B} = c_{\lambda,\mu}^{B}(a, b, c, d; t) \in \mathbb{R}$; (b) $\langle P_{\lambda}^{B}, \tilde{m}_{\mu} \rangle_{B} = 0$ for $\mu < \lambda$.

The following lemma shows that the multivariable big q-Jacobi polynomials depend essentially only on a, b, t and the ratio c/d.

LEMMA 3.4.3. For $t \in (0, 1)$ and $(a, b, c, d) \in V_B$ we have

$$(3.4.11) d^{-|\lambda|} P^B_{\lambda}(dz; a, b, c, d; t) = P^B_{\lambda}(z; a, b, c/d, 1; t), \quad \forall \lambda \in \Lambda.$$

PROOF. For $f \in A_{\mathbb{R}}^{\mathfrak{S}}$ set $f_d(z) := f(dz)$, where $dz := (dz_1, \ldots, dz_n)$. Then it suffices to show that

$$\langle f,g \rangle_{B,t}^{a,b,c,d} = \langle f_d,g_d \rangle_{B,t}^{a,b,\frac{c}{d},1}, \quad \forall f,g \in A_{\mathbb{R}}^{\mathfrak{S}}.$$

But this is a direct consequence of the formulas

(3.4.12)
$$\int_0^{\alpha\gamma} h(x)d_qx = \alpha \int_0^{\gamma} h(\alpha x)d_qx,$$

$$\underline{c}_B(c,d;t) = d^{-2\tau\binom{n}{2}-n} \underline{c}_B(c/d,1;t)$$

and $\Delta^B(dz; a, b, c, d; t) = d^{2\tau \binom{n}{2}} \Delta^B(z; a, b, c/d, 1; t)$, where $t = q^{\tau}$.

The following proposition is the analogue of Proposition 3.3.2 in case of the big q-Jacobi polynomials.

PROPOSITION 3.4.4. Let $t \in (0, 1)$ and $(a, b, c, d) \in V_B$. There exists a sequence of positive real numbers $\{\varepsilon_k\}_{k \in \mathbb{Z}_+}$ which converges to 0, such that

(3.4.13)
$$\lim_{k \to \infty} \left(\prod_{i=1}^{n} \left(-\varepsilon_k^{-2} q t^{i-1}; q \right)_{\infty} \right) \left(\varepsilon_k (cd/q)^{\frac{1}{2}} \right)^{|\lambda| + |\mu|} \langle m_\lambda, m_\mu \rangle_{\underline{t}_B(\varepsilon_k), t}$$
$$= 2^n n! \left(q; q \right)_{\infty}^{-2n} (1-q)^{-n} \langle \tilde{m}_\lambda, \tilde{m}_\mu \rangle_{B, t}^{a, b, c, d}$$

for all $\lambda, \mu \in \Lambda$, where $\langle ., . \rangle_{\underline{t},t}$ is given by (2.6.11) and \underline{t}_B is given by (3.2.10).

The proof will be given in Section 3.8. Note that $\underline{t}_B(\varepsilon) \in V_K$ for $\varepsilon \in \mathbb{R}_{>0}$ sufficiently small, so $\langle ., . \rangle_{\underline{t}_B(\varepsilon), t}$ is well defined and positive definite for $\varepsilon \in \mathbb{R}_{>0}$ sufficiently small by Lemma 2.6.4 and Corollary 2.6.5.

The arguments of Section 3.3 can now be repeated to establish full orthogonality of the big q-Jacobi polynomials with respect to $\langle ., . \rangle_B$ and to calculate their quadratic norms. For $\lambda \in \Lambda$, $(a, b, c, d) \in V_B$ and $t \in (0, 1)$ let $\mathcal{N}^B(\lambda) = \mathcal{N}^B(\lambda; a, b, c, d; t)$ be given by

(3.4.14)
$$\mathcal{N}^{B}(\lambda) := (cd)^{|\lambda|} q^{\frac{1}{2} \sum_{i=1}^{n} (\lambda_{i} - 1 + 2(n-i)\tau)\lambda_{i}} \mathcal{N}_{qJ}^{+}(\lambda; a, b; t) \mathcal{N}_{qJ}^{-}(\lambda; a, b; t)} \\ \cdot \prod_{i=1}^{n} \left(-q^{\lambda_{i} + 1} b t^{n-i} c/d, -q^{\lambda_{i} + 1} a t^{n-i} d/c; q \right)_{\infty}^{-1}$$

where $t = q^{\tau}$ and \mathcal{N}_{qJ}^+ respectively \mathcal{N}_{qJ}^- is given by (3.3.10) respectively (3.3.11). Observe that $\mathcal{N}^B(\lambda)$ is well defined and positive. We have now the following multivariable analogue of Theorem 3.2.4 and of the limit transition (3.2.11).

THEOREM 3.4.5. Let $t \in (0,1)$ and $(a, b, c, d) \in V_B$. There exists a sequence of positive real numbers $\{\varepsilon_k\}_{k \in \mathbb{Z}_+}$ which converges to 0, such that

(3.4.15)
$$\lim_{k \to \infty} \left(\varepsilon_k (cd/q)^{\frac{1}{2}} \right)^{|\lambda|} P_\lambda \left((q/cd)^{\frac{1}{2}} \varepsilon_k^{-1} z; \underline{t}_B(\varepsilon_k); t \right) = P_\lambda^B(z; a, b, c, d; t)$$

for all $\lambda \in \Lambda$. Furthermore, the polynomials $\{P_{\lambda}^{B}\}_{\lambda \in \Lambda}$ are orthogonal with respect to $\langle ., . \rangle_{B}$ and the quadratic norms of the big q-Jacobi polynomials are given by

(3.4.16)
$$\langle P_{\lambda}^{B}, P_{\lambda}^{B} \rangle_{B} = \mathcal{N}^{B}(\lambda), \quad \lambda \in \Lambda.$$

PROOF. We have the limit

$$\begin{split} \lim_{\varepsilon \to 0} \left(\prod_{i=1}^n \left(-\varepsilon^{-2} q t^{i-1}; q \right)_{\infty} \right) \left((cd/q)^{\frac{1}{2}} \varepsilon \right)^{2|\lambda|} \mathcal{N}(\lambda; \underline{t}_B(\varepsilon); t) \\ &= 2^n n! \left(q; q \right)_{\infty}^{-2n} (1-q)^{-n} \mathcal{N}^B(\lambda; a, b, c, d; t). \end{split}$$

The proof is now analogous to the proof of Theorem 3.3.3.

The quadratic norm evaluations of the big q-Jacobi polynomials for the special case a = b = 0, c = 1 and $t = q^k$ with $k \in \mathbb{N}$ are recently proved by Baker and Forrester [8, Section 4.3] using Pieri formulas. In order to see that the quadratic norms of the big q-Jacobi polynomials in this special case are in agreement with the quadratic norm evaluations [8, (4.3)], one needs to use the evaluation formula for the Macdonald polynomials [85, (6.11)] together with [55, Proposition 3.2].

Recall the monomials $\tilde{e}_r(z) = \tilde{m}_{(1^r)}(z)$ for $r \in [1, n]$, which are algebraically independent generators of $A^{\mathfrak{S}}$. Let \hat{P}^B_{λ} ($\lambda \in \Lambda$) be the unique polynomial in the *n* variables $y = (y_1, \ldots, y_n)$ satisfying

$$\hat{P}^B_\lambda(\tilde{e}_1(z),\ldots,\tilde{e}_n(z)) = P^B_\lambda(z).$$

The limit transition (3.4.15) can now be reformulated as follows (cf. Corollary 3.3.4).

COROLLARY 3.4.6. Let
$$\lambda \in \Lambda$$
, $(a, b, c, d) \in V_B$ and $t \in (0, 1)$, then

$$\lim_{k \to \infty} (s_{\varepsilon_k})^{-|\lambda|} \hat{P}_{\lambda}(s_{\varepsilon_k} y_1, \dots, (s_{\varepsilon_k})^n y_n; \underline{t}_B(\varepsilon_k); t) = \hat{P}_{\lambda}^B(y; a, b, c, d, ; t)$$

for certain sequence $\{\varepsilon_k\}_{k\in\mathbb{Z}_+}$ in $\mathbb{R}_{>0}$ converging to zero, where $s_{\varepsilon} := \varepsilon^{-1}(q/cd)^{\frac{1}{2}}$.

Let $D_B = D_{B,a,b,c,d,t}$ be the second order q-difference operator

(3.4.17)
$$D_B = \sum_{j=1}^{n} \left(\phi_{B,j}^+(z) (T_j^+ - \mathrm{Id}) + \phi_{B,j}^-(z) (T_j^- - \mathrm{Id}) \right)$$

where $\phi_{B,j}^{\pm}(z) = \phi_{B,j}^{\pm}(z; a, b, c, d; t)$ is given by

(3.4.18)

$$\phi_{B,j}^{+}(z) := qt^{n-1} \left(a - \frac{c}{qz_j}\right) \left(b + \frac{d}{qz_j}\right) \prod_{i \neq j} \frac{z_i - tz_j}{z_i - z_j}$$
$$\phi_{B,j}^{-}(z) := \left(1 - \frac{c}{z_j}\right) \left(1 + \frac{d}{z_j}\right) \prod_{i \neq j} \frac{z_j - tz_i}{z_j - z_i}.$$

Then we have the following corollary of Theorem 3.3.3.

COROLLARY 3.4.7. Let $(a, b, c, d) \in V_B$ and $t \in (0, 1)$. Then $P^B_{\lambda}(.; a, b, c, d; t)$ is an eigenfunction of $D_{B,a,b,c,d,t}$ with eigenvalue $E^{qJ}_{\lambda}(a, b; t)$ (3.3.23) for all $\lambda \in \Lambda$.

PROOF. The proof is analogous to the proof of Corollary 3.3.5.

It follows from Theorem 3.4.5 that D_B is symmetric with respect to $\langle .,. \rangle_B$. In [116] the symmetry of D_B was established by direct calculations in which the asymptotic behaviour of the weight function Δ^B (see Lemma 3.4.1) plays a crucial role. Similarly as in the little q-Jacobi polynomial case (see after Corollory 3.3.5), limits can be taken of the *n* commuting q-difference operators $D_1 := D, D_2, \ldots, D_n$ of van Diejen [16], yielding *n* commuting q-difference operators $D_{B,1} = D_B, D_{B,2}, \ldots, D_{B,n}$ which simultaneously diagonalize the big q-Jacobi polynomials. The explicit form of the q-difference operators

 $D_{B,i}$ and their eigenvalues can be computed directly (see Corollary 3.4.7 for the special case $D_{B,1} = D_B$).

The constant term identity for the big q-Jacobi polynomials can be rewritten as follows.

COROLLARY 3.4.8. Let $t \in (0, 1)$ and $(a, b, c, d) \in V_B$. Then

(3.4.19)
$$\langle 1,1 \rangle_{B,t}^{a,b,c,d} = \prod_{j=1}^{n} \left(\frac{\Gamma_q(\alpha+1+(j-1)\tau)\Gamma_q(\beta+1+(j-1)\tau)\Gamma_q(j\tau)}{\Gamma_q(\alpha+\beta+2+(n+j-2)\tau)\Gamma_q(\tau)} \right) \\ \cdot \left(-q^{\alpha+1+(j-1)\tau}d/c, -q^{\beta+1+(j-1)\tau}c/d; q \right)_{\infty}^{-1} \right)$$

where $a = q^{\alpha}, b = q^{\beta}$ and $t = q^{\tau}$.

The constant term identity (3.4.19) was conjectured by Askey [5] for $t = q^k$, $k \in \mathbb{N}$, and proved in this case by Evans [29]. See the next section for more details on the special case $t = q^k$, $k \in \mathbb{N}$. The evaluation (3.4.19) for general $t \in (0, 1)$ is equivalent to Tarasov's and Varchenko's summation formula [127, Theorem (E.10)]. The proof of Tarasov and Varchenko is by computing residues for an A type generalization of Askey-Roy's *q*-beta integral.

3.5. The special case $t = q^k$ with $k \in \mathbb{N}$

In this section the results of the previous sections are considered for deformation parameter $t = q^k$ with $k \in \mathbb{N}$. Many formulas simplify under this assumption because tis "compatible" with the base q when $t = q^k$ with $k \in \mathbb{N}$. For instance q-shifted factorials $(a;q)_k$, which for general k > 0 are given by quotients of infinite products, simplify to a finite product when $k \in \mathbb{N}$. In the remainder of this section we assume that $k \in \mathbb{N}$ is fixed.

Recall that in the definition of the orthogonality measure for the big q-Jacobi polynomials the so-called \underline{c} -weighted Jackson integral is used, where the weight for the big q-Jacobi polynomials is given by $c_{B,j} = c_B d_{B,j}$ $(j \in [0, n])$ with $d_{B,j}$ given by (3.4.4) and c_B given by (3.4.7). For $t = q^k$ it is easily verified that $d_{B,j} = 1$ for all $j \in [0, n]$. For the weight \underline{c}_B we thus have $c_{B,j} = c_B$ for $t = q^k$ independ of j, and c_B can then be rewritten as

(3.5.1)
$$c_B = \frac{q^{\binom{k}{2}\binom{n}{2}-k^2\binom{n}{3}}(c+d)^n}{\left(-d/c, -c/d; q\right)_{\infty}^n (cd)^{n+\binom{n}{2}k}} \quad (t=q^k).$$

This follows by a straightforward calculation using the relation $\theta(qx^{-1}) = \theta(x)$, (3.4.8) and

(3.5.2)
$$\sum_{i=1}^{n} (i-1)^2 = 2\binom{n}{3} + \binom{n}{2}.$$
Now set

(3.5.3)
$$\tilde{\delta}_{qJ}(z;q^k) := \prod_{1 \le i < j \le n} (z_i - z_j) z_i^{2k-1} (q^{1-k} z_j / z_i;q)_{2k-1}$$
$$= (-1)^{k\binom{n}{2}} q^{-\binom{k}{2}\binom{n}{2}} \prod_{\substack{i \ne j \\ l \in [0,k-1]}} (z_i - q^l z_j)$$

and write $\tilde{\Delta}^B(.;q^k)$ respectively $\tilde{\Delta}^L(.;q^k)$ for the weight function (3.4.3) respectively (3.3.5) with the interaction factor $\delta_{qJ}(.;q^k)$ (3.3.6) replaced by $\tilde{\delta}_{qJ}(.;q^k)$ (3.5.3).

LEMMA 3.5.1. Let $t = q^k$ $(k \in \mathbb{N})$ and fix $(a, b) \in V_L$ respectively $(a, b, c, d) \in V_B$. Then

(3.5.4)
$$\langle f,g \rangle_{L,q^k} = \frac{1}{n!} \int_{z_1=0}^1 \int_{z_2=0}^1 \dots \int_{z_n=0}^1 f(z)g(z)\tilde{\Delta}^L(z;q^k)d_qz$$

respectively

(3.5.5)
$$\langle f,g \rangle_{B,q^k} = \frac{c_B}{n!} \int_{z_1=-d}^c \int_{z_2=-d}^c \dots \int_{z_n=-d}^c f(z)g(z)\tilde{\Delta}^B(z;q^k)d_qz$$

for all $f, g \in A_{\mathbb{R}}^{\mathfrak{S}}$, where $d_q z = d_q z_n \dots d_q z_2 d_q z_1$.

PROOF. We give a proof for the big q-Jacobi orthogonality measure. Observe that $\delta_{qJ}(z;q^k) = 0$ if $z_i = q^l z_j$ for certain $i \neq j$ and $l \in [0, k-1]$ (and similarly for $\tilde{\delta}_{qJ}$). Furthermore, we have already seen that the weight $c_{B,j}$ is independent of $j \in [0, n]$, and that it is equal to c_B . It follows that $c_B^{-1} \langle f, g \rangle_{B,q^k}$ for $f, g \in A_{\mathbb{R}}^{\mathfrak{S}}$ can be rewritten as (3.5.6)

$$\sum_{j=0}^{n} \int_{z_1=0}^{c} \int_{z_2=0}^{z_1} \dots \int_{z_j=0}^{z_{j-1}} \int_{z_{j+1}=-d}^{0} \int_{z_{j+2}=z_{j+1}}^{0} \dots \int_{z_n=z_{n-1}}^{0} f(z)g(z)\Delta^B(z)d_qz.$$

For a mass point $z = (z_1, \ldots, z_j, z_{j+1}, \ldots, z_n)$ occuring in the *j*th term of (3.5.6), it is easily verified that

$$\Delta^B(z;q^k) = \tilde{\Delta}^B(z_1,\ldots,z_j,z_n,z_{n-1},\ldots,z_{n-j+1};q^k),$$

hence (3.5.6) can be rewritten as

$$\sum_{j=0}^{n} \int_{z_1=0}^{c} \int_{z_2=0}^{z_1} \dots \int_{z_j=0}^{z_{j-1}} \int_{z_{j+1}=-d}^{0} \int_{z_{j+2}=-d}^{z_{j+1}} \dots \int_{z_n=-d}^{z_{n-1}} f(z)g(z)\tilde{\Delta}^B(z)d_qz,$$

which in turn is equal to

(3.5.7)
$$\int_{z_1=-d}^c \int_{z_2=-d}^{z_1} \dots \int_{z_n=-d}^{z_{n-1}} f(z)g(z)\tilde{\Delta}^B(z)d_q z.$$

Now $\tilde{\Delta}^B(z; q^k)$ is symmetric and vanishes whenever $z_i = z_j$ for some $i \neq j$, hence the desired formula (3.5.5) follows by symmetrizing (3.5.7).

In (3.5.4) respectively (3.5.5) the weight function $\tilde{\Delta}^L(z;q^k)$ respectively $\tilde{\Delta}^B(z;q^k)$ may be replaced by

$$\hat{\Delta}^{L}(z;q^{k}) := \frac{n!}{\Gamma_{q^{k}}(n+1)} q^{-2k^{2}\binom{n}{3}-k(\alpha+1)\binom{n}{2}} \prod_{l=1}^{n} v_{L}(z_{l}) \prod_{1 \le i < j \le n} z_{i}^{2k} \left(q^{1-k} z_{j}/z_{i};q\right)_{2k}$$

where $a = q^{\alpha}$, respectively by

$$\hat{\Delta}^B(z;q^k) := \frac{n!}{\Gamma_{q^k}(n+1)} \prod_{l=1}^n v_B(z_l) \prod_{1 \le i < j \le n} z_i^{2k} (q^{1-k} z_j / z_i;q)_{2k}.$$

This follows from the fact that $\tilde{\Delta}^L(.;q^k)$ and $\tilde{\Delta}^B(.;q^k)$ are symmetric functions satisfying

$$\begin{split} \hat{\Delta}^L(z;q^k) &= \frac{n!}{\Gamma_{q^k}(n+1)} \tilde{\Delta}^L(z;q^k) \prod_{i < j} \frac{z_i - q^k z_j}{z_i - z_j}, \\ \hat{\Delta}^B(z;q^k) &= \frac{n!}{\Gamma_{q^k}(n+1)} \tilde{\Delta}^B(z;q^k) \prod_{i < j} \frac{z_i - q^k z_j}{z_i - z_j}, \end{split}$$

and from the fact that

$$\sum_{w \in \mathfrak{S}} w\left(\prod_{i < j} \frac{z_i - q^k z_j}{z_i - z_j}\right) = \Gamma_{q^k}(n+1)$$

(cf. [33] or [85, Chapter III, (1.4)]). The q-Selberg integral (3.3.25) for $t = q^k$ reduces now to the following evaluation formula.

COROLLARY 3.5.2. Let $t = q^k$ with $k \in \mathbb{N}$ and $(a, b) \in V_L$. Then

$$\begin{split} &\int_{z_1=0}^1 \dots \int_{z_n=0}^1 \prod_{1 \le i < j \le n} z_i^{2k} \big(q^{1-k} \frac{z_j}{z_i}; q \big)_{2k} \prod_{l=1}^n \frac{(qz_l;q)_{\infty}}{(qbz_l;q)_{\infty}} z_l^{\alpha} d_q z_l = \\ &q^{k(\alpha+1)\binom{n}{2}+2k^2\binom{n}{3}} \prod_{j=1}^n \frac{\Gamma_q(\alpha+1+(j-1)k)\Gamma_q(\beta+1+(j-1)k)\Gamma_q(jk+1)}{\Gamma_q(\alpha+\beta+2+(n+j-2)k)\Gamma_q(k+1)} \end{split}$$

where $a = q^{\alpha}$, $b = q^{\beta}$.

PROOF. This follows from (3.3.25), (3.5.4) and the discussion preceding the corollary since

$$\Gamma_{q^k}(n+1) = \prod_{j=1}^n \frac{\Gamma_q(jk+1)\Gamma_q(k)}{\Gamma_q(jk)\Gamma_q(k+1)}.$$

Similarly, the q-Selberg integral (3.4.19) for $t = q^k$ can be simplified as follows.

COROLLARY 3.5.3. Let $t = q^k$ with $k \in \mathbb{N}$ and $(a, b, c, d) \in V_B$. Then

$$\begin{split} \int_{z_1=-d}^{c} \cdots \int_{z_n=-d}^{c} \prod_{1 \le i < j \le n} z_i^{2k} (q^{1-k} \frac{z_j}{z_i}; q)_{2k} \prod_{l=1}^{n} \frac{(qz_l/c, -qz_l/d; q)_{\infty}}{(qaz_l/c, -qbz_l/d; q)_{\infty}} d_q z_l \\ = q^{k^2 \binom{n}{3} - \binom{k}{2} \binom{n}{2}} \prod_{j=1}^{n} \left(\frac{\Gamma_q(\alpha + 1 + (j-1)k)\Gamma_q(\beta + 1 + (j-1)k)\Gamma_q(jk+1)}{\Gamma_q(\alpha + \beta + 2 + (n+j-2)k)\Gamma_q(k+1)} \right. \\ \left. \cdot \frac{(-d/c, -c/d; q)_{\infty}(cd)^{1+(j-1)k}}{(-q^{\alpha + 1 + (j-1)k}d/c, -q^{\beta + 1 + (j-1)k}c/d; q)_{\infty}(c+d)} \right) \end{split}$$

where $a = q^{\alpha}$ and $b = q^{\beta}$.

PROOF. The proof is similar to the proof of Corollary 3.5.2, using the explicit expression (3.5.1) for the constant c_B when $t = q^k$.

Corollaries 3.5.2 and 3.5.3 were conjectured by Askey in [5]. A proof of Corollary 3.5.2 was given independently by Habsieger [33] and Kadell [54]. A proof of Corollary 3.5.3 was given by Evans in [29].

3.6. Limit transitions: an algebraic approach

In Section 3.3 and Section 3.4 limit transitions between the Koornwinder polynomials and the big respectively little *q*-Jacobi polynomials were derived by proving the limits on the level of the orthogonality measures (see Theorem 3.3.3, Proposition 3.3.2 respectively Theorem 3.4.5, Proposition 3.4.4).

In this section a different, more algebraic method is discussed for deriving limit transitions between families of multivariable orthogonal polynomials. The method is based on the fact that the multivariable orthogonal polynomials under consideration can be characterized as special type of eigenfunctions for certain explicit second order q-difference (respectively differential) operators. The limit transitions between the families can then be proved by formally computing limits of the operators and their eigenvalues. In particular, the explicit orthogonality relations are not needed for the proof of the limit transitions.

This method will be applied for computing the classical limit $q \uparrow 1$ of the big and the little q-Jacobi polynomials. We give also a limit transition from Koornwinder polynomials to generalized Jacobi polynomials which is different from the computation of the classical limit of the Koornwinder polynomials. The method for computing limit transitions between multivariable orthogonal polynomials as presented in this section can also be applied to various other families of multivariable orthogonal polynomials. See for instance [121], where the method is used to derive alternative proofs of the limit transitions from Koornwinder polynomials to multivariable big and little q-Jacobi polynomials and [19], where the method is used to derive limit transitions from Koornwinder polynomials to multivariable continuous Hahn polynomials and to BC type Heckman-Opdam polynomials.

3. LIMIT TRANSITIONS FOR MULTIVARIABLE ORTHOGONAL POLYNOMIALS

We first recall the definitions of generalized Jacobi polynomials [132] and relate them to the *BC* type Heckman Opdam polynomials [36], [37], [38]. Let V_J be set of parameters (α, β) which satisfy $\alpha, \beta > -1$, and let $\tau > 0$. Define a positive definite bilinear form $\langle ., . \rangle_{J_\tau}^{\alpha, \beta}$ on $A_{\mathbb{R}}^{\mathfrak{S}}$ by

$$\langle f,g\rangle_{J,,\alpha,\beta,\tau} := \frac{1}{n!} \int_{z_1=0}^1 \cdots \int_{z_n=0}^1 f(z)g(z)\Delta^J(z;\alpha,\beta;\tau)dz \quad f,g \in A_{\mathbb{R}}^{\mathfrak{S}}$$

with $\Delta_J(z; \alpha, \beta; \tau) := \left(\prod_{i=1}^n (1-z_i)^\beta z_i^\alpha\right) |V(z)|^{2\tau}$ and $V(z) := \prod_{i < j} (z_i - z_j)$ the Vandermonde determinant.

DEFINITION 3.6.1. Let $(\alpha, \beta) \in V_J$ and $\tau > 0$. The generalized Jacobi polynomials $P_{\lambda}^J(.; \alpha, \beta; \tau)$ $(\lambda \in \Lambda)$ are by definition the unique \mathfrak{S} -invariant polynomials which satisfy (1) $P_{\lambda}^J = \tilde{m}_{\lambda} + \sum_{\mu < \lambda; \mu \in \Lambda} d_{\lambda,\mu}^J \tilde{m}_{\mu}$ for some $d_{\lambda,\mu}^J = d_{\lambda,\mu}^J(\alpha, \beta; \tau) \in \mathbb{R}$. (2) $\langle P_{\lambda}^J(\alpha, \beta; \tau), \tilde{m}_{\mu} \rangle_{J,\alpha,\beta,\tau} = 0$ if $\mu < \lambda$.

The one variable Jacobi polynomials $\{P_m^J(z; \alpha, \beta)\}_{m \in \mathbb{Z}_+}$ are independent of τ and are explicitly given by

(3.6.1)
$$P_m^J(z;\alpha,\beta) = \frac{(-1)^m (\alpha+1)_m}{(m+\alpha+\beta+1)_m} {}_2F_1 \left(\begin{array}{c} -m,m+\alpha+\beta+1\\ \alpha+1 \end{array};z\right)$$

with

$${}_{2}F_{1}\left(\begin{array}{c}a,b\\c\end{array};z\right) = \sum_{n=0}^{\infty} \frac{(a)_{n}(b)_{n}}{(c)_{n}n!} z^{n}$$

Gauss's hypergeometric series. Usually the Jacobi polynomials are written as functions of 1 - 2z and normalized differently (cf. [28, Section 10.8]).

The generalized Jacobi polynomials are directly related to the Heckman-Opdam polynomials of type BC (cf. [36], [37], [38]) as follows. Using the notations introduced in Remark 2.3.5, we set $\langle ., . \rangle$ for the standard inner product on \mathbb{C}^n , so $\langle e_i, e_j \rangle = \delta_{i,j}$. A multiplicity function k is a function $k : R \to \mathbb{C}$ such that $k_\alpha = k_{w\alpha}$ for all $\alpha \in R, w \in \mathcal{W}$. $k = (k_\alpha)_{\alpha \in R}$ is completely determined by $k_1 := k_{e_1}, k_2 := k_{e_1+e_2}$ and $k_3 := k_{2e_1}$, so we will sometimes write $k = (k_1, k_2, k_3)$. Let V_{HO} be the set of parameters (k_1, k_2, k_3) such that $k_1 + k_3 > -\frac{1}{2}, k_3 > -\frac{1}{2}$ and $k_2 > 0$. Define a positive definite bilinear form on $A_{\mathbb{R}}^{\mathcal{W}} := \mathbb{R}[z_1^{\pm 1}, \ldots, z_n^{\pm 1}]^{\mathcal{W}}$ for $k \in V_{HO}$ by

$$\langle f,g \rangle_k := \int_{\theta_1=0}^{2\pi} \cdots \int_{\theta_n=0}^{2\pi} f(e^{i\theta})g(e^{i\theta})\Delta^{HO}(\theta;k)d\theta,$$

with $e^{i\theta} = (e^{i\theta_1}, \dots, e^{i\theta_n})$ and weight function

$$\Delta^{HO}(\theta;k) := \prod_{\alpha \in R} \left(e^{\frac{1}{2}i\langle\alpha,\theta\rangle} - e^{-\frac{1}{2}i\langle\alpha,\theta\rangle} \right)^{k_{\alpha}}$$
$$= c(k) \prod_{j=1}^{n} (\sin^{2}(\theta_{j}/2))^{k_{1}+k_{3}} (\cos^{2}(\theta_{j}/2))^{k_{3}} \prod_{l < m} |\sin^{2}(\theta_{l}/2) - \sin^{2}(\theta_{m}/2)|^{2k_{2}}$$

where $c(k) = 4^{n(k_1+2k_3)+n(n-1)k_2}$.

DEFINITION 3.6.2. Let $k = (k_1, k_2, k_3) \in V_{HO}$. The BC type Heckman-Opdam polynomials $\{P_{\lambda}^{HO}(z;k)\}_{\lambda \in \Lambda}$ are by definition the unique W-invariant Laurent polynomials which satisfy (1) $P_{\lambda}^{HO} = m_{\lambda} = \int_{-\infty}^{-\infty} dHOm_{\lambda} for some dHO \in \mathbb{R}$

(1) $P_{\lambda}^{HO} = m_{\lambda} + \sum_{\mu < \lambda; \mu \in \Lambda} d_{\lambda,\mu}^{HO} m_{\mu}$ for some $d_{\lambda,\mu}^{HO} \in \mathbb{R}$, (2) $\langle P_{\lambda}^{HO}(.;k), m_{\mu} \rangle_{k} = 0$ if $\mu < \lambda$.

Observe that

$$\tilde{m}_{\lambda}\left(\sin^{2}(\theta/2)\right) = (-4)^{-|\lambda|} m_{\lambda}\left(e^{i\theta}\right) + \sum_{\mu < \lambda} b_{\lambda,\mu} m_{\mu}\left(e^{i\theta}\right)$$

for certain constants $b_{\lambda,\mu}$, where $\sin^2(\theta/2) := (\sin^2(\theta_1/2), \dots, \sin^2(\theta_n/2))$. It follows that the defining conditions for P_{λ}^J (Definition 3.6.1) with $\alpha = k_1 + k_3 - \frac{1}{2}, \beta = k_3 - \frac{1}{2}$ and $\tau = k_2$ become the defining conditions for $P_{\lambda}^{HO}(k)$ (Definition 3.6.2) under the change of variables $x_i := \sin^2(\theta_i/2)$ ($i = 1, \dots, n$), up to the constant $(-4)^{|\lambda|}$. So the relation between Heckman-Opdam polynomials of type *BC* and the generalized Jacobi polynomials is given by

(3.6.2)
$$P_{\lambda}^{HO}(e^{i\theta};k) = (-4)^{|\lambda|} P_{\lambda}^{J}\left(\sin^{2}(\theta/2);k_{1}+k_{3}-\frac{1}{2},k_{3}-\frac{1}{2},k_{2}\right)$$

for $\lambda \in \Lambda$. Set $\partial_j := \frac{\partial}{\partial z_j}$ and let $D_{J,\tau}^{\alpha,\beta}$ be the second order partial differential operator given by

(3.6.3)
$$D_{J,\alpha,\beta,\tau} := \sum_{j=1}^{n} \left((z_j - 1) z_j \partial_j^2 + ((2 + \alpha + \beta) z_j - (\alpha + 1)) \partial_j + 2\tau (z_j - 1) z_j V(z)^{-1} (\partial_j V) (z) \partial_j \right)$$

PROPOSITION 3.6.3. ([132]) Fix $\lambda \in \Lambda$. For arbitrary $\alpha, \beta, \tau \in \mathbb{C}$ there exist constants $E^J_{\lambda,\mu}(\alpha,\beta;\tau) \in \mathbb{C}$ ($\mu \leq \lambda$) depending polynomially on α, β and τ , such that

$$D_J m_\lambda = \sum_{\mu \le \lambda} E^J_{\lambda,\mu} m_\mu.$$

The leading term $E^J_{\lambda,\lambda}(\alpha,\beta;\tau)$ will be denoted by $E^J_\lambda(\alpha,\beta;\tau)$ and is explicitly given by

(3.6.4)
$$E_{\lambda}^{J}(\alpha,\beta;\tau) := \sum_{j=1}^{n} \lambda_{j} \left(\lambda_{j} + \alpha + \beta + 1 + 2(n-j)\tau \right).$$

PROOF. This can be proved by a straightforward calculation (compare with [132, p. 132, p. 13

Furthermore, $D_{J,\alpha,\beta,\tau}$ is symmetric with respect to $\langle ., . \rangle_{J,\alpha,\beta,\tau}$ for $(\alpha,\beta) \in V_J$ and $\tau > 0$ (see [132, Theorem 4.3]). It follows from the definition of P_{λ}^J , Proposition 3.6.3 and the symmetry of D_J that

$$(3.6.5) D_{J,\alpha,\beta,\tau}P_{\lambda}^{J}(.;\alpha,\beta;\tau) = E_{\lambda}^{J}(\alpha,\beta;\tau)P_{\lambda}^{J}(.;\alpha,\beta;\tau)$$

for all $\lambda \in \Lambda$ when $(\alpha, \beta) \in V_J$ and $\tau > 0$. Furthermore, the generalized Jacobi polynomials $\{P_{\lambda}^J\}_{\lambda \in \Lambda}$ are mutual orthogonal with respect to $\langle ., . \rangle_J$ (see [36], [37] and [38], where full orthogonality is established for Heckman-Opdam polynomials associated with arbitrary root systems).

In Section 2.3 it was observed that the Koornwinder polynomials can be expressed in terms of the second order q-difference operator $D_{\underline{t},t}$ (2.3.13) and its eigenvalues by the formula

(3.6.6)
$$P_{\lambda}(.;\underline{t};t) = \left(\prod_{\mu < \lambda} \frac{D_{\underline{t},t} - E_{\mu}(\underline{t};t)}{E_{\lambda}(\underline{t};t) - E_{\mu}(\underline{t};t)}\right) m_{\lambda}$$

for parameters \underline{t} and t such that $E_{\mu}(\underline{t}; t) \neq E_{\lambda}(\underline{t}; t)$ for all $\mu < \lambda$ (E_{λ} given by (2.3.16)). Formula (3.6.6) can be used to extend the definition of the Koornwinder polynomial of degree λ to arbitrary parameter values $(\underline{t}, t) \in \mathbb{C}^5$ for which $E_{\mu}(\underline{t}; t) \neq E_{\lambda}(\underline{t}; t)$ for $\mu < \lambda$. The Koornwinder polynomials $P_{\lambda}(.; \underline{t}; t)$ ($\lambda \in \Lambda$) for generic parameters $(\underline{t}, t) \in \mathbb{C}^5$ then do no longer have interpretations as orthogonal polynomials, but they are still characterized as the unique \mathcal{W} -invariant Laurent polynomials of the form $P_{\lambda}(.; \underline{t}; t) = m_{\lambda} + \sum_{\mu < \lambda} c_{\lambda,\mu}(\underline{t}; t)m_{\mu}$ which are eigenfunctions of $D_{\underline{t},t}$ with eigenvalues $E_{\lambda}(\underline{t}; t)$ ($\lambda \in \Lambda$).

Similarly, the multivariable big and little q-Jacobi polynomials as well as the generalized Jacobi polynomials of degree λ can be expressed as

(3.6.7)
$$P_{\lambda}^{X} = \left(\prod_{\mu < \lambda} \frac{D_{X} - E_{\mu}^{X}}{E_{\lambda}^{X} - E_{\mu}^{X}}\right) \tilde{m}_{\lambda}, \quad X = B, L \text{ resp. } J$$

provided that the eigenvalues satisfy $E_{\mu}^{X} \neq E_{\lambda}^{X}$ for all $\mu < \lambda$, where $E_{\lambda}^{B} = E_{\lambda}^{L} := E_{\lambda}^{qJ}$ (3.3.23). Formula (3.6.7) for the generalized Jacobi polynomials follows from the triangularity of D_{J} (Proposition 3.6.3). Formula (3.6.7) for the little and big q-Jacobi polynomials follows from the fact that D_{L} respectively D_{B} is triangular with respect to the monomials $\{\tilde{m}_{\lambda}\}_{\lambda \in \Lambda}$ and with respect to the dominance order \leq , with leading terms given by the eigenvalues $\{E_{\lambda}^{qJ}\}_{\lambda \in \Lambda}$ (this in turn follows from Corollary 3.3.5 and Corollary 3.4.7). This triangularity property holds in fact for arbitrary complex parameters a, b, c, d, t by analytic continuation. Formula (3.6.7) will be used as definition of the polynomial P_{λ}^{X} for all parameter values such that $E_{\lambda}^{X} \neq E_{\mu}^{X}$ if $\mu < \lambda$.

REMARK 3.6.4. The triangularity of D_B and D_L can also be proved by direct computations for arbitrary complex parameter values a, b, c, d and t without using any information about their eigenfunctions (one only needs some elementary properties of the Schur functions). For details, see [116, Proposition 4.2].

There is another way of looking at the above mentioned extended definitions of multivariable polynomials. For instance, consider the Koornwinder polynomial P_{λ} =

 $m_{\lambda} + \sum_{\mu < \lambda} d_{\lambda,\mu} m_{\mu}$ of degree λ , then Proposition 2.3.6 and (3.6.6) imply that the coefficients $d_{\lambda,\mu} = d_{\lambda,\mu}(\underline{t},t)$ depend rationally on the parameters (\underline{t},t) . The extended definition of the Koornwinder polynomials to generic parameter values $(\underline{t},t) \in \mathbb{C}^5$ corresponds with extending $d_{\lambda,\mu}$ as rational function of (\underline{t},t) for all $\mu < \lambda$.

Using the formulas (3.6.6) and (3.6.7), limit transitions between multivariable orthogonal polynomials can now be proved by computing limits of the corresponding q-difference operators. We first give sufficient conditions on the parameter sets such that the eigenvalues of the triangular operators are mutually different for compatible weights (compatible with respect to the dominance order \leq). The first part of the proposition is a slight extension of [16, Lemma 5.1].

PROPOSITION 3.6.5. Let $\lambda, \mu \in \Lambda$ with $\mu < \lambda$. (i) $E_{\mu}(\underline{t};t) < E_{\lambda}(\underline{t};t)$ if $t_0t_1t_2t_3 \in [-q,1)$ and $t \in (0,1)$; (ii) $E_{\mu}^{qJ}(a,b;t) < E_{\lambda}^{qJ}(a,b;t)$ if $ab \in [-q^{-1},q^{-2})$ and $t \in (0,1)$; (iii) $E_{\mu}^{J}(\alpha,\beta;\tau) < E_{\lambda}^{J}(\alpha,\beta;\tau)$ for $(\alpha,\beta) \in V_J$ and $\tau > 0$.

For the proof of the proposition, we use the following fact.

LEMMA 3.6.6. Let $\mu, \lambda \in \Lambda$ with $\mu \leq \lambda$. Then we can walk from μ to λ while staying within Λ by successively adding an element of the subset $\tilde{R}^+ := \{e_i\}_{i=1}^n \cup \{e_i - e_j\}_{i < j}$ of the positive roots.

PROOF. Let $\mu, \lambda \in \Lambda$ such that $\mu < \lambda$. It suffices to prove that there exists an $\alpha \in \tilde{R}^+$ such that $\mu + \alpha \in \Lambda$ and $\mu + \alpha \leq \lambda$. Let

$$\lambda - \mu = \sum_{i=1}^{n-1} c_i (e_i - e_{i+1}) + c_n e_n$$

with $c_j \in \mathbb{Z}_+$ be the decomposition of $\lambda - \mu$ with respect to the simple roots $S = \{e_i - e_{i+1}\}_{i=1}^{n-1} \cup \{e_n\}$. So we have that $\lambda_k - \mu_k = c_k - c_{k-1}$ for $k = 2, \ldots, n$ and $\lambda_1 - \mu_1 = c_1$. Furthermore we have

$$\sum_{j=1}^{i} (\lambda_j - \mu_j) = c_i \quad i \in [1, n].$$

Let $\{c_p, \ldots, c_{q-1}\}$ (p < q) be a string such that $c_j > 0$ for $j = p, \ldots, q-1$ and such that $c_{p-1} = 0$ (or p = 1) and $c_q = 0$ (or q = n + 1). Then $\mu_{p-1} \ge \lambda_{p-1} \ge \lambda_p > \mu_p$ (or p = 1 and $\lambda_p > \mu_p$) and $\mu_q > \lambda_q \ge \lambda_{q+1} \ge \mu_{q+1}$ (or q = n + 1, or q = n and $\mu_q > \lambda_q$). So $\alpha = e_p - e_q \in \tilde{R}^+$ does the job for q < n + 1, and $\alpha = e_p \in \tilde{R}^+$ for q = n + 1. \Box

PROOF OF PROPOSITION 3.6.5. The inequality (ii) follows immediately from (i). For the proof of (i) and (iii) it suffices to check the inequalities for the special case that $\lambda - \mu \in \tilde{R}^+$ in view of Lemma 3.6.6. The proof follows then by direct computation, using the explicit expressions of the eigenvalues E_{λ} (2.3.16) and E_{λ}^J (3.6.4). As an example, we show that $E_{\mu} < E_{\lambda}$ when $\lambda - \mu = e_i - e_j$ and $1 \le i < j \le n$. Then $\lambda_i = \mu_i + 1$, $\lambda_j = \mu_j - 1$ and $\lambda_k = \mu_k$ for $k \neq i, j$. It follows that the difference $E_{\lambda}(\underline{t}; t) - E_{\mu}(\underline{t}; t)$ is equal to

$$(1-q)t^{i-1}q^{-\mu_i-1}\left(1-t^{j-i}q^{\mu_i-\mu_j+1}\right)\left(1+t_0t_1t_2t_3t^{2n-i-j}q^{\mu_i+\mu_j-1}\right)$$

Since i < j and $\mu_i \ge \mu_j$, we have $E_{\lambda}(\underline{t}; t) - E_{\mu}(\underline{t}; t) > 0$ for all $t \in (0, 1)$ if $t_0 t_1 t_2 t_3 \ge -q$.

For $\gamma, \delta \in \mathbb{C}$, we write $\gamma z + \delta = (\gamma z_1 + \delta, \dots, \gamma z_n + \delta)$ and $z^{-1} = (z_1^{-1}, \dots, z_n^{-1})$.

THEOREM 3.6.7. Let $(\alpha, \beta) \in V_J$, $\tau > 0$ and $\lambda \in \Lambda$. (i) Let $c, d \in \mathbb{C}$ such that $c, d \neq 0, c^2 \neq d, d \neq 1$. Then

(3.6.8)

w

$$\lim_{q\uparrow 1} P_{\lambda}\left(z; c, \frac{q^{\alpha+1}}{c}, \frac{q^{\beta+1}d}{c}, \frac{c}{d}; q^{\tau}\right) = k_{|\lambda|}^{c,d} P_{\lambda}^{J}\left(\frac{1+c^{2}-c(z+z^{-1})}{(1-d)(1-c^{2}/d)}; \alpha, \beta; \tau\right)$$

ith $k_{m}^{c,d} := \left(\frac{(d-1)(1-c^{2}/d)}{c}\right)^{m}$.

(ii) Let
$$c, d > 0$$
, then

(3.6.9)
$$\lim_{q\uparrow 1} P^B_{\lambda}(z; q^{\alpha}, q^{\beta}, c, d; q^{\tau}) = (-1)^{|\lambda|} (c+d)^{|\lambda|} P^J_{\lambda}\left(\frac{c-z}{c+d}; \alpha, \beta; \tau\right).$$

(iii) We have

(3.6.10)
$$\lim_{q\uparrow 1} P_{\lambda}^{L}\left(z;q^{\alpha},q^{\beta};q^{\tau}\right) = P_{\lambda}^{J}(z;\alpha,\beta;\tau).$$

These limit transitions should be interpreted as pointwise limits as described in Section 3.3 (after Proposition 3.3.2). The multivariable polynomials involved in the three limit transitions are well defined. Indeed, for the limits from big and little q-Jacobi polynomials to generalized Jacobi polynomials, the corresponding parameters satisfy $(q^{\alpha}, q^{\beta}, c, d) \in V_B$ (cf. Definition 3.2.3) respectively $(q^{\alpha}, q^{\beta}) \in V_L$ (cf. Definition 3.2.1). In fact, one has $q^{\alpha+\beta} \in [0, q^{-2})$ in both cases, so the multivariable big q-Jacobi polynomials involved in the limit (3.6.9) are well defined for all $c, d \in \mathbb{C}$ with $c + d \neq 0$ in view of Proposition 3.6.5(ii). So the conditions on c, d for the limit transition (3.6.9) may be weakened to the condition $c + d \neq 0$.

For the limit from Koornwinder polynomials to generalized Jacobi polynomials, the product of the four parameters \underline{t} equals $q^{\alpha+\beta+2} \in [0,1)$, hence the Koornwinder polynomials involved in the limit (3.6.8) are well defined by Proposition 3.6.5(i).

REMARK 3.6.8. In the one variable case (n = 1), the three limit transitions given in Theorem 3.6.7 can easily be derived from the explicit expressions of the polynomials in terms of (basic) hypergeometric series.

The limit transition (3.6.8) from Koornwinder polynomials to generalized Jacobi polynomials is different from taking the classical limit of the Koornwinder polynomials.

Indeed, the classical limit of the Koornwinder polynomials is given by

(3.6.11)
$$\lim_{q \uparrow 1} P_{\lambda}(z; q^{g_0}, -q^{g_1}, q^{g'_0+1/2}, -q^{g'_1+1/2}; q^g) = P_{\lambda}^{HO}(z; g_0 + g'_0 - g_1 - g'_1, g, g_1 + g'_1)$$

(see, for instance, [19]), where we assume that g > 0 and $g_0 + g'_0, g_1 + g'_1 > -1/2$.

To illustrate the computations involved in proving Theorem 3.6.7, we will give a detailed proof of the limit transition (3.6.9) from big q-Jacobi polynomials to generalized Jacobi polynomials. Under the change of variables $z_i = \frac{c-y_i}{c+d}$ ($i \in [1, n]$ and $c + d \neq 0$), the second order differential operator $D_{J,\tau}^{\alpha,\beta}$ becomes

(3.6.12)
$$D_{BJ,\alpha,\beta,c,d,\tau} := \sum_{j=1}^{n} \left((y_j - c)(y_j + d)\partial_j^2 + \hat{B}_j(y)\partial_j \right)$$

where \hat{B}_j is given by

(3.6.13)
$$\hat{B}_{j}(y) := 2\tau(y_{j} - c)(y_{j} + d)V(y)^{-1}(\partial_{j}V)(y) + (2 + \alpha + \beta)y_{j} + d(\alpha + 1) - c(\beta + 1)$$

Denote the right hand side of (3.6.9) by $P_{\lambda}^{BJ}(z) := P_{\lambda}^{BJ}(z; \alpha, \beta; c, d; \tau)$, so

(3.6.14)
$$P_{\lambda}^{BJ}(z) := (-1)^{|\lambda|} (c+d)^{|\lambda|} P_{\lambda}^{J}\left(\frac{c-z}{c+d}; \alpha, \beta; \tau\right)$$

for $\lambda \in \Lambda$.

LEMMA 3.6.9. For $(\alpha, \beta) \in V_J$, $c + d \neq 0$ and $\tau > 0$ we have

$$P_{\lambda}^{BJ} = \left(\prod_{\mu < \lambda} \frac{D_{BJ} - E_{\mu}^{J}}{E_{\lambda}^{J} - E_{\mu}^{J}}\right) \tilde{m}_{\lambda}$$

for all $\lambda \in \Lambda$, where E_{λ}^{J} (independent of c and d) is given by (3.6.4).

PROOF. For $c_1, c_2 \in \mathbb{C}$ we have

(3.6.15)
$$\tilde{m}_{\lambda}(c_1 + c_2 y_1, \dots, c_1 + c_2 y_n) = c_2^{|\lambda|} \tilde{m}_{\lambda}(y) + \sum_{\mu < \lambda} b_{\lambda,\mu} \tilde{m}_{\mu}(y)$$

for certain $b_{\lambda,\mu} \in \mathbb{C}$ depending on c_1 and c_2 . Hence P_{λ}^{BJ} (3.6.14) satisfies

$$(3.6.16) P_{\lambda}^{BJ} = \tilde{m}_{\lambda} + \sum_{\mu < \lambda} d_{\lambda,\mu}^{BJ} \tilde{m}_{\mu}$$

for certain $d_{\lambda,\mu}^{BJ} \in \mathbb{C}$. By (3.6.5) we have the eigenvalue equation

$$(3.6.17) D_{BJ}P_{\lambda}^{BJ} = E_{\lambda}^{J}P_{\lambda}^{B}$$

for all $\lambda \in \Lambda$. Hence D_{BJ} is triangular with respect to the monomials $\{\tilde{m}_{\lambda}\}_{\lambda \in \Lambda}$ and the dominance order \leq , with leading terms given by the eigenvalues $\{E_{\lambda}^{J}\}_{\lambda \in \Lambda}$. The proof

follows since the eigenvalues E_{λ}^{J} are mutual different for compatible weights (Proposition 3.6.5(iii)).

The proof of Theorem 3.6.7(ii) is now a direct consequence of the previous lemma, (3.6.7) for X = B and the following lemma.

LEMMA 3.6.10. Let $(\alpha, \beta) \in V_J$ and $\tau > 0$. For $c + d \neq 0$ we have

(3.6.18)
$$\lim_{q\uparrow 1} \frac{\left(D_{B,q^{\alpha},q^{\beta},c,d,q^{\tau}}f\right)(z)}{(1-q)^2} = \left(D_{BJ,\alpha,\beta,c,d,\tau}f\right)(z) \quad \forall f \in A^{\mathfrak{S}}.$$

The eigenvalues satisfy

(3.6.19)
$$\lim_{q\uparrow 1} \frac{E_{\lambda}^{B}(q^{\alpha}, q^{\beta}, c, d; q, q^{\tau})}{(1-q)^{2}} = E_{\lambda}^{J}(\alpha, \beta, c, d; \tau) \quad \forall \lambda \in \Lambda.$$

PROOF. The limit (3.6.18) makes sense (in terms of the definition of limit transitions as given in Section 3.3 after Proposition 3.3.2) since D_B (as well as D_{BJ}) preserves $A^{\mathfrak{S}}$. So it suffices to prove the limit transition (3.6.18) pointwise for generic $z \in \mathbb{C}^n$. We therefore assume throughout the proof that $z \in (\mathbb{C}^*)^n$ and $V(z) \neq 0$.

The second order q-difference operator D_B (3.4.17) can then be rewritten as

$$(3.6.20) \quad (D_B f)(z) = \sum_{i=1}^n \left(A_i(z) \left(T_i^- \left(\left(D_q^{i,-} \right)^2 f \right) \right)(z) + B_i(z) \left(T_i^- \left(D_q^{i,-} f \right) \right)(z) \right),$$

with $D_q^{i,-}$ the backward partial q-derivative in direction i,

$$(D_q^{i,-}f)(z) := rac{\left(f - T_i^+f
ight)(x)}{(1-q)z_i}$$

and $A_i(z) = A_i(z; a, b, c, d; t)$ respectively $B_i(z) = B_i(z; a, b, c, d; t)$ given by $A_i(z) = a^{-1}(1-a)^2 z_i^2 \phi_{\pi^+}^+ (z)$

$$\begin{aligned} H_{i}(z) &= q^{-1} (1 - q)^{2} z_{i} \psi_{B,i}(z) \\ &= (1 - q)^{2} q^{-2} (qaz_{i} - c) (qbz_{i} + d) t^{n-1} V(z)^{-1} (T_{t,i}V)(z), \\ B_{i}(z) &= q^{-1} (1 - q) z_{i} (\phi_{B,i}^{-}(z) - q\phi_{B,i}^{+}(z)) \\ &= \frac{(1 - q)}{q} t^{n-1} \left((z_{i} + (d - c) - cdz_{i}^{-1}) V(z)^{-1} (T_{t-1,i}V)(z) - (q^{2}abz_{i} + (qad - qbc) - cdz_{i}^{-1}) V(z)^{-1} (T_{t,i}V)(z) \right) \end{aligned}$$

where $\phi_{B,i}^{\pm}$ are given by (3.4.18) and with $T_{t^{\pm 1},i}$ the multiplicative $t^{\pm 1}$ -shift in the variable z_i . It is immediate that

(3.6.21)
$$\lim_{q \uparrow 1} \frac{A_i(z; q^{\alpha}, q^{\beta}, c, d; q^{\tau})}{(1-q)^2} = (z_i - c)(z_i + d).$$

To evaluate the limit for B_i , we consider for a complex variable x the map

$$(D_{u,v}^q f)(x) := \frac{(f(q^{-u}x) - f(q^v x))}{(1-q)x}, \quad (u,v \in \mathbb{R}).$$

Observe that $D_{u,v}^q$ maps polynomials to polynomials respectively Laurent polynomials to Laurent polynomials, and that

(3.6.22)
$$\lim_{q \uparrow 1} \left(D_{u,v}^q f \right)(x) = (u+v) \frac{df}{dx}(x) \quad \forall f \in \mathbb{C}[x, x^{-1}].$$

In particular,

$$\lim_{q\uparrow 1} \frac{\left(T_{q^{-\tau},i}V - T_{q^{\tau},i}V\right)(z)}{(1-q)z_j} = 2\tau \left(\partial_i V\right)(z).$$

It follows now by a straightforward computation that

(3.6.23)
$$\lim_{q\uparrow 1} \frac{B_i(z; q^{\alpha}, q^{\beta}, c, d; q^{\tau})}{(1-q)^2} = \hat{B}_i(z)$$

where B_i is given by (3.6.13). The limit (3.6.18) follows now from (3.6.12), (3.6.20), (3.6.21), (3.6.22) and (3.6.23) since $D_{0,1}^q$ is the backward partial q-derivative in the variable x. The limit (3.6.19) is checked by direct computation. It can also be proved without computation using (3.6.18) and using the triangularity properties of D_{BJ} (Lemma 3.6.9) and of D_B .

The proof of the limit transition (3.6.8) from Koornwinder polynomials to generalized Jacobi polynomials is similar, but the computations are more involved due to the less trivial change of variables $z_i = \frac{1+c^2-c(y_i+y_i^{-1})}{(1-d)(1-c^2/d)}$ ($i \in [1, n]$) which is needed. The only essential difference in the proof is then the proper analogue of (3.6.15), which is given by

$$\tilde{m}_{\lambda}\left(\frac{1+c^{2}-c(y+y^{-1})}{(1-d)(1-c^{2}/d)}\right) = \left(k_{|\lambda|}^{c,d}\right)^{-1}m_{\lambda}(y) + \sum_{\mu<\lambda}b_{\lambda,\mu}m_{\mu}(y)$$

for certain $b_{\lambda,\mu} \in \mathbb{C}$, where $k_m^{c,d}$ is defined in Theorem 3.6.7(i) (note that this formula rewrites an \mathfrak{S} -invariant monomial as linear combination of \mathcal{W} -invariant monomials m_{μ} $(\mu \leq \lambda)$ with respect to the new variables y). The limit transition (3.6.10) from little q-Jacobi polynomials to generalized Jacobi polynomials is proved similarly as the limit transition (3.6.9), the computations being simpler since no change of variables is needed. In fact, the analogue of Lemma 3.6.10 in the little q-Jacobi case is a special case of Lemma 3.6.10, since

$$(3.6.24) D_{L,a,b,t} = D_{B,b,a,1,0,t},$$

where D_L is given by (3.3.21). In fact, (3.6.24) together with (3.6.7) for X = B, L immediately implies the limit transition

(3.6.25)
$$\lim_{d \to 0} P_{\lambda}^{B}(z; b, a, 1, d; t) = P_{\lambda}^{L}(z; a, b; t)$$

for $\lambda \in \Lambda$, $(a, b) \in V_L$ and $t \in (0, 1)$ such that $E^{qJ}_{\mu}(a, b; t) \neq E^{qJ}_{\lambda}(a, b; t)$ for all $\mu < \lambda$. By a rather tricky combination of a rationality argument with respect to the parameter d and a continuity argument in t, a proof of the limit transition (3.6.25) can be given for all $\lambda \in \Lambda$ and all parameter values $(a, b) \in V_L$ and $t \in (0, 1)$ (cf. [121, Theorem 6.3]). In the proof of [121, Theorem 6.3] the orthogonality relations of the big q-Jacobi polynomials is used in an essential way.

For a more detailed discussion of the limit transitions (3.6.8), (3.6.9) and (3.6.25) we refer the reader to [121].

3.7. Limit of the orthogonality measure (little *q*-Jacobi case)

In this section a proof of Proposition 3.3.2 is given for parameters $(a, b) \in V_L$ with $b \neq 0$ and $a \leq q^{-\frac{1}{2}}$. The condition $b \neq 0$ is not essential, we only make this assumption because the formulas are more transparent when we may divide by b. For b = 0 all formulas can be rewritten in an obvious way such that they are meaningful and correct. The assumption $a \leq q^{-\frac{1}{2}}$ is less harmless. Under this assumption the parameter $t_{L,1}(\varepsilon) = -aq^{\frac{1}{2}}$ (3.2.1) has modulus ≤ 1 and hence does not contribute to the discrete parts of the symmetric form $\langle ., . \rangle_{\underline{t}_L(\varepsilon), t}$ (2.6.11) in the limit (3.3.7). When $a \in (q^{-\frac{1}{2}}, q^{-1})$, then the set of mass points $D_1(m; \underline{t}_L(\varepsilon); t)$ (2.6.2) in the discrete parts of $\langle ., . \rangle_{\underline{t}_L(\varepsilon), t}$ is ε -independent since $t_{L,1}(\varepsilon)$ is ε -independent. It can be shown that any (partly discrete) contribution to the mass points of $\langle ., . \rangle_{\underline{t}_L(\varepsilon), t}$ with some of its coordinates living in $D_1(m; \underline{t}_L(\varepsilon); t)$ with m > 0, vanishes in the limit (3.3.7). In other words, we may as well assume that there are no discrete contributions coming from the term $t_{L,1}(\varepsilon)$, which corresponds to the case $a \leq q^{-\frac{1}{2}}$. For a detailed discussion with $a > q^{-\frac{1}{2}}$, we refer the reader to [**119**].

So under these assumptions, only $t_{L,0}(\varepsilon) = \varepsilon^{-1}q^{\frac{1}{2}}$ contributes to the discrete parts of of the symmetric form $\langle ., \rangle_{\underline{t}_L(\varepsilon),t}$ in the limit (3.3.7). Set $\rho_{L,j}(\varepsilon) := t_{L,0}(\varepsilon)t^{j-1} = \varepsilon^{-1}q^{\frac{1}{2}}t^{j-1}$ for $j \in \mathbb{Z}$. Then $F(r; \underline{t}_L(\varepsilon); t)$ (2.6.3) for $\varepsilon > 0$ sufficiently small is given by

$$F(r; \underline{t}_L(\varepsilon); t) = D_0(r; \underline{t}_L(\varepsilon); t) \subset \mathbb{C}^r$$

with the set $D_0(r; \underline{t}_L(\varepsilon); t)$ (2.6.2) for r > 0 given by

$$(3.7.1) \qquad D_0(r;\underline{t}_L(\varepsilon);t) = \{\rho_L(\varepsilon)q^{\nu} \mid \nu \in P_L(r;\varepsilon)\},\\ P_L(r;\varepsilon) := \{\nu \in P(r) \mid |\rho_{L,0}(\varepsilon)q^{\nu_r}| > 1\}.$$

Using the explicit definition of the symmetric form $\langle ., . \rangle$ (2.6.11) as given in Section 2.6, as well as the definition for $m_{\lambda}(z|u)$ (3.3.8), we can write

$$(3.7.2) = \frac{\left(\prod_{i=1}^{n} \left(-\varepsilon^{-1}qt^{i-1}, -\varepsilon^{-1}qat^{i-1}; q\right)_{\infty}\right) \left(\varepsilon q^{-1/2}\right)^{|\lambda|+|\mu|} \langle m_{\lambda}, m_{\mu} \rangle_{\underline{t}_{L}(\varepsilon), t}}{\left(2\pi i\right)^{n-r}} \sum_{r, \nu} \iint_{z \in T^{n-r}} \left(m_{\lambda}m_{\mu}\right) \left(\rho_{L}q^{\nu}, \varepsilon q^{-\frac{1}{2}}z |\varepsilon q^{-\frac{1}{2}}\right) \mathcal{W}_{r}^{L}(\nu, z; \varepsilon) \frac{dz}{z}}$$

where the sum is over pairs (r, ν) with $r \in [0, n]$ and $\nu \in P(r)$ (the sum over $\nu \in P(r)$ should be ignored when r = 0) and with $\rho_{L,j} := t^{j-1}$. The renormalized weights

$$\mathcal{W}_{r}^{L}(\nu, z; \varepsilon)$$
 are defined by $\mathcal{W}_{0}^{L}(-; z; \varepsilon) := \Delta(z; \underline{t}_{L}(\varepsilon); t)$ for $r = 0$, and for $r \in [1, n]$,

$$\mathcal{W}_{r}^{L}(\nu, z; \varepsilon) = \left(\prod_{i=1}^{n} \left(-\varepsilon^{-1} q t^{i-1}, -\varepsilon^{-1} q a t^{i-1}; q\right)_{\infty}\right) \Delta_{r}^{K} \left(\rho_{L}(\varepsilon) q^{\nu}, z; \underline{t}_{L}(\varepsilon); t\right)$$

if $\nu \in P_L(r; \varepsilon)$ and zero otherwise. Split the renormalized weights in two parts,

(3.7.3)
$$\mathcal{W}_{r}^{L}(\nu, z; \varepsilon) = \Delta_{1,r}^{KL}(\nu; \varepsilon) \Delta_{2,r}^{KL}(\nu, z; \varepsilon)$$

where $\Delta_{1,r}^{KL}$ and $\Delta_{2,r}^{KL}$ are given by

$$\Delta_{1,r}^{KL}(\nu;\varepsilon) := \left(\prod_{i=1}^{r} \left(-\varepsilon^{-1}qt^{i-1}, -\varepsilon^{-1}qat^{i-1};q\right)_{\infty}\right) \Delta^{(d)}\left(\rho_{L}(\varepsilon)q^{\nu}; t_{L,0}(\varepsilon)\right)$$

if $\nu \in P_L(r; \varepsilon)$ and zero otherwise, respectively

$$\Delta_{2,r}^{KL}(\nu,z;\varepsilon) := \prod_{i=1}^{n-r} \left(-\varepsilon^{-1} q t^{r+i-1}, -\varepsilon^{-1} q a t^{r+i-1}; q \right)_{\infty}$$
$$\cdot \Delta(z; \underline{t}_L(\varepsilon); t) \delta_c \left(\rho_L(\varepsilon) q^{\nu}; z \right)$$

if $\nu \in P_L(r; \varepsilon)$, $z \in T^{n-r}$ and zero otherwise, where Δ is given by (2.3.4), $\Delta^{(d)}$ is given by (2.4.7) and δ_c is given by (2.4.11). We have used here the obvious convention that $\Delta_{1,0}^{KL}(-;\varepsilon) = 1$ and $\Delta_{2,n}^{KL}(\nu, -;\varepsilon) = 1$ if $\nu \in P_L(r;\varepsilon)$. Lebesgue's dominated convergence Theorem will now be used to pull a limit $\varepsilon_k \downarrow 0$

Lebesgue's dominated convergence Theorem will now be used to pull a limit $\varepsilon_k \downarrow 0$ in the right hand side of (3.7.2) through the integration over $z \in T^{n-r}$ and through the infinite sum over $\nu \in P(r)$ for some sequence $\{\varepsilon_k\}_{k \in \mathbb{Z}_+}$ in $\mathbb{R}_{>0}$ converging to 0. For the application of the Lebesgue's dominated convergence Theorem we need certain estimates for the functions $\Delta_{1,r}^{KL}$ and $\Delta_{2,r}^{KL}$, which are given in the following lemma.

LEMMA 3.7.1. Keep the notations and conventions as above. In particular, let $r \in [0, n]$. Then there exists a sequence $\{\varepsilon_k\}_{k \in \mathbb{Z}_+}$ in $\mathbb{R}_{>0}$ converging to 0 such that (i) if $r \geq 1$, then for all $\nu \in P(r)$,

$$\lim_{k \to \infty} \Delta_{1,r}^{KL}(\nu; \varepsilon_k) = \left(q; q\right)_{\infty}^{-2r} \Delta^L(\rho_L q^{\nu}; a, b; t) \prod_{i=1}^r \rho_{L,i} q^{\nu_i},$$

and there exists a $K \in \mathbb{R}_{>0}$ independent of $\nu \in P(r)$ such that

(3.7.4)
$$\sup_{k\in\mathbb{Z}_+} |\Delta_{1,r}^{KL}(\nu;\varepsilon_k)| \le K\Delta^L(\rho_L q^\nu;a,b;t) \prod_{i=1}^r \rho_{L,i} q^{\nu_i}$$

for all $\nu \in P(r)$.

(iii) If r < n, then $\lim_{k\to\infty} \Delta_{2,r}^{KL}(\nu,z;\varepsilon_k) = 0$ for all $z \in T^{n-r}$, $\nu \in P(r)$ and

$$\sup_{(\nu,z,k)\in P(r)\times T^{n-r}\times\mathbb{Z}_+} |\Delta_{2,r}^{KL}(\nu,z;\varepsilon_k)| < \infty.$$

Before a proof of Lemma 3.7.1 is given, we show how Lemma 3.7.1 implies Proposition 3.3.2. Since the infinite sum

(3.7.5)
$$(1-q)^{-n} \langle 1,1 \rangle_{L,t}^{a,b} = \sum_{\nu \in P(n)} \Delta^L(\rho_L q^{\nu}; a,b;t) \prod_{i=1}^n \rho_{L,i} q^{\nu_i}$$

is absolutely convergent (cf. [116, proof of Proposition 6.1]) and

$$\sup_{\nu,\varepsilon,z} |m_{\lambda} \left(\rho_L q^{\nu}, \varepsilon q^{-\frac{1}{2}} z | \varepsilon q^{-\frac{1}{2}} \right)| < \infty \quad (\lambda \in \Lambda),$$

where the supremum is taken over triples (ν, ε, z) with $\nu \in P_L(r; \varepsilon)$, $\varepsilon \in \mathbb{R}_{>0}$ and $z \in T^{n-r}$, it follows by Lebesgue's dominated convergence Theorem, (3.3.9), (3.7.2), (3.7.3) and Lemma 3.7.1 that

$$\lim_{k \to \infty} \left(\prod_{i=1}^{n} \left(-\varepsilon_{k}^{-1} q t^{i-1}, -\varepsilon_{k}^{-1} q a t^{i-1}; q \right)_{\infty} \right) \left(\varepsilon_{k} q^{-\frac{1}{2}} \right)^{|\lambda| + |\mu|} \langle m_{\lambda}, m_{\mu} \rangle_{\underline{t}_{L}(\varepsilon_{k}), t} \\ = \sum_{r, \nu} \frac{2^{r} (n-r+1)_{r}}{(2\pi i)^{n-r}} \iint_{z \in T^{n-r}} \lim_{k \to \infty} (m_{\lambda} m_{\mu}) \left(\rho_{L} q^{\nu}, \varepsilon_{k} q^{-\frac{1}{2}} z | \varepsilon_{k} q^{-\frac{1}{2}} \right) \mathcal{W}_{r}^{L}(\nu, z; \varepsilon_{k}) \frac{dz}{z} \\ = 2^{n} n! (q; q)_{\infty}^{-2n} \sum_{\nu \in P(n)} (\tilde{m}_{\lambda} \tilde{m}_{\mu}) (\rho_{L} q^{\nu}) \Delta^{L}(\rho_{L} q^{\nu}; a, b; t) \prod_{i=1}^{n} \rho_{L,i} q^{\nu_{i}} \\ = 2^{n} n! (1-q)^{-n} (q; q)_{\infty}^{-2n} \langle \tilde{m}_{\lambda}, \tilde{m}_{\mu} \rangle_{L,t}^{a,b}$$

for some sequence $\{\varepsilon_k\}_{k\in\mathbb{Z}_+}$ in $\mathbb{R}_{>0}$ converging to 0, where the sum in the second line is over four tuples (r, ν) with $r \in [0, n]$, and $\nu \in P(r)$. So for the proof of Proposition 3.3.2, it suffices to prove Lemma 3.7.1. We use the following elementary lemma.

LEMMA 3.7.2. For given $\varepsilon_0 \in \mathbb{R}_{>0}$, set $\varepsilon_k := \varepsilon_0 q^k$. (a) Let $c \in \mathbb{C}$. For $\varepsilon_0 \in \mathbb{R}_{>0}$ with $|c|\varepsilon_0 \notin \{q^{-l}\}_{l \in \mathbb{Z}_+}$ there exist positive constants $K^{\pm} > 0$ which only depend on ε_0 and |c|, such that $K^- \leq |(c\varepsilon_k; q)_{\infty}| \leq K^+$ for all $k \in \mathbb{Z}_+$. Furthermore, $\lim_{k\to\infty} (c\varepsilon_k; q)_{\infty} = 1$. (b) Let $a, b \in \mathbb{C}^*$, and set

(3.7.6)
$$f_{\{l,m\}}(\varepsilon;a,b) := \frac{\left(\varepsilon^{-1}aq^{1-m};q\right)_m}{\left(\varepsilon^{-1}bq^{1-m-l};q\right)_m}, \qquad (l,m\in\mathbb{Z}_+).$$

Let $\varepsilon_0 \in \mathbb{R}_{>0}$ such that $\varepsilon_0^{-1}|b| \notin \{q^k\}_{k \in \mathbb{Z}_+}$. Then there exists a positive constant K > 0which depends only on ε_0 , |a| and |b|, such that $|f_{\{l,m\}}(\varepsilon_k; a, b)| \leq K |q^l a/b|^m$ for all $k, l, m \in \mathbb{Z}_+$. Furthermore, $\lim_{k \to \infty} f_{\{l,m\}}(\varepsilon_k; a, b) = (q^l a/b)^m$.

(c) Let $u_i, v_j \in \mathbb{C}^*$ for $i \in [1, r]$, $j \in [1, s]$ and assume that r < s, or that r = s and $|u_1 \dots u_r| < |v_1 \dots v_r|$. Set

(3.7.7)
$$g(\varepsilon) := \frac{\left(\varepsilon^{-1}u_1, \dots, \varepsilon^{-1}u_r; q\right)_{\infty}}{\left(\varepsilon^{-1}v_1, \dots, \varepsilon^{-1}v_s; q\right)_{\infty}}$$

Let $\varepsilon_0 \in \mathbb{R}_{>0}$ such that $\varepsilon_0^{-1}|v_j| \notin \{q^l\}_{l \in \mathbb{Z}}$ for $j \in [1, s]$. Then there exists a positive constant K > 0 which depends only on ε_0 , $|u_i|$ and $|v_j|$, such that $\sup_{k \in \mathbb{Z}_+} |g(\varepsilon_k)| \leq K$. Furthermore, $\lim_{k \to \infty} g(\varepsilon_k) = 0$.

PROOF. The proof of (a) is straightforward. For (b) and (c) use (2.5.3) to rewrite $f_{\{l,m\}}$ as

$$f_{\{l,m\}}(\varepsilon;a,b) = \left(q^l a/b\right)^m \frac{\left(a^{-1}\varepsilon;q\right)_m}{\left(b^{-1}q^l\varepsilon;q\right)_m},$$

and to rewrite $g(\varepsilon_k)$ as

$$(3.7.8) \quad g(\varepsilon_k) = \left(\frac{u_1 \dots u_r}{v_1 \dots v_s} \left(-q^{(k+1)/2}\varepsilon_0\right)^{s-r}\right)^k \frac{\left(q\varepsilon_0 u_1^{-1}, \dots, q\varepsilon_0 u_r^{-1};q\right)_k}{\left(q\varepsilon_0 v_1^{-1}, \dots, q\varepsilon_0 v_s^{-1};q\right)_k}g(\varepsilon_0).$$

The limits for $f_{\{l,m\}}$ and g given in (b) respectively (c) are now immediately clear. Furthermore we have the estimate $|f_{\{l,m\}}(\varepsilon_k; a, b)| \leq K |q^l a/b|^m$ with

$$K = \frac{\left(-|a|^{-1}\varepsilon_{0};q\right)_{\infty}}{\left(|b|^{-1}\varepsilon_{0}q^{k_{0}};q\right)_{\infty}} \prod_{\{i \in \mathbb{Z}_{+} \mid 1 < |b|^{-1}\varepsilon_{0}q^{i} < 2\}} \left(|b|^{-1}\varepsilon_{0}q^{i} - 1\right)^{-1} > 0,$$

where k_0 is the smallest positive integer such that $|b|^{-1}\varepsilon_0 q^{k_0} < 1$. The estimate for $|g(\varepsilon_k)|$ in (c) is easily derived from (3.7.8), the assumptions on r, s and on the parameters u_i, v_j , and from estimates similar to the estimate for K in the proof of (b).

We proceed with the proof of Lemma 3.7.1. Set $\varepsilon_k := \varepsilon_0 q^k$ for given $\varepsilon_0 \in \mathbb{R}_{>0}$.

PROOF OF LEMMA 3.7.1(i). By (2.4.7) we have

$$(3.7.9) \quad \Delta_{1,r}^{KL}(\nu;\varepsilon) = \delta_d(\rho_L(\varepsilon)q^{\nu}) \prod_{i=1}^r \left\{ \left(-\varepsilon^{-1}qt^{i-1}, -\varepsilon^{-1}qat^{i-1}; q \right)_{\infty} \\ \cdot w_d(\rho_{L,i}(\varepsilon)q^{\nu_i}; \rho_{L,i}(\varepsilon)q^{\nu_{i-1}}) \right\}$$

with δ_d given by (2.4.8) and w_d given by (2.2.7). By (2.4.8) and (3.2.1), we have

(3.7.10)
$$\delta_d(\rho_L(\varepsilon)q^\nu) = F_1(\nu)G_1(\nu;\varepsilon)$$

with

(3.7.11)

$$F_{1}(\nu) := \prod_{1 \leq i < j \leq r} \frac{\left(t^{j-i}q^{\nu_{j}-\nu_{i}};q\right)_{\tau}}{\left(t^{i-j}q^{\nu_{i-1}-\nu_{j}};q\right)_{\nu_{i}-\nu_{i-1}}},$$

$$G_{1}(\nu;\varepsilon) := \prod_{1 \leq i < j \leq r} \frac{\left(\varepsilon^{2}t^{2-i-j}q^{-\nu_{i}-\nu_{j-1}};q\right)_{\tau}}{\left(\varepsilon^{-2}t^{i+j-2}q^{\nu_{i-1}+\nu_{j}+1};q\right)_{\nu_{i}-\nu_{i-1}}}$$

3. LIMIT TRANSITIONS FOR MULTIVARIABLE ORTHOGONAL POLYNOMIALS

for $\nu \in P(r)$, where $\nu_0 = 0$ and $t = q^{\tau}$. By applying (2.5.3) to the *q*-shifted factorials in the denominator of F_1 and using the formula

(3.7.12)
$$\sum_{i=1}^{r} (i-1)(r-i) = \binom{r}{3},$$

we obtain

(3.7.13)
$$F_{1}(\nu) = \delta_{qJ}(\rho_{L}q^{\nu})q^{-2\tau^{2}\binom{r}{3}} \prod_{j=1}^{r} \frac{(t^{j-1}q^{\nu_{j}+1};q)_{\infty}}{(q^{\nu_{j}-\nu_{j-1}+1};q)_{\infty}} t^{-2(r-j)\nu_{j}} \cdot \prod_{1 \leq i < j \leq r} (-q^{\nu_{j}-\nu_{i}+1}t^{j-i})^{\nu_{i}-\nu_{i-1}}q^{\binom{\nu_{i}-\nu_{i-1}}{2}},$$

where δ_{qJ} (3.3.6) is the interaction factor for the weight function of the little q-Jacobi polynomials. On the other hand, we have for $i \in [1, r]$ by (2.2.7),

(3.7.14)

$$\left(-\varepsilon^{-1}qt^{i-1}, -\varepsilon^{-1}qat^{i-1}; q \right)_{\infty} w_d(\rho_{L,i}(\varepsilon)q^{\nu_i}; \rho_{L,i}(\varepsilon)q^{\nu_{i-1}}) = I_{1,i}(\nu)J_{1,i}(\nu;\varepsilon)$$

with

$$I_{1,i}(\nu) := \frac{\left(t^{i-1}q^{\nu_{i-1}+1}ab\right)^{\nu_{i-1}-\nu_{i}}}{\left(q,qbt^{i-1}q^{\nu_{i}};q\right)_{\infty}\left(q;q\right)_{\nu_{i}-\nu_{i-1}}}$$

$$= v_{L}(\rho_{L,i}q^{\nu_{i}})\frac{a^{(1-i)\tau-\nu_{i}}\left(t^{i-1}q^{\nu_{i-1}+1}ab\right)^{\nu_{i-1}-\nu_{i}}}{\left(q,t^{i-1}q^{\nu_{i}+1};q\right)_{\infty}\left(q;q\right)_{\nu_{i}-\nu_{i-1}}}$$

(here v_L (3.2.13) is the one variable weight function of the little q-Jacobi polynomials) and with $J_{1,i}(\nu;\varepsilon)$ given by

$$J_{1,i}(\nu;\varepsilon) := \frac{\left(\varepsilon^{2}t^{2-2i}q^{-2\nu_{i-1}-1};q\right)_{\infty}}{\left(\varepsilon^{2}bt^{1-i}q^{-\nu_{i-1}}, -\varepsilon t^{1-i}q^{-\nu_{i-1}}, -\varepsilon at^{1-i}q^{-\nu_{i-1}};q\right)_{\infty}} \cdot \frac{\left(\varepsilon^{-2}q^{1+2\nu_{i-1}}t^{2i-2};q\right)_{\nu_{i}-\nu_{i-1}}\left(-\varepsilon^{-1}t^{i-1}q;q\right)_{\nu_{i-1}}\left(-\varepsilon^{-1}at^{i-1}q;q\right)_{\nu_{i}}}{\left(\varepsilon^{-2}b^{-1}t^{i-1}q^{\nu_{i-1}+1}, -\varepsilon^{-1}a^{-1}t^{i-1}q^{\nu_{i-1}+1};q\right)_{\nu_{i}-\nu_{i-1}}}{\cdot \frac{\left(1-\varepsilon^{-2}q^{1+2\nu_{i}}t^{2i-2}\right)}{\left(1-\varepsilon^{-2}q^{1+2\nu_{i-1}}t^{2i-2}\right)}}.$$

Thus by (3.7.9), (3.7.10), (3.7.13) and (3.7.14), we have $\Delta_{1,r}^{KL}(\nu;\varepsilon) = M_1(\nu)N_1(\nu;\varepsilon)$ for $\varepsilon \in \mathbb{R}_{>0}$ and $\nu \in P_L(r;\varepsilon)$ with

$$M_{1}(\nu) := F_{1}(\nu) \prod_{i=1}^{r} I_{1,i}(\nu)$$

$$(3.7.15) \qquad = \Delta^{L}(\rho_{L}q^{\nu}) (q;q)_{\infty}^{-2r} t^{\binom{r}{2}} \prod_{i=1}^{r} (t^{-2(r-i)}a^{-1})^{\nu_{i}} (t^{i-1}q^{\nu_{i-1}+1}ab)^{\nu_{i-1}-\nu_{i}}$$

$$\cdot \prod_{1 \le i < j \le r} (-q^{\nu_{j}-\nu_{i}+1}t^{j-i})^{\nu_{i}-\nu_{i-1}} q^{\binom{\nu_{i}-\nu_{i-1}}{2}},$$

 $(\Delta^L \text{ given by } (3.3.5))$ and with

$$N_1(\nu;\varepsilon) := G_1(\nu;\varepsilon) \prod_{i=1}^r J_{1,i}(\nu;\varepsilon).$$

Now replace the factor $\left(-arepsilon^{-1}at^{i-1}q;q
ight)_{
u_i}$ in $J_{1,i}(
u;arepsilon)$ by

$$\left(-\varepsilon^{-1}at^{i-1}q^{\nu_{i-1}+1};q\right)_{\nu_{i}-\nu_{i-1}}\left(-\varepsilon^{-1}at^{i-1}q;q\right)_{\nu_{i-1}}$$

for $i \in [1, r]$, then $N_1(\nu; \varepsilon)$ can explicitly be given by

(3.7.16)
$$N_1(\nu;\varepsilon) = N_1^1(\nu;\varepsilon)N_1^2(\nu;\varepsilon)N_1^3(\nu;\varepsilon)$$
 with

(3.7.17)
$$N_{1}^{1}(\nu;\varepsilon) := \prod_{i=1}^{r} \frac{\left(\varepsilon^{2}t^{2-2i}q^{-2\nu_{i-1}-1};q\right)_{\infty}}{\left(\varepsilon^{2}bt^{1-i}q^{-\nu_{i-1}}, -\varepsilon t^{1-i}q^{-\nu_{i-1}}, -\varepsilon at^{1-i}q^{-\nu_{i-1}};q\right)_{\infty}} \cdot \prod_{1 \le i \le r} \left(\varepsilon^{2}t^{2-i-j}q^{-\nu_{i}-\nu_{j}-1};q\right)_{\tau}$$

if $\nu \in P_L(r;\varepsilon)$ and zero otherwise,

$$(3.7.18) \qquad N_1^2(\nu;\varepsilon) := \prod_{i=1}^r \left(\frac{\left(\varepsilon^{-2}q^{1+2\nu_{i-1}}t^{2i-2}, -\varepsilon^{-1}at^{i-1}q^{1+\nu_{i-1}}; q\right)_{\nu_i - \nu_{i-1}}}{\left(\varepsilon^{-2}b^{-1}t^{i-1}q^{1+\nu_{i-1}}, -\varepsilon^{-1}a^{-1}t^{i-1}q^{1+\nu_{i-1}}; q\right)_{\nu_i - \nu_{i-1}}} \cdot \frac{\left(\varepsilon^{-2}q^{1+2\nu_i}t^{2i-2}; q\right)_1}{\left(\varepsilon^{-2}q^{1+2\nu_{i-1}}t^{2i-2}; q\right)_1}\right)$$

if $\nu \in P_L(r; \varepsilon)$ and zero otherwise, and

(3.7.19)
$$N_1^3(\nu;\varepsilon) = \frac{\prod_{i=1}^r \left(-\varepsilon^{-1}t^{i-1}q, -\varepsilon^{-1}at^{i-1}q;q\right)_{\nu_{i-1}}}{\prod_{1 \le i < j \le r} \left(\varepsilon^{-2}t^{i+j-2}q^{\nu_{i-1}+\nu_j+1};q\right)_{\nu_i-\nu_{i-1}}}$$

if $\nu \in P_L(r;\varepsilon)$ and zero otherwise. For the factor N_1^1 we have for generic $\varepsilon_0 > 0$, (3.7.20) $\lim_{k \to \infty} N_1^1(\nu;\varepsilon_k) = 1 \qquad (\nu \in P(r))$ by Lemma 3.7.2(a). Lemma 3.7.2(b) can be used to calculate the limit of the factor N_1^2 . We obtain for generic $\varepsilon_0 > 0$,

(3.7.21)
$$\lim_{k \to \infty} N_1^2(\nu; \varepsilon_k) = \prod_{i=1}^r \left(q^{\nu_{i-1}+2} t^{i-1} a^2 b \right)^{\nu_i - \nu_{i-1}}$$

for all $\nu \in P(r)$. As an example, the limit of a factor of N_1^2 will be calculated explicitly using Lemma 3.7.2(b). Consider the factor

(3.7.22)
$$N_1^{i,2}(\nu;\varepsilon) := \frac{\left(\varepsilon^{-2}q^{1+2\nu_{i-1}}t^{2i-2};q\right)_{\nu_i-\nu_{i-1}}}{\left(\varepsilon^{-2}q^{1+\nu_{i-1}}b^{-1}t^{i-1};q\right)_{\nu_i-\nu_{i-1}}}$$

of $N_1^2(\nu; \varepsilon)$ for some $i \in [1, r]$. Then for generic $\varepsilon_0 > 0$, it follows from Lemma 3.7.2(b) that

(3.7.23)
$$\lim_{k \to \infty} N_1^{i,2}(\nu; \varepsilon_k) = \lim_{k \to \infty} N_1^{i,2}(\nu; \varepsilon_{k+\nu_r})$$
$$= \lim_{k \to \infty} N_1^{i,2}(\nu; q^{\nu_r} \varepsilon_k)$$
$$= \lim_{k \to \infty} f_{\{\nu_{i-1}, \nu_i - \nu_{i-1}\}} (\varepsilon_{2\nu_r - \nu_i - \nu_{i-1} + 2k}; \varepsilon_0^{-1} t^{2i-2}, \varepsilon_0^{-1} b^{-1} t^{i-1})$$
$$= (q^{\nu_{i-1}} t^{i-1} b)^{\nu_i - \nu_{i-1}}.$$

Similarly, the limits of the other factors of N_1^2 can be computed, which yield (3.7.21). Finally, we have for generic $\varepsilon_0 > 0$,

(3.7.24)
$$\lim_{k \to \infty} N_1^3(\nu; \varepsilon_k) = \prod_{i=1}^r \left(a t^{2(i-1)} q^{\nu_{i-1}+1} \right)^{\nu_{i-1}} \cdot \prod_{1 \le i < j \le r} \left(-t^{i+j-2} q^{\nu_{i-1}+\nu_j+1} \right)^{\nu_{i-1}-\nu_i} q^{-\binom{\nu_i-\nu_{i-1}}{2}}$$

since

(3.7.25)
$$\sum_{i=1}^{r} \nu_{i-1} = \sum_{1 \le i < j \le r} (\nu_i - \nu_{i-1}) \qquad (\nu \in P(r)).$$

We thus obtain for generic $\varepsilon_0 > 0$ by (3.7.16), (3.7.20), (3.7.21) and (3.7.24),

(3.7.26)
$$\lim_{k \to \infty} N_1(\nu; \varepsilon_k) = \prod_{i=1}^r t^{(i-1)(\nu_i + \nu_{i-1})} q^{\nu_{i-1}\nu_i - \nu_{i-1} + 2\nu_i} a^{2\nu_i - \nu_{i-1}} b^{\nu_i - \nu_{i-1}} \\ \cdot \prod_{1 \le i < j \le r} \left(-t^{i+j-2} q^{\nu_{i-1} + \nu_j + 1} \right)^{\nu_{i-1} - \nu_i} q^{-\binom{\nu_i - \nu_{i-1}}{2}}$$

for all $\nu \in P(r)$. By (3.7.15) and (3.7.26) we obtain for generic $\varepsilon_0 > 0$,

$$\lim_{k \to \infty} \Delta_{1,r}^{KL}(\nu; \varepsilon_k) = M_1(\nu) \lim_{k \to \infty} N_1(\nu; \varepsilon_k)$$
$$= (q; q)_{\infty}^{-2r} \Delta^L(\rho_L q^{\nu}) t^{\binom{r}{2}} q^{|\nu|}$$
$$= (q; q)_{\infty}^{-2r} \Delta^L(\rho_L q^{\nu}) \prod_{i=1}^r \rho_{L,i} q^{\nu_i}$$

for all $\nu \in P(r)$, where Δ^L is given by (3.3.5) and $|\nu| := \nu_1 + \cdots + \nu_r$ for $\nu \in P(r)$.

To prove the estimate (3.7.4), we use the estimates of Lemma 3.7.2 (a) and (b) for (factors of) N_1 . For N_1^1 , we use Lemma 3.7.2(a) and the condition that $N_1^1(\nu; \varepsilon) = 0$ if $\nu \notin P_L(r; \varepsilon)$ to prove that $\sup_{\nu,k} |N_1^1(\nu; \varepsilon_k)| < \infty$ for generic $\varepsilon_0 > q^{\frac{1}{2}}$. Indeed, since $\nu \in P_L(r; \varepsilon)$ implies $\varepsilon < q^{\frac{1}{2}+\nu_r}$, we have for $\varepsilon_0 > q^{\frac{1}{2}}$,

(3.7.27)
$$\sup_{(\nu,k)\in P(r)\times\mathbb{Z}_{+}}|N_{1}^{1}(\nu;\varepsilon_{k})| = \sup_{(\nu,k)\in P(r)\times\mathbb{Z}_{+}}|N_{1}^{1}(\nu;\varepsilon_{k}q^{\nu_{r}})| < \infty$$

by (3.7.17) and Lemma 3.7.2(a).

For $N_1^2(\nu; \varepsilon)$ we want to establish the estimate

(3.7.28)
$$\sup_{k \in \mathbb{Z}_+} |N_1^2(\nu; \varepsilon_k)| \le K_1^2 \prod_{i=1}^r \left(q^{\nu_{i-1}+2} t^{i-1} a^2 |b| \right)^{\nu_i - \nu_{i-1}}$$

for generic $\varepsilon_0 > q^{\frac{1}{2}}$ with $K_1^2 > 0$ independent of $\nu \in P(r)$, in view of the limit (3.7.21). This can be done with the help of Lemma 3.7.2(b). As an example, consider the factor $N_1^{i,2}(\nu;\varepsilon)$ (3.7.22). In view of the limit (3.7.23), we want to prove the estimate

$$\sup_{k \in \mathbb{Z}_+} |N_1^{i,2}(\nu;\varepsilon_k)| \le K_1^{i,2} (q^{\nu_{i-1}} t^{i-1} |b|)^{\nu_i - \nu_{i-1}}$$

for generic $\varepsilon_0 > q^{\frac{1}{2}}$ with $K_1^{i,2} > 0$ independent of $\nu \in P(r)$. This follows for generic $\varepsilon_0 > q^{\frac{1}{2}}$, using the fact that $N_1^{i,2}(\nu;\varepsilon) = 0$ if $\nu \notin P_L(r;\varepsilon)$, by the estimates

$$\begin{split} \sup_{k \in \mathbb{Z}_{+}} |N_{1}^{i,2}(\nu;\varepsilon_{k})| &= \sup_{k \in \mathbb{Z}_{+}} |N_{1}^{i,2}(\nu;q^{\nu_{r}}\varepsilon_{k})| \\ &= \sup_{k \in \mathbb{Z}_{+}} |f_{\{\nu_{i-1},\nu_{i}-\nu_{i-1}\}}(\varepsilon_{2\nu_{r}-\nu_{i}-\nu_{i-1}+2k};\varepsilon_{0}^{-1}t^{2i-2},\varepsilon_{0}^{-1}b^{-1}t^{i-1})| \\ &\leq \sup_{k \in \mathbb{Z}_{+}} |f_{\{\nu_{i-1},\nu_{i}-\nu_{i-1}\}}(\varepsilon_{k};\varepsilon_{0}^{-1}t^{2i-2},\varepsilon_{0}^{-1}b^{-1}t^{i-1})| \\ &\leq K_{1}^{i,2}(q^{\nu_{i-1}}t^{i-1}|b|)^{\nu_{i}-\nu_{i-1}} \end{split}$$

with $K_1^{i,2}$ independent of ν by Lemma 3.7.2(b) (cf. (3.7.23)). In a similar manner, estimates can be given for the other factors of N_1^2 .

For N_1^3 , we want to prove that

(3.7.29)
$$\sup_{k \in \mathbb{Z}_{+}} |N_{1}^{3}(\nu; \varepsilon_{k})| \leq K_{1}^{3} \prod_{i=1}^{n} \left(at^{2(i-1)}q^{\nu_{i-1}+1}\right)^{\nu_{i-1}} \\ \prod_{1 \leq i < j \leq r} \left(t^{i+j-2}q^{\nu_{i-1}+\nu_{j}+1}\right)^{\nu_{i-1}-\nu_{i}}q^{-\binom{\nu_{i}-\nu_{i-1}}{2}}$$

for generic $\varepsilon_0 > q^{\frac{1}{2}}$ with $K_1^3 > 0$ independent of $\nu \in P(r)$, in view of the limit (3.7.24). This follows by straightforward estimates, using (2.5.3) and the fact that $N_1^3(\nu; \varepsilon) = 0$ if $\nu \notin P_L(r; \varepsilon)$. Hence by (3.7.16), (3.7.27), (3.7.28) and (3.7.29) we have the estimate

(3.7.30)
$$\sup_{k\in\mathbb{Z}_{+}} |N_{1}(\nu;\varepsilon_{k})| \leq K_{1} \prod_{i=1}^{r} t^{(i-1)(\nu_{i}+\nu_{i-1})} q^{\nu_{i-1}\nu_{i}-\nu_{i-1}+2\nu_{i}} a^{2\nu_{i}-\nu_{i-1}} |b|^{\nu_{i}-\nu_{i-1}} \\ \cdot \prod_{1\leq i< j\leq r} (t^{i+j-2}q^{\nu_{i-1}+\nu_{j}+1})^{\nu_{i-1}-\nu_{i}} q^{-\binom{\nu_{i}-\nu_{i-1}}{2}}$$

for generic $\varepsilon_0 > q^{\frac{1}{2}}$, with $K_1 > 0$ independent of $\nu \in P(r)$, so in particular,

$$\sup_{k \in \mathbb{Z}_+} |\Delta_{1,r}^{KL}(\nu;\varepsilon_k)| = |M_1(\nu)| \sup_{k \in \mathbb{Z}_+} |N_1(\nu;\varepsilon_k)| \le K \Delta^L(\rho_L q^{\nu}) \prod_{i=1}^r \rho_{L,i} q^{\nu_i}$$

with K > 0 independent of $\nu \in P(r)$. This completes the proof of Lemma 3.7.1(i). \Box

PROOF OF LEMMA 3.7.1(ii). Using the explicit formulas for the weight function Δ (2.3.4), (2.2.3), (2.3.5) and for δ_c (2.4.11) as well as the definition of $\underline{t}_L(\varepsilon)$ (3.2.1), we can rewrite $\Delta_{2,r}^{KL}(\nu, z; \varepsilon)$ as

$$\Delta^{KL}_{2,r}(\nu,z;\varepsilon) = N^1_2(z)N^2_2(\nu,z;\varepsilon)N^3_2(z;\varepsilon)N^4_2(\nu,z;\varepsilon)$$

with

(3.7.31)
$$N_2^1(z) := \delta(z;t) \prod_{i=1}^{n-r} \frac{\left(z_i^2, z_i^{-2}; q\right)_{\infty}}{\left(-q^{\frac{1}{2}} z_i, -q^{\frac{1}{2}} z_i^{-1}, -q^{\frac{1}{2}} a z_i, -q^{\frac{1}{2}} a z_i^{-1}; q\right)_{\infty}}$$

where δ is given by (2.3.5),

$$(3.7.32) \quad N_2^2(\nu, z; \varepsilon) := \prod_{j=1}^{n-r} \frac{\prod_{i=1}^r \left(\varepsilon q^{-\frac{1}{2}-\nu_i} t^{1-i} z_j, \varepsilon q^{-\frac{1}{2}-\nu_i} t^{1-i} z_j^{-1}; q\right)_{\tau}}{\left(\varepsilon b q^{\frac{1}{2}} z_j, \varepsilon b q^{\frac{1}{2}} z_j^{-1}; q\right)_{\infty}} \quad (t = q^{\tau})$$

if $\nu \in P_L(r; \varepsilon)$ and zero otherwise,

$$(3.7.33) \quad N_2^3(z;\varepsilon) := \prod_{i=1}^{n-r} \frac{\left(-\varepsilon^{-1}qt^{r+i-1}, -\varepsilon^{-1}qat^{r+i-1}, \varepsilon^{-1}q^{\frac{1}{2}}z_i, \varepsilon^{-1}q^{\frac{1}{2}}z_i^{-1}; q\right)_{\infty}}{\left(\varepsilon^{-1}q^{\frac{1}{2}}z_i, \varepsilon^{-1}q^{\frac{1}{2}}z_i^{-1}, \varepsilon^{-1}q^{\frac{1}{2}}t^r z_i, \varepsilon^{-1}q^{\frac{1}{2}}t^r z_i^{-1}; q\right)_{\infty}}$$

and

(3.7.34)
$$N_{2}^{4}(\nu, z; \varepsilon) := \prod_{\substack{0 \le i \le r-1\\1 \le j \le n-r}} \frac{\left(\varepsilon^{-1}q^{\frac{1}{2}+\nu_{i}}t^{r}z_{j}, \varepsilon^{-1}q^{\frac{1}{2}+\nu_{i}}t^{r}z_{j}^{-1}; q\right)_{\nu_{i+1}-\nu_{i}}}{\left(\varepsilon^{-1}q^{\frac{1}{2}+\nu_{i}}t^{i}z_{j}, \varepsilon^{-1}q^{\frac{1}{2}+\nu_{i}}t^{i}z_{j}^{-1}; q\right)_{\nu_{i+1}-\nu_{i}}}$$

if $\nu \in P_L(r; \varepsilon)$ and zero otherwise, where $\nu_0 = 0$. Note that N_3^1 is bounded on T^{n-r} . For N_2^2 it follows from Lemma 3.7.2 (a) that $\lim_{k\to\infty} N_2^2(\nu, z; \varepsilon_k) = 1$ for all $\nu \in P(r)$, $z \in T^{n-r}$ and that

$$\sup_{(\nu,z,k)\in P(r)\times T^{n-r}\times\mathbb{Z}_+}|N_2^2(\nu,z;\varepsilon_k)|<\infty$$

for generic $\varepsilon_0 > q^{\frac{1}{2}}$. By Lemma 3.7.2(c) and the fact that 0 < a < 1/q, we have for generic $\varepsilon_0 > 0$ that $\lim_{k\to\infty} N_2^3(z;\varepsilon_k) = 0$ for all $z \in T^{n-r}$ and

$$\sup_{(z,k)\in T^{n-r}\times\mathbb{Z}_+}|N_2^3(z;\varepsilon_k)|<\infty.$$

Finally, Lemma 3.7.2(b) can be used to prove that $\lim_{k\to\infty} N_3^4(\nu, z; \varepsilon_k) = t^{2(n-r)|\nu|}$ for all $\nu \in P(r), z \in T^{n-r}$ and that

$$\sup_{(\nu,z,k)\in P(r)\times T^{n-r}\times\mathbb{Z}_+}|N_2^4(\nu,z;\varepsilon_k)|<\infty$$

for generic $\varepsilon_0 > q^{\frac{1}{2}}$. This completes the proof of Lemma 3.7.1(ii).

3.8. Limit of the orthogonality measure (big q-Jacobi case)

In the next lemma a new expression is given for the weight \underline{c}_B which appears in the definition of $\langle ., . \rangle_B$.

LEMMA 3.8.1. The weight $\underline{c}_B \in (\mathbb{C}^*)^{n+1}$ can be rewritten as

(3.8.1)

$$c_{B,j} = (q;q)_{\infty}^{n} \prod_{i=1}^{j} \frac{\theta(-t^{i+j-n}c/d)}{\theta(-t^{1-i}d/c)\theta(-t^{i}c/d)} \prod_{i=1}^{n-j} \frac{1}{\theta(-t^{1-i}c/d)}$$

$$.q^{-2\tau^{2}((n-j)\binom{j}{2} + \binom{j}{3} + \binom{n-j}{3})} t^{-\binom{j}{2} - \binom{n-j}{2}} t^{\binom{j}{2} - \binom{n-j}{2} + j - n}$$

for $j \in [0, n]$.

PROOF. For j = 0, (3.8.1) follows from (3.4.7) since $c_{B,j} = c_B d_{B,j}$ with $d_{B,0} = 1$. For $j \in [1, n]$, write $\tilde{c}_{B,j}$ for the right hand side of (3.8.1), then by the explicit expression (3.4.4) for $d_{B,j}$ it remains to prove that

$$\frac{\tilde{c}_{B,j}}{\tilde{c}_{B,j-1}} = \prod_{m=j+1}^n \Psi_t(-t^{n-m-j+1}d/c)$$

for $j \in [1, n]$, with Ψ_t given by (3.4.5). This follows by a direct calculation.

3. LIMIT TRANSITIONS FOR MULTIVARIABLE ORTHOGONAL POLYNOMIALS

The remainder of this section is devoted to a proof of Proposition 3.4.4.

Fix in this section $(a, b, c, d) \in V_B$ with $a, b \neq 0$. With slight modifications of the formulas, the proof goes also through for a = 0 or b = 0.

For $\varepsilon \in \mathbb{R}_{>0}$ we write $\rho_{B,j}(\varepsilon) := t_{B,0}(\varepsilon)t^{j-1} = (qc/d)^{\frac{1}{2}}\varepsilon^{-1}t^{j-1}$ and we write $\sigma_{B,j}(\varepsilon) := t_{B,1}(\varepsilon)t^{j-1} = -(qd/c)^{\frac{1}{2}}\varepsilon^{-1}t^{j-1}$ for $j \in \mathbb{Z}$, where $\underline{t}_B(\varepsilon)$ is given by (3.2.10). Then for $\varepsilon > 0$ sufficiently small, we have

$$F(r;\underline{t}_B(\varepsilon);t) = \bigcup_{\substack{l+m=r\\l,m\in\mathbb{Z}_+}} D_0(l;\underline{t}_B(\varepsilon);t) \times D_1(m;\underline{t}_B(\varepsilon);t) \subset \mathbb{C}^r$$

where F(r) is given by (2.6.3) and

(3.8.2)
$$D_0(l; \underline{t}_B(\varepsilon); t) = \{ \rho_B(\varepsilon) q^{\nu} \mid \nu \in P_B^{(0)}(l; \varepsilon) \},$$
$$P_B^{(0)}(l; \varepsilon) := \{ \nu \in P(l) \mid |\rho_{B,l}(\varepsilon) q^{\nu_l}| > 1 \},$$

if l > 0, respectively

(3.8.3)
$$D_1(m; \underline{t}_B(\varepsilon); t) = \{ \sigma_B(\varepsilon) q^{\nu} \mid \nu \in P_B^{(1)}(m; \varepsilon) \},$$
$$P_B^{(1)}(m; \varepsilon) := \{ \nu \in P(m) \mid |\sigma_{B,m}(\varepsilon) q^{\nu_m}| > 1 \}$$

if m > 0. Write

$$(3.8.4) \left(\prod_{i=1}^{n} \left(-\varepsilon^{-2}qt^{i-1};q\right)_{\infty}\right) \left(\left(cd/q\right)^{\frac{1}{2}}\varepsilon\right)^{|\lambda|+|\mu|} \langle m_{\lambda},m_{\mu}\rangle_{\underline{t}_{B}}(\varepsilon),t$$
$$= \sum_{r,l,m,\nu,\nu'} \iint_{z\in T^{n-r}} \left(m_{\lambda}m_{\mu}\right) \left(\rho_{B}q^{\nu},\sigma_{B}q^{\nu'},\left(cd/q\right)^{\frac{1}{2}}\varepsilon z|\left(cd/q\right)^{\frac{1}{2}}\varepsilon\right) \mathcal{W}_{l,m;r}^{B}(\nu,\nu',z;\varepsilon)\frac{dz}{z}$$

where $\rho_{B,i} := ct^{i-1}$, $\sigma_{B,i} := -dt^{i-1}$ and $m_{\lambda}(z|u)$ is given by (3.3.8), and the sum is over five tuples (r, l, m, ν, ν') with $r \in [0, n]$, $l, m \in \mathbb{Z}_+$ with l + m = r, $\nu \in P(l)$, $\nu' \in P(m)$, and with the renormalized weight $\mathcal{W}^B_{l,m;r}(\nu, \nu', z; \varepsilon)$ given by

(3.8.5)
$$\mathcal{W}_{l,m;r}^{B}(\nu,\nu',z;\varepsilon) := \frac{2^{r}(n-r+1)_{r}}{(2\pi i)^{n-r}} \Delta_{1,l,m}^{KB}(\nu,\nu';\varepsilon) \Delta_{2,l,m}^{KB}(\nu,\nu',z;\varepsilon)$$

when r = l + m, with

(3.8.6)
$$\Delta_{1,l,m}^{KB}(\nu,\nu';\varepsilon) := \left(\prod_{i=1}^{r} \left(-\varepsilon^{-2}qt^{i-1};q\right)_{\infty}\right) \delta_{c}\left(\rho_{B}(\varepsilon)q^{\nu};\sigma_{B}(\varepsilon)q^{\nu'}\right) \\ \cdot \Delta^{(d)}\left(\rho_{B}(\varepsilon)q^{\nu};t_{B,0}(\varepsilon)\right) \Delta^{(d)}\left(\sigma_{B}(\varepsilon)q^{\nu'};t_{B,1}(\varepsilon)\right)$$

if $\nu\in P_B^{(0)}(l;\varepsilon),$ $\nu'\in P_B^{(1)}(m;\varepsilon)$ and zero otherwise,

(3.8.7)
$$\Delta_{2,l,m}^{KB}(\nu,\nu',z;\varepsilon) := \prod_{i=1}^{n-l} \left(-\varepsilon^{-2} q t^{r+i-1}; q \right)_{\infty} \cdot \Delta(z; \underline{t}_{B}(\varepsilon); t) \delta_{c} \left(\rho_{B}(\varepsilon) q^{\nu}; z \right) \delta_{c}(\sigma_{B}(\varepsilon) q^{\nu'}; z)$$

if $\nu \in P_B^{(0)}(l;\varepsilon)$, $\nu' \in P_B^{(1)}(m;\varepsilon)$ and zero otherwise, with δ_c given by (2.4.11). The obvious conventions are used when l = 0, m = 0 or r = n (compare with the little q-Jacobi case in Section 3.7). In particular, $\Delta_{2,l,n-l}^{KB}(\nu,\nu';\varepsilon) = 1$ for $\nu \in P_B^{(0)}(l;\varepsilon)$, $\nu' \in P_B^{(1)}(n-l;\varepsilon)$ and $l \in [0,n]$. The following lemma will be used to pull a limit $\varepsilon_k \downarrow 0$ in the right hand side of

The following lemma will be used to pull a limit $\varepsilon_k \downarrow 0$ in the right hand side of (3.8.4) through the integration over $z \in T^{n-r}$ and through the infinite sums over $\nu \in P(l)$ and $\nu' \in P(m)$ for some sequence $\{\varepsilon_k\}_{k \in \mathbb{Z}_+}$ in $\mathbb{R}_{>0}$ converging to 0.

LEMMA 3.8.2. Keep the notations and conventions as above. Let $l, m \in \mathbb{Z}_+$ with $l + m \in [0, n]$ and write r := l + m. Then there exists a sequence $\{\varepsilon_k\}_{k \in \mathbb{Z}_+}$ in $\mathbb{R}_{>0}$ converging to 0 such that (i) for all $\nu \in P(l), \nu' \in P(m)$

$$\lim_{k \to \infty} \Delta_{1,l,m}^{KB}(\nu,\nu';\varepsilon_k) = (q;q)_{\infty}^{-2r} c_{B,l} \Delta^B(\rho_B q^{\nu},\sigma_B q^{\nu'}) \prod_{i=1}^l \rho_{B,i} q^{\nu_i} \prod_{j=1}^m |\sigma_{B,j}| q^{\nu'_j}$$

and there exists a $K \in \mathbb{R}_{>0}$ independent of $\nu \in P(l)$ and $\nu' \in P(m)$ such that

$$\sup_{k \in \mathbb{Z}_+} |\Delta_{1,l,m}^{KB}(\nu,\nu';\varepsilon_k)| \le K c_{B,l} \Delta^B(\rho_B q^{\nu}, \sigma_B q^{\nu'}) \prod_{i=1}^l \rho_{B,i} q^{\nu_i} \prod_{j=1}^m |\sigma_{B,j}| q^{\nu'_j}.$$

for all $\nu \in P(l)$ and all $\nu' \in P(m)$, where $\Delta^B(z) = \Delta^B(z; a, b, c, d; t)$ is given by (3.4.3).

(ii) if r < n, then $\lim_{k\to\infty} \Delta_{2,l,m}^{KB}(\nu,\nu',z;\varepsilon_k) = 0$ for all $\nu \in P(l), \nu' \in P(m), z \in T^{n-r}$ and

$$\sup_{(k,\nu,\nu',z)\in\mathbb{Z}_+\times P(l)\times P(m)\times T^{n-r}} |\Delta_{2,l,m}^{KB}(\nu,\nu',z;\varepsilon_k)| < \infty$$

The proof of Proposition 3.4.4 is now an easy consequence of Lemma 3.8.2 and Lebesgue's dominated convergence Theorem. Indeed, the infinite sum

$$(3.8.8) \quad (1-q)^{-n} \langle 1,1 \rangle_B = \sum_{(\nu,\nu',l)} c_{B,l} \Delta^B(\rho_B q^{\nu}, \sigma_B q^{\nu'}) \prod_{i=1}^l \rho_{B,i} q^{\nu_i} \prod_{j=1}^{n-l} |\sigma_{B,j}| q^{\nu'_j}$$

is absolutely convergent (cf. [116, proof of Proposition 6.1]), where the sum is taken over the three tuples (ν, ν', l) with $\nu \in P(l)$ and $\nu' \in P(n-l)$ and $l \in [0, n]$. Furthermore,

$$\sup_{\nu,\nu',z,\varepsilon} |m_{\lambda}(\rho_B q^{\nu}, \sigma_B q^{\nu'}, (cd/q)^{\frac{1}{2}} \varepsilon z |(cd/q)^{\frac{1}{2}} \varepsilon)| < \infty,$$

where the supremum is taken over the four tuples $(\nu, \nu', z, \varepsilon)$ with $\nu \in P_B^{(0)}(l; \varepsilon), \nu' \in P_B^{(1)}(m; \varepsilon), z \in T^{n-r}$ (r = l + m) and $\varepsilon > 0$, hence it follows from Lebesgue's dominated convergence Theorem, (3.3.9), (3.8.4), (3.8.8) and Lemma 3.8.2 that

$$\lim_{k \to \infty} \left(\prod_{i=1}^{n} \left(-\varepsilon_{k}^{-2} q t^{i-1}; q \right)_{\infty} \right) \left((cd/q)^{\frac{1}{2}} \varepsilon_{k} \right)^{|\lambda|+|\mu|} \langle m_{\lambda}, m_{\mu} \rangle_{\underline{t}_{B}(\varepsilon_{k}), t} \\ = \frac{2^{n} n!}{(q;q)_{\infty}^{2n}} \sum_{l=0}^{n} \sum_{\substack{\nu \in P(l) \\ \nu' \in P(n-l)}} c_{B,l} \left(\tilde{m}_{\lambda} \tilde{m}_{\mu} \Delta^{B} \right) \left(\rho_{B} q^{\nu}, \sigma_{B} q^{\nu'} \right) \prod_{i=1}^{l} \rho_{B,i} q^{\nu_{i}} \prod_{j=1}^{n-l} |\sigma_{B,j}| q^{\nu'_{j}} \\ = 2^{n} n! (1-q)^{-n} \left(q; q \right)_{\infty}^{-2n} \langle \tilde{m}_{\lambda}, \tilde{m}_{\mu} \rangle_{B,t}^{a,b,c,d},$$

for some sequence $\{\varepsilon_k\}_{k\in\mathbb{Z}_+}$ in $\mathbb{R}_{>0}$ converging to 0. So for the proof of Proposition 3.4.4, it suffices to prove Lemma 3.8.2.

Proof of Lemma 3.8.2.

Using the explicit expressions for $\Delta^{(d)}$ (2.4.7), δ_c (2.4.11) and $\underline{t}_B(\varepsilon)$ (3.2.10), we can write

(3.8.9)
$$\Delta_{1,l,m}^{KB}(\nu,\nu';\varepsilon) := U_0(\nu,\nu';l,m)U_+(\varepsilon;\nu,\nu';l,m)U_-(\varepsilon;\nu,\nu';l,m)$$

with U_0, U_+ respectively U_- the factor of $\Delta_{1,l,m}^K$ consisting of products of q-shifted factorials of the form $(e;q)_s$, $(\varepsilon^2 e;q)_s$ respectively $(\varepsilon^{-2}e;q)_s$ $(s \in \mathbb{Z}_+ \cup \{\infty\})$. By a straightforward computation, the factors U_0, U_+ and U_- can be explicitly given by

(3.8.10)
$$U_{0}(\nu,\nu';l,m) := \Psi_{0}(\nu,l;a,b,c,d)\Psi_{0}(\nu',m;b,a,d,c) \\ \prod_{\substack{1 \le i \le l \\ 1 \le j \le m}} \left(-t^{i-j}q^{\nu_{i}-\nu'_{j}}c/d, -t^{j-i}q^{\nu'_{j}-\nu_{i}}d/c;q\right)_{\tau}$$

if $(\nu, \nu') \in P_0(l; \varepsilon) \times P_1(m; \varepsilon)$, and zero otherwise, with $t = q^{\tau}$ and with

$$\Psi_{0}(\nu, l; a, b, c, d) := F_{1}(\nu)$$

$$(3.8.11) \qquad \cdot \prod_{i=1}^{l} \frac{1}{(q, -t^{1-i}q^{-\nu_{i-1}}d/c, at^{i-1}q^{\nu_{i-1}+1}, -bt^{i-1}q^{\nu_{i-1}+1}c/d; q)_{\infty}}$$

$$\cdot \prod_{i=1}^{l} \frac{(at^{i-1}q^{\nu_{i-1}+1}, -bt^{i-1}q^{\nu_{i-1}+1}c/d; q)_{\nu_{i}-\nu_{i-1}}}{(q, -t^{i-1}q^{\nu_{i-1}+1}c/d; q)_{\nu_{i}-\nu_{i-1}}(abt^{i-1}q^{\nu_{i-1}+1})^{\nu_{i}-\nu_{i-1}}}$$

where $\nu_0 = 0$ and $F_1(\nu)$ is given by (3.7.11), and

(3.8.12)
$$U_{+}(\varepsilon;\nu,\nu';l,m) := \Psi_{+}(\varepsilon;\nu,l;a,b,c,d)\Psi_{+}(\varepsilon;\nu',m;b,a,d,c) \\ \prod_{\substack{1 \le i \le l \\ 1 \le j \le m}} \left(-\varepsilon^{2}t^{2-i-j}q^{-1-\nu_{i}-\nu'_{j}};q\right)_{\tau}$$

if $(\nu,\nu')\in P_B^{(0)}(l;\varepsilon)\times P_B^{(1)}(m;\varepsilon)$ and zero otherwise, with

(3.8.13)
$$\Psi_{+}(\varepsilon;\nu,l;a,b,c,d) := \prod_{i=1}^{l} \frac{\left(\varepsilon^{2}t^{2(1-i)}q^{-2\nu_{i-1}-1}d/c;q\right)_{\infty}}{\left(\varepsilon^{2}at^{1-i}q^{-\nu_{i-1}}d/c,-\varepsilon^{2}bt^{1-i}q^{-\nu_{i-1}};q\right)_{\infty}} \\ \cdot \prod_{1 \le i < j \le l} \left(\varepsilon^{2}t^{2-i-j}q^{-\nu_{i}-\nu_{j}-1}d/c;q\right)_{\tau}$$

and

(3.8.14)
$$U_{-}(\varepsilon;\nu,\nu';l,m) := \Psi_{-}(\varepsilon;\nu,l;a,b,c,d)\Psi_{-}(\varepsilon;\nu',m;b,a,d,c) \\ \cdot \left(\prod_{j=1}^{m} \frac{\left(-\varepsilon^{-2}qt^{l+j-1};q\right)_{\infty}}{\left(-\varepsilon^{-2}qt^{j-1};q\right)_{\infty}}\right) \prod_{\substack{1 \le i \le l\\ 1 \le j \le m}} \left(-\varepsilon^{-2}t^{i+j-2}q^{\nu_{i}+\nu'_{j}+1};q\right)_{\tau}$$

if $(\nu,\nu') \in P_B^{(0)}(l;\varepsilon) \times P_B^{(1)}(m;\varepsilon)$ and zero otherwise, with

$$\begin{aligned} \Psi_{-}(\varepsilon;\nu,l;a,b,c,d) &:= \prod_{i=1}^{l} \frac{\left(\varepsilon^{-2}t^{2(i-1)}q^{2\nu_{i}+1}c/d;q\right)_{1}}{\left(\varepsilon^{-2}t^{2(i-1)}q^{2\nu_{i}-1+1}c/d;q\right)_{1}} \\ (3.8.15) \qquad \cdot \prod_{i=1}^{l} \frac{\left(\varepsilon^{-2}t^{2(i-1)}q^{2\nu_{i}-1+1}c/d, -\varepsilon^{-2}t^{i-1}q^{\nu_{i}-1+1};q\right)_{\nu_{i}-\nu_{i-1}}}{\left(\varepsilon^{-2}a^{-1}t^{i-1}q^{\nu_{i}-1+1}c/d, -\varepsilon^{-2}b^{-1}t^{i-1}q^{\nu_{i}-1+1};q\right)_{\nu_{i}-\nu_{i-1}}} \\ \cdot \frac{\prod_{i=1}^{l} \left(-\varepsilon^{-2}qt^{i-1};q\right)_{\nu_{i-1}}}{\prod_{1\leq i< j\leq l} \left(\varepsilon^{-2}t^{i+j-2}q^{\nu_{i}-1+\nu_{j}+1}c/d;q\right)_{\nu_{i}-\nu_{i-1}}}. \end{aligned}$$

For given $\varepsilon_0 \in \mathbb{R}^*$, we write $\varepsilon_k := \varepsilon_0 q^k$. Then for generic $\varepsilon_0 > 0$ we have

(3.8.16)
$$\lim_{k \to \infty} U_+(\varepsilon_k; \nu, \nu'; l, m) = 1$$

for all $(\nu,\nu') \in P(l) \times P(m)$ by Lemma 3.7.2(a). By (3.7.25), we have for generic $\varepsilon_0 > 0$

(3.8.17)

$$\lim_{k \to \infty} \frac{\prod_{i=1}^{l} \left(-\varepsilon_{k}^{-2} q t^{i-1}; q\right)_{\nu_{i-1}}}{\prod_{1 \le i < j \le l} \left(\varepsilon_{k}^{-2} t^{i+j-2} q^{\nu_{i-1}+\nu_{j}+1} c/d; q\right)_{\nu_{i}-\nu_{i-1}}}$$

$$= \prod_{i=1}^{l} q^{\binom{\nu_{i-1}+1}{2}} t^{(i-1)\nu_{i-1}} \prod_{1 \le i < j \le l} \left(-t^{i+j-2} q^{\nu_{i-1}+\nu_{j}+1} c/d\right)^{\nu_{i-1}-\nu_{i}} q^{-\binom{\nu_{i}-\nu_{i-1}}{2}}$$

for the factor of Ψ_{-} in the third line of (3.8.15). The factor of U_{-} in the second line of (3.8.14) can be rewritten as

(3.8.18)
$$\begin{pmatrix} \prod_{j=1}^{m} \frac{(-\varepsilon^{-2}qt^{l+j-1};q)_{\infty}}{(-\varepsilon^{-2}qt^{j-1};q)_{\infty}} \end{pmatrix} \prod_{\substack{1 \le i \le l \\ 1 \le j \le m}} (-\varepsilon^{-2}t^{i+j-2}q^{\nu_i+\nu'_j+1};q)_{\tau} \\ = \prod_{\substack{1 \le i \le l \\ 1 \le j \le m}} \frac{(-\varepsilon^{-2}qt^{i+j-1};q)_{\nu_i+\nu'_j}}{(-\varepsilon^{-2}qt^{i+j-2};q)_{\nu_i+\nu'_j}} .$$

It follows then from (3.8.14), (3.8.15), (3.8.17), (3.8.18) and Lemma 3.7.2(b) that for generic $\varepsilon_0 > 0$,

(3.8.19)
$$\lim_{k \to \infty} U_{-}(\varepsilon_{k}; \nu, \nu'; l, m) = t^{m|\nu| + l|\nu'|} \Psi^{\infty}_{-}(\nu, l; a, b, c, d) \Psi^{\infty}_{-}(\nu', m; b, a, d, c)$$

with

(3.8.20)
$$\Psi_{-}^{\infty}(\nu, l; a, b, c, d) := \prod_{i=1}^{l} \left(t^{i-1} q^{\nu_{i-1}+2} ab \right)^{\nu_{i}-\nu_{i-1}} q^{\binom{\nu_{i-1}+1}{2}} t^{(i-1)\nu_{i-1}} \\ \cdot \prod_{1 \le i < j \le l} \left(-t^{i+j-2} q^{\nu_{i-1}+\nu_{j}+1} c/d \right)^{\nu_{i-1}-\nu_{i}} q^{-\binom{\nu_{i}-\nu_{i-1}}{2}}.$$

We will rewrite now U_0 in the form

(3.8.21)
$$U_{0}(\nu,\nu';l,m) = (q;q)_{\infty}^{-r} \prod_{i=1}^{l} \frac{\theta(-t^{i-m}c/d)}{\theta(-t^{1-i}d/c)\theta(-t^{i}c/d)} \prod_{j=1}^{m} \frac{1}{\theta(-t^{1-j}c/d)} \\ .C_{0}(\nu,\nu';l,m)\Delta^{B}(\rho_{B}q^{\nu},\sigma_{B}q^{\nu'};a,b,c,d;t)$$

and we determine the factor $C_0(\nu, \nu'; l, m)$ explicitly. Using (3.4.8) and using the formula $\theta(x) = \theta(qx^{-1})$ for the Jacobi theta function $\theta(x)$ (3.4.6), $\Psi_0(\nu, l; a, b, c, d)$ (3.8.11) can be rewritten as

(3.8.22)
$$\Psi_{0}(\nu,l;a,b,c,d) = F_{1}(\nu)$$
$$\cdot \prod_{i=1}^{l} \frac{v_{B}(\rho_{B,i}q^{\nu_{i}};a,b,c,d)(t^{i-1}c/d)^{\nu_{i-1}}q^{\binom{\nu_{i-1}+1}{2}}}{\theta(-t^{1-i}d/c)(q^{\nu_{i}+1}t^{i-1};q)_{\infty}(q;q)_{\nu_{i}-\nu_{i-1}}(abt^{i-1}q^{\nu_{i-1}+1})^{\nu_{i}-\nu_{i-1}}}$$

where v_B (3.2.16) is the one variable weight function for the big q-Jacobi polynomials. Since $v_B(-dx; a, b, c, d) = v_B(dx; b, a, d, c)$, we obtain from (3.8.22)

$$\Psi_0(\nu',m;b,a,d,c) = F_1(\nu')$$

$$(3.8.23) \qquad \cdot \prod_{j=1}^{m} \frac{v_B(\sigma_{B,j}q^{\nu'_j};a,b,c,d)(t^{j-1}d/c)^{\nu'_{j-1}}q^{\binom{\nu'_{j-1}+1}{2}}}{\theta(-t^{1-j}c/d)(q^{\nu'_j+1}t^{j-1};q)_{\infty}(q;q)_{\nu'_j-\nu'_{j-1}}(abt^{j-1}q^{\nu'_{j-1}+1})^{\nu'_j-\nu'_{j-1}}}.$$

The factor $F_1(\nu)$ (3.7.11) of $\Psi_0(\nu, l; a, b, c, d)$ can be rewritten as

(3.8.24)

$$F_{1}(\nu) = \delta_{qJ}(\rho_{B}q^{\nu})q^{-2\tau^{2}\binom{l}{3}}c^{-l(l-1)\tau}\prod_{i=1}^{l}\frac{(t^{i-1}q^{1+\nu_{i}};q)_{\infty}}{(q^{1+\nu_{i}-\nu_{i-1}};q)_{\infty}}t^{-2(l-i)\nu_{i}}$$

$$\cdot\prod_{1\leq i< j\leq l}(-q^{\nu_{j}-\nu_{i}+1}t^{j-i})^{\nu_{i}-\nu_{i-1}}q^{\binom{\nu_{i}-\nu_{i-1}}{2}}$$

for $\nu \in P(l)$ where δ_{qJ} is given by (3.3.6) and $t = q^{\tau}$. This follows from (3.7.13) since $\delta_{qJ}(\rho_B q^{\nu}) = c^{l(l-1)\tau} \delta_{qJ}(\rho_L q^{\nu})$ for $\nu \in P(l)$ (here $\rho_{L,i} = t^{i-1}$). Similarly, we have for the factor $F_1(\nu')$ of $\Psi_0(\nu', m; b, a, d, c)$,

(3.8.25)

$$F_{1}(\nu') = \delta_{qJ}(\sigma_{B}q^{\nu'})q^{-2\tau^{2}\binom{m}{3}}d^{-m(m-1)\tau} \prod_{j=1}^{m} \frac{\left(t^{j-1}q^{1+\nu'_{j}};q\right)_{\infty}}{\left(q^{1+\nu'_{j}-\nu'_{j-1}};q\right)_{\infty}}t^{-2(m-j)\nu'_{j}} \cdot \prod_{1 \le i < j \le m} \left(-q^{\nu'_{j}-\nu'_{i}+1}t^{j-i}\right)^{\nu'_{i}-\nu'_{i-1}}q^{\binom{\nu'_{i}-\nu'_{i-1}}{2}},$$

since $\delta_{qJ}(\sigma_B q^{\nu'}) = d^{m(m-1)\tau} \delta_{qJ}(\rho_L q^{\nu'})$ for $\nu' \in P(m)$. Finally, we set for $z = (z_1, \ldots, z_r)$ with r := l + m,

$$\tilde{\delta}_{qJ}^{l}(z) := \prod_{\substack{1 \le i \le l \\ l+1 \le j \le r}} |z_i - z_j| |z_i|^{2\tau - 1} (qt^{-1}z_j/z_i; q)_{2\tau - 1},$$

then the factor of U_0 in the second line of (3.8.10) can be rewritten as

(3.8.26)
$$\prod_{\substack{1 \le i \le l \\ 1 \le j \le m}} \left(-t^{i-j} q^{\nu_i - \nu'_j} c/d, -t^{j-i} q^{\nu'_j - \nu_i} d/c; q \right)_{\tau}$$

$$= \tilde{\delta}_{qJ}^{l}(\rho_{B}q^{\nu}, \sigma_{B}q^{\nu'})t^{m|\nu|-l|\nu'|} \prod_{i=1}^{l} \frac{\theta(-t^{i-m}c/d)}{\theta(-t^{i}c/d)(ct^{i-1}q^{\nu_{i}})^{2m\tau}}$$

for $\nu \in P(l)$ and $\nu' \in P(m),$ since we have for $i \in [1,l], j \in [1,m]$ that

$$(3.8.27) (-t^{i-j}q^{\nu_i-\nu'_j}c/d, -t^{j-i}q^{\nu'_j-\nu_i}d/c; q)_{\tau} = \frac{\theta(-t^{i-j}c/d)(-t^{j-i-1}q^{1+\nu'_j-\nu_i}d/c; q)_{\infty}}{\theta(-t^{i-j+1}c/d)(-t^{j-i+1}q^{\nu'_j-\nu_i}d/c; q)_{\infty}} (1+t^{j-i}q^{\nu'_j-\nu_i}d/c)t^{\nu_i-\nu'_j}$$

(formula (3.8.27) follows from a straightforward computation using (2.5.3), (3.4.8) and $\theta(x) = \theta(qx^{-1})$). Since

$$\delta_{qJ}(\rho_B q^{\nu}, \sigma_B q^{\nu'}) = \delta_{qJ}(\rho_B q^{\nu}) \delta_{qJ}(\sigma_B q^{\nu'}) \tilde{\delta}^l_{qJ}(\rho_B q^{\nu}, \sigma_B q^{\nu'})$$

for $\nu \in P(l)$ and $\nu' \in P(m)$, we obtain from (3.4.3), (3.8.10), (3.8.22), (3.8.23), (3.8.24), (3.8.25) and (3.8.26) that (3.8.21) holds with

(3.8.28)
$$C_0(\nu,\nu';l,m) = t^{-m|\nu|-l|\nu'|} c^{-2lm\tau} q^{-2m\binom{l}{2}\tau^2} \\ \hat{C}_0(\nu,l;a,b,c,d) \hat{C}_0(\nu',m;b,a,d,c)$$

where

$$(3.8.29) \qquad \hat{C}_{0}(\nu,l;a,b,c,d) := q^{-2\tau^{2}\binom{l}{3}}c^{-2\binom{l}{2}\tau} \\ \cdot \prod_{i=1}^{l} (t^{i-1}c/d)^{\nu_{i-1}}q^{\binom{\nu_{i-1}+1}{2}}t^{-2(l-i)\nu_{i}}(abt^{i-1}q^{\nu_{i-1}+1})^{\nu_{i-1}-\nu_{i}} \\ \cdot \prod_{1 \le i < j \le l} (-q^{\nu_{j}-\nu_{i}+1}t^{j-i})^{\nu_{i}-\nu_{i-1}}q^{\binom{\nu_{i}-\nu_{i-1}}{2}}.$$

By Lemma 3.8.1 (with n in the right hand side of (3.8.1) equal to r = l + m in this situation), and by (3.8.9), (3.8.16), (3.8.19), (3.8.21) and (3.8.28) we have for generic $\varepsilon_0 > 0$ that

(3.8.30)
$$\lim_{k \to \infty} \Delta_{1,l,m}^{KB}(\nu,\nu';\varepsilon_k) = U_0(\nu,\nu';l,m) \lim_{k \to \infty} U_-(\varepsilon_k;\nu,\nu';l,m) \\ = (q;q)_{\infty}^{-2r} c_{B,l} \Delta^B(\rho_B q^{\nu},\sigma_B q^{\nu'};a,b,c,d;t) \\ .E(\nu,l;a,b,c,d) E(\nu',m;b,a,d,c)$$

for $\nu \in P(l)$ and $\nu' \in P(m)$, with

(3.8.31) $E(\nu, l; a, b, c, d) := q^{2\tau^2 \binom{l}{3}} t^{\binom{l}{2}} c^{2\binom{l}{2}\tau+l} \hat{C}_0(\nu, l; a, b, c, d) \Psi^{\infty}_{-}(\nu, l; a, b, c, d).$ By (3.8.20) and (3.8.29), we obtain

$$E(\nu, l; a, b, c, d) = c^{l} t^{\binom{l}{2}} q^{|\nu|} = \prod_{i=1}^{l} \rho_{B,i} q^{\nu_{i}}, \quad \nu \in P(l)$$

In particular we have $E(\nu', m; b, a, d, c) = \prod_{j=1}^{m} |\sigma_{B,j}| q^{\nu'_j}$ for $\nu' \in P(m)$, hence by (3.8.30)

$$\lim_{k \to \infty} \Delta_{1,l,m}^{KB}(\nu,\nu';\varepsilon_k) = (q;q)_{\infty}^{-2r} c_{B,l} \Delta^B(\rho_B q^{\nu},\sigma_B q^{\nu'}) \prod_{i=1}^l \rho_{B,i} q^{\nu_i} \prod_{j=1}^m |\sigma_{B,j}| q^{\nu'_j}$$

for all $\nu \in P(l)$, $\nu' \in P(m)$. To complete the proof of Lemma 3.8.2(i), it suffices to prove that for generic $\varepsilon_0 > \max((qc/d)^{\frac{1}{2}}, (qd/c)^{\frac{1}{2}})$,

$$\sup_{k \in \mathbb{Z}_+} |\Delta_{1,l,m}^{KB}(\nu,\nu';\varepsilon_k)| \le K c_{B,l} \Delta^B(\rho_B q^{\nu}, \sigma_B q^{\nu'}; a, b, c, d; t) \prod_{i=1}^l \rho_{B,i} q^{\nu_i} \prod_{i=1}^m |\sigma_{B,j}| q^{\nu'_j}$$

.

for all $\nu \in P(l)$ and all $\nu' \in P(m)$, with K > 0 independent of ν and ν' . This can be proved by similar arguments as in the little q-Jacobi case (see proof of Lemma 3.7.1(i)). In particular, the estimates for almost all factors of $\Delta_{1,l,m}^{KB}$ can be obtained from one of

the three estimates of Lemma 3.7.2. Only for the factor in the third line of the expression of Ψ_{-} (3.8.15) one needs a seperate argument to establish the desired estimate. We may assume that this factor is zero unless $\nu \in P_B^{(0)}(l;\varepsilon)$ and $\nu' \in P_B^{(1)}(m;\varepsilon)$. In view of the limit (3.8.17), we would like to establish the estimate

$$\sup_{\substack{\{k \in \mathbb{Z}_{+} | \nu \in P_{B}^{(0)}(l;\varepsilon_{k}), \nu' \in P_{B}^{(1)}(m;\varepsilon_{k})\}}} \left| \frac{\prod_{i=1}^{l} \left(-\varepsilon_{k}^{-2}qt^{i-1};q\right)_{\nu_{i-1}}}{\prod_{1 \leq i < j \leq l} \left(\varepsilon_{k}^{-2}t^{i+j-2}q^{\nu_{j}+\nu_{i-1}+1}c/d;q\right)_{\nu_{i}-\nu_{i-1}}} \right| \\
\leq K' \prod_{i=1}^{l} q^{\binom{\nu_{i-1}+1}{2}} t^{(i-1)\nu_{i-1}} \prod_{1 \leq i < j \leq l} \left(t^{i+j-2}q^{\nu_{i-1}+\nu_{j}+1}c/d\right)^{\nu_{i-1}-\nu_{i}} q^{-\binom{\nu_{i}-\nu_{i-1}}{2}}$$

with K' > 0 independent of $\nu \in P(l)$ and $\nu' \in P(m)$, and $\varepsilon_0 > \max((qc/d)^{\frac{1}{2}}, (qd/c)^{\frac{1}{2}})$ generic. This can be done similarly as was done for the factor N_1^3 (3.7.19) in the little *q*-Jacobi case.

The proof of Lemma 3.8.2(ii) is similar to the proof of Lemma 3.7.1(iii).

. .

CHAPTER 4

Harmonic analysis on quantum Grassmannians

4.1. Introduction

In this chapter certain subfamilies of the Koornwinder polynomials and certain subfamilies of the multivariable big and little q-Jacobi polynomials are interpreted on quantum analogues of the complex Grassmannian

$$U/K := U(n)/(U(n-l) \times U(l)), \quad (l \le [n/2]),$$

where U(n) is group of n by n unitary matrices.

The first connections between q-special functions and quantum groups were described in the papers [129], [87], [63], in which the little q-Jacobi polynomials were related to matrix coefficients of irreducible representations of the quantum analogue of SU(2). In [101] Podles defined a one parameter family of quantum analogues of 2-spheres. The related zonal spherical functions were identified with big q-Jacobi polynomials by Noumi and Mimachi [93].

In [64] and [66], Koornwinder generalized these results for quantum 2-spheres by replacing the notion of invariance under a quantum subgroup by the notion of invariance under a twisted primitive element in the quantized universal enveloping algebra. The problem that the quantized function algebra on U has less quantum subgroups then one would expect from the classical setting is circumvented by this infinitesimal approach. In particular, the infinitesimal approach allowed Koornwinder to identify a two parameter family of Askey-Wilson polynomials as zonal spherical functions on the one-parameter family of Podles spheres. The special cases of big and little q-Jacobi polynomials could then be reobtained by sending one respectively two parameters to infinity. On the level of the spherical functions, this corresponds with the limit transitions from Askey-Wilson polynomials to big respectively little q-Jacobi polynomials as discussed in the previous chapter.

Analogous statements are valid for the projective space, see Noumi, Yamada and Mimachi [96] for the little q-Jacobi case, and Dijkhuizen and Noumi [24] for the general case. In this chapter, the generalizations of these results to the higher rank cases of the complex Grassmannian are considered.

Noumi [91] applied the infinitesimal method for the first time on higher rank symmetric spaces. The quantized symmetric spaces were now defined using invariance under certain 2-sided coideals in the quantized universal enveloping algebra. So far, the method has been succesfully applied for all compact symmetric spaces of classical type, e.g. [94],

[95], [92], [124], [24]. In all cases, the zonal spherical functions can be identified with Koornwinder polynomials or Macdonald polynomials.

Noumi, Dijkhuizen and Sugitani [92] introduced a one parameter family of quantum analogues of the complex Grassmannian and they announced results on the corresponding harmonic analysis. The spherical functions associated with this family of quantum Grassmannians can be identified with a two parameter subfamily of the Koornwinder polynomials. The quantum subgroup case, which is defined using invariance with respect to the obvious quantum subgroup corresponding to $K = U(n - l) \times U(l)$, can be formally obtained from the one parameter family of quantum Grassmannians by sending the parameter to infinity. In this chapter it is shown that this type of limit transitions on quantum Grassmannians is compatible with the limit transitions from Koornwinder polynomials to multivariable big and little q-Jacobi polynomials, which were proved in the previous chapter. In particular, the zonal spherical functions on the quantum complex Grassmannian for the quantum subgroup case can be identified with multivariable big and little q-Jacobi polynomials. These results will follow from a careful study of the one parameter family of quantum Grassmannians which was introduced in [92] by Noumi, Dijkhuizen and Sugitani.

This chapter is organized as follows. In Section 4.2 the harmonic analysis on the classical complex Grassmannian is recalled. Section 4.3 contains preliminary results on the quantum unitary group and the corresponding quantized universal enveloping algebra. In Section 4.4 the spherical representations corresponding to the quantized complex Grassmannian (quantum subgroup case) are classified. In Section 4.5 and Section 4.6 we recall the announced results of Noumi, Dijkhuizen and Sugitani [92] about the harmonic analysis on a one parameter family of quantum Grassmannians, and we provide detailed proofs for most of these results. Finally, in Section 4.7 the limit transitions from Koornwinder polynomials to multivariable big and little q-Jacobi polynomials, cf. (3.3.13) respectively (3.4.15), are linked to limits of the one parameter family of quantum Grassmannians.

4.2. The classical complex Grassmannian

Two general references for the contents of this section are Helgason [41], and Heckman and Schlichtkrull [39].

Throughout this paper, $n \ge 2$ and $1 \le l \le [\frac{n}{2}]$ are fixed integers. Let $G := GL(n, \mathbb{C})$ denote the general linear group with Lie algebra $\mathfrak{g} = \mathfrak{gl}(n, \mathbb{C})$, and U := U(n) the unitary group with Lie algebra u. Let $T \subset U$ denote the maximal torus consisting of diagonal matrices in U. Write $\mathfrak{h} \subset \mathfrak{g}$ for the corresponding Cartan subalgebra. Let e_{ij} $(1 \le i, j \le n)$ denote the standard matrix units. The matrices $h_i := e_{ii}$ $(1 \le i \le n)$ form a basis of \mathfrak{h} . Write $\tilde{\varepsilon}_i \in \mathfrak{h}^*$ $(1 \le i \le n)$ for the corresponding dual basis vectors and define a non degenerate symmetric bilinear form on \mathfrak{h}^* by $\langle \tilde{\varepsilon}_i, \tilde{\varepsilon}_j \rangle = \delta_{i,j}$. The usual positive system R^+ in the root system $R := R(\mathfrak{g}, \mathfrak{h})$ consists of the vectors $\tilde{\varepsilon}_i - \tilde{\varepsilon}_j$ $(1 \le i < j \le n)$. Let $P = P_n := \bigoplus_{1 \le i \le n} \mathbb{Z} \tilde{\varepsilon}_i$ denote the rational character lattice of G (equivalently, the lattice of analytically integral weights of U). Recall that the cone of

4.2. THE CLASSICAL COMPLEX GRASSMANNIAN

dominant weights $P^+ = P_n^+$ is given by

(4.2.1)
$$P_n^+ := \{ (\lambda_1, \dots, \lambda_n) \in P \mid \lambda_1 \ge \lambda_2 \ge \dots \ge \lambda_n \}$$

Denote by \leq the (partial) dominance ordering on P. One has $\mu \leq \lambda$ if and only if

(4.2.2)
$$\sum_{i=1}^{j} \mu_i \leq \sum_{i=1}^{j} \lambda_i \quad (1 \leq j \leq n-1) \quad \text{and} \quad \sum_{i=1}^{n} \mu_i = \sum_{i=1}^{n} \lambda_i.$$

We write $K := U(n-l) \times U(l)$ and $\mathfrak{k} := \mathfrak{gl}(n-l, \mathbb{C}) \oplus \mathfrak{gl}(l, \mathbb{C})$ for the corresponding complexified Lie algebra \mathfrak{k} . K is regarded as a subgroup of U via the embedding

(4.2.3)
$$U(n-l) \times U(l) \hookrightarrow U(n), \quad (A,B) \mapsto \begin{pmatrix} A & 0 \\ 0 & B \end{pmatrix}$$

The pair (U, K) is symmetric. Indeed, the involutive Lie group automorphism θ : $U \to U$, defined by $\theta(g) := JgJ$ with

(4.2.4)
$$J := \sum_{1 \le k \le n-l} e_{kk} - \sum_{1 \le k \le l} e_{k'k'}, \quad (k' := n+1-k)$$

has fixed point group K. The differential of θ at the unit element $e \in U$, extended \mathbb{C} linearly to a Lie-algebra involution $\mathfrak{g} \to \mathfrak{g}$, will also be denoted by θ . The +1-eigenspace of the involution $\theta: \mathfrak{g} \to \mathfrak{g}$ is exactly the Lie-subalgebra \mathfrak{k} . Write \mathfrak{p} for the -1-eigenspace of θ , then we have the eigenspace decomposition $\mathfrak{g} = \mathfrak{k} \oplus \mathfrak{p}$.

For certain purposes, it is more convenient to consider the involution $\theta' \colon \mathfrak{g} \to \mathfrak{g}$ defined by $\theta'(X) := J'XJ'$ with

(4.2.5)
$$J' := \sum_{l < k < l'} e_{kk} - \sum_{1 \le k \le l} e_{kk'} - \sum_{1 \le k \le l} e_{k'k}.$$

Since J' is conjugate to J, the involution θ' is conjugate to θ by an inner automorphism of \mathfrak{g} . Observe that both θ and θ' leave $\mathfrak{u} \subset \mathfrak{g}$ invariant.

Let $\mathfrak{g} = \mathfrak{k}' \oplus \mathfrak{p}'$ be the eigenspace decomposition of θ' in +1 and -1 eigenspaces. The intersection $\mathfrak{h} \cap \mathfrak{k}'$ is spanned by the elements $h_i + h_{i'}$ $(1 \le i \le l)$ and h_i (l < i < l'), whereas the intersection $\mathfrak{a} := \mathfrak{h} \cap \mathfrak{p}'$ is spanned by $h_i - h_{i'}$ $(1 \le i \le l)$ and is maximal abelian in \mathfrak{p}' . The positive system of R with respect to the lexicographic ordering of $\mathfrak{h}_{\mathbb{R}} := \sum_{j=1}^{n} \mathbb{R}h_j$ relative to the ordered basis $h_1 - h_{1'}, \ldots, h_l - h_{l'}, h_1 + h_{1'}, \ldots, h_l + h_{l'}, h_{l+1}, \ldots, h_{n-l}$ coincides with R^+ .

Write ε'_i for the restriction of $\tilde{\varepsilon}_i$ to a $(1 \le i \le l)$. The root system $R \subset \mathfrak{h}^*$ is mapped under the natural projection $\mathfrak{h}^* \twoheadrightarrow \mathfrak{a}^*$ onto the restricted root system $\Sigma' = \Sigma'(\mathfrak{g}, \mathfrak{a})$. Choose the positive system in Σ' with respect to the lexicographic ordering of $\mathfrak{a}_{\mathbb{R}} :=$ $\mathfrak{h}_{\mathbb{R}} \cap \mathfrak{a}$ relative to the ordered basis $h_1 - h_{1'}, \ldots, h_l - h_{l'}$ of $\mathfrak{a}_{\mathbb{R}}$. This ordering is compatible with the lexicographic ordering of $\mathfrak{h}_{\mathbb{R}}$ introduced above in the sense that $\lambda \in \mathfrak{h}_{\mathbb{R}}^*$ is positive if its restriction to $\mathfrak{a}_{\mathbb{R}}$ is strict positive. The positive root vectors in Σ' are

$$\varepsilon'_i \quad (1 \le i \le l), \quad \varepsilon'_i \pm \varepsilon'_j \quad (1 \le i < j \le l), \quad 2\varepsilon'_i \quad (1 \le i \le l),$$

4. HARMONIC ANALYSIS ON QUANTUM GRASSMANNIANS

the roots ε'_i $(1 \le i \le l)$ occurring only if $n \ne 2l$. Σ' is isomorphic with BC_l if $n \ne 2l$ and isomorphic with C_l if n = 2l. The root multiplicities corresponding to the short, medium, and long roots are

$$(4.2.6) mtextbf{m}_1 = 2(n-2l), mtextbf{m}_2 = 2 (l > 1), mtextbf{m}_3 = 1.$$

For later purposes, it is convenient to rescale the root system Σ' by a factor 2. So we set $\Sigma := 2\Sigma' \subset \mathfrak{a}^*, \varepsilon_i := 2\varepsilon'_i \ (1 \le i \le l)$. Then the corresponding weight lattice $P_{\Sigma} \subset \mathfrak{a}^*$ is the \mathbb{Z} -span of the $\varepsilon_i \ (1 \le i \le l)$, and the set P_{Σ}^+ of dominant weights $\mu = \sum_i \mu_i \varepsilon_i$ (taken with respect to the lexicographic ordering on $\mathfrak{a}_{\mathbb{R}}$ introduced above) is characterized by the condition $\mu_1 \ge \cdots \ge \mu_l \ge 0$. The dominance ordering $\le on P_{\Sigma}$ is explicitly given by

(4.2.7)
$$\mu \le \lambda \iff \sum_{i=1}^{j} \mu_i \le \sum_{i=1}^{j} \lambda_i \quad (1 \le j \le l).$$

Let $K' \subset U$ denote the connected subgroup corresponding to \mathfrak{k}' . The symmetric pairs (U, K) and (U, K') are Gelfand pairs, i.e. every finite dimensional irreducible representation of U has at most one K-fixed vector up to scalar multiples. According to [41, Chapter V, Theorem 4.1], a highest weight $\lambda \in P^+$ is K'-spherical, i.e. corresponds to a representation with a non-zero K'-fixed vector, if and only if the restriction of λ to $\mathfrak{h} \cap \mathfrak{k}'$ is zero and the restriction of λ to a lies in P_{Σ}^+ . Hence we get the following result.

THEOREM 4.2.1. The set $P_K^+ \subset P^+$ of K-spherical dominant weights consists of all dominant weights of the form

$$\lambda := (\lambda_1, \ldots, \lambda_l, 0, \ldots, 0, -\lambda_l, \ldots, -\lambda_1).$$

Write $\lambda^{\natural} := (\lambda_1, \ldots, \lambda_l)$ for a dominant weight $\lambda \in P_K^+$. The assignment $\lambda \mapsto \lambda^{\natural}$ defines a bijection of P_K^+ onto P_{Σ}^+ . Let $\varpi_r \in P_K^+$ be the spherical weight for which $\varpi_r^{\natural} = (1^r)$. Then $P_K^+ = \bigoplus_{1 \le r \le l} \mathbb{Z}_+ \varpi_r$. We will call $\{\varpi_r\}_{r=1}^l$ the fundamental dominant spherical weights.

Let A denote the algebra of polynomial functions on U, A(T) the algebra of polynomial functions on the maximal torus T. A(T) may be naturally identified with the algebra $\mathbb{C}[z^{\pm 1}] = \mathbb{C}[z_1^{\pm 1}, \ldots, z_n^{\pm 1}]$ in n variables z_i $(1 \le i \le n)$ in the following way. Observe that $T \simeq i\mathfrak{h}_{\mathbb{R}}/2\pi i\tilde{P}$ using the exponential mapping, where $\tilde{P} := \bigoplus_{1 \le j \le n} \mathbb{Z}h_j$. Then the coordinate functions z_j can be defined by $z_j := e^{\tilde{\varepsilon}_j}$, where for $[X] \in T$ with $X \in i\mathfrak{h}_{\mathbb{R}}$ a representative of [X], $e^{\tilde{\varepsilon}_j}([X]) := e^{\tilde{\varepsilon}_j(X)}$. More explicitly, z_j is given by

$$z_j : \operatorname{diag}(e^{i\theta_1}, e^{i\theta_2}, \dots, e^{i\theta_n}) \mapsto e^{i\theta_j}, \quad (\theta_k \in [0, 2\pi))$$

where $diag(a_1, \ldots, a_n)$ is the diagonal matrix with a_1, \ldots, a_n on the diagonal.

Let $\mathcal{H} \subset A$ denote the subalgebra of K'-biinvariant functions. One has the decomposition

(4.2.8)
$$\mathcal{H} = \bigoplus_{\lambda \in P_K^+} \mathcal{H}(\lambda), \quad \mathcal{H}(\lambda) := \mathcal{H} \cap W(\lambda),$$

 $W(\lambda) \subset A$ denoting the subspace spanned by the matrix coefficients of the irreducible representation of highest weight λ . Each of the subspaces $\mathcal{H}(\lambda)$ ($\lambda \in P_K^+$) is one dimensional, since (U, K') is a Gelfand pair. Any non-zero element $\varphi(\lambda)$ of $\mathcal{H}(\lambda)$ is called a zonal spherical function.

Set

(4.2.9)
$$T_l := \exp(i\mathfrak{a}_{\mathbb{R}})/(\exp(i\mathfrak{a}_{\mathbb{R}}) \cap K') \simeq i\mathfrak{a}_{\mathbb{R}}/2\pi i Q_{\Sigma}^{\vee},$$

with $Q_{\Sigma}^{\vee} \subset \mathfrak{a}_{\mathbb{R}}$ the coroot lattice of Σ . More explicitly, the coroot lattice Q_{Σ}^{\vee} is the \mathbb{Z} -span of the elements $\frac{1}{2}(h_i - h_{i'})$ $(i \in [1, l])$, $\exp(i\mathfrak{a}_{\mathbb{R}})$ are the diagonal matrices

diag
$$(e^{i\theta_1}, \dots, e^{i\theta_l}, 1, \dots, 1, e^{-i\theta_l}, \dots, e^{-i\theta_1}) \quad (\theta_i \in [0, 2\pi)),$$

and $\exp(i\mathfrak{a}_{\mathbb{R}}) \cap K'$ are the matrices in $\exp(i\mathfrak{a}_{\mathbb{R}})$ of order 2.

Write $\log : T_l \to i\mathfrak{a}_{\mathbb{R}}$ for the multi-valued inverse of the exponential map $\exp : i\mathfrak{a}_{\mathbb{R}} \to T_l$. Similarly as for A(T), the algebra of polynomial functions on T_l may be identified with the algebra $\mathbb{C}[x^{\pm 1}]$ of Laurent polynomials in the l variables x_j $(1 \le j \le l)$, where the identification is given by $x_j(t) := e^{\varepsilon_j (\log(t))}$. In other words, the map x_j is given by

$$x_i: \operatorname{diag}(e^{i\theta_1}, \ldots, e^{i\theta_l}, 1, \ldots, 1, e^{-i\theta_l}, \ldots, e^{-i\theta_1}) \mapsto e^{2i\theta_j}.$$

It follows that the algebra $\mathbb{C}[x^{\pm 1}]$ of polynomial functions on T_l can be naturally embedded in the algebra $\mathbb{C}[z^{\pm 1}]$ of polynomial functions on the maximal torus T by the assignment

$$(4.2.10) x_1 = z_1 z_n^{-1}, x_2 = z_2 z_{n-1}^{-1}, \dots, x_l = z_l z_{n+1-l}^{-1}.$$

Let $\mathfrak{S} := \mathfrak{S}_l$ denote the permutation group on l letters, $\mathcal{W} := \mathcal{W}_l = \mathbb{Z}_2^l \rtimes \mathfrak{S}_l$ the Weyl group of Σ . The natural action of \mathcal{W} on $\mathfrak{a}_{\mathbb{R}}$ descends to T_l . Hence \mathcal{W} acts naturally on the algebra $\mathbb{C}[x^{\pm 1}]$. Write $\mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$ for the subalgebra of \mathcal{W} -invariant Laurent polynomials. By Chevalley's restriction theorem and the above metioned natural embedding of $\mathbb{C}[x^{\pm 1}]$ into $\mathbb{C}[z^{\pm 1}]$, we have the following theorem.

THEOREM 4.2.2. Restriction to T induces an isomorphism of \mathcal{H} onto the algebra $\mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$ of \mathcal{W} -invariant Laurent polynomials in the variables x_i $(1 \le i \le l)$.

Recall from the previous chapter that the *BC* type Heckman-Opdam polynomials $\{P_{\lambda}^{HO}(.;k)\}_{\lambda \in P_{\Sigma}^{\pm}}$ for $k = (k_1, k_2, k_3) \in V_{HO}$ form a linear basis of $\mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$, and that they are orthogonal with respect to the inner product

(4.2.11)
$$\langle P, Q \rangle_{HO} := \int_{T_l} P(t) \overline{Q(t)} \Delta_{HO}(t; k) dt.$$

Here dt denotes the normalized Haar measure on the torus T_l and $t \mapsto \Delta_{HO}(t;k)$ on T_l is defined by

(4.2.12)
$$\Delta_{HO}(t;k) := \prod_{\alpha \in \Sigma} \left(e^{\frac{1}{2} \langle \alpha, \log(t) \rangle} - e^{-\frac{1}{2} \langle \alpha, \log(t) \rangle} \right)^{k_{\alpha}}.$$

The multiplicity parameters k_{α} are by definition equal to k_i (i = 1, 2, 3), depending on whether $\alpha \in \Sigma$ is a short, medium, or long root. If l = 1 there is no dependence on k_2 .

In the following theorem the zonal spherical functions on the symmetric space U/K are identified with BC type Heckman-Opdam polynomials (cf. [39, p. 76]).

THEOREM 4.2.3. Under restriction to the maximal torus T, the zonal K'-spherical function $\varphi(\lambda)$ ($\lambda \in P_K^+$) is mapped onto (a scalar multiple of) the Heckman-Opdam hypergeometric polynomial $P_{\lambda^{\rm h}}^{HO}(x;k)$ with $k_i = \frac{1}{2}m_i$ (i = 1, 2, 3).

Of course, the zonal K-spherical functions can be described in the same way, since the subgroups K and K' are conjugate.

As was shown in the previous chapter, the *BC* type Heckman-Opdam polynomials and the generalized Jacobi polynomials $\{P_{\lambda}^{J}(.; \alpha, \beta; \tau) \mid \lambda \in P_{\Sigma}^{+}\}$ are related via a simple change of variables,

(4.2.13)
$$P_{\lambda}^{HO}(x;k) = (-4)^{|\lambda|} P_{\lambda}^{J}(z(x);k_1+k_3-1/2,k_3-1/2,k_2)$$

where $z_j(x) := -\frac{1}{4}(x_j + x_j^{-1} - 2)$. In particular, the zonal spherical functions in Theorem 4.2.3 under this change of variables become generalized Jacobi polynomials with parameter values $\alpha = n - 2l$, $\beta = 0$ and $\tau = 1$.

Being matrix coefficients of irreducible representations, the zonal spherical functions are mutually orthogonal with respect to the L^2 inner product on $A \subset L^2(U, dg)$, where dg is the normalized Haar measure on U (Schur orthogonality). The restriction of the L^2 inner product to the algebra \mathcal{H} of bi-K'-invariant matrix coefficients coincides under the isomorphism of Theorem 4.2.2 with the inner product $\langle ., . \rangle_{HO}$ on $\mathbb{C}[x^{\pm}]^{\mathcal{W}}$ up to a non-zero positive constant. This non-zero positive constant can be explicitly determined using the well-known evaluation of the Selberg integral (cf. [107], [84]).

4.3. Preliminaries on the quantum unitary group

Various aspects of the quantum unitary group have been studied in many different papers. Our main references will be [96] and [91, Section 1], which are based on the R-matrix approach described in [104].

The quantized coordinate ring $A_q(Mat(n, \mathbb{C}))$ of the space of $n \times n$ complex matrices is defined as the algebra with generators t_{ij} $(1 \le i, j \le n)$ and relations

(4.3.1)
$$\begin{aligned} t_{ki}t_{kj} &= qt_{kj}t_{ki}, \quad t_{ik}t_{jk} = qt_{jk}t_{ik} \quad (i < j), \\ t_{il}t_{kj} &= t_{kj}t_{il}, \quad t_{ij}t_{kl} - t_{kl}t_{ij} = (q - q^{-1})t_{il}t_{kj} \quad (i < k, j < l). \end{aligned}$$

These equations can be written in more compact notation using the matrix

(4.3.2)
$$R := \sum_{ij} q^{\delta_{ij}} e_{ii} \otimes e_{jj} + (q - q^{-1}) \sum_{i>j} e_{ij} \otimes e_{ji}$$

where e_{ij} $(1 \le i, j \le n)$ denote the standard matrix units. Let V denote the vector space \mathbb{C}^n with canonical basis $\{v_i\}_i$ and regard e_{ij} as matrix units in End(V) with respect to
the canonical basis $\{v_i\}_i$. The relations (4.3.1) are then given by the matrix equation $(R \otimes 1)T_1T_2 = T_2T_1(R \otimes 1)$, where

$$T_1:=\sum_{i,j}e_{ij}\otimes \mathrm{id}_V\otimes t_{ij},\quad T_2:=\sum_{i,j}\mathrm{id}_V\otimes e_{ij}\otimes t_{ij}.$$

We will use the shorthand notation $RT_1T_2 = T_2T_1R$ for this matrix equation. Observe furthermore that the matrix R is invertible, with inverse given by

$$R^{-1} := \sum_{ij} q^{-\delta_{ij}} e_{ii} \otimes e_{jj} - (q - q^{-1}) \sum_{i>j} e_{ij} \otimes e_{ji}$$

and that R is a solution of the Quantum Yang-Baxter Equation

$$(4.3.3) R_{12}R_{13}R_{23} = R_{23}R_{13}R_{12},$$

where $R_{12} = R \otimes id_V$, $R_{23} = id_V \otimes R$ and

$$R_{13} = \sum_{i*} r_i \otimes \mathrm{id}_V \otimes r'_i,$$

if $R = \sum_i r_i \otimes r'_i$.

A linear basis for $A_q(Mat(n, \mathbb{C}))$ (cf. [96, Theorem 1.4], see also [60]) is given by the monomials

(4.3.4)
$$t^{A} := \prod_{(i,j)} t_{ij}^{a_{ij}} = t_{11}^{a_{11}} \cdots t_{1n}^{a_{1n}} t_{21}^{a_{21}} \cdots t_{nn}^{a_{nn}},$$

where $A = (a_{ij})$ runs through the set $\operatorname{Mat}(n, \mathbb{Z}_+)$ of $n \times n$ matrices with non-negative integer coefficients. We call $\{t_A \mid A \in \operatorname{Mat}(n, \mathbb{Z}_+)\}$ the monomial basis of $A_q(\operatorname{Mat}(n, \mathbb{C}))$. A total order \preceq on $\operatorname{Mat}(n; \mathbb{Z}_+)$ is defined by associating with $A \in \operatorname{Mat}(n, \mathbb{Z}_+)$ the sequence

(4.3.5)
$$\left(\sum_{i,j} a_{ij}, a_{11}, \dots, a_{1n}, a_{21}, \dots, a_{nn}\right) \in \mathbb{Z}_+^{n^2+1},$$

and ordering these sequences lexicographically. Every $0 \neq \phi \in A_q(Mat(n, \mathbb{C}))$ can be written uniquely as

(4.3.6)
$$\phi = c_{\phi} t^{d(\phi)} + \text{ lower order terms w.r.t. } \preceq$$

with $d(\phi) \in Mat(n, \mathbb{Z}_+)$ and $c_{\phi} \neq 0$. We call $c_{\phi}t^{d(\phi)}$ the leading term of ϕ , and $d(\phi)$ its degree. One easily sees that

(4.3.7)
$$d(\phi\psi) = d(\phi) + d(\psi), \quad 0 \neq \phi, \psi \in A_q(\operatorname{Mat}(n, \mathbb{C}))$$

(cf. [96, Lemma 1.5]). In particular, there are no zero divisors in $A_q(Mat(n; \mathbb{C}))$.

The quantized coordinate ring $A_q(G)$ of the general linear group $G = GL(n, \mathbb{C})$ is defined by adjoining to $A_q(\operatorname{Mat}(n, \mathbb{C}))$ the inverse $\operatorname{det}_q^{-1}$ of the quantum determinant

$$\det_q := \sum_{\sigma \in \mathfrak{S}_n} (-q)^{l(\sigma)} t_{1\sigma(1)} \cdots t_{n\sigma(n)} \in A_q(\operatorname{Mat}_q(n, \mathbb{C}))$$

4. HARMONIC ANALYSIS ON QUANTUM GRASSMANNIANS

(*l* denoting the length function on \mathfrak{S}), which is central in $A_q(\operatorname{Mat}(n, \mathbb{C}))$. The monomial basis of $A_q(\operatorname{Mat}(n, \mathbb{C}))$ can be naturally extended to a basis of $A_q(G)$, which we shall also call the monomial basis of $A_q(G)$.

There is a unique Hopf algebra structure on $A_q(G)$ such that (t_{ij}) becomes a matrix corepresentation, i.e.

$$\Delta(t_{ij}) = \sum_{k=1}^{n} t_{ik} \otimes t_{kj}, \quad \varepsilon(t_{ij}) = \delta_{ij}.$$

The antipode $S: A_q(G) \to A_q(G)$ is given on the generators by

$$S(t_{ij}) := (-q)^{i-j} \xi_{i^c}^{j^c} \det_q^{-1}$$

with $i^c := [1, n] \setminus \{i\}$, and with the quantum minor ξ_J^I for subsets $I = \{i_1 < \ldots < i_r\}, J = \{j_1 < \ldots < j_r\} \subset [1, n]$ defined by

$$\xi_J^I := \sum_{\sigma \in \mathfrak{S}_r} (-q)^{l(\sigma)} t_{i_1 j_{\sigma(1)}} \cdots t_{i_r j_{\sigma(r)}}.$$

It is easy to see that the ξ_J^I are all non-zero. $A_q(G)$ becomes a Hopf *-algebra by requiring (t_{ij}) to be a unitary matrix corepresentation, i.e. $t_{ij}^* := S(t_{ji})$. We write $A_q(U) = A_q(U(n))$ for $A_q(G)$ endowed with this *-operation. The mapping $\tau := * \circ S$ is a conjugate linear involution on $A_q(U)$ such that $\tau(t_{ij}) = t_{ji}$.

The quantized Borel subgroups $A_q(B^{\pm})$ of upper respectively lower triangular matrices are defined as the Hopf quotients of $A_q(G)$ by the relations

 $t_{ij} = 0 \ (i > j)$ respectively $t_{ij} = 0 \ (i < j)$.

The corresponding projections will be denoted by $\pi_{\pm} \colon A_q(G) \to A_q(B^{\pm})$. Observe that the $z_i := \pi_{\pm}(t_{ii})$ $(1 \le i \le n)$ in $A_q(B^{\pm})$ are invertible. Corresponding to the diagonal subgroup $T \subset U(n)$ we have a natural surjective Hopf *-algebra morphism |T| of $A_q(U)$ onto the Laurent polynomial algebra $A(T) := \mathbb{C}[z_1^{\pm 1}, \ldots, z_n^{\pm 1}]$.

Next, we briefly recall the global description of $A_q(D)$. For every $\lambda \in P$, $z^{\lambda} := z_1^{\lambda_1} \dots z_n^{\lambda_n}$ defines a linear character (i.e. one dimensional corepresentation) of $A_q(B^{\pm})$ and of A(T). Using these linear characters it is straightforward to define (highest) weight vectors in left or right $A_q(U)$ -comodules. We take highest weight vectors of right and left $A_q(U)$ -comodules with respect to $A_q(B^+)$ and $A_q(B^-)$, respectively. For instance, a highest weight vector of weight $\lambda \in P^+$ in a right $A_q(U)$ -comodule M is a non-zero vector $v \in M$ such that

$$(\mathrm{id}\otimes\pi_+)\circ\rho_M(v)=v\otimes z^\lambda,$$

 $\rho_M \colon M \to M \otimes A_q(U)$ denoting the comodule mapping. Finite dimensional $A_q(U)$ comodules are then completely reducible and unitarizable (see, for instance, [96], [23]).
Recall that a right $A_q(U)$ -comodule M endowed with a positive definite inner product

(taken to be conjugate linear in the second variable) is called unitary if

$$\sum_{(v)(w)} \langle v_{(1)}, w_{(1)} \rangle w_{(2)}^* v_{(2)} = \langle v, w \rangle 1 \quad (v, w \in M),$$

where the symbolic notation $\sum_{(v)} v_{(1)} \otimes v_{(2)} := \rho_M(v)$ is used. The irreducible finite dimensional $A_q(U)$ -comodules are parametrized by dominant weights $\lambda \in P^+$ as in the classical case (cf. [96, Theorem 2.12]). The irreducible right $A_q(U)$ -comodule with highest weight $\lambda \in P^+$ is denoted by $V_R(\lambda)$. The vector space $V_L(\lambda) := \text{Hom}(V_R(\lambda), \mathbb{C})$ has a natural left $A_q(U)$ -comodule structure, which is also irreducible of highest weight λ . If no confusion is possible, we will write $V(\lambda)$ for the left comodule $V_L(\lambda)$ respectively for the right comodule $V_R(\lambda)$.

REMARK 4.3.1. Let M be a finite dimensional right $A_q(U)$ -comodule with comodule mapping $R: M \to M \otimes A_q(U)$. Write M° for the vector space complex conjugate to $M, \sigma: M \otimes A_q(U) \to A_q(U) \otimes M$ for the flip. Then the mapping

$$(4.3.8) R^{\circ}: M^{\circ} \to A_q(U) \otimes M^{\circ}, \quad R^{\circ}:= (\tau \otimes \mathrm{id}) \circ \sigma \circ R,$$

where $\tau = * \circ S$, defines a left $A_q(U)$ -comodule structure on M° . In (4.3.8) R is considered as a conjugate linear map from M° to $M \otimes A_q(U)$ and $\tau \otimes$ id as a conjugate linear map from $A_q(U) \otimes M$ to $A_q(U) \otimes M^\circ$.

The assignment $M \mapsto M^{\circ}$ is a 1-1 correspondence between right and left $A_q(U)$ comodules preserving weights and highest weights. Hence M° is isomorphic to the left $A_q(U)$ -comodule Hom (M, \mathbb{C}) . A right $A_q(U)$ -comodule intertwiner $\Psi : M \to N$ also
intertwines the left $A_q(U)$ -comodule structures of M° and N° (i.e. when Ψ is considered
as map from M° to N°).

Recall that the comultiplication $\Delta \colon A_q(U) \to A_q(U) \otimes A_q(U)$ defines a bicomodule structure on $A_q(U)$. Let $W(\lambda) \subset A_q(U)$ ($\lambda \in P^+$) denote the subspace spanned by the matrix coefficients of either $V_R(\lambda)$ or $V_L(\lambda)$. The irreducible decomposition of the bicomodule $A_q(U)$ reads

(4.3.9)
$$A_q(U) = \bigoplus_{\lambda \in P^+} W(\lambda), \quad W(\lambda) \simeq V_L(\lambda) \otimes V_R(\lambda).$$

Let h be the normalized Haar functional on $A_q(U)$. It can be characterized as the unique linear functional on $A_q(U)$ which is zero on $W(\lambda)$ for $0 \neq \lambda \in P^+$ and which sends $1 \in A_q(U)$ to $1 \in \mathbb{C}$. The subspaces $W(\lambda)$ are mutually orthogonal with respect to the inner product $\langle \varphi, \psi \rangle := h(\psi^* \varphi)$.

We consider now in some more detail the vector corepresentation, its dual representation, and their exterior powers. Recall the notation $\{v_i\}_i$ for the canonical basis of $V = \mathbb{C}^n$. V becomes a right $A_q(U)$ -comodule (called vector corepresentation) with

(4.3.10)
$$R: V \mapsto V \otimes A_q(U), \quad R(v_j) := \sum_{i=1}^n v_i \otimes t_{ij} \quad (1 \le j \le n).$$

V is irreducible with highest weight $\tilde{\varepsilon}_1$ and highest weight vector v_1 . Observe that the vectors v_i have weight $\tilde{\varepsilon}_i$. The corepresentation V is unitary with respect to the inner product $\langle v_i, v_j \rangle = \delta_{ij}$.

Let V^* denote the linear dual of V with dual basis (v_i^*) . V^* becomes a right $A_q(U)$ comodule (the dual corepresentation) with

(4.3.11)
$$R: V^* \mapsto V^* \otimes A_q(U), \quad R(v_j^*) := \sum_{i=1}^n v_i^* \otimes t_{ij}^* \quad (1 \le j \le n).$$

 V^* is irreducible with highest weight $-\tilde{\varepsilon}_n$ and highest weight vector v_n^* . Note that the vectors v_i^* have weight $-\tilde{\varepsilon}_i$. The corepresentation V^* is unitary with respect to the inner product $\langle v_i^*, v_j^* \rangle := q^{-\langle 2\rho, \tilde{\varepsilon}_i \rangle} \delta_{ij}$, where $\rho := \sum_{k=1}^n (n-k)\tilde{\varepsilon}_k$. This follows from the well-known fact that $S^2(t_{ij}) = q^{\langle 2\rho, \tilde{\varepsilon}_j - \tilde{\varepsilon}_i \rangle} t_{ij}$ $(1 \le i, j \le n)$.

Let $\Lambda_q(V)$ respectively $\Lambda_q(V^*)$ denote the associative algebra which is generated by v_1, \ldots, v_n respectively v_1^*, \ldots, v_n^* with relations

$$(4.3.12) v_i \wedge v_i = 0 \ (1 \le i \le n), \quad v_i \wedge v_j = -q^{-1}v_j \wedge v_i \ (i < j)$$

respectively

(4.3.13)
$$v_i^* \wedge v_i^* = 0 \ (1 \le i \le n), \quad v_j^* \wedge v_i^* = -q^{-1}v_i^* \wedge v_j^* \ (i < j).$$

Then $\Lambda_q(V)$ respectively $\Lambda_q(V^*)$ inherits a natural right $A_q(U)$ -comodule structure from V respectively V^* by extending the comodule mapping R as a unital algebra homomorphism. $\Lambda_q(V)$ respectively $\Lambda_q(V^*)$ has a natural grading such that the generators v_i respectively v_i^* have degree 1:

$$\Lambda_q(V) = \bigoplus_{r=0}^n \Lambda_q^r(V), \quad \Lambda_q(V^*) = \bigoplus_{r=0}^n \Lambda_q^r(V^*).$$

Write $v_I := v_{i_1} \wedge \cdots \wedge v_{i_r}$ and $v_I^* := v_{i_r}^* \wedge \cdots \wedge v_{i_1}^*$ if $I = \{i_1 < \cdots < i_r\} \subset [1, n]$. Write |I| for the cardinality of I. Then the v_I and v_I^* (|I| = r) form a basis of $\Lambda_q^r(V)$ and $\Lambda_q^r(V^*)$ respectively. One has the multiplicative property $(I, J \subset [1, n])$

$$v_I \wedge v_J = \mathbf{s}_{\mathbf{q}}(I;J)v_{I\cup J}, \quad v_I^* \wedge v_J^* := \mathbf{s}_{\mathbf{q}}(J;I)v_{I\cup J}^*$$

where

$$\mathbf{s}_{\mathbf{q}}(I;J) := \begin{cases} 0 & \text{if } I \cap J \neq \emptyset, \\ (-q)^{l(I;J)} & \text{if } I \cap J = \emptyset, \end{cases}$$

and $l(I; J) := |\{(i, j) \in I \times J | i > j\}|$. The comodules $\Lambda_q^r(V)$ and $\Lambda_q^r(V^*)$ are irreducible subcomodules of $\Lambda_q(V)$, and the coactions satisfy

(4.3.14)
$$R(v_J) = \sum_{|I|=r} v_I \otimes \xi_J^I, \quad R(v_J^*) = \sum_{|I|=r} v_I^* \otimes (\xi_J^I)^* \quad (|J|=r).$$

From this it follows immediately that

(4.3.15)
$$\Delta(\xi_J^I) = \sum_{|K|=r} \xi_K^I \otimes \xi_J^K \quad (|I|, |J|=r)$$

and hence

(4.3.16)
$$\sum_{|K|=r} \xi_K^I S(\xi_J^K) = \delta_{I,J} \quad (|I|, |J|=r).$$

The $A_q(U)$ -comodule $\Lambda_q^r(V)$ respectively $\Lambda_q^r(V^*)$ has highest weight $(1^r) := \tilde{\varepsilon}_1 + \cdots + \tilde{\varepsilon}_r$ respectively $-\tilde{\varepsilon}_{n-r+1} - \cdots - \tilde{\varepsilon}_n$ with highest weight vector $v_{[1,r]} = v_1 \wedge \cdots \wedge v_r$ respectively $v_{[n-r+1,n]}^* = v_n^* \wedge \cdots \wedge v_{n-r+1}^*$. The inner product on $\Lambda_q^r(V)$ defined by $\langle v_I, v_J \rangle = \delta_{I,J}$ is $A_q(U)$ -invariant. On the space $\Lambda_q^r(V^*)$ we have the invariant inner product $\langle v_I^*, v_J^* \rangle := \delta_{I,J} q^{-\langle 2\rho, \tilde{\varepsilon}_I \rangle}$, where $\tilde{\varepsilon}_I := \sum_{i \in I} \tilde{\varepsilon}_i$.

The Laplace expansions for quantum minor determinants now easily follow from the fact that $\Lambda_q(V)$ is an $A_q(U)$ -comodule algebra (cf. [96, Proposition 1.1]). Fix $I, J \subset [1, n]$ such that |I| = |J| > 0. Fix two subsets $J_1, J_2 \subset J$ such that $|J_1| + |J_2| = |J|$. Then

(4.3.17)
$$s_{q}(J_{1};J_{2})\xi_{J}^{I} = \sum_{\substack{I_{1},I_{2} \subset I; \\ |I_{1}| = |J_{1}|, |I_{2}| = |J_{2}|}} s_{q}(I_{1};I_{2})\xi_{J_{1}}^{I_{1}}\xi_{J_{2}}^{I_{2}},$$

(4.3.18)
$$s_{q}(J_{1};J_{2})\xi_{I}^{J} = \sum_{\substack{I_{1},I_{2} \subset I;\\|I_{1}|=|J_{1}|,|I_{2}|=|J_{2}|}} s_{q}(I_{1};I_{2})\xi_{I_{1}}^{J_{1}}\xi_{I_{2}}^{J_{2}}.$$

It can now be shown that (cf. [96, (3.2)])

(4.3.19)
$$(\xi_J^I)^* = S(\xi_I^J) = \frac{\mathbf{s}_q(J;J^c)}{\mathbf{s}_q(I;I^c)} \xi_{J^c}^{I^c} \det_q^{-1} \quad (|I| = |J| = r),$$

where $I^c := \{1, \ldots, n\} \setminus I$.

For the following results on the quantized universal enveloping algebra associated with $\mathfrak{gl}(n, \mathbb{C})$ we refer the reader to Noumi's paper [91, Section 1] and references therein. Let $U_q(\mathfrak{g}) = U_q(\mathfrak{gl}(n, \mathbb{C}))$ denote the quantized universal enveloping algebra (cf. Drinfeld [27], Jimbo [47]) associated with the Lie algebra $\mathfrak{g} = \mathfrak{gl}(n, \mathbb{C})$. The algebra $U_q(\mathfrak{g})$ is generated by elements q^h $(h \in P)$ and X_i^{\pm} $(1 \le i \le n-1)$ subject to the relations

$$q^{0} = 1, \quad q^{h}q^{h'} = q^{h+h'}$$

$$q^{h}X_{i}^{\pm}q^{-h} = q^{\pm \langle h, \alpha_{i} \rangle}X_{i}^{\pm}$$

$$(4.3.20) \qquad [X_{i}^{+}, X_{j}^{-}] = \delta_{i,j}\frac{q^{\alpha_{i}} - q^{-\alpha_{i}}}{q - q^{-1}}$$

$$(X_{i}^{\pm})^{2}X_{j}^{\pm} - (q + q^{-1})X_{i}^{\pm}X_{j}^{\pm}X_{i}^{\pm} + X_{j}^{\pm}(X_{i}^{\pm})^{2} = 0, \quad (|i - j| = 1)$$

$$X_{i}^{\pm}X_{j}^{\pm} = X_{j}^{\pm}X_{i}^{\pm}, \quad (|i - j| > 1),$$

where $\alpha_i := \tilde{\varepsilon}_i - \tilde{\varepsilon}_{i+1}$ $(i \in [1, n-1])$ are the simple roots for $R = R(\mathfrak{g}, \mathfrak{h})$. The following formulas uniquely determine a Hopf-*-algebra structure on $U_q(\mathfrak{g})$:

$$\begin{split} \Delta(X_{i}^{+}) &= X_{i}^{+} \otimes 1 + q^{\alpha_{i}} \otimes X_{i}^{+}, \ \Delta(X_{i}^{-}) = X_{i}^{-} \otimes q^{-\alpha_{i}} + 1 \otimes X_{i}^{-} \\ \Delta(q^{h}) &= q^{h} \otimes q^{h}, \\ (4.3.21) \qquad S(q^{h}) &= q^{-h}, \ S(X_{i}^{+}) = -q^{-\alpha_{i}}X_{i}^{+}, \ S(X_{i}^{-}) = -X_{i}^{-}q^{\alpha_{i}}, \\ \varepsilon(q^{h}) &= 1, \ \varepsilon(X_{i}^{\pm}) = 0, \\ (q^{h})^{*} &= q^{h}, \ (X_{i}^{+})^{*} = q^{-1}X_{i}^{-}q^{\alpha_{i}}, \ (X_{i}^{-})^{*} = qq^{-\alpha_{i}}X_{i}^{-1}. \end{split}$$

More useful for the purposes of this chapter are the *L*-operators $L_{ij}^{\pm} \in U_q(\mathfrak{g})$ $(1 \leq i, j \leq n)$, which should be thought of as quantum analogues of the root vectors for $(\mathfrak{gl}(n, \mathbb{C}), \mathfrak{h})$. The *L*-operators are constructed as follows. Define elements $E_{ij} \in U_q(\mathfrak{g})$ $(1 \leq i \neq j \leq n)$ inductively by the formulas

(4.3.22)
$$E_{i,i+1} = X_i^+, \quad E_{ik} = E_{ij}E_{jk} - qE_{jk}E_{ij}, \\ E_{i+1,i} = X_i^-, \quad E_{ki} = E_{kj}E_{ji} - q^{-1}E_{ji}E_{kj}$$

where $1 \le i < j < k \le n$. The *L*-operators L_{ij}^{\pm} are now defined by the formulas

(4.3.23)
$$L_{ii}^{+} = q^{\tilde{\varepsilon}_{i}}, \ L_{ij}^{+} = (q - q^{-1})q^{\tilde{\varepsilon}_{i}}E_{ji}, \ L_{ji}^{+} = 0, \\ L_{ii}^{-} = q^{-\tilde{\varepsilon}_{i}}, \ L_{ji}^{-} = -(q - q^{-1})E_{ij}q^{-\tilde{\varepsilon}_{i}}, \ L_{ij}^{-} = 0$$

for $1 \leq i < j \leq n$. The L_{ij}^{\pm} generate $U_q(\mathfrak{g})$ and satisfy commutation relations which can be expressed by means of the matrix R (4.3.2). Indeed, set $L^{\pm} := \sum_{i,j} e_{ij} \otimes L_{ij}^{\pm}$, where $e_{ij} \in \operatorname{End}(V)$ is regarded as the matrix units with respect to the fixed chosen basis $\{v_i\}_{i=1}^n$ of V. Write $R^+ := PRP$, where $P := \sum_{i,j} e_{ij} \otimes e_{ji}$ is the permutation operator. Then we have the commutation relations

$$R^{+}L_{1}^{\pm}L_{2}^{\pm} = L_{2}^{\pm}L_{1}^{\pm}R^{+}, \quad R^{+}L_{1}^{+}L_{2}^{-} = L_{2}^{-}L_{1}^{+}R^{+}$$

Furthermore, the action of the comultiplication, counit and *-structure on the L-operators are given by

(4.3.24)

$$\Delta(L_{ij}^{\pm}) = \sum_{k} L_{ik}^{\pm} \otimes L_{kj}^{\pm}, \quad \varepsilon(L_{ij}^{\pm}) = \delta_{ij}, \quad (L_{ij}^{\pm})^* = S(L_{ji}^{\mp}) \quad (1 \le i, j \le n).$$

The involution $\tau = * \circ S \colon U_q(\mathfrak{g}) \to U_q(\mathfrak{g})$ acts on the generators as

(4.3.25)
$$\tau(L_{ij}^{\pm}) = L_{ji}^{\mp} \quad (1 \le i, j \le n).$$

There is a natural Hopf *-algebra duality $\langle \cdot, \cdot \rangle$ between $U_q(\mathfrak{g})$ and $A_q(U)$. Writing $T = \sum_{i,j} e_{ij} \otimes t_{ij}$, the Hopf-algebra duality $\langle \cdot, \cdot \rangle$ is determined by the formulas

(4.3.26)
$$\langle L_1^{\pm}, T_2 \rangle = R^{\pm}, \quad \langle L^{\pm}, \det_q \rangle = q^{\pm 1} \operatorname{id}_V.$$

The relations satisfied by $\langle ., . \rangle$ are explicitly given by

(4.3.27)
$$\begin{array}{l} \langle XY,\phi\rangle = \langle X\otimes Y,\Delta(\phi)\rangle, \quad \langle 1,\phi\rangle = \varepsilon(\phi)\\ \langle X,\phi\psi\rangle = \langle \Delta(X),\phi\otimes\psi\rangle, \quad \langle X,1\rangle = \varepsilon(X),\\ \langle S(X),\phi\rangle = \langle X,S(\phi)\rangle,\\ \langle X,\phi^*\rangle = \overline{\langle \tau(X),\phi\rangle}, \quad \langle X^*,\phi\rangle = \overline{\langle X,\tau(\phi)\rangle}, \end{array}$$

where $X, Y \in U_q(\mathfrak{g}), \phi, \psi \in A_q(U)$ and where $\langle X \otimes Y, \phi \otimes \psi \rangle := \langle X, \phi \rangle \langle Y, \psi \rangle$.

Write $U_q(\mathfrak{h})$ for the subalgebra generated by the q^h $(h \in P)$. It is Laurent polynomial in the elements $q^{\tilde{\varepsilon}_i}$ $(1 \leq i \leq n)$, and may be considered as the (quantized) algebra of functions on the Cartan subalgebra \mathfrak{h} . There is an induced Hopf *-algebra duality between $U_q(\mathfrak{h})$ and A(T) such that

$$\langle q^h\,,\,z^\lambda\rangle:=q^{\langle h,\lambda\rangle},\quad z^\lambda=z_1^{\lambda_1}\cdots z_n^{\lambda_n}\quad (h,\lambda\in P).$$

For a right $A_q(U)$ -comodule (R, M), the $A_q(U)$ -coaction R can be differentiated using the Hopf-algebra pairing $\langle ., . \rangle$. This then yields a left $U_q(\mathfrak{g})$ -module structure on M(cf. [96]). To be precise, the left $U_q(\mathfrak{g})$ -action on M is defined by

(4.3.28)
$$X.m := \sum_{(m)} \langle X, m_{(2)} \rangle m_{(1)}, \quad (X \in U_q(\mathfrak{g}), m \in M),$$

where $R(m) =: \sum_{(m)} m_{(1)} \otimes m_{(2)} \in M \otimes A_q(U)$ for $m \in M$. For example, by differentiating the vector corepresentation (4.3.10) we obtain the algebra homomorphism $\rho_V: U_q(\mathfrak{g}) \to \operatorname{End}(V)$ which is uniquely determined by the formulas

(4.3.29)
$$R^{\pm} = \sum_{ij} e_{ij} \otimes \rho_V(L_{ij}^{\pm}), \quad (R^{\pm})^{-1} = \sum_{ij} e_{ij} \otimes \rho_V(S(L_{ij}^{\pm})),$$

where $R^- := R^{-1}$. By differentiation of right $A_q(U)$ -coactions, a 1-1 correspondence is obtained between right $A_q(U)$ -comodule structures on a finite dimensional vector space M and P-weighted left $U_q(\mathfrak{g})$ -module structures on M. Recall that M is P-weighted if it is spanned by vectors that transform under $U_q(\mathfrak{h})$ according to $q^h \cdot v = q^{\langle h, \lambda \rangle} v$ $(\lambda \in P)$. A highest weight vector v of highest weight λ in a left $U_q(\mathfrak{g})$ -module M is then characterized by the conditions $L_{ij}^- \cdot v = 0$ (i > j) (or, equivalently, $X_i^+ v = 0$ for $i \in [1, n - 1]$) and $q^h \cdot v = q^{\langle h, \lambda \rangle} v$. There is a similar relationship between left $A_q(U)$ comodules and right $U_q(\mathfrak{g})$ -modules. For a right $U_q(\mathfrak{g})$ -module M, a weight vector $0 \neq$ $v \in M$ is a highest weight vector if $v.L_{ij}^+ = 0$ (i < j) (or, equivalently, $v.X_i^- = 0$ for $i \in [1, n - 1]$).

REMARK 4.3.2. For a right $A_q(U)$ -comodule (R, M) and for an element $m \in M$, we write m° if we consider m as element in the left $A_q(U)$ -comodule (R°, M°) (cf. Remark 4.3.1). Then, the differentiated right $U_q(\mathfrak{g})$ -module structure on M° is related to the differentiated left $U_q(\mathfrak{g})$ -module structure on M by $m^{\circ} \cdot X = (X^* \cdot m)^{\circ}$, where $m \in M$ and $X \in U_q(\mathfrak{g})$.

4. HARMONIC ANALYSIS ON QUANTUM GRASSMANNIANS

The coalgebra structure of $A_q(U)$ induces in a natural way a $A_q(U)$ -bicomodule structure on $A_q(U)$. By differentiating this $A_q(U)$ -bicomodule structure, $A_q(U)$ becomes a $U_q(\mathfrak{g})$ -bimodule with $U_q(\mathfrak{g})$ -symmetry. The action of the *L*-operators is then given by

(4.3.30)
$$L_1^{\pm} \cdot T_2 = T_2 R^{\pm}, \quad T_2 \cdot L_1^{\pm} = R^{\pm} T_2.$$

The irreducible decomposition of the $U_q(\mathfrak{g})$ -bimodule $A_q(U)$ is given by (4.3.9). This decomposition may also be characterized as the simultaneous eigenspace decomposition of $A_q(U)$ under the action of the center $\mathcal{Z} \subset U_q(\mathfrak{g})$.

It can be shown that the pairing $\langle \cdot, \cdot \rangle$ (4.3.27) is doubly non-degenerate. In particular, $A_q(U)$ can be embedded as Hopf-*-algebra into the Hopf-*-algebra dual of $U_q(\mathfrak{g})$. The image under this embedding is the Hopf-subalgebra spanned by matrix elements of finite dimensional *P*-weighted $U_q(\mathfrak{g})$ -modules (cf. Chapter 5).

4.4. Spherical corepresentations

We call $A_q(K) := A_q(U(n-l)) \otimes A_q(U(l))$ the quantized coordinate ring of the subgroup $K = U(n-l) \times U(l)$ of U = U(n). Corresponding to the embedding (4.2.3) there is an obvious surjective Hopf *-algebra morphism $\pi_K : A_q(U(n)) \to A_q(K)$. Write $A_q(U/K)$ for the right $A_q(K)$ -fixed elements in $A_q(U)$, i.e.

(4.4.1)
$$A_q(U/K) := \{ \phi \in A_q(U) \mid (\mathrm{id} \otimes \pi_K) \circ \Delta(\phi) = \phi \otimes 1 \}.$$

Observe that $A_q(U/K)$ is a left $A_q(U)$ -comodule *-subalgebra of $A_q(U)$. The algebra $A_q(U/K)$ can be interpreted as the quantized algebra of functions on the complex Grassmannian U/K.

For the study of $A_q(U/K)$ it is important to obtain explicit information about $A_q(K)$ spherical corepresentations of $A_q(U)$, i.e. finite dimensional right $A_q(U)$ -comodules with
non-zero $A_q(K)$ -fixed vectors. Recall that a vector v in a right $A_q(U)$ -comodule M with
comodule mapping $R: M \to M \otimes A_q(U)$ is $A_q(K)$ -fixed if

(4.4.2)
$$(\operatorname{id} \otimes \pi_K) \circ R(v) = v \otimes 1.$$

One defines $A_q(K)$ -fixed vectors in left $A_q(U)$ -comodules in a similar way. This section is devoted to a proof of the following theorem on $A_q(K)$ -spherical representations.

THEOREM 4.4.1. Every finite dimensional irreducible corepresentation of $A_q(U)$ has at most one $A_q(K)$ -fixed vector (up to scalar multiples). The finite dimensional corepresentations with non-zero $A_q(K)$ -fixed vectors are parametrized by the classical sublattice P_K^+ of spherical dominant weights (cf. Section 4.2).

REMARK 4.4.2. Let M be a finite dimensional right $A_q(U)$ -comodule. It follows from Remark 4.3.1 that a vector $v \in M$ is $A_q(K)$ -fixed if and only if $v \in M^\circ$ is $A_q(K)$ fixed. Hence, any statement about $A_q(K)$ -fixed vectors in right $A_q(U)$ -comodules immediately translates to a corresponding statement for left $A_q(U)$ -comodules and vice-versa.

For the proof of Theorem 4.4.1 it suffices to show that the irreducible decomposition of $V_R(\lambda)$ as a right $A_q(K)$ -comodule is the same as the decomposition of the irreducible finite dimensional U(n)-highest weight representation of highest weight λ , decomposed as representation of the subgroup K. One way of establishing this result is by differentiating the coaction of $A_q(U)$ on $V_R(\lambda)$ using the doubly non degenerate Hopf-algebra pairing between $A_q(U)$ and $U_q(\mathfrak{g})$. Then, the desired result follows from well-known results on the representation theory of quantized universal enveloping algebras. This approach is quite general, and will be treated in more detail in Chapter 6.

In this section another proof of Theorem 4.4.1 is given which does not use the quantized universal enveloping algebra. The strategy will be to relate the decomposition of the restriction to $A_q(K)$ of the right $A_q(U)$ -comodule $V_R(\lambda)$ ($\lambda \in P^+$) to characters on the maximal torus T. The following general result about corepresentation theory of semisimple coalgebras is needed (a coalgebra is said to be semisimple if every finite dimensional A-comodule is completely reducible).

PROPOSITION 4.4.3. Let A and B be semisimple coalgebras. Then every finite dimensional $A \otimes B$ -comodule is completely reducible. Denote $\{V_{\alpha} \mid \alpha \in \hat{A}\}$ and $\{V_{\beta} \mid \beta \in \hat{B}\}$ for a complete set of mutually inequivalent, irreducible, finite dimensional right Aand B-comodules, respectively. Then

$$\{V_{\alpha} \boxtimes V_{\beta} \mid \alpha \in \hat{A}, \beta \in \hat{B}\}$$

is a complete set of mutually inequivalent, irreducible, finite dimensional right $A \otimes B$ comodules. Here $V_{\alpha} \boxtimes V_{\beta} = V_{\alpha} \otimes V_{\beta}$ as a linear space and it has right comodule structure given by $R_{\alpha} \boxtimes R_{\beta} := \sigma_{2,3} \circ (R_{\alpha} \otimes R_{\beta})$, where $\sigma_{2,3}$ is the flip of the second and third tensor component and where R_{α} and R_{β} are the right comodule mapping of V_{α} and V_{β} , respectively.

PROOF. It is known that $V_{\alpha} \boxtimes V_{\beta}$ is an irreducible right $A \otimes B$ -comodule for $\alpha \in \hat{A}, \beta \in \hat{B}$ (cf. [67]). The right A and B-comodule structure on the first and second tensor component of $V_{\alpha} \boxtimes V_{\beta}$ can be recovered by the coactions

(4.4.4)
$$R_1^{\alpha} := \left(\operatorname{id}_{V_{\alpha}} \otimes \operatorname{id}_{V_{\beta}} \otimes \operatorname{id}_A \otimes \varepsilon_B \right) \circ \left(R_{\alpha} \boxtimes R_{\beta} \right),$$

respectively

$$(4.4.5) R_2^\beta := (\mathrm{id}_{V_\alpha} \otimes \mathrm{id}_{V_\beta} \otimes \varepsilon_A \otimes \mathrm{id}_B) \circ (R_\alpha \boxtimes R_\beta)$$

where ε_A and ε_B are the counits of A and B, respectively. So if $V_{\alpha} \boxtimes V_{\beta}$ is isomorphic to $V_{\alpha'} \boxtimes V_{\beta'}$ as $A \otimes B$ -comodules, then they are isomorphic as A and as B-comodules (with coactions given by R_1^{α} respectively R_2^{β}), which can only happen when $\alpha = \alpha'$ and $\beta = \beta'$. So (4.4.3) consists of mutually inequivalent, irreducible $A \otimes B$ -comodules. It remains to prove that an arbitrary finite dimensional right $A \otimes B$ -comodule (R, V)decomposes as a direct sum of irreducibles, each irreducible component being isomorphic to $V_{\alpha} \boxtimes V_{\beta}$ for some $\alpha \in \hat{A}, \beta \in \hat{B}$. Let R_A and R_B be the associated right A and B-comodule structures on V for the finite dimension right $A \otimes B$ -comodule (R, V):

$$R_A := (\mathrm{id}_V \otimes \mathrm{id}_A \otimes \varepsilon_B) \circ R, \quad R_B := (\mathrm{id}_V \otimes \varepsilon_A \otimes \mathrm{id}_B) \circ R.$$

Observe that

$$(4.4.6) \qquad (R_A \otimes \mathrm{id}_B) \circ R_B = R, \quad (R_B \otimes \mathrm{id}_A) \circ R_A = (\mathrm{id}_V \otimes \sigma_{A,B}) \circ R,$$

where $\sigma_{A,B} : A \otimes B \to B \otimes A$ is the flip. When V is considered as corepresentation space of B via the coaction R_B , we write $V^{(B)}$ for V. For $\beta \in \hat{B}$, let $\text{Hom}_B(V_\beta, V^{(B)})$ be the linear space of intertwiners between the B-comodules V_β and $V^{(B)}$:

$$\operatorname{Hom}_{B}(V_{\beta}, V^{(B)}) := \{ \phi \in \operatorname{Hom}_{\mathbf{C}}(V_{\beta}, V^{(B)}) \mid R_{B} \circ \phi = (\phi \otimes \operatorname{id}_{B}) \circ R_{\beta} \}.$$

Then, due to Schur's lemma for corepresentations [67], we have a well defined \mathbb{C} -linear bijection

(4.4.7)
$$d := \bigoplus_{\beta \in \hat{B}} d_{\beta} : \bigoplus_{\beta \in \hat{B}} \left(\operatorname{Hom}_{B}(V_{\beta}, V^{(B)}) \otimes V_{\beta} \right) \to V,$$

given component wise by $d_{\beta}(\phi_{\beta} \otimes v) := \phi_{\beta}(v) \ (v \in V_{\beta}, \phi_{\beta} \in \text{Hom}_{B}(V_{\beta}, V^{(B)})).$

We give now a right A-comodule structure $R_{\beta}^{(A)}$ on $\operatorname{Hom}_B(V_{\beta}, V^{(B)})$ such that d_{β} intertwines of the corresponding right $A \otimes B$ -comodule action on $\operatorname{Hom}_B(V_{\beta}, V^{(B)}) \boxtimes V_{\beta}$ and the right $A \otimes B$ -comodule action R on V. Since A is semisimple, this will complete the proof of the proposition.

Let $W(A) \subseteq A$ be the finite dimensional subspace of matrix elements of the right Acomodule action R_A on V, and let $\{a_i\}_{i \in I}$ be a linear basis for W(A). Fix an intertwiner $\phi \in \operatorname{Hom}_B(V_\beta, V^{(B)})$, then for every $v \in V_\beta$ there exist unique elements $\phi_i(v) \in V$ such that

$$R_A(\phi(v)) = \sum_{i \in I} \phi_i(v) \otimes a_i$$

The maps $\phi_i : V_\beta \to V \ (i \in I)$ are linear, and it is easily checked that

$$\phi_i \in \operatorname{Hom}_B(V_\beta, V^{(B)}) \quad (\forall i \in I)$$

using (4.4.6) and the definition of the ϕ_i . The map

$$R_{\beta}^{(A)} : \operatorname{Hom}_{B}(V_{\beta}, V^{(B)}) \to \operatorname{Hom}_{B}(V_{\beta}, V^{(B)}) \otimes A,$$

given by

$$R_{\beta}^{(A)}(\phi) := \sum_{i \in I} \phi_i \otimes a_i, \quad \phi \in \operatorname{Hom}_B(V_{\beta}, V^{(B)}),$$

defines a right A-comodule structure on $\operatorname{Hom}_B(V_\beta, V^{(B)})$. Using (4.4.6) it follows that d_β is an intertwiner for the $A \otimes B$ -comodule actions $\sigma_{2,3} \circ (R_\beta^{(A)} \otimes R_\beta)$ and R, which completes the proof of the proposition.

For $\lambda \in P^+$ with $\lambda_n \ge 0$, define the Schur polynomial $s_\lambda(z) \in A(T)$ by

$$s_{\lambda}(z) := \Delta(z)^{-1} \sum_{w \in \mathfrak{S}_n} (-1)^{l(w)} z^{w(\lambda+\rho)},$$

with $\Delta(z) := \prod_{i < j} (z_i - z_j)$ the Vandermonde determinant. For arbitrary $\lambda \in P^+$ with $\lambda_n \geq -m$ $(m \in \mathbb{Z})$ define $s_{\lambda}(z) := z^{-(m^n)}s_{\lambda+(m^n)}(z) \in A(T)$. Then the s_{λ} ($\lambda \in P^+$) are well-defined and form a basis of the subalgebra $A(T)^{\mathfrak{S}_n}$ of symmetric Laurent polynomials. Recall that the character of a finite dimensional $A_q(U)$ -comodule M is defined by $\chi_M := \sum_i \pi_{ii} \in A_q(U)$, where the $\pi_{ij} \in A_q(U)$ are the matrix coefficients of M with respect to a basis of M. The character χ_M is independent of the particular choice of basis for M. As shown in [96, (3.22)], the character $\chi_{\lambda} \in A_q(U)$ of the irreducible comodule $V_R(\lambda)$ satisfies

(4.4.8)
$$(\chi_{\lambda})|_{T}(z) = s_{\lambda}(z) \quad (\lambda \in P^{+}),$$

as in the classical case (q = 1).

PROPOSITION 4.4.4. Let $\lambda \in P^+$. The restriction of the $A_q(U)$ -comodule $V_R(\lambda)$ to $A_q(K)$ decomposes as

(4.4.9)
$$V_R(\lambda) \simeq \bigoplus_{\mu,\nu} (V_R(\mu) \boxtimes V_R(\nu))^{\oplus c_{\mu,\nu}^{\lambda}},$$

the sum ranging over $\mu \in P_{n-l}^+$, $\nu \in P_l^+$. Here the $c_{\mu,\nu}^{\lambda}$ are the non-negative integers characterized by

(4.4.10)
$$s_{\lambda}(z_1,\ldots,z_n) = \sum_{\mu,\nu} c_{\mu,\nu}^{\lambda} s_{\mu}(z_1,\ldots,z_{n-l}) s_{\nu}(z_{n-l+1},\ldots,z_n),$$

the sum ranging over $\mu \in P_{n-l}^+$, $\nu \in P_l^+$.

PROOF. There exists a decomposition (4.4.9) for certain uniquely determined nonnegative integers $c_{\mu,\nu}^{\lambda}$ by the previous proposition. It follows from (4.4.8) that the $c_{\mu,\nu}^{\lambda}$ satisfy (4.4.10), since $\chi_{M\boxtimes N} = \chi_M \otimes \chi_N \in A_q(K)$ for a finite dimensional right $A_q(U(n-l))$ -comodule M and a finite dimensional right $A_q(U(l))$ -comodule N. \Box

It follows from Proposition 4.4.4 that the abstract decomposition of an arbitrary finite dimensional right $A_q(U)$ -comodule M into irreducible $A_q(K)$ -comodules is the same as in the classical (q = 1) case. Hence, Theorem 4.4.1 is now a consequence of Theorem 4.2.1.

REMARK 4.4.5. The proof of Theorem 4.4.1 can also be derived from Proposition 4.4.4 using the Littlewood-Richardson rule, which is a combinatorial rule for computing the coefficients $c_{\mu,\nu}^{\lambda}$ in (4.4.10).

4. HARMONIC ANALYSIS ON QUANTUM GRASSMANNIANS

4.5. A one parameter family of quantum Grassmannians

In this section a family of quantum Grassmannians is defined which depends on one real parameter $-\infty < \sigma < \infty$ (cf. [92, Section 2]). The key ingredient in the definition will be the $n \times n$ complex matrix J^{σ} defined by

$$(4.5.1) J^{\sigma} := \sum_{1 \le k \le l} (1 - q^{2\sigma}) e_{kk} + \sum_{l < k < l'} e_{kk} - \sum_{1 \le k \le l} q^{\sigma} e_{kk'} - \sum_{1 \le k \le l} q^{\sigma} e_{k'k},$$

where k' := n - k + 1 $(1 \le k \le n)$. Observe that $\lim_{\sigma \to \infty} J^{\sigma} = J^{\infty}$, where J^{∞} is defined by

(4.5.2)
$$J^{\infty} := \sum_{k=1}^{n-l} e_{kk}$$

The subspace $\mathfrak{k}^{\sigma} \subset U_q(\mathfrak{g})$ is by definition spanned by the coefficients of the matrix

(4.5.3)
$$L^+ J^{\sigma} - J^{\sigma} L^- \in \operatorname{End}(V) \otimes U_q(\mathfrak{g}).$$

It follows from (4.3.24) that \mathfrak{k}^{σ} is a two-sided coideal in $U_q(\mathfrak{g})$, i.e. $\Delta(\mathfrak{k}^{\sigma}) \subset U_q(\mathfrak{g}) \otimes \mathfrak{k}^{\sigma} + \mathfrak{k}^{\sigma} \otimes U_q(\mathfrak{g})$ and $\varepsilon(\mathfrak{k}^{\sigma}) = 0$. This is in fact true when J^{σ} is replaced by any $n \times n$ matrix J in the definition of \mathfrak{k}^{σ} . Moreover, since J^{σ} is a symmetric matrix, it follows from (4.3.25) that \mathfrak{k}^{σ} is τ -invariant.

Define the subalgebra $A_q(\mathfrak{t}^{\sigma} \setminus U) \subset A_q(U)$ as the subspace of all left \mathfrak{t}^{σ} -invariant elements in $A_q(U)$, i.e. all $a \in A_q(U)$ such that $\mathfrak{t}^{\sigma} \cdot a = 0$. As is well-known (cf. for instance [22, Proposition 1.9]), the fact that \mathfrak{t}^{σ} is a τ -invariant two-sided coideal implies that $A_q(\mathfrak{t}^{\sigma} \setminus U)$ is a *-subalgebra which is invariant under the right $U_q(\mathfrak{g})$ -action on $A_q(U)$ (or, equivalently, the left coaction of $A_q(U)$ on itself). Important for the study of $A_q(\mathfrak{t}^{\sigma} \setminus U)$ is the fact that $X = J^{\sigma}$ is a solution of the reflection equation

$$(4.5.4) R_{12}X_1R_{12}^{-1}X_2 = X_2R_{21}^{-1}X_1R_{21},$$

where $R_{12} := R$, $R_{21} := PRP(=R^+)$, $X_1 = X \otimes id_V$ and $X_2 := id_V \otimes X$. This fact can be verified by direct computations.

REMARK 4.5.1. The algebra $A_q(\mathfrak{t}^{\sigma} \setminus U)$ can be considered as a quantized coordinate ring on the complex Grassmannian $U(n)/(U(n-l) \times U(l))$ in the following way (see [92] for more details). The quantum function space on the $n \times n$ -Hermitean matrices is defined as the algebra generated by $X = (x_{ij})_{ij}$ with relations given by the reflection equation (4.5.4). It can be endowed with a *-structure and a left $A_q(U)$ -coaction (the quantum analogue of the adjoint action). Since J^{σ} is a solution of (4.5.4) it gives rise to a (*-invariant) character of the quantum function algebra of Hermitean matrices. In other words, J^{σ} corresponds with a classical point in the quantum space of Hermitean matrices. Then $A_q(\mathfrak{t}^{\sigma} \setminus U)$ may be considered as the quantized function algebra on the adjoint orbit of the classical point corresponding to J^{σ} (see [92, Proposition 2.4]). Since J^{σ} has two different eigenvalues 1 and $-q^{2\sigma}$ with multiplicity n - l and l respectively, this quantum adjoint orbit is associated with the complex Grassmannian $U(n)/(U(n - l) \times U(l))$.

4.5. A ONE PARAMETER FAMILY OF QUANTUM GRASSMANNIANS

The quantized function algebra $A_q(U/K)$ (4.4.1) can formally be interpreted as the algebra $A_q(\mathfrak{t}^{\sigma}\setminus U)$ with $\sigma \to \infty$. To make this a little bit more explicit, we write

$$L^{+} = \begin{pmatrix} ^{11}L^{+} & ^{12}L^{+} & ^{13}L^{+} \\ 0 & ^{22}L^{+} & ^{23}L^{+} \\ 0 & 0 & ^{33}L^{+} \end{pmatrix}, \quad L^{-} = \begin{pmatrix} ^{11}L^{-} & 0 & 0 \\ ^{21}L^{-} & ^{22}L^{-} & 0 \\ ^{31}L^{-} & ^{32}L^{-} & ^{33}L^{-} \end{pmatrix},$$

where ${}^{11}L^+$ is an $l \times l$ matrix, ${}^{22}L^+$ an $(n-2l) \times (n-2l)$ matrix etc. Let D be the $l \times l$ matrix with 1's on the antidiagonal and 0's everywhere else. The coefficients of the matrix $L^+J^{\sigma} - J^{\sigma}L^-$ coincide with the coefficients of the following six matrices up to a sign:

(i)
$$q^{\sigma}(D \cdot {}^{31}L^{-} - {}^{13}L^{+} \cdot D) + (1 - q^{2\sigma})({}^{11}L^{+} - {}^{11}L^{-})$$

$$(ii) \quad {}^{12}L^+ + q^5D \cdot {}^{32}L^-$$

$$(iii) \quad {}^{23}L^+ \cdot q^{\sigma}D + {}^{21}L^-,$$

(4.5.5) (iv)
$${}^{22}L^+ - {}^{22}L^-,$$

(v) ${}^{11}L^+ \cdot q^{\sigma}D - q^{\sigma}D \cdot {}^{33}L^-,$
(vi) ${}^{33}L^+ \cdot q^{\sigma}D - q^{\sigma}D \cdot {}^{11}L^-.$

Obviously, the coefficients of the following matrix are also contained in \mathfrak{k}^{σ} :

(4.5.6)
$$(vii) \quad q^{\sigma}(D \cdot {}^{13}L^+ - {}^{31}L^- \cdot D) + (1 - q^{2\sigma})({}^{33}L^+ - {}^{33}L^-)$$

For later use, observe that the following elements of the "Cartan subalgebra" $U_q(\mathfrak{h})$ belong to \mathfrak{k}^{σ} :

$$(4.5.7) \quad L_{ii}^{+} - L_{ii}^{-} \ (l < i < l'), \quad L_{ii}^{+} - L_{i'i'}^{-} \ (1 \le i \le l), \quad L_{ii}^{-} - L_{i'i'}^{+} \ (1 \le i \le l).$$

It is clear from (4.5.5) and (4.5.6) that, in the limit $\sigma \to \infty$, the matrices in (i)–(vii) tend either to zero or to the following matrices

$$(4.5.8) \qquad {}^{11}L^+ - {}^{11}L^-, \quad {}^{12}L^+, \quad {}^{21}L^-, \quad {}^{22}L^+ - {}^{22}L^-, \quad {}^{33}L^+ - {}^{33}L^-.$$

Again, the subspace $\mathfrak{k}^{\infty} \subset U_q(\mathfrak{g})$ spanned by the coefficients of the matrices in (4.5.8) is a τ -invariant two-sided coideal. Now, on the one hand, \mathfrak{k}^{∞} -invariance in a left or right $U_q(\mathfrak{g})$ -module M is obviously the same as invariance with respect to the Hopf *-subalgebra

$$U_q(\mathfrak{k}) := U_q(\mathfrak{gl}(n-l,\mathbb{C})) \otimes U_q(\mathfrak{gl}(l,\mathbb{C})) \hookrightarrow U_q(\mathfrak{gl}(n,\mathbb{C})),$$

where invariance of $v \in M$ with respect to $u \in U_q(\mathfrak{g})$ should be interpreted as $u \cdot v = \varepsilon(u) \cdot v$ (if M is a left $U_q(\mathfrak{g})$ -module). Using the Hopf-algebra duality between $A_q(U)$ and $U_q(\mathfrak{g})$ it can be easily shown that invariance of $v \in M$ with respect to $U_q(\mathfrak{k})$ is the same as invariance with respect to $A_q(K)$ (cf. [22, Proposition 1.12]). It follows that $A_q(K)$ -invariance is equivalent to \mathfrak{k}^{∞} -invariance, hence $A_q(\mathfrak{k}^{\infty} \setminus U) = A_q(U/K)$.

REMARK 4.5.2. It should be observed that the matrix J^{∞} also satisfies the reflection equation (4.5.4), but the subspace spanned by the coefficients of the matrix $L^+J^{\infty} - J^{\infty}L^-$ is strictly smaller than \mathfrak{k}^{∞} and of little use for the purposes of this paper.

4. HARMONIC ANALYSIS ON QUANTUM GRASSMANNIANS

The following lemma is now a direct consequence of the arguments given above.

LEMMA 4.5.3. Let M be a finite dimensional right $A_q(U)$ -comodule with linear basis $\{m_i\}_i$. Consider M as left $U_q(\mathfrak{g})$ -module using the differentiated action (4.3.28). Suppose that $v_{\sigma} := \sum_i c_i(\sigma)m_i \ (c_i(\sigma) \in \mathbb{C})$ is a \mathfrak{k}^{σ} -fixed vector for all $\sigma \in \mathbb{R}$ and that $c_i := \lim_{\sigma \to \infty} c_i(\sigma)$ exists for all i. Then $\sum_i c_i m_i$ is a $A_q(K)$ -fixed vector in M.

REMARK 4.5.4. In some suitable algebraic sense (cf. [11, Proposition 9.2.3]) the algebra $U_q(\mathfrak{g})$ "tends" to $U(\mathfrak{g})$ when q tends to 1. The corresponding limits of the L-operators are given by

$$L_{ij}^{\pm}/(q-q^{-1}) \rightarrow \pm e_{ji} \ (i \leq j), \quad (q^{\varepsilon_i} - q^{-\varepsilon_i})/(q-q^{-1}) \rightarrow e_{ii}$$

(cf. [91, (1.10), (1.11)]). Hence, by (4.5.8) respectively (4.5.5), the subspace $\mathfrak{k}^{\sigma} \subset U_q(\mathfrak{g})$ ($\sigma = \infty$ respectively $\sigma = 0$) tends to the Lie subalgebra $\mathfrak{k} = \mathfrak{gl}(n - l, \mathbb{C}) \oplus \mathfrak{gl}(l, \mathbb{C}) \subset \mathfrak{g}$ respectively $\mathfrak{k}' \subset \mathfrak{g}$ (cf. Section 4.2) in the limit $q \to 1$.

Reflection equations play an important role in the quantization of symmetric spaces (cf. [91, Section 2], [94]). For the purposes of this chapter, the importance of this equation lies in the following fact. Recall that a vector w in a left $U_q(\mathfrak{g})$ -module M is called \mathfrak{k}^{σ} -fixed if $\mathfrak{k}^{\sigma} \cdot w = 0$ (a similar definition can be given for right $U_q(\mathfrak{g})$ -modules).

PROPOSITION 4.5.5. ([95, Prop. 3.1], [92]) Let J be any $n \times n$ complex matrix. Write $\mathfrak{k}^J \subset U_q(\mathfrak{g})$ for the two-sided coideal spanned by the coefficients of $L^+J - JL^-$. The element

$$w^J := \sum_{i,j} J_{ij} v_i \otimes v_j^* \in V \otimes V^*$$

in the left $U_q(\mathfrak{g})$ -module $V \otimes V^*$ is a \mathfrak{k}^J -fixed vector if and only if J satisfies the reflection equation (4.5.4).

PROOF. In the proof the same notational conventions as in [91, Proof of Proposition 2.3] will be used. Recall that the $U_q(\mathfrak{g})$ -module structure on V^* corresponding to the dual $A_q(U)$ -comodule V^* is given by

$$u \cdot v^*(v) := v^*(S(u) \cdot v) \quad (u \in U_q(\mathfrak{g}), v^* \in V^*, v \in V).$$

Set $\mathbf{v} := (v_1, \ldots, v_n)$, then it follows from (4.3.29) that

(4.5.9)
$$L_1^{\pm} \cdot \mathbf{v}_2 = \mathbf{v}_2 \cdot R_{12}^{\pm}, \quad L_1^{\pm} \cdot \mathbf{v}_2^* = \mathbf{v}_2^* \cdot (R_{21}^-)^{t_2}, \quad L_1^- \cdot \mathbf{v}_2^* = \mathbf{v}_2^* \cdot (R_{21}^+)^{t_2}.$$

Here t_2 denotes transposition with respect to the second tensor factor. An equation like $L_1^+ \cdot \mathbf{v}_2 = \mathbf{v}_2 \cdot R_{12}^+$ should be interpreted as $L_{ij}^+ \cdot v_k = \sum_{l=1}^n (R_{12}^+)_{jk}^{il} v_l$ for all $1 \le i, j, k \le n$, where $R_{12}^+ = \sum_{i,j,k,l} (R_{12}^+)_{jl}^{ik} e_{ij} \otimes e_{kl}$. Using the identities (4.5.9) one computes in shorthand notation,

$$L^{+}J \cdot w^{J} = L_{1}^{+} \cdot (\mathbf{v}_{2}J_{2} \otimes (\mathbf{v}^{*})_{2}^{t})J_{1} = (L_{1}^{+} \cdot \mathbf{v}_{2})J_{2} \otimes L_{1}^{+^{*}} \cdot (\mathbf{v}^{*})_{2}^{t}J_{1}$$

= $\mathbf{v}_{2}R_{12}^{+}J_{2}R_{21}^{-}J_{1} \otimes (\mathbf{v}^{*})_{2}^{t},$

since by (4.5.9) one has $L_1^+ \cdot (\mathbf{v}^*)_2^t = R_{21}^- \cdot (\mathbf{v}^*)_2^t$. On the other hand,

$$JL^{-} \cdot w^{J} = J_{1}L_{1}^{-} \cdot (\mathbf{v}_{2}J_{2} \otimes (\mathbf{v}^{*})_{2}^{t}) = J_{1}\mathbf{v}_{2}R_{12}^{-}J_{2} \otimes L_{1}^{-} \cdot (\mathbf{v}^{*})_{2}^{t}$$
$$= \mathbf{v}_{2}J_{1}R_{12}^{-}J_{2}R_{21}^{+} \otimes (\mathbf{v}^{*})_{2}^{t},$$

since by (4.5.9) one has $L_1^- \cdot (\mathbf{v}^*)_2^t = R_{21}^+ (\mathbf{v}^*)_2^t$. It follows from the two preceding computations that w^J is \mathfrak{t}^J -fixed if and only if $R_{12}^+ J_2 R_{21}^- J_1 = J_1 R_{12}^- J_2 R_{21}^+$. Multiplying this last equation from the left and from the right by the permutation operator P gives (4.5.4), which proves the proposition.

By Proposition 4.5.5 and the fact that the matrix J^{σ} satisfies the reflection equation (4.5.4), it follows that

(4.5.10)
$$w^{\sigma} := \sum_{ij} J^{\sigma}_{ij} v_i \otimes v^*_j \in V \otimes V^*$$

is a \mathfrak{k}^{σ} -fixed vector in the left $U_q(\mathfrak{g})$ -module $V \otimes V^*$. Observe that $\lim_{\sigma \to \infty} w^{\sigma} = w^{\infty}$, with w^{∞} the right $A_q(U/K)$ -fixed vector defined in

(4.5.11)
$$w^{\infty} := \sum_{i,j}^{n} J_{ij}^{\infty} v_i \otimes v_j^* = \sum_{i=1}^{n-l} v_i \otimes v_i^*.$$

Since $V \otimes V^* \simeq V(\varpi_1) \oplus V(0)$ as left $U_q(\mathfrak{g})$ -modules (where V(0) is the trivial module) and since w^{σ} has a non-zero weight component of weight ϖ_1 , it follows that $V(\varpi_1)$ has a non-zero \mathfrak{k}^{σ} -fixed vector.

Next we construct a right \mathfrak{k}^{σ} -fixed vector in $V^{\circ} \otimes (V^{*})^{\circ}$. Observe that a vector $\tilde{w} = \sum_{i,j} \tilde{J}_{ij} v_i \otimes v_j^{*} \in V^{\circ} \otimes (V^{*})^{\circ}$ for a real matrix $\tilde{J} = \sum_{ij} \tilde{J}_{ij} e_{ij}$ is right \mathfrak{k}^{σ} -fixed if and only if \tilde{w} is left $S(\mathfrak{k}^{\sigma})$ -fixed as element in $V \otimes V^{*}$ by the τ -invariance of \mathfrak{k}^{σ} and by Remark 4.3.2. Reasoning as in the proof of Proposition 4.5.5, it follows that \tilde{w} is left $S(\mathfrak{k}^{\sigma})$ -fixed if \tilde{J} is a solution of the linear equation

(4.5.12)
$$J_1^{\sigma}(R_{21}^{-})^{t_1}\tilde{J}_2((R_{21}^{-})^{t_1})^{-1} = R^{t_1}\tilde{J}_2(R^{t_1})^{-1}J_1^{\sigma}$$

where J^{σ} is given by (4.5.1). A solution $\tilde{J} = \tilde{J}^{\sigma}$ of (4.5.12) is given by

(4.5.13)
$$\tilde{J}^{\sigma} := \sum_{1 \le k \le l} (1 - q^{2(n-2l)} q^{2\sigma}) e_{kk} + \sum_{l < k < l'} e_{kk} - q^{\sigma-1} \sum_{1 \le k \le l} q^{2(k-l)} e_{kk'} - q^{\sigma-1} \sum_{1 \le k \le l} q^{2(k'-l)} e_{k'k}.$$

We write $\tilde{w}^{\sigma} = \sum_{ij} \tilde{J}_{ij}^{\sigma} v_i \otimes v_j^*$ for the corresponding right \mathfrak{k}^{σ} -fixed vector in $V^{\circ} \otimes (V^*)^{\circ}$. Similarly as was shown for left \mathfrak{k}^{σ} -fixed vectors it follows that $V(\varpi_1)^{\circ}$ has a non-zero right \mathfrak{k}^{σ} -fixed vector. Observe that $\lim_{\sigma \to \infty} \tilde{w}^{\sigma} = w^{\infty}$, with w^{∞} the $A_q(K)$ -fixed vector given by (4.5.11).

Recall from the previous section that $V(\lambda)$ has at most one \mathfrak{k}^{∞} -fixed vector up to scalar multiples, and that $V(\lambda)$ has non-zero \mathfrak{k}^{∞} -fixed vectors if and only if $\lambda \in P_K^+$

(cf. Theorem 4.4.1). We have the following analogous statement for \mathfrak{k}^{σ} -fixed vectors $(-\infty < \sigma < \infty)$.

THEOREM 4.5.6. ([92, Theorem 2.6]) Let $\lambda \in P^+$ and fix $-\infty < \sigma < \infty$. The irreducible left $U_q(\mathfrak{g})$ -module $V(\lambda)$ with highest weight λ has at most one \mathfrak{t}^{σ} -fixed vector (up to scalar multiples). There exist non-zero \mathfrak{t}^{σ} -vectors in $V(\lambda)$ if and only if $\lambda \in P_K^+$. The same statement holds for right \mathfrak{t}^{σ} -fixed vectors in $V(\lambda)^\circ$.

In the remainder of this section a proof of Theorem 4.5.6 is given. Fix a parameter $-\infty < \sigma < \infty$. First of all, we have the following crucial lemma.

LEMMA 4.5.7. Let $\lambda \in P^+$ and fix $-\infty < \sigma < \infty$. Then any non-zero \mathfrak{t}^{σ} -fixed vector in the left $U_q(\mathfrak{g})$ -module $V(\lambda)$ has a non-zero weight component of highest weight λ . The same statement holds for the right $U_q(\mathfrak{g})$ -module $V(\lambda)^{\circ}$.

The proof of the lemma follows by analyzing the particular form of the two-sided coideal \mathfrak{k}^{σ} . The details are omitted here, since the proof is analogous to the proof of [91, Lemma 3.2] and [24, Proposition 3.2].

Since the linear subspace of $V(\lambda)$ (respectively $V(\lambda)^{\circ}$) consisting of weight vectors of weight λ is one dimensional, it follows from Lemma 4.5.7 that every irreducible finite dimensional *P*-weighted $U_q(\mathfrak{g})$ -module has at most one \mathfrak{k}^{σ} -fixed vector up to scalar multiples.

Set $P_K = \bigoplus_{1 \le r \le l} \mathbb{Z} \varpi_r$, where ϖ_r are the fundamental spherical weights (cf. Section 4.2). Observe that the assignment $\lambda \mapsto \lambda^{\natural}$ as defined in Section 4.2 extends to an orderpreserving bijection from P_K onto P_{Σ} . For $\mu \in P_{\Sigma}$, we write $\lambda_{\mu} \in P_K$ for the inverse of μ under the bijection \natural .

The following lemma is immediate from the fact that the Cartan type elements listed in (4.5.7) belong to \mathfrak{k}^{σ} .

LEMMA 4.5.8. Let $\lambda \in P_K^+$, $-\infty < \sigma < \infty$ and assume that $v \in V(\lambda)$ is a non-zero left \mathfrak{k}^{σ} -fixed vector. Let $v = \sum_{\mu \leq \lambda} v_{\mu}$ be the decomposition of v in weight vectors, where v_{μ} has weight $\mu \in P$. Then $v_{\mu} = 0$ unless $\mu \in P_K$. The same statement is valid for the right $U_q(\mathfrak{g})$ -module $V(\lambda)^{\circ}$.

It follows from Lemma 4.5.7 and Lemma 4.5.8 that if $V(\lambda)$ (respectively $V(\lambda)^{\circ}$) has a non-zero \mathfrak{k}^{σ} -fixed vector, then $\lambda \in P_{K}^{+}$. To finish the proof of Theorem 4.5.6 we have to show that all modules $V(\lambda)$ and

To finish the proof of Theorem 4.5.6 we have to show that all modules $V(\lambda)$ and $V(\lambda)^{\circ}$ ($\lambda \in P_{K}^{+}$) have non-zero \mathfrak{k}^{σ} -fixed vectors. The existence of non-trivial \mathfrak{k}^{σ} -fixed vectors in $V(\varpi_{1})$ and in $V(\varpi_{1})^{\circ}$ is already proved. Explicit intertwining operators

$$\widehat{\Psi}_r \colon (V \otimes V^*)^{\otimes r} \to \Lambda^r_a(V) \otimes \Lambda^r_a(V^*), \quad (1 \le r \le l)$$

will be constructed to prove the existence of \mathfrak{k}^{σ} -fixed vectors in higher fundamental spherical representations. The proof of Theorem 4.5.6 is then completed by computing socalled principal term of $\widehat{\Psi}_r((w^{\sigma})^{\otimes r})$, with $w^{\sigma} \in V(\varpi_1)$ the \mathfrak{k}^{σ} -fixed vector given by (4.5.10). Before giving the construction of $\widehat{\Psi}_r$, we first introduce the notion of principal term of a vector $v \in \Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ (cf. [95], [124]). For the present setting it is convenient to use a slightly modified definition of Noumi's and Sugitani's notion of principal term (cf. [95], [124]). The definition is based on certain specific properties of the comodule $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$. The comodule $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ has a multiplicity free decomposition

(4.5.14)
$$\Lambda_q^r(V) \otimes \Lambda_q^r(V^*) \cong \bigoplus_{s=0}^r V(\varpi_s) \quad (1 \le r \le l)$$

as right $A_q(U)$ -comodules, where $\varpi_0 := 0 \in P_K^+$. The decomposition (4.5.14) can be proved by computing the restriction of the character of the module $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ to the torus and using the classical Pieri formula for Schur functions [85, I, (5.17)] (cf. Proposition 4.4.4). Due to the multiplicity free decomposition (4.5.14), the module $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ is extremely useful for the study of \mathfrak{k}^{σ} -fixed vectors in $V(\varpi_r)$, as will be shown in the remainder of this chapter as well as in the next chapter. It follows from (4.5.14) that all the weights $\mu \in P$ of the module $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ are $\leq \varpi_r$ with respect to the dominance order. The vector $v_{[1,r]} \otimes v_{[n-r+1,n]}^* \in \Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ is the highest weight vector of the unique copy of $V(\varpi_r)$ within $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$. Suppose now that $v = \sum_{\mu \leq \varpi_r} v_{\mu}$ is the weight space decomposition of a vector $v \in \Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$, where v_{μ} is the weight component of weight $\mu \in P$. Then the *principal term* of v is defined by

$$(4.5.15) [v] := \sum_{\nu \in \mathcal{W}(1^r)} v_{\lambda_{\nu}}$$

(cf. [95], [124]), where $\mathcal{W} = \mathcal{W}_l$ acts on $(1^r) \in P_{\Sigma}^+ \subset P_{\Sigma} = \mathbb{Z}^l$ by permutations and inversions (cf. Section 4.2). Since $\{\lambda_{\nu} | \nu \in \mathcal{W}(1^r)\}$ lies in the \mathfrak{S}_n -orbit of the highest weight $\varpi_r \in P^+$ of $V(\varpi_r)$, it follows that the principal term of a vector $v \in \Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ lies in the unique copy of $V(\varpi_r)$ within $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$. If v is a non-zero \mathfrak{k}^{σ} -fixed vector in $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ and if v - [v] has a non-zero weight component of weight ν , then, by Lemma 4.5.8, $\nu \in P_K$ and $\nu^{\natural} \in C(\varpi_r)$, where

$$(4.5.16) C(\mu) := \{ \mu' \in P_{\Sigma} \mid w\mu' \lneq \mu \; \forall w \in \mathcal{W} \} \quad (\mu \in P_{\Sigma}^+)$$

is the strict integral convex hull of $\mathcal{W}\mu$. In the next proposition the principal term of a \mathfrak{k}^{σ} -fixed vector in $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ (respectively in $\Lambda_q^r(V)^{\circ} \otimes \Lambda_q^r(V^*)^{\circ}$) is compared with the elements u_r , \tilde{u}_r ($1 \le r \le l$), which are defined by

$$(4.5.17) u_r := \sum_{\substack{I \subset [1,l] \cup [l',n] \\ |I| = r, I \cap I' = \emptyset}} v_I \otimes v_{I'}^*, \quad \tilde{u}_r := \sum_{\substack{I \subset [1,l] \cup [l',n] \\ |I| = r, I \cap I' = \emptyset}} q^{\langle 2\rho, \tilde{\varepsilon}_{I'} \rangle} v_I \otimes v_{I'}^*$$

where $I' := \{i' \mid i \in I\}$. The element u_r lies in the unique copy of $V(\varpi_r)$ within $\Lambda_q^r(V) \otimes \Lambda_q^l(V^*)$, whereas \tilde{u}_r lies in the unique copy of $V(\varpi_r)^\circ$ within $\Lambda_q^r(V)^\circ \otimes \Lambda_q^r(V^*)^\circ$. Observe that by the explicit form of the \mathfrak{k}^σ -fixed vectors w^σ respectively \tilde{w}^σ , we have

(4.5.18)
$$[w^{\sigma}] = -q^{\sigma}u_1, \quad [\tilde{w}^{\sigma}] = -q^{\sigma-1}q^{2(1-l)}\tilde{u}_1.$$

For the construction of the intertwiner $\widehat{\Psi}_r$, consider now the linear bijection $\beta \colon V^* \otimes V \to V \otimes V^*$ determined by

(4.5.19)
$$\beta(v_i^* \otimes v_j) = q^{-\delta_{ij}} v_j \otimes v_i^* + (q^{-1} - q) \delta_{ij} \sum_{k < j} v_k \otimes v_k^*.$$

Write $V_i := V, V_i^* := V^*$ $(1 \le i \le r)$. Define a linear bijection

$$\Psi_r \colon (V_1 \otimes V_1^*) \otimes \cdots \otimes (V_r \otimes V_r^*) \to (V_1 \otimes \cdots \otimes V_r) \otimes (V_1^* \otimes \cdots \otimes V_r^*)$$

by

(4.5.20)
$$\Psi_r := \beta_{1,r} \circ \beta_{2,r} \circ \cdots \circ \beta_{r-1,r} \circ \cdots \circ \beta_{13} \circ \beta_{23} \circ \beta_{12},$$

where β_{ij} acts by definition as the identity on all factors of the tensor product except for $V_i^* \otimes V_j$, on which it is equal to β . Write

$$\operatorname{pr}_r \colon V^{\otimes r} \to \Lambda^r_q(V), \quad \operatorname{pr}^*_r \colon (V^*)^{\otimes r} \to \Lambda^r_q(V^*)$$

for the canonical projections. We have now the following generalization of (4.5.18).

PROPOSITION 4.5.9. Let $1 \le r \le l$. The operator

$$\widehat{\Psi}_r \colon (V \otimes V^*)^{\otimes r} \to \Lambda^r_q(V) \otimes \Lambda^r_q(V^*)$$

defined by $\widehat{\Psi}_r := (\mathrm{pr}_r \otimes \mathrm{pr}_r^*) \circ \Psi_r$ is a surjective intertwiner and

$$\begin{split} & [\widehat{\Psi}_r((w^{\sigma})^{\otimes r})] = c_r(\sigma)u_r, \quad c_r(\sigma) := \left(\frac{q^{\sigma}}{q^2 - 1}\right)^r \left(q^2; q^2\right)_r, \\ & [\widehat{\Psi}_r((\tilde{w}^{\sigma})^{\otimes r})] = \tilde{c}_r(\sigma)\tilde{u}_r, \quad \tilde{c}_r(\sigma) := \left(\frac{q^{\sigma-1}q^{2(1-l)}}{q^2 - 1}\right)^r \left(q^2; q^2\right)_r. \end{split}$$

Before giving a proof of Proposition 4.5.9, we first show how it implies Theorem 4.5.6. Since \mathfrak{t}^{σ} is a two-sided coideal and Ψ_r an intertwining operator, Proposition 4.5.9 shows that $\Psi_r((w^{\sigma})^{\otimes r})$ is a non-zero \mathfrak{t}^{σ} -fixed vector. Proposition 4.5.9 implies that the leading term of $\Psi_r((w^{\sigma})^{\otimes r})$ is non-zero, hence it follows that $V(\varpi_r)$ ($1 \leq r \leq l$) has a non-zero \mathfrak{t}^{σ} -fixed vector. Since any $\lambda \in P_K^+$ can be written as a positive integral linear combination of the fundamental spherical weights $\{\varpi_r\}_{1\leq r\leq l}$, it follows by an easy argument using tensor products and Lemma 4.5.7 that any $\lambda \in P_K^+$ is actually spherical. For right \mathfrak{t}^{σ} -fixed vectors the same argument holds, since $\widehat{\Psi}_r$ is also an intertwiner as map from the module $(V^{\circ} \otimes (V^*)^{\circ})^{\otimes r}$ to $\Lambda_q^r(V)^{\circ} \otimes \Lambda_q^r(V^*)^{\circ}$ (cf. Remark 4.3.1).

So it remains to prove Proposition 4.5.9. The proof of this proposition proceeds by induction on r. The proof is broken up into a couple of lemmas.

LEMMA 4.5.10. For $2 \le r \le n+1$ the linear mapping $\widehat{\Phi}_r \colon \Lambda_a^{r-1}(V^*) \otimes V \to V \otimes \Lambda_a^{r-1}(V^*)$

4.5. A ONE PARAMETER FAMILY OF QUANTUM GRASSMANNIANS

defined on the basis vectors $v_I^* \otimes v_j$ ($|I| = r - 1, 1 \le j \le n$) by

$$\widehat{\Phi}_{r}(v_{I}^{*} \otimes v_{j}) = \begin{cases} v_{j} \otimes v_{I}^{*} & \text{if } j \notin I, \\ q^{-1}v_{j} \otimes v_{I}^{*} - (q - q^{-1}) \sum_{m < j} \frac{\mathrm{s}_{q}(I \setminus j; m)}{\mathrm{s}_{q}(I \setminus j; j)} v_{m} \otimes v_{(I \setminus j) \cup m}^{*} & \text{if } j \in I \end{cases}$$

is an intertwining operator of right $A_q(U)$ -comodules.

PROOF. Let $P: V \otimes V \to V \otimes V$ denote the flip. Define a linear bijection $\gamma: V \otimes V \to V \otimes V$ by $\gamma := PR$, with R as in (4.3.2). The action of γ on the basis vectors $v_i \otimes v_j$ $(1 \le i, j \le n)$ is given by

(4.5.21)
$$\gamma(v_i \otimes v_j) = q^{\delta_{ij}} v_j \otimes v_i + (q - q^{-1})\theta_{i,j} v_i \otimes v_j$$

with $\theta_{i,j} := 1$ if i < j and $\theta_{i,j} := 0$ otherwise. The fact that the commutation relations between the $t_{ij} \in A_q(U)$ can be written as $RT_1T_2 = T_2T_1R$ (cf. Section 4.3) implies that γ is an intertwining operator. Since R is a solution of the Quantum Yang-Baxter Equation, γ satisfies

(4.5.22)
$$\gamma_1 \circ \gamma_2 \circ \gamma_1 = \gamma_2 \circ \gamma_1 \circ \gamma_2,$$

with $\gamma_i \in \operatorname{End}(V^{\otimes 3})$ acting as γ on the *i*th and (i+1)th tensor factors and as the identity on the remaining factor. Note furthermore that the exterior algebra $\Lambda_q(V)$ is isomorphic as right $A_q(U)$ -comodule algebra with T(V)/I, where T(V) is the tensor algebra of V and $I \subset T(V)$ the two-sided ideal generated by $\operatorname{ker}(\operatorname{id} - q^{-1}\gamma) \subset V^{\otimes 2} \subset T(V)$. Consider now the intertwiner $\Gamma_k \colon V^{\otimes (k-1)} \otimes V \to V \otimes \Lambda_q^{k-1}(V)$ ($2 \leq k \leq n+1$) defined by

$$\Gamma_k = (\mathrm{id} \otimes \mathrm{pr}_{k-1}) \circ \gamma_1 \circ \gamma_2 \circ \cdots \circ \gamma_{k-1}.$$

Application of [35, Lemma 4.9 (1)] to the Yang-Baxter operator $q^{-1}\gamma$ shows that there exists a unique bijective intertwiner

$$\widehat{\Gamma}_k \colon \Lambda_q^{k-1}(V) \otimes V \to V \otimes \Lambda_q^{k-1}(V)$$

such that $\Gamma_k = \widehat{\Gamma}_k \circ (\operatorname{pr}_{k-1} \otimes \operatorname{id})$. By a straightforward computation one verifies that

 $\widehat{\Gamma}_k(v_I\otimes v_j)=q^{|I\cap j|}v_j\otimes v_I+$

$$+ (1-q^2)(-q)^{-k+1} \operatorname{s_q}(I;j) \sum_{\substack{i \in I \\ i < j}} \operatorname{s_q}(i;I \setminus i) v_i \otimes v_{(I \setminus i) \cup j}$$

for $I \subset [1, n]$ with |I| = k - 1 and $1 \le j \le n$.

Next, the linear mapping $\delta_k : \Lambda_q^k(V^*) \to \Lambda_q^{n-k}(V) \otimes \mathbb{C} \det_q^{-1} (1 \le k \le n)$ defined on the basis elements v_I^* (|I| = k) by $\delta_k(v_I^*) := s_q(I; I^c) v_{I^c} \otimes \det_q^{-1}$ is a bijective

4. HARMONIC ANALYSIS ON QUANTUM GRASSMANNIANS

intertwiner by (4.3.19). With the canonical identification $V \otimes \mathbb{C} \det_q^{-1} \cong \mathbb{C} \det_q^{-1} \otimes V$ we have an intertwining operator $\widehat{\Phi}_r \colon \Lambda_q^{r-1}(V^*) \otimes V \to V \otimes \Lambda_q^{r-1}(V^*)$ defined by

$$\widehat{\Phi}_r := q^{-1}(\mathrm{id}\otimes\delta_{r-1}^{-1})\circ(\widehat{\Gamma}_{n-r+2}\otimes\mathrm{id})\circ(\delta_{r-1}\otimes\mathrm{id}).$$

Starting from the explicit expressions for $\widehat{\Gamma}_{n-r+2}$ and δ_{r-1} , a straightforward calculation shows that $\widehat{\Phi}_r$ acts on the basis vectors $v_I^* \otimes v_j$ as required.

COROLLARY 4.5.11. The linear mappings β , Ψ_r , and $\widehat{\Psi}_r$ are right $A_q(U)$ -comodule homomorphisms.

PROOF. The assertion follows from the previous lemma since $\beta = \overline{\Phi}_2$ and since the natural projections pr_r and pr_r^* intertwine the right $A_q(U)$ -comodule actions.

LEMMA 4.5.12. Let $1 \le r \le l$. The bijective intertwining operator

 $\Phi_r \colon (V_1^* \otimes \cdots \otimes V_{r-1}^*) \otimes V_1 \to V_1 \otimes (V_1^* \otimes \cdots \otimes V_{r-1}^*)$

defined by $\Phi_r := \beta_{12} \circ \beta_{23} \circ \cdots \circ \beta_{r-2,r-1} \circ \beta_{r-1,r}$ satisfies

$$(\mathrm{id}\otimes\mathrm{pr}_{r-1}^*)\circ\Phi_r=\widehat{\Phi}_r\circ(\mathrm{pr}_{r-1}^*\otimes\mathrm{id}).$$

PROOF. For $I = \{i_1 < \ldots < i_r\} \subset [1, n]$, set $\tilde{v}_I^* := v_{i_r}^* \otimes \ldots \otimes v_{i_2}^* \otimes v_{i_1}^*$. It is clear from the definitions that

$$(\mathrm{id} \otimes \mathrm{pr}_{r-1}^*) \circ \Phi_r(\tilde{v}_I^* \otimes v_j) = v_j \otimes v_I^* \quad \text{if } j \notin I.$$

If $j \in I$, then

$$(\mathrm{id}\otimes\mathrm{pr}_{r-1}^*)\circ\Phi_r(\tilde{v}_I^*\otimes v_j)=q^{-1}v_j\otimes v_I^*-(q-q^{-1})\sum_{m< j}c(m,j)v_m\otimes v_{(I\setminus j)\cup m}^*,$$

where $c(m, j) := (-q)^{|\{i \in I | m < i < j\}|}$ if $m \notin I$, and c(m, j) := 0 otherwise. Using the definition of the *q*-signum s_q , it follows that $c(m, j) = s_q(I \setminus j; m) s_q(I \setminus j; j)^{-1}$ if m < j, which concludes the proof of the lemma.

Observe that the multiplication maps

$$\mu \colon \Lambda_q(V) \otimes \Lambda_q(V) \to \Lambda_q(V), \quad \mu^* \colon \Lambda_q(V^*) \otimes \Lambda_q(V^*) \to \Lambda_q(V^*)$$

are intertwiners of the $A_q(U)$ -coactions, since $\Lambda_q(V)$ and $\Lambda_q(V^*)$ are $A_q(U)$ -comodule algebras.

LEMMA 4.5.13. The intertwining operator

$$\widehat{\Theta}_r \colon \Lambda^{r-1}_q(V) \otimes \Lambda^{r-1}_q(V^*) \otimes V \otimes V^* \to \Lambda^r_q(V) \otimes \Lambda^r_q(V^*)$$

defined by $\widehat{\Theta}_r := (\mu \otimes \mu^*) \circ (\mathrm{id}_{\Lambda_q^{r-1}(V)} \otimes \widehat{\Phi}_r \otimes \mathrm{id}_{V^*})$ satisfies

$$\begin{aligned} [\widehat{\Theta}_r(u_{r-1} \otimes w^{\sigma})] &= -q^{\sigma} \frac{1 - q^{2r}}{1 - q^2} u_r \\ [\widehat{\Theta}_r(\widetilde{u}_{r-1} \otimes \widetilde{w}^{\sigma})] &= -q^{\sigma-1} q^{2(1-l)} \frac{1 - q^{2r}}{1 - q^2} \widetilde{u}_r \end{aligned}$$

for $2 \leq r \leq l$.

PROOF. If v is a vector of weight μ in the domain of $\widehat{\Theta}_r$, then $\widehat{\Theta}_r(v)$ is again a weight vector of weight μ since $\widehat{\Theta}_r$ intertwines the right $A_q(U)$ -coaction. Hence, for a fixed $I \subset [1, l] \cup [l', n]$ with $I \cap I' = \emptyset$ and |I| = r - 1, we have that $[\widehat{\Theta}_r(v_I \otimes v_{I'}^* \otimes v_s \otimes v_t^*)] = 0$ unless $s, t \notin I \cup I'$ and $s \neq t$. By the explicit formulas for the action of $\widehat{\Phi}_r$ (cf. Lemma 4.5.10), it follows that

$$[\widehat{\Theta}_r(u_{r-1}\otimes w^{\sigma})] = -q^{\sigma}\sum_{I,k} v_I \wedge v_k \otimes v_{I'}^* \wedge v_{k'}^* = -q^{\sigma}\sum_J c_J v_J \otimes v_{J'}^*$$

where the first sum is taken over pairs (I, k) with $I \subset [1, l] \cup [l', n]$, $k \in [1, l] \cup [l', n]$, |I| = r - 1, $I \cap I' = \emptyset$ and $k \notin I \cup I'$, and the second sum is taken over subsets $J \subset [1, l] \cup [l', n]$ with $J \cap J' = \emptyset$ and |J| = r. The corresponding constant c_J is given by

$$c_J = \sum_{k \in J} s_q(J \setminus k; k) s_q(k'; J' \setminus k') = \sum_{k \in J} (s_q(J \setminus k; k))^2 = \sum_{s=0}^{r-1} q^{2s} = \frac{1 - q^{2r}}{1 - q^2}.$$

The proof for the leading term of $\widehat{\Theta}_r(\tilde{u}_{r-1} \otimes \tilde{w}^{\sigma})$ is similar.

Proposition 4.5.9 can now be proved by induction to r, using the previous lemma for the induction step.

PROOF OF PROPOSITION 4.5.9. Define an intertwiner

$$\Theta_r \colon V^{\otimes (r-1)} \otimes (V^*)^{\otimes (r-1)} \otimes V \otimes V^* \to V^{\otimes r} \otimes (V^*)^{\otimes r}$$

by

 $\Theta_r := \mathrm{id}_{V^{\otimes (r-1)}} \otimes \Phi_r \otimes \mathrm{id}_{V^*}.$

It follows from Lemma 4.5.12 that

(4.5.23)
$$(\mathrm{pr}_r \otimes \mathrm{pr}_r^*) \circ \Theta_r = \Theta_r \circ (\mathrm{pr}_{r-1} \otimes \mathrm{pr}_{r-1}^* \otimes \mathrm{id}_V \otimes \mathrm{id}_{V^*}).$$

From the definitions of Ψ_r and Φ_r it follows that

$$\Psi_r = \Theta_r \circ (\Psi_{r-1} \otimes \mathrm{id})$$

and hence by (4.5.23)

(4.5.24) $\widehat{\Psi}_r = \widehat{\Theta}_r \circ (\widehat{\Psi}_{r-1} \otimes \mathrm{id}).$

121

This allows us to prove the proposition by induction to r. The proposition is trivial for r = 1. Suppose that $r \ge 2$. By the induction hypotheses and Lemma 4.5.8 we have

$$\widehat{\Psi}_{r-1}((w^{\sigma})^{\otimes r-1}) = c_{r-1}(\sigma)u_{r-1} + \sum_{\nu \in C((1^{r-1}))} v_{\lambda_{\nu}}$$

where $v_{\lambda_{\nu}}$ is some weight vector of weight λ_{ν} and $C(\mu)$ is defined by (4.5.16). For $\nu \in C((1^{r-1}))$ we have $[\widehat{\Theta}_r(v_{\lambda_{\nu}} \otimes w^{\sigma})] = 0$, hence the induction step for the computation of $[\widehat{\Psi}_r((w^{\sigma})^{\otimes r})]$ follows by combining Lemma 4.5.13 with (4.5.24). The leading term $[\widehat{\Psi}_r((\tilde{w}^{\sigma})^{\otimes r})]$ can be computed in a similar way.

REMARK 4.5.14. It should be observed that the proof of Theorem 4.5.6 differs in important details from the proof of Theorem 4.4.1. Observe for instance that Lemma 4.5.7 does not hold with \mathfrak{k}^{σ} -fixed replaced by \mathfrak{k}^{∞} -fixed, since any \mathfrak{k}^{∞} -fixed vector lies automatically in the zero weight space of the module.

4.6. Zonal (σ, τ) -spherical functions

In this section the \mathfrak{k}^{τ} -invariant $(-\infty < \tau < \infty)$ functions are studied in the quantized coordinate ring $A_q(\mathfrak{k}^{\sigma} \setminus U)$ $(-\infty < \sigma < \infty)$. The results of this section were announced in [92, Section 3]. The rank 1 case of these results were earlier derived by Koornwinder [63] for n = 2 and for arbitrary projective space by Noumi and Dijkhuizen [24].

Let $-\infty < \sigma, \tau \leq \infty$ and denote $\mathcal{H}^{\sigma,\tau}$ for the *-subalgebra of left \mathfrak{k}^{σ} -invariant and right \mathfrak{k}^{τ} -invariant functions in $A_q(U)$. From Theorem 4.4.1, Theorem 4.5.6 and (4.3.9) we obtain the decomposition

(4.6.1)
$$\mathcal{H}^{\sigma,\tau} = \bigoplus_{\lambda \in P_K^+} \mathcal{H}^{\sigma,\tau}(\lambda), \quad \mathcal{H}^{\sigma,\tau}(\lambda) := W(\lambda) \cap \mathcal{H}^{\sigma,\tau},$$

the subspaces $\mathcal{H}^{\sigma,\tau}(\lambda)$ ($\lambda \in P_K^+$) being one dimensional. A non-zero element $\varphi^{\sigma,\tau}(\lambda) \in \mathcal{H}^{\sigma,\tau}(\lambda)$ is called a zonal (σ,τ)-spherical function. Since the decomposition (4.3.9) is orthogonal with respect to the inner product $\langle \varphi, \psi \rangle = h(\psi^*\varphi)$, the zonal spherical functions $\varphi^{\sigma,\tau}(\lambda)$ ($\lambda \in P_K^+$) are mutually orthogonal with respect to $\langle ., . \rangle$.

Let M denote a right $A_q(U)$ -comodule with comodule mapping R_M and an invariant inner product $\langle ., . \rangle$. With any two elements $v, w \in M$ we associate the matrix coefficient

The map θ_M induces a linear map (denoted by the same symbol)

$$\theta_M \colon M^{\circ} \otimes M \to A_q(U),$$

which is surjective onto the subspace spanned by the matrix coefficients of M. If no confusion can arise we sometimes write $\theta := \theta_M$. The following lemma is a direct consequence of these definitions (cf. [91, Lemma 4.8]).

LEMMA 4.6.1. Let M be a unitary right $A_q(U)$ -comodule. The map $\theta_M \colon M^{\circ} \otimes M \to A_q(U)$ satisfies the following properties:

(i) θ_M is a $A_q(U)$ -bicomodule homomorphism, i.e.

(4.6.3)
$$\Delta \circ \theta_M = (\theta_M \otimes \mathrm{id}) \circ (\mathrm{id} \otimes R_M), \quad \Delta \circ \theta_M = (\mathrm{id} \otimes \theta_M) \circ (R_M^\circ \otimes \mathrm{id}),$$

where R_M° is defined as in Remark 4.3.1.

- (ii) $\theta_M(v, w) = \tau(\theta_M(w, v)) \ (v, w \in M).$
- (iii) If M is irreducible of highest weight $\lambda \in P^+$, then $\theta_M \colon M^\circ \otimes M \to W(\lambda)$ is an isomorphism of $A_q(U)$ -bicomodules.

Lemma 4.6.1 can be used to construct zonal (σ, τ) -spherical functions as follows. Let $v_{\sigma}(\lambda) \in V(\lambda)$ respectively $\tilde{v}_{\tau}(\lambda) \in V(\lambda)^{\circ}$ be a non-zero \mathfrak{k}^{σ} -fixed respectively \mathfrak{k}^{τ} -fixed vector $(\lambda \in P_{K}^{+})$. Let $\langle ., . \rangle$ be a unitary inner product on $V(\lambda)$, and write θ_{λ} for the map θ in Lemma 4.6.1 with respect to the unitary comodule $(V(\lambda), \langle ., . \rangle)$. Then

(4.6.4)
$$\varphi^{\sigma,\tau}(\lambda) := \theta_{\lambda}(\tilde{v}_{\tau}(\lambda), v_{\sigma}(\lambda)) \in \mathcal{H}^{\sigma,\tau}(\lambda)$$

is a zonal (σ, τ) -spherical function by Lemma 4.6.1. This leads to the following lemma.

LEMMA 4.6.2. Let $-\infty < \sigma, \tau < \infty$ and $\lambda \in P_K^+$. The image of $\varphi^{\sigma,\tau}(\lambda)$ under the restriction map $|_T: A_q(U) \to A(T)$ is of the form

(4.6.5)
$$\varphi^{\sigma,\tau}(\lambda)_{|T} = c_{\lambda^{\natural}} m_{\lambda^{\natural}}(x) + \sum_{\nu \in C(\lambda^{\natural})} c_{\nu} x^{\nu}, \quad c_{\nu} \in \mathbb{C},$$

with $c_{\lambda^{\natural}} \neq 0$ and $C(\nu)$ given by (4.5.16). Here the notation $x^{\nu} := x_1^{\nu_1} x_2^{\nu_2} \dots x_l^{\nu_l}$ for $\nu = (\nu_1, \dots, \nu_l) \in P_{\Sigma}$ is used, where the x_i $(1 \leq i \leq l)$ are defined by (4.2.10).

PROOF. Since any $\lambda \in P_K^+$ can be written as a positive integral linear combination of the fundamental spherical weights $\{\varpi_r\}_{1 \le r \le l}$, it follows by an easy argument using tensor products, Lemma 4.5.7 and Lemma 4.5.8 that (4.6.5) for arbitrary $\lambda \in P_K^+$ follows from (4.6.5) for the fundamental spherical weights $\{\varpi_r\}_{r=1}^l$.

So fix a fundamental weight ϖ_r $(1 \le r \le l)$. Consider the unitary inner product

(4.6.6)
$$\langle v_I \otimes v_J^*, v_K \otimes v_L^* \rangle = q^{-\langle 2\rho, \varepsilon_J \rangle} \delta_{I,K} \delta_{J,L}$$

on $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ (cf. Section 4.3) and write θ for the map (4.6.2) associated with the module $(\Lambda_q^r(V) \otimes \Lambda_q^r(V^*), \langle ., . \rangle)$. By (4.5.14), the module $V(\varpi_r)$ may be considered as irreducible component of $\Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ with unitary structure given by the restriction of $\langle ., . \rangle$ to $V(\varpi_r)$. Then, by Proposition 4.5.9 and the fact that $u_r \in \Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ (respectively $\tilde{u}_r \in \Lambda_q^r(V)^\circ \otimes \Lambda_q^r(V^*)^\circ$) lies in the unique irreducible component $V(\varpi_r)$ (respectively $V(\varpi_r)^\circ$), the principal terms of the \mathfrak{k}^{σ} -fixed vector $v_{\sigma}(\varpi_r)$ and the \mathfrak{k}^{τ} -fixed vector $\tilde{v}_{\tau}(\varpi_r)$ are given by

$$(4.6.7) [v_{\sigma}(\varpi_r)] = c_r u_r, \quad [\tilde{v}_{\tau}(\varpi_r)] = \tilde{c}_r \tilde{u}_r$$

for non-zero constants $c_r, \tilde{c}_r \in \mathbb{C}$. For $v_\mu \in \Lambda^r_q(V) \otimes \Lambda^r_q(V^*)$ of weight μ and $\tilde{v}_\nu \in \Lambda^r_q(V)^\circ \otimes \Lambda^r_q(V^*)^\circ$ of weight ν we have $\theta(\tilde{v}_\nu, v_\mu)_{|T} = 0$ if $\mu \neq \nu$, and $\theta(\tilde{v}_\nu, v_\mu)_{|T}$ is a

multiple of z^{μ} if $\mu = \nu$. Using furthermore Lemma 4.5.8 and using the fact that $\mathbb{C}[x^{\pm 1}]$ is the subalgebra of A(T) spanned by the monomials $z^{\mu}(=x^{\mu^{\natural}})$ ($\mu \in P_K$), we obtain from (4.6.7),

$$\begin{split} \varphi^{\sigma,\tau}(\lambda)_{|T} &= \theta\big(\tilde{v}_{\tau}(\varpi_r), v_{\sigma}(\varpi_r)\big)_{|T} \\ &= \theta\big([\tilde{v}_{\tau}(\varpi_r)], [v_{\sigma}(\varpi_r)]\big)_{|T} + \sum_{\nu \in C((1^r))} d_{\nu} x^{\nu} \\ &= d_{(1^r)} m_{(1^r)}(x) + \sum_{\nu \in C((1^r))} d_{\nu} x^{\nu} \end{split}$$

with $d_{(1^r)} = c_r \tilde{c}_r \neq 0$, since $\theta(\tilde{u}_r, u_r)|_T = m_{(1^r)}(x)$. This completes the proof of (4.6.5) for the fundamental spherical weights.

Lemma 4.6.2 has the following important consequence.

COROLLARY 4.6.3. The restriction of the map $_{|T}: A_q(U) \to A(T)$ to $\mathcal{H}^{\sigma,\tau}$ defines an injection from $\mathcal{H}^{\sigma,\tau}$ into $\mathbb{C}[x^{\pm 1}]$ for $-\infty < \sigma, \tau < \infty$. In particular, $\mathcal{H}^{\sigma,\tau}$ is a commutative algebra for $-\infty < \sigma, \tau < \infty$.

Recall from [104] and [92, Section 3] the Casimir operator

$$C := \sum_{ij} q^{2(n-i)} L_{ij}^+ S(L_{ji}^-) \in U_q(\mathfrak{g}).$$

Since C is central, it acts on $W(\lambda)$ ($\lambda \in P^+$) as a scalar $\chi_{\lambda}(C)$, which is given by

$$\chi_{\lambda} = \sum_{k=1}^{n} q^{2(\lambda_k + n - k)}.$$

Also, C maps $\mathcal{H}^{\sigma,\tau}$ into itself. Therefore, if $-\infty < \sigma, \tau < \infty$, the restricted Casimir operator $C: \mathcal{H}^{\sigma,\tau} \to \mathcal{H}^{\sigma,\tau}$ induces an operator

$$L\colon \mathcal{H}^{\sigma,\tau}|_T \to \mathcal{H}^{\sigma,\tau}|_T \subset \mathbb{C}[x^{\pm 1}],$$

which is called the radial part of C. Explicitly, L is the map which satisfies

$$L(\phi_{|T}) = (C\phi)_{|T}, \quad \forall \phi \in \mathcal{H}^{\sigma, \tau}.$$

Crucial for the identification of the zonal (σ, τ) -spherical functions is the realization of the radial part L of the Casimir element C as the restriction to $\mathcal{H}^{\sigma,\tau}|_T$ of an explicit second order q^2 -difference operator on $\mathbb{C}[x^{\pm 1}]$. Without proof we will state here the result (see [92, Section 3]).

THEOREM 4.6.4. ([92]) Let $-\infty < \sigma, \tau < \infty$ and $\lambda \in P_K^+$. The operator $L - \chi_{\lambda}(C)$ id coincides on $\mathcal{H}^{\sigma,\tau}|_T \subset \mathbb{C}[x^{\pm 1}]$ with a constant multiple of Koornwinder's second order q^2 -difference operator $D - E_{\lambda^{\natural}}$ id in the variables $x = (x_1, \ldots, x_l)$ with base

 q^2 and parameters $(\underline{t},t) = (\underline{t}^{\sigma,\tau},q^2)$, given by

(4.6.8)
$$t_0^{\sigma,\tau} = -q^{\sigma+\tau+1}, \quad t_1^{\sigma,\tau} = -q^{-\sigma-\tau+1}, \\ t_2^{\sigma,\tau} = q^{\sigma-\tau+1}, \quad t_3^{\sigma,\tau} = q^{-\sigma+\tau+2(n-2l)+1}$$

For a proof of the theorem for rank 1, see [24]. In [94] a proof can be found for the special case n = 2l and $\sigma = \tau = 0$.

Note that $\underline{t}^{\sigma,\tau} \in V_K$ for $-\infty < \sigma, \tau < \infty$ (cf. Section 2.6) and that $t_0^{\sigma,\tau} t_1^{\sigma,\tau} t_2^{\sigma,\tau} t_3^{\sigma,\tau} \in (0,1)$. In particular, the eigenvalues $E_{\lambda^{\natural}}$ are mutually different for compatible weights when $-\infty < \sigma, \tau < \infty$ (see Proposition 3.6.5).

We write $D_{\sigma,\tau}$ for Koornwinder's second order q^2 -difference operator in base q^2 with parameters $(\underline{t}^{\sigma,\tau}, q^2)$, and $E^{\sigma,\tau}_{\mu}$ ($\mu \in P^+_{\Sigma}$) for the corresponding eigenvalues. We furthermore write $P^{\sigma,\tau}_{\mu}(x) := P_{\mu}(x; \underline{t}^{\sigma,\tau}; q^2)$ ($\mu \in P^+_{\Sigma}$) for the corresponding monic Koornwinder polynomials in base q^2 . By Theorem 4.6.4, $\varphi^{\sigma,\tau}(\lambda)_{|T} \in \mathbb{C}[x^{\pm 1}]$ is an eigenfunction of $D_{\sigma,\tau}$ with eigenvalue $E^{\sigma,\tau}_{\lambda^{\natural}}$ for $\lambda \in P^+_K$. By [94, Lemma 6.2], any eigenfunction $\phi(x) \in \mathbb{C}[x^{\pm 1}]$ of $D_{\sigma,\tau}$ with eigenvalue $E^{\sigma,\tau}_{\mu}$ ($\mu \in P^+_{\Sigma}$) which is of the particular form

$$\phi(x)=c_\mu m_\mu(x)+\sum_{
u\in C(\mu)}c_
u x^
u$$

is a constant multiple of the Koornwinder polynomial $P^{\sigma,\tau}_{\mu}(x)$. Combined with Lemma 4.6.2, the following main result of the paper [**92**] is obtained.

THEOREM 4.6.5. ([92]) Let $-\infty < \sigma, \tau < \infty$. The restriction $\varphi^{\sigma,\tau}(\lambda)_{|T}$ of the zonal spherical function $\varphi^{\sigma,\tau}(\lambda) \in \mathcal{H}^{\sigma,\tau}(\lambda)$ ($\lambda \in P_K^+$) is equal to the Koornwinder polynomial $P_{\lambda^{\natural}}^{\sigma,\tau}(x)$, up to a non-zero scalar multiple. In particular, |T| defines an algebra isomorphism from $\mathcal{H}^{\sigma,\tau}$ onto $\mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$.

REMARK 4.6.6. It should be observed here that the assumption $-\infty < \sigma, \tau < \infty$ in the preceding arguments is absolutely essential. In fact, the map $|_T : A_q(U) \to A(T)$ factors through the projection $\pi_K : A_q(U) \to A_q(K)$. This implies that the image of $\mathcal{H}^{\sigma,\tau}$ under $|_T$ is one dimensional as soon as either σ or τ is infinite.

4.7. Limit transitions on quantum Grassmannians

In this section the right \mathfrak{k}^{τ} -invariant $(-\infty < \tau \leq \infty)$ functions in the quantized coordinate ring $A_q(U/K) = A_q(\mathfrak{k}^{\infty} \setminus U)$ are studied. The harmonic analysis for $\sigma = \infty$ and/or $\tau = \infty$ will be derived as limit case of the harmonic analysis for $-\infty < \sigma, \tau < \infty$, using explicit knowledge of the limit transitions from Koornwinder polynomials to multivariable big and little q-Jacobi polynomials. The rank 1 case of these results were earlier derived by Koornwinder [63] for 2-spheres and for arbitrary projective space by Noumi and Dijkhuizen [24].

For the proper interpretation of the limit transitions of the zonal spherical functions, a careful study is needed of the pre-images of the W-invariant functions $e_s(x) := m_{(1^s)}(x)$

 $(1 \leq s \leq l)$ under the isomorphism $|_T : \mathcal{H}^{\sigma,\tau} \to \mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$. For $-\infty < \sigma, \tau < \infty$, write $e_r^{\sigma,\tau}$ for the unique element in $\mathcal{H}^{\sigma,\tau}$ such that its restriction to the torus is equal to $e_r(x)$ $(1 \leq r \leq l)$. It is convenient to put $e_0(x) := 1$ and $e_0^{\sigma,\tau} := 1$. The \mathcal{W} -invariant functions $\{e_r\}_{r=1}^l$ are algebraically independent generators of $\mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$. In other words, the assignment

(4.7.1)
$$\hat{P}(e_1(x), \dots, e_l(x)) := P(x), \quad P \in \mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$$

defines an algebra isomorphism $\mathbb{C}[x^{\pm 1}]^{\mathcal{W}} \to \mathbb{C}[y]$, where $y = (y_1, \ldots, y_l)$ are l independent variables. It follows now from Theorem 4.6.5 that the elements $\{e_r^{\sigma,\tau}\}_{r=1}^l$ are algebraically independent generators of the algebra $\mathcal{H}^{\sigma,\tau}$.

Using Theorem 4.6.5 it is now easy to derive an explicit form of the restriction of the normalized Haar functional h to $\mathcal{H}^{\sigma,\tau}$. Recall that the parameters $\underline{t}^{\sigma,\tau}$ lies in the parameter domain V_K for $-\infty < \sigma, \tau < \infty$ (see Definition 2.6.1 for the definition of V_K). Let $\langle ., \rangle_{\underline{t}^{\sigma,\tau},q^2}$ be the non degenerate bilinear form for which the Koornwinder polynomials $P_{\mu}^{\sigma,\tau}(x)$ ($\mu \in P_{\Sigma}^+$) are mutually orthogonal (see Theorem 2.6.6). Write $\langle \phi \rangle_{\sigma,\tau} := \langle \phi, 1 \rangle_{\underline{t}^{\sigma,\tau},q^2}$ for the constant term of $\phi \in \mathbb{C}[x^{\pm 1}]^W$. Observe that $\langle 1 \rangle_{\sigma,\tau} = \mathcal{N}(0; \underline{t}^{\sigma,\tau}; q^2)$ is known explicitly by Gustafson's evaluation of the multidimensional Askey-Wilson integral (cf. (2.1.1)). In particular, the constant term $\langle 1 \rangle_{\sigma,\tau}$ is non-zero for all $-\infty < \sigma, \tau < \infty$.

COROLLARY 4.7.1. Let $-\infty < \sigma, \tau < \infty$. The restricted Haar functional $h : \mathcal{H}^{\sigma,\tau} \to \mathbb{C}$ is explicitly given by

$$h(\hat{P}(e_1^{\sigma,\tau}, e_2^{\sigma,\tau}, \dots, e_l^{\sigma,\tau})) = \frac{\langle P \rangle_{\sigma,\tau}}{\langle 1 \rangle_{\sigma,\tau}} \quad (P \in \mathbb{C}[x^{\pm 1}]^{\mathcal{W}}).$$

PROOF. The left and the right hand side are equal to zero for $P = P_{\mu}^{\sigma,\tau}$ with $0 \neq \mu \in P_{\Sigma}^+$, and equal to 1 for P = 1. The corollary follows now by linearity, since the Koornwinder polynomials $P_{\mu}^{\sigma,\tau}(x)$ ($\mu \in P_{\Sigma}^+$) form a linear basis of $\mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$. \Box

Recall the intertwiner

$$\widehat{\Psi}_r \colon (V \otimes V^*)^{\otimes r} \to \Lambda^r_a(V) \otimes \Lambda^r_a(V^*)$$

defined in Proposition 4.5.9. Introduce left \mathfrak{k}^{σ} -fixed vectors $w_r^{\sigma} \in \Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ and right \mathfrak{k}^{τ} -fixed vectors $\tilde{w}_r^{\tau} \in \Lambda_q^r(V)^{\circ} \otimes \Lambda_q^r(V^*)^{\circ}$ by

$$w_r^{\sigma} := \widehat{\Psi}_r \big((w^{\sigma})^{\otimes r} \big), \quad \widetilde{w}_r^{\tau} := \widehat{\Psi}_r \big((\widetilde{w}^{\tau})^{\otimes r} \big) \quad (1 \le r \le l).$$

Here we have used the notation $\tilde{w}^{\infty} := w^{\infty}$ (4.5.11) in case of $\tau = \infty$, which is compatible with the definition of \tilde{w}^{τ} for $-\infty < \tau < \infty$ since $\lim_{\tau \to \infty} \tilde{w}^{\tau} = w^{\infty}$. Consider now the (σ, τ) -spherical elements

(4.7.2)
$$\varphi_r^{\sigma,\tau} := \theta(\tilde{w}_r^{\tau}, w_r^{\sigma}) \in \bigoplus_{s=0}^r \mathcal{H}^{\sigma,\tau}(\varpi_s) \quad (1 \le r \le l)$$

(cf. (4.5.14)), where θ is the map (4.6.2) associated with the unitary module $(\Lambda_q^r(V) \otimes \Lambda_q^r(V^*), \langle ., . \rangle)$ (see (4.6.6) for the definition of $\langle ., . \rangle$). It is convenient to put $\varphi_0^{\sigma, \tau} := 1$. By

Theorem 4.6.5 and (4.7.2), $\varphi_r^{\sigma,\tau}|_T$ is a linear combination of the \mathcal{W} -invariant functions $e_s(x) \in \mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$ $(0 \le s \le r)$.

LEMMA 4.7.2. Let $-\infty < \sigma, \tau < \infty, 1 \le r \le l$. In the expansion

$$\varphi_r^{\sigma,\tau}|_T = a_r^r(q^\sigma, q^\tau)e_r + \dots + a_0^r(q^\sigma, q^\tau)e_0,$$

each coefficient a_i^r is a polynomial in q^{σ} and q^{τ} which is the sum of monomials of partial degree $\geq i$ in each of the variables. Moreover, $a_r^r(q^{\sigma}, q^{\tau}) = c_r(\sigma)\tilde{c}_r(\tau) = cq^{r\sigma+r\tau}$ with $c \neq 0$ independent of q^{σ} and q^{τ} , where $c_r(\sigma)$ and $\tilde{c}_r(\tau)$ are defined in Proposition 4.5.9.

PROOF. It is obvious from the definitions that the coefficients are polynomial in q^{σ} and q^{τ} . To prove the estimates on the partial degrees, we study the action of the intertwiner $\hat{\Psi}_r$ on the vectors $(w^{\sigma})^{\otimes r}$ and $(\tilde{w}^{\tau})^{\otimes r}$ in detail. We proceed in a number of steps.

1) Let $1 \leq i_1 \leq \cdots \leq i_r \leq n$ and $1 \leq j_1 \leq \cdots \leq j_r \leq n$ be integers. We use the shorthand notation $\underline{i} := (i_1, \ldots, i_r), \underline{j} := (j_1, \ldots, j_r)$. Call a tensor t in some tensor product space made up of factors V or V^* (the total number of factors V being equal to the total number of factors V^*) a basic tensor of type $(\underline{i}, \underline{j})$ if t is the tensor product in any given order of the vectors v_{i_1}, \ldots, v_{i_r} and $v_{j_1}^*, \ldots, v_{j_r}^*$. Let $n_k(\underline{i})$ denote the cardinality of the set $\{p \in [1, r] \mid i_p = k\}$. For a basic tensor t of type $(\underline{i}, \underline{j})$ define

$$n(t) := \sum_{k=1}^{n} \min(n_k(\underline{i}), n_k(\underline{j})).$$

From an informal point of view, n(t) is the number of factors v_i in t that "cancel" against a factor v_i^* . Recall the intertwiner $\Psi_r : (V \otimes V^*)^{\otimes r} \to V^{\otimes r} \otimes (V^*)^{\otimes r}$ defined in (4.5.20). Let t be a basic tensor in $(V \otimes V^*)^{\otimes r}$. Since Ψ_r is the composition of the intertwiner β (see (4.5.20)) it follows by inspection of (4.5.19) that $\Psi_r(t)$ is a linear combination of basic tensors t' in $V^{\otimes r} \otimes (V^*)^{\otimes r}$ with n(t') = n(t).

(2) A basic tensor $t \in (V \otimes V^*)^{\otimes r}$ is called typical if it is a product of tensors in $V \otimes V^*$ of type $v_i \otimes v_i^*$ $(1 \le i \le n-l)$, $v_i \otimes v_{i'}^*$, $v_{i'} \otimes v_i^*$ $(1 \le i \le l)$. We call a typical tensor $t \in (V \otimes V^*)^{\otimes r}$ k-typical if the number of factors of type $v_i \otimes v_{i'}^*$ $(i \in [1, l] \cup [l', n])$ is equal to k. If t is a k-typical tensor then $\widehat{\Psi}_r(t)$ is a linear combination of elements $v_I \otimes v_J^*$ where $I, J \subset [1, n]$ are such that |I| = |J| = r and $|I \cap J| \ge r - k$. In fact, this follows from (1) and the definition of $\widehat{\Psi}_r$, since $n(t) \ge r - k$.

(3) It is an immediate consequence of the definition of the coactions on $\Lambda_q^r(V)$ and $\Lambda_q^r(V^*)$ and of (4.3.19) that

$$\theta(v_I \otimes v_J^*, v_K \otimes v_L^*)|_T = q^{-\langle 2\rho, \tilde{\varepsilon}_J \rangle} \delta_{I,K} \delta_{J,L} z^{\tilde{\varepsilon}_I - \tilde{\varepsilon}_J},$$

for $I, J \subset [1, n]$ with |I| = |J| = r.

(4) Let t be a k-typical tensor and t' a m-typical tensor. Fix a weight $\mu \in \mathcal{W}(1^i) \subset P_{\Sigma}$ $(i \in [1, r])$ and suppose that the coefficient of $z^{\lambda_{\mu}}$ is non-zero in the expansion of $\theta(\widehat{\Psi}_r(t), \widehat{\Psi}_r(t'))_{|T}$ with respect to the basis $\{z^{\lambda}\}_{\lambda \in P}$ of A(T). Then, $k \geq i$ and $m \geq i$. This is a straightforward consequence of (2) and (3).

4. HARMONIC ANALYSIS ON QUANTUM GRASSMANNIANS

(5) There is a unique expansion $(w^{\sigma})^{\otimes r} = \sum_{k=0}^{r} \sum_{t_k} c_{t_k} t_k$ where t_k runs over all k-typical tensors in $(V \otimes V^*)^{\otimes r}$ and where the $c_{t_k} \neq 0$ are linear combinations of monomials $(q^{\sigma})^i$ with $i \geq k$. Similarly, there is a unique expansion $(\tilde{w}^{\tau})^{\otimes r} =$ $\sum_{k=0}^{r} \sum_{t'_k} d_{t'_k} t'_k$ where t_k runs over all k-typical tensors in $(V \otimes V^*)^{\otimes r}$ and where the $d_{t'_k} \neq 0$ are linear combinations of monomials $(q^{\tau})^i$ with $i \geq k$. Hence,

$$\varphi_r^{\sigma,\tau}|_T = \sum_{k,m=0}^r \sum_{t_k,t'_m} c_{t_kt'_m} \theta(\widehat{\Psi}_r(t'_m),\widehat{\Psi}_r(t_k))|_T$$

with $c_{t_k t'_m} = c_{t_k} \overline{d_{t'_m}}$ a linear combination of monomials $(q^{\sigma})^i (q^{\tau})^j$ with $i \ge k$ and $j \ge m$. Combined with (4) this yields the desired lower bounds on the partial degrees of the monomials $(q^{\sigma})^i (q^{\tau})^j$ occurring in $a_i^r (q^{\sigma}, q^{\tau})$. The explicit expression for $a_r^r (q^{\sigma}, q^{\tau})$ follows immediately from Proposition 4.5.9.

As a corollary we obtain the following crucial lemma.

LEMMA 4.7.3. Let
$$-\infty < \tau < \infty$$
. The limits
$$\lim_{\sigma \to \infty} q^{r\sigma} e_r^{\sigma,\tau}, \quad \lim_{\sigma \to \infty} q^{2r\sigma} e_r^{\sigma,\sigma} \quad (1 \le r \le l)$$

exist in $A_q(U)$. In other words, the coefficients of $q^{r\sigma}e_r^{\sigma,\tau}$ respectively $q^{2r\sigma}e_r^{\sigma,\sigma}$ in the expansion with respect to the monomial basis of $A_q(U)$ tend to finite values in the limit $\sigma \to \infty$.

PROOF. Fix $1 \le r \le l$ and let $-\infty < \sigma, \tau < \infty$. From Lemma 4.7.2 it is readily deduced that

$$q^{r\sigma+r\tau}e_r = b_r^r(q^\sigma, q^\tau)\varphi_r^{\sigma,\tau}|_T + \dots + b_0^r(q^\sigma, q^\tau)\varphi_0^{\sigma,\tau}|_T,$$

with b_i^r ($0 \le i \le r$) some polynomial in two variables and b_r^r a non-zero constant polynomial (the important fact here is that b_i^r is a polynomial and not a Laurent polynomial). Hence

$$(4.7.3) \qquad e_r^{\sigma,\tau} = q^{-r\sigma-r\tau} \left(b_r^r (q^\sigma, q^\tau) \varphi_r^{\sigma,\tau} + \dots + b_0^r (q^\sigma, q^\tau) \varphi_0^{\sigma,\tau} \right)$$

Since $\varphi_i^{\sigma,\tau} \to \varphi_i^{\infty,\tau}$ and $\varphi_i^{\sigma,\sigma} \to \varphi_i^{\infty,\infty}$ when $\sigma \to \infty$, the lemma follows.

In view of Lemma 4.7.3 we may set for $1 \le r \le l$ and $-\infty < \tau < \infty$,

$$(4.7.4) \quad \tilde{e}_r^{\infty,\tau} := \lim_{\sigma \to \infty} q^{r(\sigma+\tau-1)} (-1)^r e_r^{\sigma,\tau}, \quad \tilde{e}_r^{\infty,\infty} := \lim_{\sigma \to \infty} q^{r(2\sigma-1)} (-1)^r e_r^{\sigma,\sigma}.$$

It is clear from the definitions that $\tilde{e}_r^{\infty,\tau} \in \mathcal{H}^{\infty,\tau}$ $(1 \leq r \leq l, -\infty < \tau \leq \infty)$. Observe that the elements $(-1)^r e_r^{\sigma,\tau}$ are mapped onto $e_r(-x) \in \mathbb{C}[x^{\pm 1}]^{\mathcal{W}}$ under the restriction mapping $|_T$. It will be shown later on that the limit transitions (4.7.4) on the level of quantum Grassmannians turn out to be the proper analogue of the limit transitions

(4.7.5)
$$\lim_{u \to 0} u^r e_r(u^{-1}y) = \tilde{e}_r(y), \quad (1 \le r \le l)$$

 $(y = (y_1, \ldots, y_l))$, where $\tilde{e}_r(y) = \tilde{m}_{(1^r)}(y) \in \mathbb{C}[y]^{\mathfrak{S}}$ is the *r*th elementary symmetric polynomial $(1 \leq r \leq l)$. It was shown in the previous chapter that the limit transitions

(4.7.5) play a crucial role for the limits from Koornwinder polynomials to multivariable big and little *q*-Jacobi polynomials.

The elements \tilde{e}_r $(1 \leq r \leq l)$ are algebraically independent generators of the algebra $\mathbb{C}[x]^{\mathfrak{S}}$. In other words, we have an algebra isomorphism $\mathbb{C}[x]^{\mathfrak{S}} \to \mathbb{C}[y]$ which will be denoted in the same way as the algebra isomorphism $\mathbb{C}[x^{\pm 1}]^{\mathcal{W}} \to \mathbb{C}[y]$ (4.7.1):

$$(4.7.6) \qquad \qquad P(\tilde{e}_1(x),\ldots,\tilde{e}_l(x)) := P(x), \quad (P \in \mathbb{C}[x]^{\mathfrak{S}}).$$

THEOREM 4.7.4. Let $-\infty < \tau < \infty$. The elements $\tilde{e}_r^{\infty,\tau}$ $(1 \le r \le l)$ mutually commute and are algebraically independent generators of the algebra $\mathcal{H}^{\infty,\tau}$. Any zonal (∞, τ) -spherical function $\varphi^{\infty,\tau}(\lambda) \in \mathcal{H}^{\infty,\tau}(\lambda)$ $(\lambda \in P_K^+)$ is equal to a non-zero scalar multiple of

$$\hat{P}^B_{\lambda^{\natural}}(\tilde{e}_1^{\infty,\tau},\ldots,\tilde{e}_l^{\infty,\tau};1,q^{2(n-2l)},1,q^{2\tau+2(n-2l)};q^2),$$

where $P^B_{\mu}(.; a, b, c, d; t)$ is the multivariable big q-Jacobi polynomial in base q^2 .

PROOF. The elements $e_r^{\sigma,\tau}$ $(1 \le r \le l)$ mutually commute for σ finite by Corollary 4.6.3. Combined with the definition of the elements $\tilde{e}_r^{\infty,\tau}$ (4.7.4), it follows that the $\tilde{e}_r^{\infty,\tau}$ $(1 \le r \le l)$ mutually commute. Hence the element $Q(\tilde{e}_1^{\infty,\tau},\ldots,\tilde{e}_l^{\infty,\tau}) \in \mathcal{H}^{\infty,\tau}$ for a polynomial $Q \in \mathbb{C}[y]$ is well defined.

For σ finite and $\lambda \in P_K^+$, a zonal (σ, τ) -spherical function $\varphi^{\sigma, \tau}(\lambda) \in \mathcal{H}^{\sigma, \tau}(\lambda)$ is given by

(4.7.7)
$$\varphi^{\sigma,\tau}(\lambda) := (-q^{\sigma+\tau-1})^{|\lambda^{\mathfrak{h}}|} \hat{P}_{\lambda^{\mathfrak{h}}}(e_{1}^{\sigma,\tau},\ldots,e_{l}^{\sigma,\tau};\underline{t}^{\sigma,\tau};q^{2})$$

with $P_{\mu}(.; \underline{t}; t)$ the Koornwinder polynomial in base q^2 (cf. Theorem 4.6.5). Using the elementary properties of the Koornwinder polynomials given in Remark 2.3.11, this zonal (σ, τ) -spherical function can be rewritten as

$$\varphi^{\sigma,\tau}(\lambda) = (s_{\varepsilon})^{-|\lambda^{\natural}|} \hat{P}_{\lambda^{\natural}} (s_{\varepsilon} f_{1}^{\varepsilon}, \dots, s_{\varepsilon}^{l} f_{l}^{\varepsilon}; \underline{t}_{B}(\varepsilon); q^{2}),$$

where $s_{\varepsilon} := q/\varepsilon(cd)^{\frac{1}{2}}, f_r^{\varepsilon} := (-s_{\varepsilon})^{-r} e_r^{\sigma,\tau}, \varepsilon := q^{\sigma-(n-2l)}$ and

$$\underline{t}_B(\varepsilon) = \left(\varepsilon^{-1}(q^2c/d)^{\frac{1}{2}}, -\varepsilon^{-1}(q^2d/c)^{\frac{1}{2}}, \varepsilon a(q^2d/c)^{\frac{1}{2}}, -\varepsilon b(q^2c/d)^{\frac{1}{2}}\right),$$

with parameters a, b, c, d given by

$$(4.7.8) a := 1, \ b := q^{2(n-2l)}, \ c := 1, \ d := q^{2\tau + 2(n-2l)}.$$

Observe that $s_{\varepsilon} = q^{1-\sigma-\tau}$, hence by the definition (4.7.4) of $\tilde{e}_r^{\infty,\tau}$, $\lim_{\varepsilon \downarrow 0} f_r^{\varepsilon} = \tilde{e}_r^{\infty,\tau}$ for all r. Combined with the limit transition from Koornwinder polynomials to multivariable big q-Jacobi polynomials (cf. Corollary 3.4.6), we have that $\varphi^{\infty,\tau}(\lambda) := \lim_{\sigma \to \infty} \varphi^{\sigma,\tau}(\lambda)$ exists as limit in $A_q(U)$, and that

(4.7.9)
$$\varphi^{\infty,\tau}(\lambda) = \hat{P}^B_{\lambda^{\natural}}(\tilde{e}_1^{\infty,\tau},\ldots,\tilde{e}_l^{\infty,\tau};1,q^{2(n-2l)},1,q^{2\tau+2(n-2l)};q^2)$$

with $P^B_{\mu}(.; a, b, c, d; t)$ the big q-Jacobi polynomial in base q^2 . It is clear that $\varphi^{\infty,\tau}(\lambda) \in \mathcal{H}^{\infty,\tau}(\lambda)$, but it may be zero since the algebraic independence of the elements $\tilde{e}_r^{\infty,\tau}$ ($r \in [1, l]$) is not yet established. To prove that $\varphi^{\infty,\tau}(\lambda)$ is non-zero, we compute the quadratic norm $\|\varphi^{\sigma,\tau}(\lambda)\|^2$ with respect to the inner product $\langle \phi, \psi \rangle := h(\psi^*\phi)$, where h is the

normalized Haar functional. Since all highest weights $\lambda \in P_K^+$ are self dual (i.e. $V(\lambda)$ is isomorphic to its dual representation), and since the two-sided coideal \mathfrak{k}^{σ} is τ -invariant, we have $(\varphi^{\sigma,\tau}(\lambda))^* = \varphi^{\sigma,\tau}(\lambda)$. Then it follows from the definition (4.7.7) of $\varphi^{\sigma,\tau}(\lambda)$, Corollary 4.7.1 and Theorem 2.6.6 that

(4.7.10)
$$\|\varphi_{\varepsilon}^{\sigma,\tau}(\lambda)\|^2 = h\big((\varphi^{\sigma,\tau}(\lambda))^2\big) = s_{\varepsilon}^{-2|\lambda^{\natural}|} \frac{\mathcal{N}(\lambda^{\natural};\underline{t}_B(\varepsilon);q^2)}{\mathcal{N}(0;\underline{t}_B(\varepsilon);q^2)},$$

where $\mathcal{N}(\mu; \underline{t}; t)$ for $\underline{t} \in V_K$ and 0 < t < 1 is the quadratic norm of the Koornwinder polynomial $P_{\mu}(.; \underline{t}; t)$ in base q^2 with respect to the corresponding inner product $\langle ., . \rangle_{\underline{t}, t}$ (cf. Theorem 2.6.6). The limit $\varepsilon \downarrow 0$ (equivalently, $\sigma \to \infty$) in (4.7.10) can now be computed in the left hand side and in the right hand side (cf. the proof of Theorem 3.4.5). It follows that

$$\|\varphi^{\infty,\tau}(\lambda)\|^2 = \frac{\mathcal{N}^B(\lambda^{\natural}; 1, q^{2(n-2l)}, 1, q^{2\tau+2(n-2l)}; q^2)}{\mathcal{N}^B(0; 1, q^{2(n-2l)}, 1, q^{2\tau+2(n-2l)}; q^2)}$$

where $\mathcal{N}^B(\mu; a, b, c, d; t)$ is the quadratic norm of the multivariable big q-Jacobi polynomial $P^B_{\mu}(.; a, b, c, d; t)$ in base q^2 (see Theorem 3.4.5). It follows in particular that the quadratic norm $\|\varphi^{\infty, \tau}(\lambda)\|^2$ is non-zero, hence $\varphi^{\infty, \tau}(\lambda) \neq 0$ for all $\lambda \in P^+_K$. Hence the elements $\varphi^{\infty, \tau}(\lambda) \in \mathcal{H}^{\infty, \tau}(\lambda)$ are zonal (∞, τ) -spherical functions for all $\lambda \in P^+_K$.

It remains to prove that the $\tilde{e}_r^{\infty,\tau}$ $(1 \le r \le l)$ are algebraically independent. Consider the finite dimensional subspaces

$$\mathcal{H}_m := \bigoplus_{\lambda \in P_K^+ : \lambda \le m \varpi_l} \mathcal{H}^{\infty, \tau}(\lambda), \quad (m \in \mathbb{Z}_+).$$

The dimension of the linear subspace \mathcal{H}_m is equal to the number of positive integers $\underline{m} = (m_1, \ldots, m_l) \in \mathbb{Z}_+^{\times l}$ with $|\underline{m}| := \sum_i m_i \leq m$, since $\varpi_r \leq \varpi_l$ for all $r \in [0, l]$. For such a sequence of positive integers \underline{m} , set $Q_{\underline{m}}(y) := y_1^{m_1} \ldots y_l^{m_l}$. Since $\tilde{e}_r^{\infty, \tau} \in \bigoplus_{s=0}^r \mathcal{H}^{\infty, \tau}(\varpi_s)$ and

$$\mathcal{H}^{\infty,\tau}(\mu).\mathcal{H}^{\infty,\tau}(\mu') \subseteq \bigoplus_{\nu \in P_K^+: \nu \le \mu + \mu'} \mathcal{H}^{\infty,\tau}(\nu), \quad (\mu,\mu' \in P_K^+)$$

we have $Q_{\underline{m}}(\tilde{e}_1^{\infty,\tau},\ldots,\tilde{e}_l^{\infty,\tau}) \in \mathcal{H}_m$ for all \underline{m} with $|\underline{m}| \leq m$. Hence the algebraic independence of $\tilde{e}_r^{\infty,\tau}$ $(1 \leq r \leq l)$ will follow from the fact that $Q_{\underline{m}}(\tilde{e}_1^{\infty,\tau},\ldots,e_l^{\infty,\tau})$ $(|\underline{m}| \leq m)$ span \mathcal{H}_m for all $m \in \mathbb{Z}_+$.

Observe that \mathcal{H}_m is spanned by the zonal spherical functions $\varphi^{\infty,\tau}(\lambda)$ $(\lambda \leq m\omega_l)$. Since $P^B_{\mu}(x)$ $(\mu \in P^+_{\Sigma})$ is of the form $\tilde{m}_{\mu}(x) + \sum_{\nu \in P^+_{\Sigma}: \nu < \mu} c_{\nu} \tilde{m}_{\nu}(x)$ for certain constants c_{ν} , it follows from the explicit expression (4.7.9) for $\varphi^{\infty,\tau}(\lambda)$ that each $\varphi^{\infty,\tau}(\lambda)$ with $\lambda \leq m\omega_l$ can be written as a linear combination of $Q_{\underline{m}}(\tilde{e}_1^{\infty,\tau},\ldots,\tilde{e}_l^{\infty,\tau})$ $(|\underline{m}| \leq m)$. Hence, the monomials $Q_{\underline{m}}(\tilde{e}_1^{\infty,\tau},\ldots,\tilde{e}_l^{\infty,\tau})$ $(|\underline{m}| \leq m)$ span \mathcal{H}_m .

THEOREM 4.7.5. The elements $\tilde{e}_r^{\infty,\infty}$ $(1 \leq r \leq l)$ mutually commute and are algebraically independent generators of the algebra $\mathcal{H}^{\infty,\infty}$. Any zonal spherical function

$$\varphi^{\infty,\infty}(\lambda)$$
 $(\lambda \in P_K^+)$ is equal to a non-zero scalar multiple of

$$P^L_{\lambda^{\natural}}(\tilde{e}_1^{\infty,\infty},\ldots,\tilde{e}_l^{\infty,\infty};q^{2(n-2l)},1;q^2)$$

where $P_{\mu}^{L}(.; a, b; t)$ is the multivariable little q-Jacobi polynomial in base q^{2} .

PROOF. By Theorem 4.6.5 and by symmetry properties of the Koornwinder polynomials (cf. Remark 2.3.11), a zonal (σ, σ) -spherical function $\varphi^{\sigma,\sigma}(\lambda) \in \mathcal{H}^{\sigma,\sigma}(\lambda)$ $(\lambda \in P_K^+)$ is explicitly given by

$$\varphi^{\sigma,\sigma}(\lambda) := (s_{\varepsilon})^{-|\lambda^{\mathfrak{q}}|} \hat{P}_{\lambda^{\mathfrak{q}}}(s_{\varepsilon}f_{1}^{\varepsilon},\ldots,s_{\varepsilon}^{l}f_{l}^{\varepsilon};\underline{t}_{L}(\varepsilon);q^{2}),$$

where $P_{\mu}(.;\underline{t};t)$ is the Koornwinder polynomial in base q^2 and $\varepsilon := q^{2\sigma}$, $s_{\varepsilon} := q/\varepsilon$, $f_r^{\varepsilon} := (-s_{\varepsilon})^{-r} e_r^{\sigma,\sigma}$, and $\underline{t}_L(\varepsilon) := (\varepsilon^{-1}q, -aq, \varepsilon bq, -q)$ with $a := q^{2(n-2l)}$ and b := 1. Using Corollary 3.3.4 together with the observation that $\lim_{\varepsilon \downarrow 0} f_r^{\varepsilon} = \tilde{e}_r^{\infty,\infty}$, the proof is analogous to the proof of Theorem 4.7.4.

REMARK 4.7.6. Using the limits $\tilde{e}_r^{\infty,\infty} = \lim_{\tau\to\infty} \tilde{e}_r^{\infty,\tau}$ $(1 \le r \le l)$, Theorem 4.7.5 can also be proved by sending $\tau \to \infty$ in the results of Theorem 4.7.4. On the level of multivariable orthogonal polynomials this limit corresponds with the limit from multivariable big *q*-Jacobi polynomials to multivariable little *q*-Jacobi polynomials (3.6.25).

REMARK 4.7.7. As a corollary of Theorem 4.7.4 and Theorem 4.7.5, the restricted Haar functional $h : \mathcal{H}^{\sigma,\tau} \to \mathbb{C}$ for $\sigma = \infty, -\infty < \tau < \infty$ respectively for $\sigma = \tau = \infty$ can be expressed in terms of the orthogonality measure of the multivariable big respectively little *q*-Jacobi polynomials, similarly as was proved for the case $-\infty < \sigma, \tau < \infty$ in Corollary 4.7.1.

In the last two sections we have interpreted for $-\infty < \sigma, \tau < \infty, \mu \in P_K^+$ the Koornwinder, the multivariable big and the multivariable little *q*-Jacobi polynomial

(4.7.11)
$$P_{\mu}(.;-q^{\sigma+\tau+1},-q^{-\sigma-\tau+1},q^{\sigma-\tau+1},q^{-\sigma+\tau+2(n-2l)+1};q^{2}),$$
$$P_{\mu}^{B}(.;1,q^{2(n-2l)},1,q^{2\tau+2(n-2l)};q^{2}),$$
$$P_{\mu}^{L}(.;q^{2(n-2l)},1;q^{2})$$

in base q^2 as zonal spherical function on quantum analogues of the complex Grassmannian. For each of these polynomials, the limit $q \uparrow 1$ can be computed using the limit transitions (3.6.11), (3.6.9) respectively (3.6.10). The limits can be written in terms of the generalized Jacobi polynomial $P_{\mu}^{J}(.; n - 2l, 0; 1)$ or, equivalently, in terms of the BC type Heckman-Opdam polynomial $P_{\mu}^{HO}(.; n - 2l, 1, 1/2)$. This corresponds nicely with the classical interpretation of the Heckman-Opdam polynomial $P_{\mu}^{HO}(.; n - 2l, 1, 1/2)$ as zonal spherical function on the complex Grassmannian U/K (see Section 4.2).

· · · · ·

CHAPTER 5

Quantum Plücker coordinates

5.1. Introduction

In this chapter the algebraic properties of the quantized algebra $A_q(U/K)$ of regular functions on the complex Grassmannian

$$U/K = U(n)/(U(n-l) \times U(l)) \simeq SU(n)/S(U(n-l) \times U(l)), \quad (l \le [n/2])$$

are studied in terms of quantum analogues of Plücker coordinates. The Plücker coordinates on the complex Grassmannian U/K were discussed in detail in Section 1.3. For the generalization to the quantum setting it is convenient to consider the Plücker coordinates from a representation theoretic viewpoint as follows.

The Dynkin diagram of K is obtained from the Dynkin diagram U by deleting its (n-l)th node. Consider the irreducible finite dimensional U-highest weight representation $M := V((1^{n-l}))$ corresponding to the fundamental weight (1^{n-l}) of the deleted node and let $v \in M$ be a highest weight vector. Then $K \subset U$ is the sub-group of U which stabilizes the line $\overline{v} \in \mathbb{P}(M)$. This induces an embedding of the homogeneous space U/K into $\mathbb{P}(M)$, called the Plücker embedding. Consider the linear dual M^* as functions on U by the embedding $\phi \mapsto \phi(v)$ ($\phi \in M^*$), and let $\{\phi_i\}_i$ be a basis of M^* consisting of weight vectors. Since the weight spaces are one dimensional, the basis elements are unique up to rescaling. Then, the elements $\phi_i(.v)$ are the holomorphic *Plücker coordinates* on U/K. The holomorphic Plücker coordinates are sections of a particular line bundle over U/K. From a physical point of view, the holomorphic Plücker coordinates correspond to coherent states on the complex Grassmannian U/K (for the terminology, see Perelomov [100]). The connection of this description of the holomorphic Plücker coordinates with the description of Plücker coordinates in Section 1.3 can be made using the explicit realization of M as the (n - l)th graded piece of the exterior algebra on \mathbb{C}^n . The holomorphic Plücker coordinates correspond then to the dual Plücker coordinates of Section 1.3.

This construction of holomorphic Plücker coordinates can be immediately generalized to the quantum setting (cf. [111], [52]). Indeed, consider the irreducible finite dimensional $U_q(\mathfrak{g})$ -representation $M_q := V((1^{n-l}))$ of highest weight (1^{n-l}) , and let v be a highest weight vector. Furthermore, let $\{\phi_i\}_i$ be a basis of weight vectors of the linear dual M_q^* . Then, the elements $\phi_i(.v)$ are called the quantum holomorphic Plücker coordinates on U/K or, in the terminology of [52], quantum coherent states on U/K. The quantum Plücker coordinates $\phi_i(.v)$ lie in the linear dual of $U_q(\mathfrak{g})$. Using the doubly non-degenerate Hopf-*-algebra pairing between $U_q(\mathfrak{g})$ and $A_q(U)$, they may considered as elements in $A_q(U)$. Products of the form $\phi_i(.v)(\phi_j(.v))^*$ lie in $A_q(U/K)$. The elements $(\phi_j(.v))^*$ will be called the quantum anti-holomorphic Plücker coordinates on U/K. The quantum holomorphic and anti-holomorphic Plücker coordinates can be expressed in terms of quantum minors using the realization of M_q as the (n-l)th graded piece of the quantum analogue of the exterior algebra on \mathbb{C}^n .

As was shown in the previous chapter, the modules $W_r := \Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ $(r \in [1, l])$ play a crucial role in the analysis of $A_q(U/K)$. In fact, the irreducible decomposition of W_r is multiplicity free and the irreducible representations it contains are exactly all copies of the fundamental spherical representations $V(\varpi_s)$ $(s \in [0, r])$.

In Section 5.2 the copy of $V(\varpi_r)$ within W_r will be realized as the kernel of an explicitly defined surjective intertwiner $\Omega_{r,r} : W_r \to W_{r-1}$. This intertwiner may be considered as a *q*-analogue of the contraction map. The contraction map $\Omega_{r,r}$ is a type of lowering operator, in the sense that it "removes" the highest fundamental spherical part of W_r .

In the previous chapter, a surjective intertwiner $\widehat{\Psi}_r : W_1^{\otimes r} \to W_r$ (cf. Proposition 4.5.9) was constructed. This map will be used in this chapter as a type of raising operator. Using the lowering and raising operator, it is shown in Section 5.3 that there are 2l sets of elements which each generate the algebra $A_q(U/K)$. More precisely, for each fundamental spherical weight ϖ_r ($r \in [1, l]$), two sets of generators exist which can be realized as matrix elements of the module W_r . One set of generators corresponding to the largest module W_l is exactly the set consisting of products $\phi_i(.v)(\phi_j(.v))^*$ of quantum holomorphic and anti-holomorphic Plücker coordinates on U/K.

In Section 5.4 it is shown that the algebra generated by the quantum anti-holomorphic Plücker coordinates is exactly the subalgebra of $A_q(U(n-l)) \otimes A_q(SU(l))$ -fixed elements in $A_q(\operatorname{Mat}(n, \mathbb{C}))$ and that the algebra generated by the quantum holomorphic and anti-holomorphic Plücker coordinates is exactly the subalgebra of $A_q(U(n-l)) \otimes A_q(SU(l))$ -fixed elements in $A_q(U(n))$.

The algebraic structure of the algebra generated by the quantum anti-holomorphic Plücker coordinates was clarified by Taft and Towber [126]. In Section 5.5, their results are shortly reviewed. We replace certain arguments from [126] by representation theoretic arguments from [96].

In Section 5.6 the algebra generated by the quantum holomorphic and anti-holomorphic Plücker coordinates is studied in detail. A linear basis of the algebra in terms of "straightened" monomials of quantum Plücker coordinates is constructed, and defining relations between the quantum holomorphic and anti-holomorphic Plücker coordinates are given. The algebra $A_q(U/K)$ can be extracted as the subalgebra of "zero-weighted" elements, i.e. it is the subalgebra spanned by straightened monomials for which the number of occurrences of quantum holomorphic Plücker coordinates is equal to the number of occurrences of quantum anti-holomorphic Plücker coordinates.

In Section 5.7, the subalgebra of bi- $A_q(K)$ -fixed elements is studied in terms of the quantum Plücker coordinates.

5.2. THE CONTRACTION MAP

5.2. The contraction map

In the following proposition the q-analogue of the contraction map is defined. Recall the construction of the right $A_q(U)$ -comodules $\Lambda_q^r(V)$ and $\Lambda_q^r(V^*)$ in Section 4.3.

PROPOSITION 5.2.1. Fix $1 \le r, s \le n$. The linear mapping

$$\Omega_{r,s} \colon \Lambda^r_q(V) \otimes \Lambda^s_q(V^*) \to \Lambda^{r-1}_q(V) \otimes \Lambda^{s-1}_q(V^*)$$

defined by

$$\Omega_{r,s}(v_I \otimes v_J^*) := \sum_{i \in I \cap J} \frac{\mathrm{s}_{\mathbf{q}}(i; J^c)}{\mathrm{s}_{\mathbf{q}}(I^c; i)} v_{I \setminus i} \otimes v_{J \setminus i}^* \quad (|I| = r, |J| = s)$$

is an intertwining operator of right $A_q(U)$ -comodules.

PROOF. Denote R for the comodule mapping on $\Lambda_q^r(V) \otimes \Lambda_q^s(V^*)$ as well as for the comodule mapping on $\Lambda_q^{r-1}(V) \otimes \Lambda_q^{s-1}(V^*)$. Then, on the one hand, we have

$$R \circ \Omega_{r,s}(v_I \otimes v_J^*) = \sum_{\substack{|K|=r-1\\|L|=s-1}} v_K \otimes v_L^* \otimes \sum_{i \in I \cap J} \frac{\mathbf{s}_{\mathbf{q}}(i;J^c)}{\mathbf{s}_{\mathbf{q}}(I^c;i)} \xi_{I \setminus i}^K (\xi_{J \setminus i}^L)^*,$$

and, on the other hand,

$$(\Omega_{r,s} \otimes \mathrm{id}) \circ R(v_I \otimes v_J^*) = \sum_{\substack{|K|=r-1\\|L|=s-1}} v_K \otimes v_L^* \otimes \sum_{k \in K^c \cap L^c} \frac{\mathrm{s}_{\mathrm{q}}(k; L^c \setminus k)}{\mathrm{s}_{\mathrm{q}}(K^c \setminus k; k)} \xi_I^{K \cup k} (\xi_J^{L \cup k})^*.$$

Hence it suffices to prove that for |K| = r - 1 and |L| = s - 1,

(5.2.1)
$$\sum_{i\in I\cap J}\frac{\mathrm{s}_{\mathbf{q}}(i;J^c)}{\mathrm{s}_{\mathbf{q}}(I^c;i)}\xi_{I\setminus i}^K(\xi_{J\setminus i}^L)^* = \sum_{k\in K^c\cap L^c}\frac{\mathrm{s}_{\mathbf{q}}(k;L^c\setminus k)}{\mathrm{s}_{\mathbf{q}}(K^c\setminus k;k)}\xi_I^{K\cup k}(\xi_J^{L\cup k})^*.$$

Formula (5.2.1) will be proved by applying twice the Laplace expansions for quantum minors (see Section 4.3). First observe that the *q*-signum s_q has the following elementary properties:

(5.2.2)
$$\begin{aligned} & s_{q}(I_{1} \cup I_{2}; J) = s_{q}(I_{1}; J) \, s_{q}(I_{2}; J) & \text{if } I_{1} \cap I_{2} = \emptyset, \\ & s_{q}(I; J_{1} \cup J_{2}) = s_{q}(I; J_{1}) \, s_{q}(I; J_{2}) & \text{if } J_{1} \cap J_{2} = \emptyset, \\ & s_{q}(I; J) \, s_{q}(J; I) = (-q)^{|I||J|} & \text{if } I \cap J = \emptyset. \end{aligned}$$

From (5.2.2) it follows that for $i \in I \cap J$, |I| = r, |J| = s,

(5.2.3)
$$\frac{\operatorname{s}_{\mathbf{q}}(J\setminus i; J^{c}\cup i)}{\operatorname{s}_{\mathbf{q}}(I^{c}; i)} = (-q)^{s-n} \operatorname{s}_{\mathbf{q}}(J; J^{c}) \operatorname{s}_{\mathbf{q}}(I\setminus i; i)$$

and for $k \in K^c \cap L^c$, |K| = r - 1, |L| = s - 1,

(5.2.4)
$$\operatorname{s}_{q}(K^{c} \setminus k; k)^{-1} = (-q)^{s-n} \frac{\operatorname{s}_{q}(K; k) \operatorname{s}_{q}(L \cup k; L^{c} \setminus k)}{\operatorname{s}_{q}(L; L^{c})}.$$

5. QUANTUM PLÜCKER COORDINATES

Observe that (5.2.4) follows by taking $I := K \cup k$, $J := L \cup k$ and i := k in (5.2.3). Now we compute,

$$\sum_{i \in I \cap J} \frac{\mathbf{s}_{\mathbf{q}}(i; J^{c})}{\mathbf{s}_{\mathbf{q}}(I^{c}; i)} \xi_{I \setminus i}^{K} (\xi_{J \setminus i}^{L})^{*} =$$

$$= \sum_{i \in I \cap J} \frac{\mathbf{s}_{\mathbf{q}}(J \setminus i; J^{c} \cup i) \mathbf{s}_{\mathbf{q}}(i; J^{c})}{\mathbf{s}_{\mathbf{q}}(I^{c}; i) \mathbf{s}_{\mathbf{q}}(L; L^{c})} \xi_{I \setminus i}^{K} \xi_{i \cup J^{c}}^{L^{c}} \det_{q}^{-1}$$

$$= \sum_{k \in L^{c}} \frac{\mathbf{s}_{\mathbf{q}}(k; L^{c} \setminus k)}{\mathbf{s}_{\mathbf{q}}(L; L^{c})} \left(\sum_{i \in I \cap J} \frac{\mathbf{s}_{\mathbf{q}}(J \setminus i; J^{c} \cup i)}{\mathbf{s}_{\mathbf{q}}(I^{c}; i)} \xi_{I \setminus i}^{K} t_{ki} \right) \xi_{J^{c}}^{L^{c} \setminus k} \det_{q}^{-1}$$

the first equality following by (4.3.19), the second by the Laplace expansion (4.3.17),

$$= (-q)^{s-n} \frac{\mathrm{s}_{\mathbf{q}}(J; J^c)}{\mathrm{s}_{\mathbf{q}}(L; L^c)} \sum_{k \in L^c} \mathrm{s}_{\mathbf{q}}(k; L^c \setminus k) \Big(\sum_{i \in I \cap J} \mathrm{s}_{\mathbf{q}}(I \setminus i; i) \xi_{I \setminus i}^K t_{ki} \Big) \xi_{J^c}^{L^c \setminus k} \det_q^{-1}$$

by (5.2.3). Now, again by (4.3.17) it follows that $\sum_{k \in L^c} s_q(k; L^c \setminus k) t_{ki} \xi_{J^c}^{L^c \setminus k} = 0$ if $i \in J^c$. Hence, in the last line of the above computation, the sum may be taken over $i \in I$ instead of over $i \in I \cap J$. Moreover, by the Laplace expansion (4.3.18) it follows that

$$\sum_{i \in I} \operatorname{s}_{q}(I \setminus i; i) \xi_{I \setminus i}^{K} t_{ki} = \begin{cases} \operatorname{s}_{q}(K; k) \xi_{I}^{K \cup k} & \text{if } k \in K^{c}, \\ 0 & \text{if } k \in K. \end{cases}$$

Hence,

$$\sum_{i \in I \cap J} \frac{\mathbf{s}_{\mathbf{q}}(i; J^c)}{\mathbf{s}_{\mathbf{q}}(I^c; i)} \xi_{I \setminus i}^K (\xi_{J \setminus i}^L)^* = \\ = (-q)^{s-n} \frac{\mathbf{s}_{\mathbf{q}}(J; J^c)}{\mathbf{s}_{\mathbf{q}}(L; L^c)} \sum_{k \in K^c \cap L^c} \mathbf{s}_{\mathbf{q}}(k; L^c \setminus k) \, \mathbf{s}_{\mathbf{q}}(K; k) \xi_I^{K \cup k} \xi_{J^c}^{L^c \setminus k} \, \det_q^{-1} \,.$$

Now (5.2.1) follows by applying (4.3.19) and (5.2.4).

Throughout this chapter, let W_r be the right $A_q(U)$ -comodule

(5.2.5)
$$W_r := \Lambda_a^r(V) \otimes \Lambda_a^r(V^*) \quad (1 \le r \le l).$$

By the previous proposition, we have an explicit intertwiner $\Omega_{r,r}: W_r \to W_{r-1}$ of right $A_q(U)$ -comodules. Our next objective is to prove that $\Omega_{r,r}$ is surjective. To establish this result, we study the action of $\Omega_{r,r}$ on $A_q(K)$ -fixed vectors. In the next proposition an explicit linear basis for the space of $A_q(K)$ -fixed vectors in W_r is given.

PROPOSITION 5.2.2. Fix $1 \le r \le l$. Write

$$\mathcal{I}_r^{(i)} := \{ I \subset [1,n] \mid |I| = r, \ |I \cap [1,n-l]| = i \} \quad (0 \le i \le r),$$
5.2. THE CONTRACTION MAP

(5.2.6)
$$w_r^{(i)} := \sum_{I \in \mathfrak{I}_r^{(i)}} v_I \otimes v_I^* \in W_r.$$

Then the vectors $w_r^{(i)}$ $(0 \le i \le r)$ form a basis for the subspace of $A_q(K)$ -fixed vectors in W_r .

PROOF. It is obvious from the definition that the vectors $w_r^{(i)}$ $(0 \le i \le r)$ are linearly independent. Moreover, it follows from Theorem 4.4.1 and the decomposition (4.5.14) that the subspace of $A_q(K)$ -fixed vectors has dimension r+1. It therefore suffices to show that the elements $w_r^{(i)}$ are $A_q(K)$ -fixed. Let $I, J \subset [1, n]$ be such that |I| = |J| = r. Then

(5.2.7)
$$\pi_{K}(\xi_{J}^{I}) = \begin{cases} 0 & \text{if } I \in \mathfrak{I}_{r}^{(i)}, J \in \mathfrak{I}_{r}^{(j)}, i \neq j, \\ \xi_{J_{1}}^{I_{1}} \otimes \xi_{J_{2}}^{I_{2}} & \text{if } I, J \in \mathfrak{I}_{r}^{(i)}, \end{cases}$$

with $I_1 := I \cap [1, n - l]$, $I_2 := \{j - n + l \mid j \in I \cap [n - l + 1, n]\}$, and similarly for J. By applying (4.3.14), (5.2.7), (4.3.19) and (4.3.16) it follows that

$$(\mathrm{id}\otimes\pi_K)\big(R(w_r^{(i)})\big) = \sum_{K,L\in\mathfrak{I}_r^{(i)}} v_K \otimes v_L^* \otimes \sum_{I\in\mathfrak{I}_r^{(i)}} \pi_K(\xi_I^K(\xi_I^L)^*)$$
$$= \sum_{K\in\mathfrak{I}_r^{(i)}} v_K \otimes v_K^* \otimes 1 = w_r^{(i)} \otimes 1,$$

which concludes the proof of the proposition.

Observe that $\Omega_{r,r}(w_r^{(i)})$ is an $A_q(K)$ -fixed vector in W_{r-1} by Proposition 5.2.1 and Proposition 5.2.2.1 here it can be uniquely written as a linear combination of the basis elements $\{w_{r-1}^{(i)}\}_{i=0}^{r-1}$. More precisely, we have the following lemma.

LEMMA 5.2.3. Fix $1 \le r \le l$. Then, $\Omega_{r,r}(w_r^{(i)}) = c_r^{(i)} w_{r-1}^{(i-1)} + d_r^{(i)} w_{r-1}^{(i)}$ with

$$c_r^{(i)} = \begin{cases} 0 & \text{if } i = 0, \\ (-q)^{r-n} \frac{(1-q^{2(n-l-i+1)})}{1-q^2} & \text{if } 1 \le i \le r, \end{cases}$$
$$d_r^{(i)} = \begin{cases} (-q)^{r-n} \frac{(q^{2(n-l-i)}-q^{2(n-r+1)})}{1-q^2} & \text{if } 0 \le i \le r-1, \\ 0 & \text{if } i = r. \end{cases}$$

PROOF. Observe that

$$\Omega_{r,r}(w_r^{(i)}) = \sum_{I \in \mathfrak{I}_{r-1}^{(i-1)}} \sum_{\substack{j \leq n-l \\ j \notin I}} \frac{\mathrm{s}_{\mathbf{q}}(j; I^c \setminus j)}{\mathrm{s}_{\mathbf{q}}(I^c \setminus j; j)} v_I \otimes v_I^* + \sum_{I \in \mathfrak{I}_{r-1}^{(i)}} \sum_{\substack{j > n-l \\ j \notin I}} \frac{\mathrm{s}_{\mathbf{q}}(j; I^c \setminus j)}{\mathrm{s}_{\mathbf{q}}(I^c \setminus j; j)} v_I \otimes v_I^*,$$

where the first term should be omitted when i = 0, and the second term should be omitted when i = r. Now

$$s_q(I^c \setminus j; j)^{-1} s_q(j; I^c \setminus j) = (-q)^{r-n} s_q(j; I^c \setminus j)^2$$

for $I \subset [1, n], |I| = r - 1$. Also, for $I \in \mathfrak{I}_{r-1}^{(i-1)}$ one has

$$\sum_{\substack{j \in [1, n-l] \\ j \notin I}} s_{q}(j; I^{c} \setminus j)^{2} = \sum_{s=0}^{n-l-i} q^{2s} = \frac{1 - q^{2(n-l-i+1)}}{1 - q^{2}},$$

whereas for $I \in \mathfrak{I}_{r-1}^{(i)}$ one has

$$\sum_{\substack{i \in [n-l+1,n] \\ i \notin I}} s_q(j; I^c \setminus j)^2 = \sum_{s=0}^{l+i-r} q^{2(n-l-i+s)} = \frac{q^{2(n-l-i)} - q^{2(n-r+1)}}{1 - q^2}.$$

This proves the assertion.

COROLLARY 5.2.4. The intertwiner $\Omega_{r,r} \colon W_r \to W_{r-1}$ is surjective. The kernel of $\Omega_{r,r}$ is equal to the unique copy of $V(\varpi_r)$ within W_r .

PROOF. It follows from Proposition 5.2.2 and Lemma 5.2.3 that the image of $\Omega_{r,r}$ contains the subspace of $A_q(K)$ -fixed vectors in W_{r-1} . Since $\Omega_{r,r}$ is an intertwiner (cf. Proposition 5.2.1), it follows from the irreducible decomposition of the comodules W_s (4.5.14) and from the fact that each $V(\varpi_s)$ has non-zero $A_q(K)$ -fixed vectors (cf. Theorem 4.4.1) that $\Omega_{r,r}$ is surjective. Again by (4.5.14) and Proposition 5.2.1, it follows that the kernel of $\Omega_{r,r}$ is the unique copy of $V(\varpi_r)$ within W_r .

From Lemma 5.2.3 the following explicit expression for a non-zero $A_q(K)$ -fixed vector in $V(\varpi_r) \subset W_r$ can be derived.

COROLLARY 5.2.5. Fix $1 \leq r \leq l$. A non-zero $A_q(K)$ -fixed vector in $V(\varpi_r) \subset \Lambda_q^r(V) \otimes \Lambda_q^r(V^*)$ is given by

$$\sum_{i=0}^{r} \frac{\left(q^{2(l-r+1)}; q^2\right)_i}{\left(q^{2(l-n)}; q^2\right)_i} w_r^{(i)} \in V(\varpi_r).$$

PROOF. The above vector is obviously non-zero, $A_q(K)$ -fixed and an easy computation using Lemma 5.2.3 shows that the vector lies in the kernel of $\Omega_{r,r}$. The result follows now from Corollary 5.2.4.

In the next proposition it is shown that the $A_q(K)$ -fixed vectors $w_r^{(r)}$ $(1 \le r \le l)$ can be reconstructed from $w_1^{(1)}$ using the "raising operator" $\widehat{\Psi}_r$ (cf. Proposition 4.5.9).

PROPOSITION 5.2.6. Let
$$1 \le r \le l$$
. Then $\widehat{\Psi}_r((w_1^{(1)})^{\otimes r}) = d_r w_r^{(r)}$ with

$$d_r = \frac{(-q)^{\binom{r}{2}}}{(1-q^2)^r} (q^2; q^2)_r,$$

where $\widehat{\Psi}_r$ is the intertwiner defined in Proposition 4.5.9.

PROOF. By induction to r, the case r = 1 being trivial. For the induction step, recall the intertwiner $\widehat{\Theta}_r := (\mathrm{pr}_{r-1,1} \otimes \mathrm{pr}_{r-1,1}^*) \circ (\mathrm{id}_{\Lambda_q^{r-1}(V)} \otimes \widehat{\Phi}_r \otimes \mathrm{id}_{V^*})$ defined in Lemma 4.5.13. Then the induction step will follow from the formula

(5.2.8)
$$\widehat{\Theta}_r(w_{r-1}^{(r-1)} \otimes w_1^{(1)}) = (-q)^{r-1} \frac{1-q^{2r}}{1-q^2} w_r^{(r)}$$

for $2 \le r \le l$ (compare with the proof of Proposition 4.5.9). Now it follows easily from Lemma 4.5.10 that for |I| = r - 1, $1 \le i \le n$,

$$\begin{split} \widehat{\Theta}_{r}(v_{I} \otimes v_{I}^{*} \otimes v_{i} \otimes v_{i}^{*}) &= \\ \begin{cases} (-q)^{r-1} v_{I \cup i} \otimes v_{I \cup i}^{*} & \text{if } i \notin I, \\ (-q)^{r-1} (q^{2}-1) \sum_{m < i} \left(\frac{\mathrm{s}_{q}(I \setminus i; m)}{\mathrm{s}_{q}(I \setminus i; i)} \right)^{2} v_{I \cup m} \otimes v_{I \cup m}^{*} & \text{if } i \in I. \end{cases} \end{split}$$

Hence

$$\widehat{\Theta}_{r}(w_{r-1}^{(r-1)} \otimes w_{1}^{(1)}) = (-q)^{r-1} \sum_{\substack{J \subset [1, n-l] \\ |J| = r}} c_{J} v_{J} \otimes v_{J}^{*},$$

where

$$c_J := \sum_{i \in J} \left(1 + (q^2 - 1) \sum_{\substack{j \in J \\ j > i}} \left(\frac{\operatorname{s}_{\mathbf{q}}(J \setminus (i \cup j); i)}{\operatorname{s}_{\mathbf{q}}(J \setminus (i \cup j); j)} \right)^2 \right).$$

A straightforward computation using (5.2.2) shows that

$$c_J = \sum_{s=0}^{r-1} q^{2s} = \frac{1-q^{2r}}{1-q^2}.$$

This proves the proposition.

Observe that Proposition 5.2.6 can be used to prove that the "constant term" a_0^r of $\varphi_r^{\sigma,\tau}|_T$ (cf. Lemma 4.7.2) is non-zero. In fact, a_0^r can be computed explicitly using Proposition 5.2.6.

5.3. Algebraic generators of $A_q(U/K)$

Recall that the subspace of right $A_q(K)$ -invariant functions

$$A_q(U/K) := \{ \varphi \in A_q(U) \, | \, (\mathrm{id} \otimes \pi_K) \circ \Delta(\varphi) = \varphi \otimes 1 \},\$$

is a *-subalgebra and a left $A_q(U)$ -subcomodule of $A_q(U)$. The goal of this section is to give several explicit sets of generators for $A_q(U/K)$.

5. QUANTUM PLÜCKER COORDINATES

Define an inner product on W_r by

(5.3.1)
$$\langle v_I \otimes v_J^*, v_K \otimes v_L^* \rangle := \langle v_I, v_K \rangle \langle v_J^*, v_L^* \rangle = \delta_{I,K} \delta_{J,L} q^{-\langle 2\rho, \tilde{\varepsilon}_J \rangle}$$

(cf. Section 4.3). Write θ_r for the map of Lemma 4.6.1 associated with the unitary comodule $(W_r, \langle ., . \rangle)$. Observe that

(5.3.2)
$$\theta_r(v_I \otimes v_J^*, v_K \otimes v_L^*) = q^{-\langle 2\rho, \tilde{\varepsilon}_J \rangle} \xi_K^I (\xi_L^J)^*,$$

where I, J, K, L are subsets of [1, n] of cardinality r.

PROPOSITION 5.3.1. Fix $r \in [1, l]$. The elements

$$\psi_r^{(i)} := \sum_{I \in \mathfrak{I}_r^{(i)}} \xi_I^{[1,r]} (\xi_I^{[n-r+1,n]})^* \in A_q(U) \quad (i \in [0,r])$$

are highest weight vectors of highest weight ϖ_r with respect to the natural left $A_q(U)$ coaction on $A_q(U)$. Furthermore, the $\psi_r^{(i)}$ $(i \in [0,r])$ are right $A_q(K)$ -invariant and
they differ only by a non-zero scalar multiple.

PROOF. Write u_r for the highest weight vector $v_{[1,r]} \otimes v_{[n-r+1,n]}^*$ of the unique copy of $V(\varpi_r)$ within W_r . The elements $\tilde{\psi}_r^{(i)} := \theta_r(u_r, w_r^{(i)})$ $(i \in [0,r])$ are right $A_q(K)$ invariant and satisfy $(\pi_- \otimes \mathrm{id}) \circ \Delta(\tilde{\psi}_r^{(i)}) = z^{\varpi_r} \otimes \tilde{\psi}_r^{(i)}$ by Lemma 4.6.1, Proposition 5.2.2 and Remark 4.3.1. We claim that the elements $\tilde{\psi}_r^{(i)}$ are all non-zero and that they differ only by a non-zero scalar multiple. The proof of the proposition is then completed by observing that $\psi_r^{(i)}$ is a non-zero multiple of $\tilde{\psi}_r^{(i)}$ for all $i \in [0, r]$ (cf. (5.2.6), (5.3.2)).

For any $i \in [0, r]$ let $v_r^{(i)} \in V(\varpi_i) \subset W_r$ denote a non-zero $A_q(K)$ -fixed vector. By Corollary 5.2.5 one has $v_r^{(r)} = \sum_{i=0}^r \lambda_i w_r^{(i)}$ with $\lambda_i \neq 0$ for all i. Since the inner product $\langle \cdot, \cdot \rangle$ on W_r is invariant under the coaction of $A_q(U)$, the vectors $v_r^{(i)}$ $(0 \leq i \leq r)$ are mutually orthogonal. We may therefore assume that the $v_r^{(i)}$ have been rescaled such that they form an orthonormal basis of the subspace of $A_q(K)$ -fixed vectors in W_r . Then $w_r^{(i)} = \sum_{j=0}^r \langle w_r^{(i)}, v_r^{(j)} \rangle v_r^{(j)}$ with $\langle w_r^{(i)}, v_r^{(r)} \rangle = \sum_{j=0}^r \lambda_j \langle w_r^{(i)}, w_r^{(j)} \rangle = \lambda_i \sum_{I \in \mathfrak{I}_r^{(i)}} q^{-\langle 2\rho, \varepsilon_I \rangle} \neq 0$ for all $0 \leq i \leq r$. Now $\tilde{\psi}_r^{(i)} = \langle w_r^{(i)}, v_r^{(r)} \rangle \theta_r(u_r, v_r^{(r)})$, since the $V(\varpi_j)$ $(0 \leq j \leq r-1)$ lie in the orthogonal complement of $V(\varpi_r)$ with respect to the inner product $\langle \cdot, \cdot \rangle$ on W_r . In particular, the $\tilde{\psi}_r^{(i)}$ $(0 \leq i \leq r)$ differ only by a scalar multiple. Moreover, $\theta_r(u_r, v_r^{(r)}) \neq 0$, since the restriction of θ_r to $V(\varpi_r) \subset W_r$ is injective. It follows that $\tilde{\psi}_r^{(i)} \neq 0$ for all $i \in [0, r]$.

By the Peter-Weyl decomposition (4.3.9) and Theorem 4.4.1, the left $A_q(U)$ -comodule $A_q(U/K)$ has a multiplicity-free irreducible decomposition,

$$A_q(U/K) = \bigoplus_{\lambda \in P_K^+} V_L(\lambda), \quad V_L(\lambda) := W(\lambda) \cap A_q(U/K),$$

where $V_L(\lambda)$ is irreducible of highest weight λ . Combined with Proposition 5.3.1 and the fact that $A_q(U)$ has no zero divisors we obtain the following proposition.

PROPOSITION 5.3.2. For $\mu \in P_{\Sigma}^+$ a highest weight vector for the left $A_q(U)$ -subcomodule $V_L(\lambda_{\mu}) \subset A_q(U/K)$ is given by

$$\psi_{\mu} := \psi_1^{a_1} \psi_2^{a_2} \cdots \psi_l^{a_l} \in A_q(U).$$

where $\psi_r := \psi_r^{(r)}$ $(1 \le r \le l)$, $a_l := \mu_l$ and $a_i := \mu_i - \mu_{i+1} \ge 0$ $(1 \le i \le l-1)$.

Using Proposition 5.3.2 it is now possible to give several sets of algebraic generators of the subalgebra $A_q(U/K)$. Fix $r \in [1, l]$. For $I, J \subset [1, n]$ such that |I| = |J| = r there are uniquely determined elements $x_{IJ}^{\pm} \in A_q(U)$ such that

(5.3.3)
$$R(w_r^{(r)}) = \sum_{|I|=|J|=r} v_I \otimes v_J^* \otimes x_{IJ}^+, \ R(w_r^{(0)}) = \sum_{|I|=|J|=r} v_I \otimes v_J^* \otimes x_{IJ}^-.$$

By (4.3.14), the matrix elements x_{IJ}^{\pm} are explicitly given by

(5.3.4)
$$x_{IJ}^{+} = \sum_{\substack{K \subset [1,n-l] \\ |K|=r}} \xi_{K}^{I} (\xi_{K}^{J})^{*}, \quad x_{IJ}^{-} = \sum_{\substack{K \subset [n-l+1,n] \\ |K|=r}} \xi_{K}^{I} (\xi_{K}^{J})^{*}$$

If x_{IJ}^{ε} is non-zero, then it is equal to $\theta(v_I \otimes v_J^*, w_r^{(r)})$ if $\varepsilon = +$ respectively $\theta(v_I \otimes v_J^*, w_r^{(0)})$ if $\varepsilon = -$ up to a non-zero constant. This implies by Lemma 4.6.1 that the x_{IJ}^{ε} are right $A_q(K)$ -invariant.

Observe that the elements x_{IJ}^{ε} for r = 1 and $I = \{i\}, J = \{j\}$ satisfy

(5.3.5)
$$x_{ij}^- = \delta_{ij} - x_{ij}^+ \quad (1 \le i, j \le n)$$

1

by (4.3.16). The contraction map $\Omega_{r,r}: W_r \to W_{r-1}$ and the intertwiner $\widehat{\Psi}_r: (W_1)^{\otimes r} \to W_r$ can now be used as lowering operator respectively raising operator to prove the following main result of this section.

THEOREM 5.3.3. Fix $r \in [1, l]$ and $\varepsilon \in \{+, -\}$. Then, the elements x_{IJ}^{ε} (|I| = |J| = r) generate the subalgebra $A_q(U/K)$.

PROOF. For $r \in [1, l]$ and $\varepsilon \in \{+, -\}$, let $B_r^{\varepsilon} \subset A_q(U/K)$ be the unital subalgebra generated by the $x_{IJ}^{\varepsilon}(|I| = |J| = r)$.

Fix $r \in [2, l]$. By Lemma 5.2.3 we have $\Omega_{r,r}(w_r^{(0)}) = d_r^{(0)} w_{r-1}^{(0)}$ with $d_r^{(0)}$ a non-zero constant. Now let $R \circ \Omega_{r,r}$ act on the vector $w_r^{(0)}$, where R is the right $A_q(U)$ -coaction on W_{r-1} , and use that $\Omega_{r,r}$ intertwines the $A_q(U)$ -coaction, then it follows that

$$\sum_{I|=|J|=r} \Omega_{r,r}(v_I \otimes v_J^*) \otimes x_{IJ}^- = d_r^{(0)} \sum_{|K|=|L|=r-1} v_K \otimes v_L^* \otimes x_{KL}^-.$$

Since $v_K \otimes v_L^*$ (K| = |L| = r - 1) form a linear basis of W_{r-1} , it follows that $B_{r-1}^- \subseteq B_r^-$. Similarly, it follows from $\Omega_{r,r}(w_r^{(r)}) = c_r^{(r)}w_{r-1}^{(r-1)}$ with $c_r^{(r)}$ a non-zero constant (cf. Lemma 5.2.3) that $B_{r-1}^+ \subseteq B_r^+$. Furthermore, by (5.3.5) we have $B_1^+ = B_1^-$. Hence we have the following inclusions of subalgebras,

$$A_q(U/K) \supseteq B_l^- \supseteq B_{l-1}^- \supseteq \ldots \supseteq B_1^- = B_1^+ \subseteq \ldots \subseteq B_{l-1}^+ \subseteq B_l^+ \subseteq A_q(U/K).$$

So it remains to prove that $B_1^+ = A_q(U/K)$. Since B_1^+ is a left $A_q(U)$ -comodule subalgebra of $A_q(U/K)$, it suffices to prove that $\psi_r^{(r)} \in B_1^+$ for $r \in [1, l]$ (cf. Proposition 5.3.2). Now $\psi_r^{(r)} = x_{[1,r][n-r+1,n]}^+$, so by the definition of B_1^{ε} we have $\psi_1^{(1)} \in B_1^+$. For $r \in [2, l]$ we have $\widehat{\Psi}_r((w_1^{(1)})^{\otimes r}) = d_r w_r^{(r)}$ for some non-zero constant d_r by Proposition 5.2.6. Let $R \circ \widehat{\Psi}_r$ act on $(w_1^{(1)})^{\otimes r}$, where R is the right $A_q(U)$ -coaction on W_r , and use that $\widehat{\Psi}_r$ intertwines the $A_q(U)$ -coactions, then it follows that

$$\sum_{\substack{i_1,\ldots,i_r\\i_1,\ldots,i_r}} \widehat{\Psi}_r(v_{i_1} \otimes v_{j_1}^* \otimes \cdots \otimes v_{i_r} \otimes v_{j_r}^*) \otimes x_{i_1j_1}^+ \cdots x_{i_rj_r}^+ = d_r \sum_{|I|=|J|=r} v_I \otimes v_J^* \otimes x_{IJ}^+.$$

Since the elements $v_I \otimes v_J^*$ (|I| = |J| = r) form a linear basis of W_r , it follows that $\psi_r^{(r)} = x_{[1,r][n-r+1,n]}^+ \in B_1^+$. This concludes the proof of the theorem. \Box

5.4. Algebras generated by quantum Plücker coordinates

To simplify the notations, we set

(5.4.1)
$$t_I := \xi_{[n-l+1,n]}^I = s_q(I^c; I) \left(\xi_{[1,n-l]}^{I^c}\right)^* \det_q, \quad (I \subset [1,n], \ |I| = l).$$

In the terminology of the introduction, the elements $(\xi_{[1,n-l]}^I)^*$ (|I| = n - l) are the quantum anti-holomorphic Plücker coordinates. Since det_q lies in the center of $A_q(U)$, the study of the algebraic relations between quantum holomorphic and anti-holomorphic quantum Plücker coordinates is equivalent with the study of the algebraic relations between the elements t_I , t_J^* (|I| = |J| = l). This "ambiguity" is caused by the fact that we work here with the reductive group U(n) instead of its semisimple part SU(n) (which, in the quantum setting, amounts to dividing $A_q(U)$ out by the relation det_q = 1). In the reductive setting it turns out to be more convenient to work with the elements t_I and t_J^* , and we will therefore reserve the names quantum anti-holomorphic respectively holomorphic Plücker coordinates to the elements t_I respectively t_J^* . In terms of these quantum analogues of the Plücker coordinates, the algebraic generators x_{IJ}^- of $A_q(U/K)$ can be rewritten as $x_{IJ}^- = t_I t_J^*$ for $I, J \subset [1, n], |I| = |J| = l$ (cf. Theorem 5.3.3).

For the study of the algebraic properties of $A_q(U/K)$ it is convenient to study first the algebra generated by the quantum anti-holomorphic Plücker coordinates t_I (|I| = l) and the algebra generated by the Plücker coordinates t_I , t_J^* (|I| = |J| = l). In this section these related algebras are characterized by certain invariance properties.

Let $A_q(SU(n))$ be the algebra $A_q(U(n))$ divided out by the two-sided ideal generated by det_q - 1, and set $\pi : A_q(U(n)) \to A_q(SU(n))$ for the natural projection. There exists a unique Hopf-*-algebra structure on $A_q(SU(n))$ such that π is an homomorphism of Hopf-*-algebras. Any right $A_q(U)$ -comodule (R, M) has a right $A_q(SU(n))$ comodule structure with comodule map given by (id $\otimes \pi$) $\circ R$. If M is irreducible as $A_q(U)$ -comodule, then it remains irreducible as $A_q(SU(n))$ -comodule. All irreducible $A_q(SU(n))$ -comodules arise in this way. Furthermore, $V(\lambda) \simeq V(\lambda')$ as $A_q(SU(n))$ -comodule if and only if $\lambda - \lambda' \in \mathbb{Z}(1^n)$. Set

(5.4.2)
$$K_0 := U(n-l) \times SU(l), \quad A_q(K_0) := A_q(U(n-l)) \otimes A_q(SU(l))$$

and let $\pi_K^0 : A_q(U(n)) \to A_q(K_0)$ be the surjective Hopf-*-algebra homomorphism given by

(5.4.3)
$$\pi_K^0 := \left(\operatorname{id}_{A_q(U(n-l))} \otimes \pi \right) \circ \pi_K.$$

Then

(5.4.4)
$$A_q(U/K_0) := \{ \phi \in A_q(U) \mid (\mathrm{Id} \otimes \pi_K^0) \circ \Delta(\phi) = \phi \otimes 1 \}$$

is a *-subalgebra and a left $A_q(U)$ -comodule *-subalgebra of $A_q(U)$.

THEOREM 5.4.1. The left $A_q(U)$ -comodule $A_q(U/K_0)$ has the multiplicity-free irreducible decomposition

$$A_q(U/K_0) = \bigoplus_{\lambda \in P_{K_0}^+} V_L(\lambda), \quad V_L(\lambda) := W(\lambda) \cap A_q(U/K_0)$$

where $V_L(\lambda)$ is irreducible of highest weight λ . The set $P_{K_0}^+$ of spherical weights are explicitly given by $P_{K_0}^+ := \{\lambda_{\mu,s} \mid \mu \in P_{\Sigma}^+, s \ge -\mu_l\}$ where

$$\lambda_{\mu,s} := (\mu_1, \dots, \mu_l, 0, \dots, 0, -s - \mu_l, \dots - s - \mu_1) \in P_n^+.$$

A highest weight vector of the left $A_q(U)$ -subcomodule $V_L(\lambda_{\mu,s}) \subset A_q(U/K_0)$ is given by

$$\psi_{\mu,s} := \begin{cases} \psi_{\mu} \left(t_{[1,l]} \right)^{-s} & \text{if } s \le 0, \\ \psi_{\mu} \left(\left(t_{[n-l+1,n]} \right)^{s} \right)^{s} & \text{if } s \ge 0 \end{cases}$$

where ψ_{μ} is defined in Proposition 5.3.2. Furthermore, $A_q(U/K_0)$ is generated as algebra by the elements t_I , t_J^* (|I| = |J| = l).

A proof of Theorem 5.4.1 can be given by repeating the arguments which were used in the previous sections and in Chapter 4 for proving the analogous results about $A_q(U/K)$. The details are omitted here.

We identify the bialgebra $A_q(\operatorname{Mat}(n, \mathbb{C}))$ with its image under the natural bialgebra embedding $A_q(\operatorname{Mat}(n, \mathbb{C})) \hookrightarrow A_q(U)$. Then

$$(5.4.5) \quad A_q(\operatorname{Mat}(n,\mathbb{C})/K_0) := \{ \phi \in A_q(\operatorname{Mat}(n,\mathbb{C})) \mid (\operatorname{Id} \otimes \pi_K^0) \circ \Delta(\phi) = \phi \otimes 1 \}$$

is a left $A_q(U)$ -comodule subalgebra of $A_q(\operatorname{Mat}(n, \mathbb{C}))$.

COROLLARY 5.4.2. (*cf.* [**89**, formula (17)])

The left $A_q(U)$ -comodule $A_q(Mat(n)/K_0)$ has the multiplicity-free irreducible decomposition

(5.4.6)
$$A_q(Mat(n,\mathbb{C})/K_0) = \bigoplus_{s\in\mathbb{Z}_+} V_L((s^l)),$$

5. QUANTUM PLÜCKER COORDINATES

where $V_L((1^s)) := W((s^l)) \cap A_q(Mat(n, \mathbb{C})/K_0)$ is an irreducible left $A_q(U)$ -comodule of highest weight (s^l) . A highest weight vector of the left $A_q(U)$ -subcomodule $V_L((s^l))$ is $(t_{[1,l]})^s$. Furthermore, $A_q(Mat(n, \mathbb{C})/K_0)$ is generated as unital algebra by the quantum Plücker coordinates $t_I(|I| = l)$.

PROOF. Recall that the irreducible decomposition of $A_q(Mat(n, \mathbb{C}))$ under its natural $A_q(U)$ -bicomodule structure is given by

$$A_q(\operatorname{Mat}(n,\mathbb{C})) = \bigoplus_{\lambda \in P_n^+ : \lambda_n > 0} W(\lambda)$$

(cf. [96]). The decomposition (5.4.6) and the statement about the highest weight vector in $V_L((s^l))$ are now direct consequences of Theorem 5.4.1. Let *B* be the unital subalgebra of $A_q(\operatorname{Mat}(n, \mathbb{C})/K_0)$ generated by the t_I (|I| = l). Since *B* contains the highest weight vectors $(t_{[1,l]})^s$ and is stable under the left $A_q(U)$ -coaction, it follows that $B = A_q(\operatorname{Mat}(n, \mathbb{C})/K_0)$.

COROLLARY 5.4.3. There is a natural grading $A_q(U/K_0) = \bigoplus_{m \in \mathbb{Z}} \mathcal{A}_m$, where

$$\mathcal{A}_m := \{ \phi \in A_q(U/K_0) \,|\, (id \otimes_{|T}) \circ \Delta(\phi) = \phi \otimes (z_{n-l+1} \dots z_n)^m \}.$$

Furthermore,

$$\mathcal{A}_m = \{ \phi \in A_q(U) \, | \, (id \otimes \pi_K) \circ \Delta(\phi) = \phi \otimes (1 \otimes det_q^m) \}.$$

In particular, $A_a(U/K) = \mathcal{A}_0$.

PROOF. By direct calculations it is verified that $t_I \in A_1$ and $t_J^* \in A_{-1}$ for all subsets $I, J \subset [1, n]$ of cardinality l. Since $A_q(U/K_0)$ is generated as algebra by the quantum Plücker coordinates t_I, t_J^* (|I| = |J| = l) (cf. Theorem 5.4.1), it follows that $A_q(U/K_0) = \bigoplus_{m \in \mathbb{Z}} A_m$. The remaining assertions are straightforward.

By Theorem 5.3.3, Theorem 5.4.1, Corollary 5.4.2 and Corollary 5.4.3 we may regard $A_q(\operatorname{Mat}(n, \mathbb{C})/K_0)$ as the quantum algebra of anti-holomorphic polynomials on U/K and $A_q(U/K_0)$ as the quantum algebra of complex-valued polynomials on U/K. Then, by Corollary 5.4.3, $A_q(U/K)$ corresponds to the quantum algebra of zero weighted complex valued polynomials on U/K.

The subalgebra $A_q(\operatorname{Mat}(n, \mathbb{C})/K_0)$ has been studied extensively by Taft and Towber [126]. In particular, in [126] an algebraic characterization is obtained for the subalgebra $A_q(\operatorname{Mat}(n, \mathbb{C})/K_0)$ in terms of the quantum Plücker coordinates t_I (|I| = l), and a linear basis is constructed in terms of straightened monomials in the t_I 's. Some of these results can be recovered from the explicit basis of the irreducible finite dimensional $A_q(U)$ -comodules as given in [96]. Since these results will play an important role in the remainder of this chapter, they will be treated in some detail in the next section, following mainly the line of arguments from [96].

5.5. Algebraic properties of anti-holomorphic Plücker coordinates

In the following lemma we give explicit quadratic relations for the quantum Plücker coordinates t_I .

LEMMA 5.5.1. For $r_1, r_2 \in [0, l]$, $J_1, J_2, K \subset [1, n]$, $|J_1| = l - r_1$, $|J_2| = l - r_2$ and $|K| = r_1 + r_2$, set

$$R_q(r_1, r_2; J_1, J_2; K) := \sum_{\substack{L \subset K \\ |L| = r_2}} \mathrm{s}_{\mathbf{q}}(J_1; K \backslash L) \, \mathrm{s}_{\mathbf{q}}(K \backslash L; L) \, \mathrm{s}_{\mathbf{q}}(L; J_2) t_{J_1 \cup (K \backslash L)} t_{L \cup J_2}.$$

Then, $R_q(r_1, r_2; J_1, J_2; K) = 0$ if $r_1 + r_2 \ge l + 1$.

PROOF. Follows easily from the generalized Plücker relations given in [96, Proposition 1.2] and from the definition of the quantum Plücker coordinates t_I .

Among the relations in Lemma 5.5.1, there are two types of quadratic relations which play an important role for the algebraic structure of $A_q(Mat(n, \mathbb{C})/K_0)$, namely the socalled *q*-Garnir relations and the Young symmetry relations. The *q*-Garnir relations are by definition the quadratic relations

(5.5.1)
$$R_q(l+1-r,r;J_1,J_2;K) = 0,$$
$$(r \in [1,l], |J_1| = r-1, |J_2| = l-r, |K| = l+1)$$

and the Young symmetry relations are by definition the quadratic relations

(5.5.2)
$$R_q(l,r;\emptyset,J_2;K) = 0, \quad (r \in [1,l], |J_2| = l - r, |K| = l + r).$$

It turns out that the q-Garnir relations, as well as the Young symmetry relations, characterize the algebraic structure of $A_q(\operatorname{Mat}(n, \mathbb{C})/K_0)$ completely. Before stating the precise result, we need to introduce some more notations and terminology.

For $\lambda \in P_n^+$ with $\lambda_n \geq 0$, let SSTAB_n(λ) be the semistandard tableaux of shape λ with coefficients in [1, n]. In other words, $T \in \text{SSTAB}_n(\lambda)$ is a sequence of numbers $T_{ij} \in [1, n]$, where for fixed $1 \leq i \leq n, j$ runs through $[1, \lambda_i]$, such that $T_{ij} < T_{i+1,j}$ and $T_{ij} \leq T_{i,j+1}$ for all the relevant i and j. For $T \in \text{SSTAB}_n(\lambda)$, let $T^{(j)} \subset [1, n]$ be the set of numbers in the jth column of T. Observe that the cardinality of $T^{(j)}$ is equal to the number of boxes in the jth column of T. For $s \in \mathbb{Z}_+$ and $T \in \text{SSTAB}_n(s^l)$, set

$$t_T := t_{T^{(1)}} t_{T^{(2)}} \dots t_{T^{(s)}} \in A_q(\operatorname{Mat}(n, \mathbb{C})/K_0).$$

The elements t_T ($T \in SSTAB_n((s^l)), s \in \mathbb{Z}_+$) are called the *straightened monomials*.

The subspace $\Lambda_q^l(V)$ spanned by the vectors $\{v_I\}_{|I|=l}$ has a left $A_q(U)$ -comodule structure given by the formula

(5.5.3)
$$L(v_I) = \sum_{|J|=l} \xi_J^I \otimes v_J \quad (|I|=l).$$

With this left $A_q(U)$ -comodule structure, $\Lambda_q^l(V)$ is irreducible of highest weight (1^l) with highest weight vector $v_{[1,l]}$ and lowest weight vector $v_{[n-l+1,n]}$. The inner product

defined by $\langle v_I, v_J \rangle = \delta_{I,J}$ for |I| = |J| = l is left $A_q(U)$ -invariant, where we use the convention that a left $A_q(U)$ -comodule (L, M) with inner product $\langle ., . \rangle$ is left invariant if

$$\sum_{m),(n)} m_{(1)}(n_{(1)})^* \langle m_{(2)}, n_{(2)} \rangle = \langle m, n \rangle 1 \quad \forall m, n \in M$$

 $(\sum_{(m)} m_{(1)} \otimes m_{(2)} := L(m))$. Let $T(\Lambda_q^l(V)) := \bigoplus_{s \in \mathbb{Z}_+} \Lambda_q^l(V)^{\otimes s}$ be the tensor algebra of the comodule $\Lambda_q^l(V)$. The tensor algebra $T(\Lambda_q^l(V))$ has a unique left $A_q(U)$ -comodule algebra structure such that the coaction on the tensors of degree 1 coincides with the coaction (5.5.3) on $\Lambda_q^l(V)$. Write L for the corresponding comodule mapping on $T(\Lambda_q^l(V))$. The comodule map L preserves the natural grading of the tensor algebra, i.e. the subspaces $\Lambda_q^l(V)^{\otimes s}$ are invariant subspaces. The linear map defined by

$$\Psi_a: T(\Lambda_q^l(V)) \to A_q(\operatorname{Mat}(n, \mathbb{C})/K_0), \quad \Psi_a(v_{I_1} \otimes v_{I_2} \otimes \ldots \otimes v_{I_s}) := t_{I_1} t_{I_2} \ldots t_{I_s}$$

for $|I_j| = l$ is a surjective left $A_q(U)$ -comodule algebra homomorphism (the subindex a of Ψ_a stands for anti-holomorphic). By Lemma 5.5.1, the elements

$$\begin{split} \hat{R}_q(r_1, r_2; J_1, J_2; K) &:= \\ \sum_{\substack{L \subset K \\ |L| = r_2}} \mathbf{s}_q(J_1; K \setminus L) \, \mathbf{s}_q(K \setminus L; L) \, \mathbf{s}_q(L; J_2) v_{J_1 \cup (K \setminus L)} \otimes v_{L \cup J_2} \in \Lambda_q^l(V)^{\otimes 2} \end{split}$$

for $r_1, r_2 \in [0, l]$, $|J_1| = l - r_2$, $|J_2| = l - r_2$ and $|K| = r_1 + r_2$ with $r_1 + r_2 \ge l + 1$ lie in the kernel of Ψ_a . The following theorem covers the essential results from [126]. A proof of the theorem is sketched in which certain arguments from [126] are replaced by representation theoretic arguments from [96].

THEOREM 5.5.2. Let $s \in \mathbb{Z}_+$.

(

(i) $\Psi_a \text{ maps } \Lambda_q^l(V)^{\otimes s}$ surjectively onto $V_L((1^s)) := W((1^s)) \cap A_q(Mat(n, \mathbb{C})/K_0)$. (ii) The set $\mathcal{B}_a(s) := \{t_T \mid T \in SSTAB_n((s^l))\}$ is a linear basis of $V_L((s^l))$. The disjoint union $\mathcal{B}_a := \bigcup_{s \in \mathbb{Z}_+} \mathcal{B}_a(s)$ is a linear basis of $A_q(Mat(n, \mathbb{C})/K_0)$.

(iii) The kernel of Ψ_a is generated as two-sided ideal by the q-Garnir type elements

$$R_q(l+1-r,r;J_1,J_2;K) \quad (r \in [1,l], |J_1| = r-1, |J_2| = l-r, |K| = r+1)$$

(iv) The kernel of Ψ_a is generated as two-sided ideal by the Young symmetry type elements

$$R_q(l,r;\emptyset,J_2;K)$$
 $(r \in [1,l], |J_2| = l - r, |K| = l + r).$

PROOF. (i) Fix arbitrary subsets $I_j \subset [1, n]$ of cardinality l and set $\tilde{t} := t_{I_1} t_{I_2} \dots t_{I_s}$. We have to show that $\tilde{t} \in W((1^s))$.

The left $A_q(U)$ -invariant inner product $\langle ., . \rangle$ on $\Lambda_q^l(V)$ induces in a natural way a left $A_q(U)$ -invariant inner product on $\Lambda_q^l(V)^{\otimes s}$, which will also be denoted by $\langle ., . \rangle$. Then

$$\tilde{t} = \langle L((v_{[n-l+1,n]})^{\otimes s}), v_{I_1} \otimes v_{I_2} \otimes \ldots \otimes v_{I_s} \rangle,$$

where we have used the notation

$$\langle \sum_{i} a_i \otimes \phi_i, \psi \rangle := \sum_{i} a_i \langle \phi_i, \psi \rangle, \quad a_i \in A_q(U), \ \phi_i, \psi \in \Lambda_q^l(V)^{\otimes s}.$$

Since $(v_{[n-l+1,n]})^{\otimes s} \in \Lambda_q^l(V)^{\otimes s}$ is a lowest weight vector of the unique copy of $V_L((s^l))$ in $\Lambda_q^l(V)^{\otimes s}$, it follows from the unitarity of $\langle ., . \rangle$ that $\tilde{t} \in W((s^l))$.

(ii) In [126] it is shown that an arbitrary product $\tilde{t} = t_{I_1}t_{I_2}...t_{I_s}$ of Plücker coordinates can be rewritten as linear combination of straightened monomials t_T ($T \in \text{SSTAB}_n((s^l))$) by repeated application of the q-Garnir relations (5.5.1) as straightening relations. Together with (i) this implies that $V_L((1^s))$ is spanned by the set $\mathcal{B}_a(s)$ of straightened monomials. Since the dimension of $V_L((s^l))$ is equal to the cardinality of $\text{SSTAB}_n((s^l))$ (see [96]), (ii) follows.

(iii) In the proof of (ii), only the q-Garnir relations (5.5.1) are needed to rewrite an arbitrary product $\tilde{t} = t_{I_1} t_{I_2} \dots t_{I_s}$ of Plücker coordinates as linear combination of straightened monomials t_T ($T \in \text{SSTAB}_n((s^l))$). Since the straightened monomials form a basis of $A_q(\text{Mat}(n, \mathbb{C})/K_0)$, (iii) follows.

(iv) In [126] it was stated that the Young symmetry relations are sufficient to straighten any product of quantum Plücker coordinates, but the proof given in [126] is not complete. A complete proof of this fact can be given using [35, Lemma 6.15]. The details will be omitted here. \Box

REMARK 5.5.3. Theorem 5.5.2 implies in particular that the restriction of Ψ_a to the $A_q(U)$ -invariant subspace $\Lambda_q^l(V)$ yields a left $A_q(U)$ -comodule isomorphism from $\Lambda_q^l(V)$ onto $\text{Span}\{t_I\}_{|I|=l}$ which maps v_I to t_I for all I.

The tensor algebra $T(\Lambda_a^l(V))$ decomposes as a direct sum

(5.5.4)
$$T(\Lambda_q^l(V)) = \bigoplus_{s,t \in \mathbb{Z}_+} T_{s,t},$$

where $T_{s,t}$ is the span of tensors $\tilde{v} := v_{I_1} \otimes v_{I_2} \otimes \ldots \otimes v_{I_s}$ for which the weight $w(\tilde{v}) := \sum_{j=1}^s \sum_{i \in I_j} i$ is equal to t. This defines a grading on $T(\Lambda_q^l(V))$, namely

$$T_{s,t} \otimes T_{s',t'} \subset T_{s+s',t+t'}, \quad (s,s',t,t' \in \mathbb{Z}_+).$$

Observe that

$$\hat{R}_q(r_1, r_2; J_1, J_2; K) \in T_{2,t}$$

with t the sum of the entries of J_1, J_2 and K. Combined with Theorem 5.5.2 (iii), it follows that $\text{Ker}(\Psi_a)$ is graded,

(5.5.5)
$$\operatorname{Ker}(\Psi_a) = \bigoplus_{s,t \in \mathbb{Z}_+} \left(\operatorname{Ker}(\Psi_a) \cap T_{s,t} \right).$$

On the other hand, set

(5.5.6)
$$A_a(s,t) := \text{Span}\{t_T \mid T \in \text{SSTAB}_n((s^l)), w(T) = t\}$$

where $w(T) := \sum_{i,j} T_{i,j}$ is the weight of T, then we have the direct sum decomposition

(5.5.7)
$$A_q(\operatorname{Mat}(n,\mathbb{C})/K_0) = \bigoplus_{s,t \in \mathbb{Z}_+} A_a(s,t)$$

by Theorem 5.5.2. Observe that $A_a(s,t) \subseteq \Psi_a(T_{s,t})$ for all $s, t \in \mathbb{Z}_+$. In fact, by (5.5.5), we have the following lemma.

LEMMA 5.5.4. For all $s, t \in \mathbb{Z}_+$, we have $A_a(s, t) = \Psi_a(T_{s,t})$. In particular, the direct sum decomposition (5.5.5) is a grading,

$$A_a(s,t).A_a(s',t') \subset A_a(s+s',t+t'), \quad (s,s',t,t' \in \mathbb{Z}_+)$$

and for subsets $I_j \subset [1, n]$ with $|I_j| = l$ and $t = \sum_{j=1}^s \sum_{i \in I_j} i$, we have $t_{I_1} t_{I_2} \dots t_{I_s} \in A_a(s, t)$.

5.6. The algebraic structure of $A_q(U/K_0)$ and $A_q(U/K)$

In this section the algebraic structure of the algebra $A_q(U/K_0)$ of complex-valued polynomial functions on U/K is characterized in terms of the Plücker coordinates t_I, t_J^* .

The algebraic structure of the subalgebra $A_q^*(\operatorname{Mat}(n)/K_0)$ of $A_q(U)$ generated by the quantum holomorphic Plücker coordinates $t_I^*(|I| = l)$ follows directly from Theorem 5.5.2. Indeed, applying the *-involution to Lemma 5.5.1, quadratic relations for the Plücker coordinates t_I^* are obtained. In particular, "dual" q-Garnir relations and "dual" Young symmetry relations for the t_I^* are obtained by applying the *-involution on the relations (5.5.1) respectively (5.5.2). The map Ψ_a in Theorem 5.5.2 should be replaced by the map

$$\Psi_h^{\scriptscriptstyle +}: T(\Lambda_a^l(V^*)) \to A_a^*(\operatorname{Mat}(n)/K_0), \quad \Psi_h(v_{I_1}^* \otimes \ldots \otimes v_{I_n}^*) := t_{I_1}^* \ldots t_{I_n}^*$$

for $|I_j| = l$, which is a surjective left $A_q(U)$ -comodule algebra homomorphism if the tensor algebra $T(\Lambda_q^l(V^*))$ is given the left $A_q(U)$ -comodule algebra structure induced from the left coaction

$$L(v_I^*) := \sum_{|J|=l} \left(\xi_J^I\right)^* \otimes v_J^*, \quad (|I|=l)$$

on $\Lambda_q^l(V^*)$ (here the subindex h of Ψ_h stands for holomorphic). In particular, $\Lambda_q^l(V^*)$ is isomorphic as left $A_q(U)$ -comodule to the linear span of the Plücker coordinates t_I^* (|I| = l) (with comodule action given by the restriction of the comultiplication Δ on $A_q(U)$). The kernel of Ψ_h is generated as two-sided ideal by the dual q-Garnir type or, equivalently, by the dual Young symmetry type elements in $\Lambda_q^l(V^*)^{\otimes 2}$.

So the algebraic relations between the quantum holomorphic Plücker coordinates and the antiholomophic Plücker coordinates have to be studied in order to clarify the algebraic structure of $A_q(U/K_0)$. Set

(5.6.1)
$$W := \Lambda_q^l(V) \oplus \Lambda_q^l(V^*),$$

and consider W as left $A_q(U)$ -comodule. Let T(W) be the tensor algebra of W and extend the left $A_q(U)$ -comodule structure on W to a left $A_q(U)$ -comodule algebra structure on T(W). The tensor algebras $T(\Lambda_q^l(V))$ and $T(\Lambda_q^l(V^*))$ can be naturally embedded into T(W) as left $A_q(U)$ -comodule subalgebras. They generate T(W) as algebra.

Let $\Psi: T(W) \to A_q(U/K_0)$ be the surjective left $A_q(U)$ -comodule algebra homomorphism which coincides on $T(\Lambda_q^l(V))$ (respectively $T(\Lambda_q^l(V^*)))$ with Ψ_a (respectively Ψ_h). We look for elements in $T(W)_{\leq 2} := \bigoplus_{s \leq 2} W^{\otimes s}$ which generate $\operatorname{Ker}(\Psi) \subset T(W)$ as a two-sided ideal.

Let M and N be two finite dimensional left $A_q(U)$ -comodules and fix linear bases $\{m_i\}_i$ and $\{n_j\}_j$ for M respectively N. Let $t_{i,j}^M, t_{s,t}^N \in A_q(U)$ be the matrix coefficients of M respectively N. The matrix coefficients are uniquely determined by the requirement that $L(m_i) = \sum_j t_{ij}^M \otimes m_j$ and $L(n_s) = \sum_t t_{st}^N \otimes n_t$. Fix an intertwiner $\Phi \in \operatorname{Hom}_{A_q(U)}(M, N)$ and let $c_{is} \in \mathbb{C}$ be the coefficients in the expansion $\Phi(m_i) = \sum_s c_{is} n_s$. Then the matrix coefficients satisfy the relations

$$\sum_{j} t^{M}_{ij} c_{jt} = \sum_{s} c_{is} t^{N}_{st}, \quad \forall \, i, t$$

in $A_q(U)$. Now recall from the proof of Theorem 5.5.2 that the t_I (respectively the t_J^*) can be obtained as matrix coefficients of the irreducible left $A_q(U)$ -comodule $\Lambda_q^l(V)$ (respectively $\Lambda_q^l(V^*)$). The trivial one dimensional corepresentation occurs with multiplicity one in $W_l = \Lambda_q^l(V) \otimes \Lambda_q^l(V^*) \subset W^{\otimes 2}$ since $\Lambda_q^l(V^*)$ is the dual representation of $\Lambda_q^l(V)$. Explicitly, the representation space $U_0 \simeq \mathbb{C}$ of the trivial corepresentation is spanned by the element

$$\sum_{|I|=l} (\mathbf{s}_{\mathbf{q}}(I;I^c))^2 v_I \otimes v_I^*.$$

This follows easily using the Laplace expansions for quantum minors, see (4.3.17) and (4.3.19). So the non-zero intertwiner in $\operatorname{Hom}_{A_q(U)}(\mathbb{C}, \Lambda_q^l(V) \otimes \Lambda_q^l(V^*))$, which is unique up to a non-zero constant, yields the commutation relation

(5.6.2)
$$\sum_{|I|=l} \left(s_{q}(I;I^{c}) \right)^{2} t_{I} t_{I}^{*} = q^{2l(n-l)}$$

in $A_q(U/K_0)$. Other commutation relations between the quantum holomorphic and antiholomorphic Plücker coordinates arise from the fact that

(5.6.3)
$$\Lambda_q^l(V) \otimes \Lambda_q^l(V^*) \simeq \Lambda_q^l(V^*) \otimes \Lambda_q^l(V)$$

as left $A_q(U)$ -comodules. The comodules are indeed isomorphic due to the quasi-triangular structure of the universal enveloping algebra $U_q(\mathfrak{g})$ (see [11] for more details). Here an isomorphism (5.6.3) will be constructed using the intertwiner $\gamma : V \otimes V \rightarrow V \otimes V$ (4.5.21) of left $A_q(U)$ -comodules. Essentially the same techniques as in the proof of Lemma 4.5.10 can be used. The construction of the isomorphism is as follows. For $i \in [1, n-1]$, let $\gamma_i \in \operatorname{End}_{A_q(U)}(V^{\otimes n})$ be the bijective intertwiner which acts as γ on the *i*th and (i + 1)th component, and as the identity on the other components. Let $s_i \in \mathfrak{S}_n$ be the simple transposition $s_i = (i, i + 1)$. Fix $\sigma \in \mathfrak{S}_n$ and let $\sigma = s_{i_1}s_{i_2}\ldots s_{i_l}$ be a reduced expression for σ . Set

$$\gamma(\sigma) := \gamma_{i_1} \circ \gamma_{i_2} \circ \ldots \circ \gamma_{i_l} \in \operatorname{End}_{A_{\sigma}(U)}(V^{\otimes n}).$$

5. QUANTUM PLÜCKER COORDINATES

Now $\gamma_i \circ \gamma_j = \gamma_j \circ \gamma_i$ if |i - j| > 1 and γ satisfies the Yang-Baxter equation (4.5.22), so $\gamma(\sigma)$ does not depend on the choice of reduced expression by Iwahori's Theorem. Consider the element $\chi_l \in \mathfrak{S}_n$ defined by

(5.6.4)
$$\chi_l := \begin{pmatrix} 1 & 2 & \dots & l & l+1 & l+2 & \dots & n \\ n-l+1 & n-l+2 & \dots & n & 1 & 2 & \dots & n-l \end{pmatrix}.$$

The length of χ_l is equal to l(n-l), and

(5.6.5)
$$\chi_l = (s_{n-l}s_{n-l+1} \dots s_{n-1}) \dots (s_2s_3 \dots s_{l+1})(s_1s_2 \dots s_l)$$

is a reduced expression for χ_l . It is not difficult to prove, using [35, Lemma 4.9], that there exists a unique bijective intertwiner

$$\tilde{\gamma}: \Lambda^l_q(V) \otimes \Lambda^{n-l}_q(V) \to \Lambda^{n-l}_q(V) \otimes \Lambda^l_q(V)$$

such that $\tilde{\gamma} \circ (\mathrm{pr}_l \otimes \mathrm{pr}_{n-l}) = (\mathrm{pr}_{n-l} \otimes \mathrm{pr}_l) \circ \gamma(\chi_l)$. An isomorphism (5.6.3) is now obtained using the isomorphism

(5.6.6)
$$\Lambda_q^l(V^*) \simeq \Lambda_q^{n-l}(V) \otimes \mathbb{C} \mathrm{det}_q^{-1}$$

as left $A_q(U)$ -comodules (cf. proof of Lemma 4.5.10). To formulate the corresponding commutation relations of the matrix elements, set for subsets $I, J, K, L \subset [1, n]$ of cardinality l,

(5.6.7)
$$C_{I,J}^{K,L} := (-q)^{-l(n-l)} \operatorname{s}_{q}(K^{c};K) \operatorname{s}_{q}(I;I^{c}) \langle \tilde{\gamma}(v_{J} \otimes v_{I^{c}}), v_{K^{c}} \otimes v_{L} \rangle,$$

where $\langle ., . \rangle$ is the $A_q(U)$ -invariant inner product on $\Lambda_q^{n-l}(V) \otimes \Lambda_q^l(V)$ defined by $\langle v_{I^c} \otimes v_J, v_{K^c} \otimes v_L \rangle = \delta_{I,K} \delta_{J,L}$.

LEMMA 5.6.1. The quantum holomorphic and anti-holomorphic Plücker coordinates satisfy the commutation relations

(5.6.8)
$$t_L^* t_K = \sum_{|I|=|J|=l} C_{I,J}^{K,L} t_I t_J^*$$

for all subsets $K, L \subset [1, n]$ of cardinality l.

PROOF. From the explicit expression of γ (4.5.21) and the explicit reduced expression (5.6.5) of χ_l it follows that

$$\tilde{\gamma}(v_{[n-l+1,n]} \otimes v_{[1,n-l]}) = v_{[1,n-l]} \otimes v_{[n-l+1,n]}$$

Applying the left $A_q(U)$ -comodule action and using that $\tilde{\gamma}$ is an intertwiner, one obtains

$$\xi_{K^{c}}^{[1,n-l]}\xi_{L}^{[n-l+1,n]} = \sum_{|I|=|J|=l} \langle \tilde{\gamma}(v_{J} \otimes v_{I^{c}}), v_{K^{c}} \otimes v_{L} \rangle \xi_{J}^{[n-l+1,n]} \xi_{I^{c}}^{[1,n-l]}$$

for |K| = |L| = l. The lemma follows now by applying the antipode S on both sides of this equation and using (4.3.19).

For subsets $I, J \subset [1, n]$ of equal cardinality, say $I = \{i_1 < i_2 < \ldots < i_r\}$ and $J = \{j_1 < j_2 < \ldots < j_r\}$, write $I \leq J$ if $i_s \leq j_s$ for all $s \in [1, r]$. The following properties for the coefficients $C_{I,J}^{K,L}$ (5.6.7) can now be derived.

LEMMA 5.6.2. The coefficients $C_{I,J}^{K,L}$ (5.6.7) satisfy $C_{I,J}^{K,L} = C_{J,I}^{L,K}$. Furthermore, the coefficients $C_{I,J}^{K,L}$ are zero unless $I \leq K$ and $J \leq L$. The diagonal coefficients are explicitly given by $C_{I,J}^{I,J} = q^{|I \cap J^c|}$.

PROOF. The proof involves a straightforward induction argument using the explicit construction of $\tilde{\gamma}$.

Observe that, by the results so far obtained, the following four type of elements lie in the kernel of Ψ :

$$\sum_{\substack{M \subset I: |M| = r}} \mathbf{s}_{\mathbf{q}}(I \setminus M; M) \, \mathbf{s}_{\mathbf{q}}(M; J) v_{I \setminus M} \otimes v_{M \cup J},$$

$$\sum_{\substack{M \subset I: |M| = r}} \mathbf{s}_{\mathbf{q}}(I \setminus M; M) \, \mathbf{s}_{\mathbf{q}}(M; J) v_{M \cup J}^{*} \otimes v_{I \setminus M}^{*},$$

$$v_{L}^{*} \otimes v_{K} - \sum_{|S| = |T| = l} C_{S,T}^{K,L} v_{S} \otimes v_{T}^{*},$$

$$q^{2l(n-l)} - \sum_{|M| = l} \mathbf{s}_{\mathbf{q}}(M; M^{c})^{2} v_{M} \otimes v_{M}^{*},$$

(5.6.9)

where $r \in [1, l]$ and $I, J, K, L \subset [1, n]$ with |I| = l + r, |J| = l - r and |K| = |L| = l.

THEOREM 5.6.3. (i) For $s, t \in \mathbb{Z}_+$, let $\mathcal{B}(s,t)$ be the set of elements $t_S(t_T)^*$ for which the pairs (S,T) satisfy $S \in SSTAB_n((s^l))$, $T \in SSTAB_n((t^l))$ and $S^{(s)} \neq [n-l+1,n]$ or $T^{(t)} \neq [n-l+1,n]$. Then, $\mathcal{B} := \bigcup_{s,t \in \mathbb{Z}_+} \mathcal{B}(s,t)$ is a disjoint union and \mathcal{B} is a linear basis of $A_q(U/K_0)$.

(ii) The elements (5.6.9) generate $Ker(\Psi)$ as a two-sided ideal.

PROOF. (i) $A_q(U/K_0)$ is spanned by the elements $t_S(t_T)^*$ with $S \in \text{SSTAB}_n((s^l))$, $T \in SSTAB_n((t^l))$ and $s, t \in \mathbb{Z}_+$, in view of Theorem 5.5.2 and Lemma 5.6.1. Fix now arbitrary $S \in \text{SSTAB}_n((s^l)), T \in SSTAB_n((t^l))$ with $S^{(s)} = T^{(t)} = [n - l + 1, n]$. To show that \mathcal{B} spans $A_q(U/K_0)$, it suffices to prove that $t_S(t_T)^*$ can be rewritten as linear combination of elements in \mathcal{B} . This follows inductively from the fact that $t_S(t_T)^*$ can be rewritten as a linear combination of elements $t_U(t_V)^*$ with $U \in \text{SSTAB}_n((u^l))$, $V \in \text{SSTAB}_n((v^l))$ where $u \leq s, v \leq t$ and $w(U) \leq w(S), w(V) \leq w(T)$, which in turn is a consequence of (5.6.2), Theorem 5.5.2 and Lemma 5.5.4.

The linear independence of the set \mathcal{B} can be established by comparing the leading terms of the elements in \mathcal{B} (cf. Section 4.3). The technical details, which are similar to the leading term type arguments in [96, Section 2.2], will be omitted here.

(ii) In the proof of (i) only the Young symmetry relations, dual Young symmetry relations, (5.6.2) and (5.6.8) are used to prove that $A_q(U/K_0)$ is spanned by \mathcal{B} . Since \mathcal{B} is a basis of $A_q(U/K_0)$, the desired result follows.

COROLLARY 5.6.4. The elements $\bigcup_{s \in \mathbb{Z}_+} \mathcal{B}(s,s)$ form a linear basis of $A_q(U/K)$.

5. QUANTUM PLÜCKER COORDINATES

PROOF. The set of elements $\bigcup_{s,t\in\mathbb{Z}_+:s-t=m}\mathcal{B}(s,t)$ form a linear basis of \mathcal{A}_m (cf. Corollary 5.4.3, Theorem 5.6.3). The corollary follows now from the fact that $\mathcal{A}_0 = A_q(U/K)$ (cf. Corollary 5.4.3).

For rank 1, the coefficients $C_{I,J}^{K,L}$ (5.6.7) can be computed explicitly, and we obtain the following complete list of commutation relations between the quantum Plücker coordinates t_{in} and t_{in}^* $(i, j \in [1, n])$ (cf. Theorem 5.6.3 (ii)),

$$t_{in}t_{jn} = qt_{jn}t_{in}, \quad t_{jn}^{*}t_{in}^{*} = qt_{in}^{*}t_{jn}^{*} \quad (i < j),$$

$$t_{in}^{*}t_{jn} = qt_{jn}t_{in}^{*} \quad (i \neq j),$$

$$\sum_{i=1}^{n} q^{2(i-1)}t_{in}t_{in}^{*} = q^{2(n-1)},$$

$$t_{in}^{*}t_{in} = t_{in}t_{in}^{*} + (1 - q^{2})\sum_{k < i} q^{2(k-i)}t_{kn}t_{kn}^{*}.$$

For rank 1, Theorem 5.6.3 has been proved before, see for instance [130], [96].

5.7. Some remarks on the subalgebra of bi- $A_q(K)$ -fixed elements

The subalgebra of bi- $A_q(K)$ -fixed elements $\mathcal{H}^{\infty,\infty}$ in $A_q(U)$ was studied in the previous chapter using suitable limit arguments. In particular, we have seen that $\mathcal{H}^{\infty,\infty}$ is a commutative algebra generated by l algebraically independent elements and that the zonal spherical functions are given in terms of multivariable little q-Jacobi polynomials (cf. Theorem 4.7.5).

For $r \in [1, l]$, let θ_r be the map (4.6.2) associated with the unitary module $(W_r, \langle ., . \rangle)$, where $\langle v_I \otimes v_J^*, v_K \otimes v_L^* \rangle = \delta_{I,K} \delta_{J,L} q^{-\langle 2\rho, \tilde{\varepsilon}_J \rangle}$. Then, the elements

(5.7.1)
$$\Psi_r^{i,j} := \theta_r(w_r^{(i)}, w_r^{(j)}) = \sum_{I \in \mathfrak{I}_r^{(i)}, J \in \mathfrak{I}_r^{(j)}} q^{-\langle 2\rho, \tilde{\varepsilon}_I \rangle} \xi_J^I(\xi_J^I)^*, \quad i, j \in [0, r]$$

all lie in the subspace $\oplus_{s=0}^{r} \mathcal{H}^{\infty,\infty}(\varpi_s) \subset \mathcal{H}^{\infty,\infty}$ (cf. Lemma 4.6.1). Recall that the zonal spherical functions $\varphi^{\infty,\infty}(\varpi_r) \in \mathcal{H}^{\infty,\infty}(\varpi_r)$ corresponding to the fundamental spherical weights ϖ_r ($r \in [1, l]$) can be realized as the matrix coefficients

$$\theta_r(v_\infty(\varpi_r), v_\infty(\varpi_r)),$$

where $v_{\infty}(\varpi_r)$ is a non-zero $A_q(K)$ -fixed vector in the unique copy of $V(\varpi_r)$ within W_r . By Corollary 5.2.5, the following explicit expressions for the zonal spherical functions $\varphi^{\infty,\infty}(\varpi_r) \in \mathcal{H}^{\infty,\infty}(\varpi_r)$ $(r \in [1, l])$ are obtained in terms of the bi- $A_q(U/K)$ -fixed elements $\Psi_r^{i,j}$ $(i, j \in [0, r])$.

LEMMA 5.7.1. *Let* $r \in [1, l]$ *. Then*

$$\sum_{i,j=0}^{r} \frac{\left(q^{2(l-r+1)};q^2\right)_i \left(q^{2(l-r+1)};q^2\right)_j}{\left(q^{2(l-n)};q^2\right)_i \left(q^{2(l-n)};q^2\right)_j} \Psi_r^{i,j} \in \mathcal{H}^{\infty,\infty}(\varpi_r)$$

is a zonal spherical function.

The bi- $A_q(K)$ -fixed elements $\psi_l^{i,0}$ are explicitly given in terms of products of Plücker coordinates by

(5.7.2)
$$\Psi_l^{i,0} = \sum_{I \in \mathfrak{I}_l^{(i)}} q^{-\langle 2\rho, \tilde{\varepsilon}_I \rangle} t_I t_I^* \in \bigoplus_{r=0}^i \mathcal{H}^{\infty,\infty}(\varpi_r), \quad (i \in [0,l]).$$

PROPOSITION 5.7.2. The elements $\Psi_l^{i,0}$ $(i \in [1,l])$ are algebraically independent generators of the algebra $\mathcal{H}^{\infty,\infty}$ of bi- $A_q(K)$ -fixed elements in $A_q(U)$.

PROOF. By Theorem 4.7.5 it suffices to prove that $\{\Psi_l^{i,0}\}_{i=1}^l \cup \{1\}$ is a linear basis of the (l+1)-dimensional subspace $\bigoplus_{r=0}^l \mathcal{H}^{\infty,\infty}(\varpi_r)$. This is an immediate consequence of Corollary 5.6.4.

REMARK 5.7.3. The algebraic independence of the elements $\Psi_l^{i,0}$ $(i \in [1, l])$ can also be proved using the monomial basis of $A_q(U/K)$ which was constructed in the previous section (cf. Corollary 5.6.4). On the other hand, the fact that $\mathcal{H}^{\infty,\infty}$ is a commutative algebra seems to be difficult to prove directly. The indirect proof using limit arguments involving the one-parameter family of quantum Grassmannians (cf. previous chapter) seems to be the only proof known up to this moment.

CHAPTER 6

Quantized flag manifolds and irreducible *-representations

6.1. Introduction

The irreducible *-representations of the "standard" quantization $\mathbb{C}_q[U]$ of the algebra of functions on a compact connected simple Lie group U were classified by Soibelman [110]. He showed that a 1-1 correspondence between the equivalence classes of irreducible *-representations of $\mathbb{C}_q[U]$ and the symplectic leaves of the underlying Poisson bracket on U exists (cf. [109], [110]). This Poisson bracket is sometimes called Bruhat-Poisson, because its symplectic foliation is a refinement of the Bruhat decomposition of U (cf. Soibelman [109], [110]). The symplectic leaves are naturally parametrized by $W \times T$, where $T \subset U$ is a maximal torus and W is the analytic Weyl group associated with (U, T).

The 1-1 correspondence between equivalence classes of irreducible *-representations of $\mathbb{C}_q[U]$ and symplectic leaves of U can be formally explained by the observation that in the semi-classical limit the kernel of an irreducible *-representation should tend to a maximal Poisson ideal. The quotient of the Poisson algebra of polynomial functions on U by this ideal is isomorphic to the Poisson algebra of functions on the symplectic leaf.

The results referred to above raise the obvious question whether the irreducible *representations of quantized function algebras on U-homogeneous spaces can be classified and related to the symplectic foliation of the underlying Poisson bracket. This
question was already raised by Lu and Weinstein [80, Question 4.8], who studied certain
Poisson brackets on U-homogeneous spaces that arise as a quotient of the Bruhat-Poisson
bracket on U.

As far as I know, affirmative answers to the above mentioned question have been given so far for only three types of U-homogeneous spaces, namely Podles's family of quantum 2-spheres [101], odd-dimensional complex quantum spheres SU(n+1)/SU(n) (cf. Vaksman and Soibelman [130]), and Stiefel manifolds U(n)/U(n-l) (cf. Podkolzin and Vainerman [102]). The relation between the irreducible *-representations of Podles spheres and the symplectic foliation of certain covariant Poisson brackets on the 2-sphere was observed by Lu and Weinstein [81].

In this chapter, the irreducible *-representations of a certain quantized *-algebra of functions on a generalized flag manifold are studied. More specifically, let G denote the complexification of U. The standard parabolic subgroups $P \subset G$ and the Bruhat-Poisson

bracket on G are defined with respect to a fixed choice of Cartan subalgebra and system of positive roots. The generalized flag manifold U/K ($K := U \cap P$) naturally becomes a Poisson U-homogeneous space (cf. Lu and Weinstein [80]). The quotient Poisson bracket on U/K is also called Bruhat-Poisson in [80]. The symplectic leaves of U/K coincide with the Schubert cells of the flag manifold $G/P \simeq U/K$.

It is straightforward to realize a quantum analogue $\mathbb{C}_q[K]$ of the algebra of polynomial functions on K as a quantum subgroup of $\mathbb{C}_q[U]$. The corresponding *-subalgebra $\mathbb{C}_q[U/K]$ of $\mathbb{C}_q[K]$ -invariant functions in $\mathbb{C}_q[U]$ may be regarded as a quantization of the Poisson algebra of functions on U/K endowed with the Bruhat-Poisson bracket. The main result in this chapter is a classification of all the irreducible *-representations of $\mathbb{C}_q[U/K]$ for an important subclass of flag manifolds, which contains in particular the irreducible compact Hermitian symmetric spaces. For this subclass it will be shown that the equivalence classes of irreducible *-representations are parametrized by the Schubert cells of U/K. It should be emphasized that the flag manifold U/K is regarded here as a real manifold. This means that the algebra of functions on U/K has a natural *-structure, which survives quantization and allows us to study *-representations in a way analogous to Soibelman's approach [110].

For an arbitrary generalized flag manifold U/K we describe in detail how irreducible *-representations of $\mathbb{C}_q[U]$ decompose under restriction to $\mathbb{C}_q[U/K]$. This decomposition corresponds precisely to the way symplectic leaves in U project to Schubert cells in the flag manifold U/K. It leads immediately to a classification of the irreducible *representations of the C^* -algebra $C_q(U/K)$, where $C_q(U/K)$ is obtained by taking the closure of $\mathbb{C}_q[U/K]$ with respect to the universal C^* -norm on $\mathbb{C}_q[U]$. The equivalence classes of the irreducible *-representations of $C_q(U/K)$ are naturally parametrized by the symplectic leaves of U/K endowed with the Bruhat-Poisson bracket.

For the classification of the irreducible *-representations of the quantized function algebra $\mathbb{C}_q[U/K]$ itself it is important to have some sort of Poincaré-Birkhoff-Witt (PBW) factorization of $\mathbb{C}_q[U/K]$, which in turn is closely related to the irreducible decomposition of tensor products of certain finite dimensional irreducible U-modules. Such a factorization is needed in order to develop some sort of highest weight representation theory for $\mathbb{C}_q[U/K]$. In Soibelman's paper [110], a crucial role is played by a similar factorization of $\mathbb{C}_q[U]$. From Soibelman's results a factorization of the algebra $\mathbb{C}_q[U/T]$, corresponding to P minimal parabolic in G, can be derived.

In this chapter a PBW type factorization for a different subclass of flag manifolds is derived using the so-called Parthasarathy-Ranga Rao-Varadarajan (PRV) conjecture. This conjecture was formulated as a follow-up to certain results in the paper [99] and was independently proved by Kumar [73] and Mathieu [88] (see also Littelmann [77]). The subclass of flag manifolds U/K considered here, can be characterized by the conditions that (U, K) is a Gelfand pair and that the Dynkin diagram of the subgroup K can be obtained from the Dynkin diagram of U by deleting one node (cf. Koornwinder [68]). These two conditions are satisfied for the irreducible compact Hermitian symmetric pairs (U, K).

6.1. INTRODUCTION

From an informal point of view, the PBW factorization in the above mentioned cases states that the quantized function algebra $\mathbb{C}_q[U/K]$ coincides with the quantized algebra of zero-weighted complex valued polynomials on U/K. Equivalently, the PBW factorization states that $\mathbb{C}_q[U/K]$ is generated as algebra by the product of the generalized quantum holomorphic Plücker coordinates and the generalized anti-quantum holomorphic Plücker coordinates (cf. Section 1.4). Recall that this result was already proved for the complex Grassmannian in the previous chapter (cf. Theorem 5.3.3).

The quantized algebra of zero-weighted complex valued polynomials can be naturally defined for arbitrary generalized flag manifold U/K. It is always a *-subalgebra of $\mathbb{C}_q[U/K]$ and invariant under the $\mathbb{C}_q[U]$ -coaction. It will be called the factorized *algebra associated with U/K. The factorized *-algebra is closely related to the quantized algebra of holomorphic polynomials on generalized flag manifolds studied by Soibelman [111], Lakshmibai and Reshetikhin [74], [75], and Jurco and Stovicek [52] (for the classical groups), as well as to the function spaces considered by Korogodsky [70].

In this chapter the irreducible *-representations of the factorized *-algebra associated with an arbitrary flag manifold U/K are classified and it is shown that the equivalence classes of irreducible *-representations are naturally parametrized by the symplectic leaves of U/K endowed with the Bruhat-Poisson bracket. In particular, a complete classification of the irreducible *-representations of $\mathbb{C}_q[U/K]$ is obtained whenever a PBW type factorization holds for $\mathbb{C}_q[U/K]$, i.e. whenever $\mathbb{C}_q[U/K]$ is equal to its factorized *-algebra. By yet unpublished results of the author, it turns out that $\mathbb{C}_q[U/K]$ equals its factorized *-algebra for *all* flag manifolds U/K; combined with the results of this chapter, a complete classification of the irreducible *-representations of $\mathbb{C}_q[U/K]$ is thus obtained.

The chapter is organized as follows. In Section 6.2 the results by Lu and Weinstein [80] and Soibelman [110] concerning the Bruhat-Poisson bracket on U and the quotient Poisson bracket on a flag manifold are reviewed. In Section 6.3 we recall some wellknown results on the "standard" quantization of the universal enveloping algebra of a simple complex Lie algebra and on its finite dimensional representations. Furthermore, the construction of the corresponding quantized function algebra $\mathbb{C}_{q}[U]$ is recalled and certain commutation relations between matrix coefficients of irreducible corepresentations of $\mathbb{C}_q[U]$ are presented. They will play a crucial role in the classification of the irreducible *-representations of the factorized *-algebra. In Section 6.4 the quantized algebra $\mathbb{C}_{q}[U/K]$ of functions on a flag manifold U/K and its associated factorized *subalgebra are defined. Furthermore it is shown that the factorized *-algebra is equal to $\mathbb{C}_{q}[U/K]$ for the subclass of flag manifolds referred to above. In Section 6.5 the restriction of an arbitrary irreducible *-representation of $\mathbb{C}_q[U]$ to $\mathbb{C}_q[U/K]$ is studied. We use here Soibelman's explicit realization of the irreducible *-representations of $\mathbb{C}_q[U]$ as tensor products of irreducible *-representations of $\mathbb{C}_q[SU(2)]$ (cf. [110], see also [60], [130] for SU(n)). As a corollary, a complete classification of the irreducible *-representations of the C^* -algebra $C_q(U/K)$ is obtained.

6. QUANTIZED FLAG MANIFOLDS AND IRREDUCIBLE *-REPRESENTATIONS

Section 6.6 is devoted to the classification of the irreducible *-representations of the factorized *-algebra associated with an arbitrary flag manifold. The techniques used in Section 6.6 are similar to those used by Soibelman [110] for the classification of the irreducible *-representations of $\mathbb{C}_q[U]$, and to those used by Joseph [49] to handle the more general problem of determining the primitive ideals of $\mathbb{C}_q[U]$.

6.2. Bruhat-Poisson brackets on flag manifolds

In this section results of Soibelman [110] and of Lu and Weinstein [80] about the Bruhat-Poisson structure on generalized flag manifolds are reviewed. Detailed discussions about the terminology related to the theory of Poisson-Lie groups are omitted. The reader who is unfamiliar with the terminology is referred to [11] and [80].

Let \mathfrak{g} be a complex simple Lie algebra with a fixed Cartan subalgebra $\mathfrak{h} \subset \mathfrak{g}$. Let G be the connected simply connected Lie group with Lie algebra \mathfrak{g} (regarded here as a real analytic Lie group).

Let $R \subset \mathfrak{h}^*$ be the root system associated with $(\mathfrak{g}, \mathfrak{h})$ and write \mathfrak{g}_{α} for the root space associated with $\alpha \in R$. Let $\Delta = \{\alpha_1, \ldots, \alpha_r\}$ be a basis of simple roots for R, and let R^+ (respectively R^-) be the set of positive (respectively negative) roots relative to Δ . We identify \mathfrak{h} with its dual by the Killing form κ . The non degenerate symmetric bilinear form on \mathfrak{h}^* induced by κ is denoted by (\cdot, \cdot) . Let $W \subset \operatorname{GL}(\mathfrak{h}^*)$ be the Weyl group of the root system R and write $s_i = s_{\alpha_i}$ for the simple reflection associated with $\alpha_i \in \Delta$.

For $\alpha \in R$ write $d_{\alpha} := (\alpha, \alpha)/2$. Let $H_{\alpha} \in \mathfrak{h}$ be the element associated with the coroot $\alpha^{\vee} := d_{\alpha}^{-1}\alpha \in \mathfrak{h}^*$ under the identification $\mathfrak{h} \simeq \mathfrak{h}^*$. Choose nonzero $X_{\alpha} \in \mathfrak{g}_{\alpha}$ $(\alpha \in R)$ such that for all $\alpha, \beta \in R$ we have $[X_{\alpha}, X_{-\alpha}] = H_{\alpha}, \kappa(X_{\alpha}, X_{-\alpha}) = d_{\alpha}^{-1}$ and $[X_{\alpha}, X_{\beta}] = c_{\alpha,\beta}X_{\alpha+\beta}$ with $c_{\alpha,\beta} = -c_{-\alpha,-\beta} \in \mathbb{R}$ whenever $\alpha + \beta \in R$. Let \mathfrak{h}_0 be the real form of \mathfrak{h} defined as the real span of the H_{α} 's $(\alpha \in R)$. Then

(6.2.1)
$$\mathfrak{u} := \sum_{\alpha \in R^+} \mathbb{R}(X_{\alpha} - X_{-\alpha}) \oplus \sum_{\alpha \in R^+} \mathbb{R}i(X_{\alpha} + X_{-\alpha}) \oplus i\mathfrak{h}_0$$

is a compact real form of g.

Set $\mathfrak{b} := \mathfrak{h}_0 \oplus \mathfrak{n}_+$ with $\mathfrak{n}_+ := \sum_{\alpha \in R^+}^{\oplus} \mathfrak{g}_{\alpha}$. Then, by the Iwasawa decomposition for \mathfrak{g} , the triple $(\mathfrak{g}, \mathfrak{u}, \mathfrak{b})$ is a Manin triple with respect to the imaginary part of the Killing form κ (cf. [80, Section 4]). Hence \mathfrak{u} , \mathfrak{b} and \mathfrak{g} naturally become Lie bialgebras. The dual Lie algebra \mathfrak{u}^* is isomorphic to \mathfrak{b} , and \mathfrak{g} may be identified with the classical double of \mathfrak{u} . The cocommutator $\delta : \mathfrak{g} \to \mathfrak{g} \land \mathfrak{g}$ of the Lie bialgebra \mathfrak{g} is coboundary, i.e.,

$$\delta(X) = (\mathrm{ad}_X \otimes 1 + 1 \otimes \mathrm{ad}_X)r,$$

with the classical r-matrix $r \in \mathfrak{g} \wedge \mathfrak{g}$ given by the following well-known skew solution of the Modified Classical Yang-Baxter Equation,

(6.2.2)
$$r = i \sum_{\alpha \in R^+} d_{\alpha} (X_{-\alpha} \otimes X_{\alpha} - X_{\alpha} \otimes X_{-\alpha}) \in \mathfrak{u} \wedge \mathfrak{u}.$$

The cocommutator on u coincides with the restriction of δ to u.

The corresponding Sklyanin bracket on the connected subgroup $U \subset G$ with Lie algebra \mathfrak{u} has

(6.2.3)
$$\Omega_g = l_g^{\otimes 2} r - r_g^{\otimes 2} r, \quad (g \in U)$$

as its associated Poisson tensor. Here l_g respectively r_g denote infinitesimal left respectively right translation. This particular Sklyanin bracket is often called Bruhat-Poisson, since its symplectic foliation is closely related to the Bruhat decomposition of G, as will be shown in a moment.

Let B be the connected subgroup of G with Lie algebra \mathfrak{b} , let $T \subset U$ be the maximal torus in U with Lie algebra $i\mathfrak{h}_0$, and set $B_+ := TB$. The analytic Weyl group $N_U(T)/T$, where $N_U(T)$ is the normalizer of T in U, is isomorphic to W. More explicitly, the isomorphism sends the simple reflection s_i to $\exp(\frac{\pi}{2}(X_{\alpha_i} - X_{-\alpha_i}))/T$. The double B_+ -cosets in G are parametrized by the elements of W. Hence we have the Bruhat decomposition

$$G = \coprod_{w \in W} B_+ w B_+.$$

By [133, Proposition 1.2.3.6] the Bruhat decomposition has the refinement

(6.2.4)
$$G = \coprod_{m \in N_U(T)} BmB$$

For $m \in N_U(T)$ we set $\Sigma_m := U \cap BmB$. Then $\Sigma_m \neq \emptyset$ for all $m \in N_U(T)$, and we have the disjoint union

$$(6.2.5) U = \coprod_{m \in N_U(T)} \Sigma_m.$$

Now recall that multiplication $U \times B \to G$ is a global diffeomorphism by the Iwasawa decomposition of G. So for any $b \in B$ and $u \in U$ there exists a unique $u^b \in U$ such that $bu \in u^b B$. As is easily verified, the map

(6.2.6)
$$U \times B \to U, \quad (u,b) \mapsto u^{b^{-1}}$$

is a right action of B on U, and the corresponding decomposition of U into B-orbits coincides with the decomposition (6.2.5). On the other hand, if we regard B as the Poisson-Lie group dual to U, the action (6.2.6) becomes the right dressing action of the dual group on U (cf. [80, Theorem 3.14]). Since the orbits in U under the right dressing action are exactly the symplectic leaves of the Poisson bracket on U (cf. [108, Theorem 13], [80, Theorem 3.15]), it follows that (6.2.5) coincides with the decomposition of U into symplectic leaves (cf. [110, Theorem 2.2]).

Next, we recall some results by Lu and Weinstein [80] concerning certain quotient Poisson brackets on generalized flag manifolds. Let $S \subset \Delta$ be a set of simple roots, and let P_S be the corresponding standard parabolic subgroup of G. The Lie algebra \mathfrak{p}_S of P_S is given by

$$\mathfrak{p}_{S} := \mathfrak{h} \oplus \bigoplus_{\alpha \in \Gamma_{S}} \mathfrak{g}_{\alpha}$$

with $\Gamma_S := R^+ \cup \{ \alpha \in R \mid \alpha \in \operatorname{span}(S) \}$. Let \mathfrak{l}_S be the Levi factor of \mathfrak{p}_S ,

(6.2.8)
$$\mathfrak{l}_{S} := \mathfrak{h} \oplus \bigoplus_{\alpha \in \Gamma_{S} \cap (-\Gamma_{S})} \mathfrak{g}_{\alpha},$$

and set $\mathfrak{k}_S := \mathfrak{p}_S \cap \mathfrak{u} = \mathfrak{l}_S \cap \mathfrak{u}$. Then \mathfrak{k}_S is a compact real form of \mathfrak{l}_S . Set $K_S := U \cap P_S \subset U$, then $K_S \subset U$ is a Poisson-Lie subgroup of U with Lie algebra \mathfrak{k}_S (cf. [80, Theorem 4.7]). Hence there is a unique Poisson bracket on U/K_S such that the natural projection $\pi : U \to U/K_S$ is a Poisson map. This bracket is also called Bruhat-Poisson. It is covariant in the sense that the natural left action $U \times U/K_S \to U/K_S$ is a Poisson map.

Let W_S be the subgroup of W generated by the simple reflections in S, then we have $P_S = B_+W_SB_+$ (cf. [133, Theorem 1.2.1.1]). It follows that the double cosets B_+xP_S ($x \in G$) are parametrized by the elements of W/W_S . Hence we have the Schubert cell decomposition of $U/K_S \simeq G/P_S$:

(6.2.9)
$$U/K_S = \coprod_{\overline{w} \in W/W_S} X_{\overline{w}}, \quad X_{\overline{w}} := (U \cap B_+ w P_S)/K_S \simeq B_+ w/P_S,$$

where $\overline{w} \in W/W_S$ is the right W_S -coset in W which contains w.

Now, by [80, Proposition 4.5], the subgroup K_S is invariant under the action of B, which implies that the B-action descends to U/K_S . The orbits in U/K_S coincide exactly with the Schubert cells. By [80, Theorem 4.6] the symplectic leaves of the Poisson manifold U/K_S are exactly the orbits under the B-action. We conclude (cf. [80, Theorem 4.7]):

THEOREM 6.2.1. The Schubert cells of the flag manifold U/K_S are the symplectic leaves of U/K_S endowed with the Bruhat-Poisson bracket.

Consider now the set of minimal coset representatives

$$(6.2.10) W^S := \{ w \in W \mid l(ws_\alpha) > l(w) \quad \forall \alpha \in S \}$$

 W^S is a complete set of coset representatives for W/W_S , i.e. any element $w \in W$ can be uniquely written as a product $w = w_1 w_2$ with $w_1 \in W^S$, $w_2 \in W_S$. The elements of W^S are minimal in the sense that

$$(6.2.11) \qquad l(w_1w_2) = l(w_1) + l(w_2), \quad (w_1 \in W^S, w_2 \in W_S),$$

where $l(w) := #(R^+ \cap wR^-)$ is the length function on W.

Observe that π maps the symplectic leaf $\Sigma_m \subset U$ onto the symplectic leaf $X_{\overline{w(m)}} \subset U/K_S$, where $w(m) := m/T \in W$. We write $\pi_m : \Sigma_m \to X_{\overline{w(m)}}$ for the surjective Poisson map obtained by restricting π to the symplectic leaf Σ_m . The minimality condition (6.2.10) translates to the following property of the map π_m .

PROPOSITION 6.2.2. Let $m \in N_U(T)$. Then $\pi_m : \Sigma_m \to X_{\overline{w(m)}}$ is a symplectic automorphism if and only if $w(m) \in W^S$.

PROOF. For $w \in W$ set

$$\mathfrak{n}_w := \bigoplus_{\alpha \in R^+ \cap wR^-} \mathfrak{g}_{\alpha}, \quad N_w := \exp(\mathfrak{n}_w).$$

Observe that the complex dimension of N_w is equal to l(w). Write $\operatorname{pr}_U : G \simeq U \times B \to U$ for the canonical projection. It is well-known that for $m \in N_U(T)$ and for $w \in W^S$ with representative $m_w \in N_U(T)$, the maps

$$\begin{split} \phi_m &: N_{w(m)} \to \Sigma_m, \quad n \mapsto \operatorname{pr}_U(nm), \\ \psi_w &: N_w \to X_{\overline{w}}, \quad n \mapsto \pi \left(\operatorname{pr}_U(nm_w) \right) \end{split}$$

are surjective diffeomorphisms (see for example [9, Proposition 1.1 and 5.1]). The map ψ_w is independent of the choice of representative m_w for w. It follows now from (6.2.11) by a dimension count that π_m can only be a diffeomorphism if $w(m) \in W^S$. On the other hand, if $m \in N_U(T)$ such that $w(m) \in W^S$, then $\pi_m = \psi_{w(m)} \circ \phi_m^{-1}$ and hence π_m is a diffeomorphism.

Soibelman [110] gave a description of the symplectic leaves Σ_m ($m \in N_U(T)$) as a product of two-dimensional leaves which turns out to have a nice generalization to the quantized setting (cf. Section 6.5). For $i \in [1, r]$, let $\gamma_i : SU(2) \hookrightarrow U$ be the embedding corresponding to the *i*th node of the Dynkin diagram of U. After a possible renormalization of the Bruhat-Poisson structure on SU(2), γ_i becomes an embedding of Poisson-Lie groups. Recall that the two-dimensional leaves of SU(2) are given by

$$S_t := \left\{ \begin{pmatrix} \alpha & \beta \\ -\overline{\beta} & \overline{\alpha} \end{pmatrix} \in \operatorname{SU}(2) \mid \operatorname{arg}(\beta) = \operatorname{arg}(t) \right\} \quad (t \in \mathbb{T})$$

where $\mathbb{T} \subset \mathbb{C}$ is the unit circle in the complex plane. The restriction of the embedding γ_i to $S_1 \subset SU(2)$ is a symplectic automorphism from S_1 onto the symplectic leaf $\Sigma_{m_i} \subset U$, where $m_i = \exp(\frac{\pi}{2}(X_{\alpha_i} - X_{-\alpha_i}))$. Recall that $m_i \in N_U(T)$ is a representative of the simple reflection $s_i \in W$.

For $m \in N_U(T)$ let $w(m) = s_{i_1}s_{i_2}\cdots s_{i_l}$ be a reduced expression for $w(m) := m/T \in W$, and let $t_m \in T$ be the unique element such that $m = m_{i_1}m_{i_2}\cdots m_{i_r}t_m$. Note that t_m depends on the choice of reduced expression for w(m). The map

$$(g_1,\ldots,g_l)\mapsto\gamma_{i_1}(g_1)\gamma_{i_2}(g_2)\cdots\gamma_{i_l}(g_l)t_m$$

defines a symplectic automorphism from $S_1^{\times l}$ onto the symplectic leaf $\Sigma_m \subset U$ (cf. [110, Section 2], [114]). Observe that the image of this map is independent of the choice of reduced expression for w(m), although the map itself is not.

Combined with Proposition 6.2.2 we now obtain the following description of the symplectic leaves of the generalized flag manifold U/K_S .

PROPOSITION 6.2.3. Let $m \in N_U(T)$ and set $w := m/T \in W$. Let $w_1 \in W^S$, $w_2 \in W_S$ be such that $w = w_1w_2$ and choose reduced expressions $w_1 = s_{i_1} \cdots s_{i_p}$ and $w_2 = s_{i_{p+1}} \cdots s_{i_l}$. Then the map

$$(g_1, g_2, \ldots, g_l) \mapsto \gamma_{i_1}(g_1)\gamma_{i_2}(g_2)\cdots\gamma_{i_l}(g_l)/K_S$$

is a surjective Poisson map from $S_1^{\times l}$ onto the Schubert cell $X_{\overline{w}}$. It factorizes through the projection $pr: S_1^{\times l} = S_1^{\times p} \times S_1^{\times (l-p)} \to S_1^{\times p}$. The quotient map from $S_1^{\times p}$ onto $X_{\overline{w}}$ is a symplectic automorphism. In particular, we have

$$X_{\overline{w}} = \left(\Sigma_{m_{i_1}} \Sigma_{m_{i_2}} \cdots \Sigma_{m_{i_p}} \right) / K_S. \quad \Box$$

See Lu [79] for more details in the case of the full flag manifold $(K_S = T)$.

6.3. Preliminaries on the quantized function algebra $\mathbb{C}_q[U]$

In this section some notations are introduced which are needed throughout the remainder of this chapter. First, we recall the definition of the quantized universal enveloping algebra associated with the simple complex Lie algebra g. We use the notations introduced in the previous section.

Set $d_i := d_{\alpha_i}$ and $H_i := H_{\alpha_i}$ for $i \in [1, r]$. Let $A = (a_{ij})$ be the Cartan matrix, i.e. $a_{ij} := d_i^{-1}(\alpha_i, \alpha_j)$. Note that $H_i \in \mathfrak{h}$ is the unique element such that $\alpha_j(H_i) = a_{ij}$ for all j. The weight lattice is given by

(6.3.1)
$$P = \{\lambda \in \mathfrak{h}^* \mid \lambda(H_i) = (\lambda, \alpha_i^{\vee}) \in \mathbb{Z} \quad \forall i\}$$

The fundamental weights $\varpi_{\alpha_i} = \varpi_i$ $(i \in [1, r])$ are characterized by $\varpi_i(H_j) = \delta_{ij}$ for all j. The set of dominant weights P_+ respectively regular dominant weights P_{++} is equal to \mathbb{K} -span $\{\varpi_{\alpha}\}_{\alpha \in \Delta}$ with $\mathbb{K} = \mathbb{Z}_+$ respectively \mathbb{N} .

The quantized universal enveloping algebra $U_q(\mathfrak{g})$ associated with the simple Lie algebra \mathfrak{g} is the unital associative algebra over \mathbb{C} with generators $K_i^{\pm 1}$, X_i^{\pm} (i = [1, r]) and relations

$$K_{i}K_{j} = K_{j}K_{i}, \quad K_{i}K_{i}^{-1} = K_{i}^{-1}K_{i} = 1$$

$$K_{i}X_{j}^{\pm}K_{i}^{-1} = q_{i}^{\pm\alpha_{j}(H_{i})}X_{j}^{\pm}$$

$$(6.3.2) \qquad X_{i}^{+}X_{j}^{-} - X_{j}^{-}X_{i}^{+} = \delta_{ij}\frac{K_{i} - K_{i}^{-1}}{q_{i} - q_{i}^{-1}}$$

$$\sum_{s=0}^{1-a_{ij}} (-1)^{s} \binom{1-a_{ij}}{s}_{q_{i}} (X_{i}^{\pm})^{1-a_{ij}-s}X_{j}^{\pm}(X_{i}^{\pm})^{s} = 0 \quad (i \neq j)$$

where $q_i := q^{d_i}$,

$$[a]_q := \frac{q^a - q^{-a}}{q - q^{-1}} \quad (a \in \mathbb{N}), \quad [0]_q := 1,$$

 $[a]_q! := [a]_q[a-1]_q \dots [1]_q$, and

$$\binom{a}{n}_q := rac{[a]_q!}{[a-n]_q! [n]_q!}$$

A Hopf algebra structure on $U_q(\mathfrak{g})$ is uniquely determined by the formulas

(6.3.3)
$$\begin{aligned} \Delta(X_i^+) &= X_i^+ \otimes 1 + K_i \otimes X_i^+, \quad \Delta(X_i^-) = X_i^- \otimes K_i^{-1} + 1 \otimes X_i^-, \\ \Delta(K_i^{\pm 1}) &= K_i^{\pm 1} \otimes K_i^{\pm 1}, \\ S(K_i^{\pm 1}) &= K_i^{\pm 1}, \quad S(X_i^+) = -K_i^{-1}X_i^+, \quad S(X_i^-) = -X_i^-K_i, \\ \varepsilon(K_i^{\pm 1}) &= 1, \quad \varepsilon(X_i^{\pm}) = 0. \end{aligned}$$

In fact, $U_q(\mathfrak{g})$ may be regarded as a quantization of the co-Poisson-Hopf algebra structure (cf. [11, Ch. 6]) on $U(\mathfrak{g})$ induced by the Lie bialgebra $(\mathfrak{g}, -i\delta)$, δ being the cocommutator of \mathfrak{g} associated with the *r*-matrix (6.2.2). $U_q(\mathfrak{g})$ becomes a Hopf *-algebra with *-structure on the generators given by

(6.3.4)
$$(K_i^{\pm 1})^* = K_i^{\pm 1}, \quad (X_i^+)^* = q_i^{-1} X_i^- K_i, \quad (X_i^-)^* = q_i K_i^{-1} X_i^+.$$

In the classical limit $q \rightarrow 1$, the *-structure becomes an involutive, conjugate-linear anti-automorphism of \mathfrak{g} with -1 eigenspace equal to the compact real form \mathfrak{u} defined in (6.2.1).

Let $U^{\pm} = U_q(\mathfrak{n}_{\pm})$ be the subalgebra of $U_q(\mathfrak{g})$ generated by X_i^{\pm} (i = [1, r]) and write $U^0 := U_q(\mathfrak{h})$ for the commutative subalgebra generated by $K_i^{\pm 1}$ (i = [1, r]). Let Q (respectively Q^+) be the integral (respectively positive integral) span of the positive roots. We have the direct sum decomposition

$$U^{\pm} = \bigoplus_{\alpha \in Q^+} U^{\pm}_{\pm \alpha},$$

where $U_{\pm\alpha}^{\pm} := \{\phi \in U^{\pm} \mid K_i \phi K_i^{-1} = q_i^{\pm \alpha(H_i)} \phi\}$. The Poincaré-Birkhoff-Witt Theorem for $U_q(\mathfrak{g})$ states that multiplication defines an isomorphism of vector spaces

$$U^- \otimes U^0 \otimes U^+ \to U_q(\mathfrak{g}).$$

In particular, $U_q(\mathfrak{g})$ is spanned by elements of the form $b_{-\eta}K^{\alpha}a_{\zeta}$ where $b_{-\eta} \in U^-_{\eta}, a_{\zeta} \in U^+_{\zeta}$ $(\eta, \zeta \in Q_+)$ and $\alpha \in Q$. Here we used the notation $K^{\alpha} = K_1^{k_1} \cdots K_r^{k_r}$ if $\alpha = \sum_i k_i \alpha_i$.

For a left $U_q(\mathfrak{g})$ -module V, we say that $0 \neq v \in V$ has weight $\mu \in \mathfrak{h}^*$ if $K_i \cdot v = q_i^{\mu(H_i)}v = q^{(\mu,\alpha_i)}v$ for all i. We write V_{μ} for the corresponding weight space. Recall that a P-weighted finite dimensional irreducible representation of $U_q(\mathfrak{g})$ is a highest weight module $V = V(\lambda)$ with highest weight $\lambda \in P_+$. If $v_\lambda \in V(\lambda)$ is a highest weight vector, we have $V(\lambda) = \sum_{\alpha \in Q^+}^{\oplus} U_{-\alpha}^- v_\lambda$ by the PBW Theorem, hence the set of weights $P(\lambda)$ of $V(\lambda)$ is a subset of the weight lattice P satisfying $\mu \leq \lambda$ for all $\mu \in P(\lambda)$. Here \leq is the dominance order on P (i.e. $\mu \leq \nu$ if $\nu - \mu \in Q^+$ and $\mu < \nu$ if $\mu \leq \nu$ and $\mu \neq \nu$).

We define irreducible finite dimensional P-weighted right $U_q(\mathfrak{g})$ -modules with respect to the opposite Borel subgroup. So the irreducible finite dimensional right $U_q(\mathfrak{g})$ -module $V(\lambda)$ with highest weight $\lambda \in P^+$ has the weight space decomposition $V(\lambda) = \sum_{\alpha \in Q^+}^{\oplus} v_{\lambda} U_{\alpha}^+$, where $v_{\lambda} \in V(\lambda)$ is the highest weight vector of $V(\lambda)$. The weights of the right $U_q(\mathfrak{g})$ -module $V(\lambda)$ coincide with the weights of the left $U_q(\mathfrak{g})$ -module $V(\lambda)$ and the dimensions of the corresponding weight spaces are the same.

The quantized algebra $\mathbb{C}_q[G]$ of functions on the connected simply connected complex Lie group G with Lie algebra \mathfrak{g} is the subspace in the linear dual $U_q(\mathfrak{g})^*$ spanned by the matrix coefficients of the finite dimensional irreducible representations $V(\lambda)$ ($\lambda \in P_+$). The Hopf *-algebra structure on $U_q(\mathfrak{g})$ induces a Hopf *-algebra structure on the quantized function algebra $\mathbb{C}_q[G] \subset U_q(\mathfrak{g})^*$ by the formulas

(6.3.5)
$$\begin{aligned} (\phi\psi)(X) &= (\phi\otimes\psi)\Delta(X), \quad 1(X) = \varepsilon(X) \\ \Delta(\phi)(X\otimes Y) &= \phi(XY), \quad \varepsilon(\phi) = \phi(1) \\ S(\phi)(X) &= \phi(S(X)), \quad (\phi^*)(X) = \overline{\phi(S(X)^*)}, \end{aligned}$$

where $\phi, \psi \in \mathbb{C}_q[G] \subset U_q(\mathfrak{g})^*$ and $X, Y \in U_q(\mathfrak{g})$. The algebra $\mathbb{C}_q[G]$ can be regarded as a quantization of the Poisson algebra of polynomial functions on the algebraic Poisson-Lie group G, where the Poisson structure on G is given by the Sklyanin bracket associated with the classical r-matrix -ir (cf. (6.2.2)). Since the *-structure (6.3.5) on $\mathbb{C}_q[G]$ is associated with the compact real form U of G in the classical limit, we will write $\mathbb{C}_q[U]$ for $\mathbb{C}_q[G]$ with this particular choice of *-structure. Note that $\mathbb{C}_q[U]$ is a $U_q(\mathfrak{g})$ -bimodule with the left respectively right action given by

(6.3.6)
$$(X.\phi)(Y) := \phi(YX), \quad (\phi.X)(Y) := \phi(XY)$$

where $\phi \in \mathbb{C}_q[U]$ and $X, Y \in U_q(\mathfrak{g})$. The finite dimensional irreducible $U_q(\mathfrak{g})$ -module $V(\lambda)$ of highest weight $\lambda \in P_+$ is unitarizable. Write (.,.) for a unitary inner product on $V(\lambda)$. Choose an orthonormal basis of $V(\lambda)$ with respect to (.,.),

(6.3.7)
$$\{v_{\mu}^{(i)} \mid \mu \in P(\lambda), i = [1, \dim(V(\lambda)_{\mu})]\},\$$

consisting of weight vectors $v_{\mu}^{(i)} \in V(\lambda)_{\mu}$. The index *i* will be omitted if dim $(V(\lambda)_{\mu}) = 1$. Set

(6.3.8)
$$C^{\lambda}_{\mu,i;\nu,j}(X) := (X \cdot v^{(j)}_{\nu}, v^{(i)}_{\mu}), \quad X \in U_q(\mathfrak{g}),$$

for $\mu, \nu \in P(\lambda)$ and $1 \leq i \leq \dim(V(\lambda)_{\mu}), 1 \leq j \leq \dim(V(\lambda)_{\nu})$. If $\dim(V(\lambda)_{\mu}) = 1$ respectively $\dim(V(\lambda)_{\nu}) = 1$ then the dependence on *i* respectively *j* in (6.3.8) is omitted. It is sometimes also convenient to use the notation

$$C_{v;w}^{\lambda}(X) := (X.w, v), \quad v, w \in V(\lambda), \ X \in U_q(\mathfrak{g}).$$

Note that when λ runs through P_+ and μ, i, ν and j run through the above-mentioned sets the matrix elements (6.3.8) form a linear basis of $\mathbb{C}_q[G]$. Furthermore, we have the

 $\circ C\lambda$

formulas

$$\Delta(C^{\lambda}_{\mu,i;\nu,j}) = \sum_{\sigma,s} C^{\lambda}_{\mu,i;\sigma,s} \otimes C^{\lambda}_{\sigma,s;\nu,j},$$
$$\varepsilon(C^{\lambda}_{\mu,i;\nu,j}) = \delta_{\mu,\nu}\delta_{i,j}, \quad (C^{\lambda}_{\mu,i;\nu,j})^* = S(C^{\lambda}_{\nu,j;\mu,i}),$$

where sums for which the summation sets are not specified are taken over the "obvious" choice of summation sets. Using the relations (6.3.9) and the Hopf algebra axiom for the antipode S we obtain

(6.3.10)
$$\sum_{\sigma,s} (C^{\lambda}_{\sigma,s;\mu,i})^* C^{\lambda}_{\sigma,s;\nu,j} = \delta_{\mu,\nu} \delta_{i,j}.$$

The elements $(C^{\lambda}_{\mu,i;\nu,j})^*$ are matrix coefficients of the dual representation $V(\lambda)^* \simeq$ $V(-\sigma_0\lambda)$ (here σ_0 is the longest element in W). In fact, let $\pi: U_q(\mathfrak{g}) \to \operatorname{End}(V(\lambda))$ be the representation of highest weight λ , and let (\cdot, \cdot) be an inner product with respect to which π is unitarizable. Fix an orthonormal basis of weight vectors $\{v_{\mu}^{(r)}\}$. Write π^* for the dual representation, i.e. $\pi^*(X)\phi = \phi \circ \pi(S(X))$ for $X \in U_q(\mathfrak{g})$ and $\phi \in V(\lambda)^*$. For $u \in V(\lambda)$ set $u^* := (\cdot, u) \in V(\lambda)^*$. Define an inner product on $V(\lambda)^*$ by

$$(u^*, v^*) := (\pi(K^{-2\rho})v, u), \quad u, v \in V(\lambda),$$

where $\rho = 1/2 \sum_{\alpha \in R^+} \alpha \in \mathfrak{h}^*$. Since $S^2(u) = K^{-2\rho} u K^{2\rho}$ for $u \in U_q(\mathfrak{g})$ and $(* \circ S)^2 = V^2 u K^{2\rho}$ Id on $U_q(\mathfrak{g})$, it follows that π^* is unitarizable with respect to the inner product (\cdot, \cdot) on $V(\lambda)^*$ and that $\{\phi_{-\mu}^{(i)} := q^{(\mu,\rho)}(v_{\mu}^{(i)})^*\}$ is an orthonormal basis of $V(\lambda)^*$ consisting of weight vectors (here $\phi_{-\mu}^{(i)}$ has weight $-\mu$). Defining the matrix coefficients $C_{-\mu,i;-\nu,j}^{-\sigma_0\lambda}$ of $(\pi^*, V(\lambda)^*)$ with respect to the orthonormal basis $\{\phi_{-\mu}^{(i)}\}$, we then have

(6.3.11)
$$(C^{\lambda}_{\mu,i;\nu,j})^* = q^{(\mu-\nu,\rho)} C^{-\sigma_0 \lambda}_{-\mu,i;-\nu,j}$$

(cf. [110, Proposition 3.3]). A fundamental role in Soibelman's theory of irreducible *representations of $\mathbb{C}_q[U]$ is played by a Poincaré-Birkhoff-Witt (PBW) type factorization of $\mathbb{C}_q[U]$. For $\lambda \in P_+$, set

(6.3.12)
$$B_{\lambda} := \operatorname{span}\{C_{v;v_{\lambda}}^{\lambda} \mid v \in V(\lambda)\}$$

Note that B_{λ} is a right $U_q(\mathfrak{g})$ -submodule of $\mathbb{C}_q[U]$ isomorphic to $V(\lambda)$. Set

(6.3.13)
$$A^+ := \bigoplus_{\lambda \in P_+} B_{\lambda}, \quad A^{++} := \bigoplus_{\lambda \in P_{++}} B_{\lambda}.$$

The subalgebra and right $U_q(\mathfrak{g})$ -module A^+ is equal to the subalgebra of left U^+ -invariant elements in $\mathbb{C}_q[U]$ (cf. [49]). The existence of a PBW type factorization of $\mathbb{C}_q[U]$ now amounts to the following statement.

THEOREM 6.3.1. [110, Theorem 3.1] The multiplication map $m : (A^{++})^* \otimes A^{++} \rightarrow A^{++}$ $\mathbb{C}_q[U]$ is surjective.

A detailed proof can be found in [49, Proposition 9.2.2]. The proof is based on certain results concerning decompositions of tensor products of irreducible finite dimensional $U_q(\mathfrak{g})$ -modules which can be traced back to Kostant in the classical case [71, Theorem 5.1]. The close connection between Theorem 6.3.1 and the decomposition of tensor products of irreducible $U_q(\mathfrak{g})$ -modules becomes clear by observing that

(6.3.14)
$$(B_{\lambda})^* B_{\mu} \simeq V(\lambda)^* \otimes V(\mu)$$

as right $U_q(\mathfrak{g})$ -modules.

Important for the study of *-representations of $\mathbb{C}_q[U]$ is some detailed information about the commutation relations between matrix elements in $\mathbb{C}_q[U]$. In view of Theorem 6.3.1, we are especially interested in commutation relations between the $C^{\lambda}_{\mu,i;\lambda}$ and $C^{\Lambda}_{\nu,j;\Lambda}$ respectively between the $C^{\lambda}_{\mu,i;\lambda}$ and $(C^{\Lambda}_{\nu,j;\Lambda})^*$, where $\lambda, \Lambda \in P_+$. To state these commutation relations we need to introduce certain vector subspaces of $\mathbb{C}_q[U]$. Let $\lambda, \Lambda \in P_+$ and $\mu \in P(\lambda), \nu \in P(\Lambda)$, then we set

(6.3.15)
$$N(\mu,\lambda;\nu,\Lambda) := \operatorname{span}\{C^{\lambda}_{v;v_{\lambda}}C^{\Lambda}_{w;v_{\lambda}} \mid (v,w) \in sN\},\\ N^{opp}(\mu,\lambda;\nu,\Lambda) := \operatorname{span}\{C^{\Lambda}_{w;v_{\lambda}}C^{\lambda}_{v;v_{\lambda}} \mid (v,w) \in sN\}$$

where $sN := sN(\mu, \lambda; \nu, \Lambda)$ is the set of pairs $(v, w) \in V(\lambda)_{\mu'} \times V(\Lambda)_{\nu'}$ with $\mu' > \mu$, $\nu' < \nu$ and $\mu' + \nu' = \mu + \nu$. Furthermore, set

(6.3.16)
$$O(\mu,\lambda;\nu,\Lambda) := \operatorname{span}\{(C^{\lambda}_{v;v_{\lambda}})^{*}C^{\Lambda}_{w;v_{\Lambda}} \mid (v,w) \in sO\},\ O^{opp}(\mu,\lambda;\nu,\Lambda) := \operatorname{span}\{C^{\Lambda}_{w;v_{\Lambda}}(C^{\lambda}_{v;v_{\lambda}})^{*} \mid (v,w) \in sO\}$$

where $sO := sO(\mu, \lambda; \nu, \Lambda)$ is the set of pairs $(v, w) \in V(\lambda)_{\mu'} \times V(\Lambda)_{\nu'}$ with $\mu' < \mu$, $\nu' < \nu$ and $\mu - \mu' = \nu - \nu'$. If sN (respectively sO) is empty, then let $N = N^{opp} = \{0\}$ (respectively $O = O^{opp} = \{0\}$).

PROPOSITION 6.3.2. Let $\lambda, \Lambda \in P_+$ and $v \in V(\lambda)_{\mu}$, $w \in V(\Lambda)_{\nu}$. (i) The matrix elements $C^{\lambda}_{v;v_{\lambda}}$ and $C^{\Lambda}_{w;v_{\lambda}}$ satisfy the commutation relation

$$C_{v;v_{\lambda}}^{\lambda}C_{w;v_{\Lambda}}^{\Lambda} = q^{(\lambda,\Lambda)-(\mu,\nu)}C_{w;v_{\Lambda}}^{\Lambda}C_{v;v_{\lambda}}^{\lambda} \mod N(\mu,\lambda;\nu,\Lambda).$$

Moreover, we have $N = N^{opp}$.

(ii) The matrix elements $(C_{v;v_{\lambda}}^{\lambda})^*$ and $C_{w;v_{\lambda}}^{\Lambda}$ satisfy the commutation relation

$$(C_{v;v_{\lambda}}^{\lambda})^{*}C_{w;v_{\Lambda}}^{\Lambda} = q^{(\mu,\nu)-(\lambda,\Lambda)}C_{w;v_{\Lambda}}^{\Lambda}(C_{v;v_{\lambda}}^{\lambda})^{*} \mod O(\mu,\lambda;\nu,\Lambda).$$

Moreover, we have $O = O^{opp}$.

Soibelman [110] derived commutation relations using the universal *R*-matrix whereas Joseph [49, Section 9.1] used the Poincaré-Birkhoff-Witt Theorem for $U_q(\mathfrak{g})$ and the left respectively right action (6.3.6) of $U_q(\mathfrak{g})$ on $\mathbb{C}_q[U]$. Although the commutation relations formulated here are slightly sharper, the proof can be derived in a similar manner and will therefore be omitted.

6.4. QUANTIZED FUNCTION ALGEBRAS ON GENERALIZED FLAG MANIFOLDS

COROLLARY 6.3.3. Let $\lambda, \Lambda \in P_+$ and $v \in V(\lambda)_{\mu}, w \in V(\Lambda)_{\nu}$. Then

(6.3.17)
$$C_{v;v_{\lambda}}^{\lambda}C_{w;v_{\Lambda}}^{\Lambda} = q^{(\mu,\nu)-(\lambda,\Lambda)}C_{w;v_{\Lambda}}^{\Lambda}C_{v;v_{\lambda}}^{\lambda} \mod N(\nu,\Lambda;\mu,\lambda).$$

Note that Proposition 6.3.2(i) and Corollary 6.3.3 give two different ways to rewrite $C_{v;v_{\lambda}}^{\lambda} C_{w;v_{\lambda}}^{\Lambda}$ as elements of the vector space

$$W_{\lambda,\Lambda} := \operatorname{span}\{C_{w';v_{\Lambda}}^{\Lambda} C_{v';v_{\lambda}}^{\lambda} \mid v' \in V(\lambda), \ w' \in V(\Lambda)\}.$$

We will need both "inequivalent" commutation relations (Proposition 6.3.2(i) and Corollary 6.3.3) in later sections. It follows in particular that, when $v' \in V(\lambda)$ and $w' \in V(\Lambda)$ run through a basis, the elements $C_{w';v_{\Lambda}}^{\Lambda} C_{v';v_{\Lambda}}^{\lambda}$ are (in general) linearly dependent. This also follows from the following two observations. On the one hand, $W_{\lambda,\Lambda} \simeq V(\lambda + \Lambda)$ as right $U_q(\mathfrak{g})$ -modules. On the other hand, $V(\lambda + \Lambda)$ occurs with multiplicity one in $V(\lambda) \otimes V(\Lambda)$, whereas in general $V(\lambda) \otimes V(\Lambda)$ has other irreducible components too.

By contrast, the commutation relation given in Proposition 6.3.2(ii) is unique in the sense that, when $v \in V(\lambda)$ and $w \in V(\Lambda)$ run through a basis, the $C^{\Lambda}_{w;v_{\Lambda}}(C^{\lambda}_{v;v_{\lambda}})^*$ are linearly independent (cf. (6.3.14)).

We end this section by recalling the special case $\mathfrak{g} = \mathfrak{sl}(2,\mathbb{C})$. Set

(6.3.18)
$$\begin{aligned} t_{11} &:= C_{\varpi_1;\varpi_1}^{\varpi_1}, \quad t_{12} := C_{\varpi_1;-\varpi_1}^{\varpi_1}, \\ t_{21} &:= C_{-\varpi_1;\varpi_1}^{\varpi_1}, \quad t_{22} := C_{-\varpi_1;-\varpi_1}^{\varpi_1}. \end{aligned}$$

Then it is well-known that the t_{ij} 's generate the algebra $\mathbb{C}_q[SU(2)]$. The commutation relations

(6.3.19)
$$t_{k1}t_{k2} = qt_{k2}t_{k1}, \quad t_{1k}t_{2k} = qt_{2k}t_{1k} \quad (k = 1, 2),$$
$$t_{12}t_{21} = t_{21}t_{12}, \quad t_{11}t_{22} - t_{22}t_{11} = (q - q^{-1})t_{12}t_{21},$$
$$t_{11}t_{22} - qt_{12}t_{21} = 1$$

characterize the algebra structure of $\mathbb{C}_q[SU(2)]$ in terms of the generators t_{ij} . The *structure is uniquely determined by the formulas $t_{11}^* = t_{22}$, $t_{12}^* = -qt_{21}$.

6.4. Quantized function algebras on generalized flag manifolds

Let S be any subset of the simple roots Δ . Sometimes S will be identified with the index set $\{i \mid \alpha_i \in S\}$. Let $\mathfrak{p}_S \subset \mathfrak{g}$ be the corresponding standard parabolic subalgebra, given explicitly by (6.2.7). Define the quantized universal enveloping algebra $U_q(\mathfrak{l}_S)$ associated with the Levi factor \mathfrak{l}_S of \mathfrak{p}_S as the subalgebra of $U_q(\mathfrak{g})$ generated by $K_i^{\pm 1}$ $(i \in [1, r])$ and X_i^{\pm} $(i \in S)$. Observe that $U_q(\mathfrak{l}_S)$ is a Hopf *-subalgebra of $U_q(\mathfrak{g})$.

For later use in this section we briefly discuss the finite dimensional representation theory of $U_q(\mathfrak{l}_S)$. Recall that \mathfrak{l}_S is a reductive Lie algebra with centre

(6.4.1)
$$Z(\mathfrak{l}_S) = \bigcap_{i \in S} \operatorname{Ker}(\alpha_i) \subset \mathfrak{h}.$$

Moreover, we have direct sum decompositions

$$\mathfrak{h} = Z(\mathfrak{l}_S) \oplus \mathfrak{h}_S, \quad \mathfrak{l}_S = Z(\mathfrak{l}_S) \oplus \mathfrak{l}_S^0$$

where $\mathfrak{h}_S = \operatorname{span} \{H_i\}_{i \in S}$ and \mathfrak{l}_S^0 is the semisimple part of \mathfrak{l}_S . The semisimple part \mathfrak{l}_S^0 is explicitly given by

(6.4.3)
$$\mathfrak{l}^0_S := \mathfrak{h}_S \oplus \bigoplus_{\alpha \in \Gamma_S \cap (-\Gamma_S)} \mathfrak{g}_\alpha.$$

Define the quantized universal enveloping algebra $U_q(\mathfrak{l}_S^0)$ associated with the semisimple part \mathfrak{l}_S^0 of \mathfrak{l}_S as the subalgebra of $U_q(\mathfrak{g})$ generated by $K_i^{\pm 1}$ and X_i^{\pm} for all $i \in S$. Observe that $U_q(\mathfrak{l}_S^0)$ is a Hopf *-subalgebra of $U_q(\mathfrak{g})$.

PROPOSITION 6.4.1. Any finite dimensional $U_q(\mathfrak{l}_S)$ -module V which is completely reducible as $U_q(\mathfrak{h})$ -module, is completely reducible as $U_q(\mathfrak{l}_S)$ -module.

PROOF. Let V be a finite dimensional left $U_q(\mathfrak{l}_S)$ -module which is completely reducible as $U_q(\mathfrak{h})$ -module. Then the linear subspace

$$V^+ := \{ v \in V \mid X_i^+ v = 0 \quad \forall i \in S \}$$

is $U_q(\mathfrak{h})$ -stable and splits as a direct sum of weight spaces. Let $\{v_i\}_i$ be a linear basis of V^+ consisting of weight vectors, and set $V_i := U_q(\mathfrak{l}_S^0)v_i$. Since $U_q(\mathfrak{l}_S^0)$ is the quantized universal enveloping algebra associated with a semisimple Lie algebra, it follows that $V = \sum_i^{\oplus} V_i$ is a decomposition of V into irreducible $U_q(\mathfrak{l}_S^0)$ -modules. On the other hand, the V_i are $U_q(\mathfrak{l}_S)$ -stable since the vectors v_i are weight vectors. Hence $V = \sum_i^{\oplus} V_i$ is a decomposition of V into irreducible $U_q(\mathfrak{l}_S)$ -modules.

There are obvious notions of weight vectors and weights for $U_q(\mathfrak{l}_S)$ -modules. With a suitably extended interpretation of the notion of highest weight, the irreducible finite dimensional $U_q(\mathfrak{l}_S)$ -modules may be characterized in terms of highest weights. We shall only be interested in irreducible $U_q(\mathfrak{l}_S)$ -modules with weights in the lattice P. For instance, the restriction of an irreducible P-weighted $U_q(\mathfrak{g})$ -module to $U_q(\mathfrak{l}_S)$ decomposes into such irreducible $U_q(\mathfrak{l}_S)$ -modules.

Branching rules for the restriction of finite dimensional representations of $U_q(\mathfrak{g})$ to $U_q(\mathfrak{l}_S)$ are determined by the behaviour of the corresponding characters. Since the characters for *P*-weighted irreducible finite dimensional representations of $U_q(\mathfrak{g})$ and $U_q(\mathfrak{l}_S)$ are the same as for the corresponding representations of \mathfrak{g} and \mathfrak{l}_S , we have the following proposition.

PROPOSITION 6.4.2. Let $\lambda \in P_+$. The multiplicity of any *P*-weighted irreducible $U_q(\mathfrak{l}_S)$ -module in the irreducible decomposition of the restriction of the $U_q(\mathfrak{g})$ -module $V(\lambda)$ to $U_q(\mathfrak{l}_S)$ is the same as in the classical case.

Next, we define the quantized algebra of functions on U/K_S . Let $\iota_S^* : U_q(\mathfrak{g})^* \to U_q(\mathfrak{l}_S)^*$ be the dual of the Hopf *-embedding $\iota_S : U_q(\mathfrak{l}_S) \hookrightarrow U_q(\mathfrak{g})$, and set

$$\mathbb{C}_q[L_S] := \iota_S^*(\mathbb{C}_q[G]) = \{\phi \circ \iota_S \mid \phi \in \mathbb{C}_q[G]\}.$$

The formulas (6.3.5) uniquely determine a Hopf *-algebra structure on $\mathbb{C}_q[L_S]$, and ι_S^* then becomes a Hopf *-algebra morphism. We write $\mathbb{C}_q[K_S]$ for $\mathbb{C}_q[L_S]$ with this

particular choice of *-structure. Assume now that $S \neq \Delta$. Define a *-subalgebra $\mathbb{C}_q[U/K_S] \subset \mathbb{C}_q[U]$ by

(6.4.4)
$$\mathbb{C}_{q}[U/K_{S}] := \{ \phi \in \mathbb{C}_{q}[U] \mid (\mathrm{id} \otimes \iota_{S}^{*})\Delta(\phi) = \phi \otimes 1 \} \\ = \{ \phi \in \mathbb{C}_{q}[U] \mid X.\phi = \varepsilon(X)\phi, \quad \forall X \in U_{q}(\mathfrak{l}_{S}) \}$$

The algebra $\mathbb{C}_q[U/K_S]$ is a left $\mathbb{C}_q[U]$ -subcomodule of $\mathbb{C}_q[U]$. We call it the quantized algebra of functions on the generalized flag manifold U/K_S .

Similarly, the quantized function algebra $\mathbb{C}_q[K_S^0]$ corresponding to the semisimple part K_S^0 of K_S can be defined as the image of the dual of the natural embedding $U_q(\mathfrak{l}_S^0) \hookrightarrow U_q(\mathfrak{g})$. Its Hopf *-algebra structure is again given by the formulas (6.3.5). The subalgebra $\mathbb{C}_q[U/K_S^0]$ then consists by definition of all right $\mathbb{C}_q[K_S^0]$ -invariant elements in $\mathbb{C}_q[U]$. Note that $\mathbb{C}_q[U/K_S^0] \subset \mathbb{C}_q[U]$ is a left $U_q(\mathfrak{h})$ -submodule and that $\mathbb{C}_q[U/K_S]$ coincides with the subalgebra of $U_q(\mathfrak{h})$ -invariant elements in $\mathbb{C}_q[U/K_S^0]$.

We now turn to PBW type factorizations of the algebra $\mathbb{C}_q[U/K_S]$. Let P(S), $P_+(S)$, respectively $P_{++}(S)$ be the \mathbb{K} -span $\{\varpi_{\alpha}\}_{\alpha \in S}$ with $\mathbb{K} = \mathbb{Z}, \mathbb{Z}_+$ respectively \mathbb{N} . Set $S^c := \Delta \setminus S$. The quantized algebra A_S^{hol} of holomorphic polynomials on U/K_S is defined by

(6.4.5)
$$A_S^{hol} := \bigoplus_{\lambda \in P_+(S^c)} B_\lambda \subset \mathbb{C}_q[U],$$

where B_{λ} is given by (6.3.12) (cf. [74], [75], [111], [52] and [70]). Note that A_{S}^{hol} is a right $U_{q}(\mathfrak{g})$ -comodule subalgebra of $\mathbb{C}_{q}[U]$, (6.4.5) being the (multiplicity free) decomposition of A_{S}^{hol} into irreducible $U_{q}(\mathfrak{g})$ -modules. A_{S}^{hol} is generated as algebra by the elements $C_{\mu,i;\varpi_{s}}^{\varpi_{s}}$ ($\mu \in P(\varpi_{s}), i \in [1, \dim(V(\varpi_{s})_{\mu}], s \in S^{c}$), which are called the (generalized) quantum holomorphic Plücker coordinates on U/K_{S} . The right $U_{q}(\mathfrak{g})$ -module algebra $(A_{S}^{hol})^{*} \subset \mathbb{C}_{q}[U]$ is called the quantized algebra of anti-holomorphic polynomials on U/K_{S} . The elements $(C_{\mu,i;\varpi_{s}}^{\varpi_{s}})^{*}$ ($\mu \in P(\varpi_{s}), i \in [1, \dim(V(\varpi_{s})_{\mu}], s \in S^{c}$) are called the (generalized) quantum anti-holomorphic Plücker coordinates.

LEMMA 6.4.3. *The linear subspace*

$$A_S^0 := m((A_S^{hol})^* \otimes A_S^{hol}) \subset \mathbb{C}_q[U],$$

where m is the multiplication map of $\mathbb{C}_q[U]$, is a right $U_q(\mathfrak{g})$ -submodule *-subalgebra of $\mathbb{C}_q[U]$.

PROOF. Proposition 6.3.2(ii) implies that A_S^0 is a subalgebra of $\mathbb{C}_q[U]$. The other assertions are immediate.

The subalgebra A_S^0 is generated as algebra by the quantum holomorphic and anti-holomorphic Plücker coordinates on U/K_S .

REMARK 6.4.4. In the classical setting (q = 1), the algebra A_S^0 (# $S^c = 1$) can be interpreted as algebra of functions on the product of an affine spherical *G*-variety with its dual. The *G*-module structure on A_S^0 is then related to the doubled *G*-action (see [97],

[98] for the terminology). These (and related) G-varieties have been studied in several papers, see for example [98], [97] and [78].

The algebra $A_S^0 \subset \mathbb{C}_q[U]$ is stable under the left $U_q(\mathfrak{h})$ -action, so we can speak of $U_q(\mathfrak{h})$ -weighted elements in A_S^0 . Let A_S be the left $U_q(\mathfrak{h})$ -invariant elements of A_S^0 . Then $A_S \subset \mathbb{C}_q[U]$ is a right $U_q(\mathfrak{g})$ -module *-subalgebra of $\mathbb{C}_q[U]$.

LEMMA 6.4.5. We have $A_S^0 \subset \mathbb{C}_q[U/K_S^0]$, so in particular $A_S \subset \mathbb{C}_q[U/K_S]$. Furthermore,

(6.4.6)
$$A_S = span\{(C_{v;v_{\lambda}}^{\lambda})^* C_{w;v_{\lambda}}^{\lambda} \mid \lambda \in P_+(S^c), \ v, w \in V(\lambda)\}.$$

PROOF. Choose $\lambda \in P_+(S^c)$ and $i \in S$. Then we have $X_i^+ \cdot v_\lambda = 0$ and $K_i \cdot v_\lambda = v_\lambda$. It follows that $\mathbb{C}v_\lambda \subset V(\lambda)$ is a one dimensional $U_{q_i}(\mathfrak{sl}(2;\mathbb{C}))$ -submodule, where we consider the $U_{q_i}(\mathfrak{sl}(2;\mathbb{C}))$ action on $V(\lambda)$ via the embedding $\phi_i \colon U_{q_i}(\mathfrak{sl}(2;\mathbb{C})) \hookrightarrow U_q(\mathfrak{g})$. It follows that $X_i^- \cdot v_\lambda = 0$. This readily implies that $A_S^0 \subset \mathbb{C}_q[U/K_S^0]$. The remaining assertions are immediate.

In view of Lemma 6.4.5, we may consider the subalgebra A_S^0 as the quantum analogue of the algebra of complex-valued polynomial functions on the real manifold U/K_S^0 and the algebra A_S as the quantum analogue of the algebra of zero-weighted complex-valued polynomials functions on U/K_S^0 . Note that A_S is generated as algebra by the products $C_{\mu,i;\varpi_s}^{\varpi_s}(C_{\nu,j;\varpi_s}^{\varpi_s})^*$ $(\mu,\nu \in P(\varpi_s), i \in [1, \dim(V(\varpi_s)_{\mu})], j \in [1, \dim(V(\varpi_s)_{\nu})]$ and $s \in S^c$.

DEFINITION 6.4.6. $A_S \subset \mathbb{C}_q[U/K_S]$ is called the factorized *-subalgebra associated with U/K_S .

In view of Theorem 6.3.1, there is reason to expect that the factorized algebra A_S is equal to $\mathbb{C}_q[U/K_S]$ for any generalized flag manifold U/K_S . We prove this fact (see Theorem 6.4.10) for a certain subclass of generalized flag manifolds that we shall define and classify in the following proposition. For the proof in these cases we use the so-called Parthasarathy-Ranga Rao-Varadarajan (PRV) conjecture, which was proved independently by Kumar [73] and Mathieu [88]. The PRV conjecture gives information about which irreducible constituents occur in tensor products of irreducible finite dimensional g-modules.

Recall the notations introduced in Section 6.2. The following proposition was observed by Koornwinder [68].

PROPOSITION 6.4.7. ([68]) Let U be a connected, simply connected compact Lie group with Lie algebra u, and let $\mathfrak{p} \subset \mathfrak{g}$ be a standard maximal parabolic subalgebra. Let $K \subset U$ be the connected subgroup with Lie algebra $\mathfrak{k} := \mathfrak{p} \cap \mathfrak{u}$. Then (U, K) is a Gelfand pair if and only if one of the following three conditions are satisfied: (i) (U, K) is an irreducible compact Hermitian symmetric pair; (ii) $(U, K) \simeq (SO(2l+1), U(l)), \quad (l \ge 2);$ (iii) $(U, K) \simeq (Sp(l), U(1) \times Sp(l-1)), \quad (l \ge 2).$

PROOF. For a list of the irreducible compact Hermitian symmetric pairs see [40, Ch. X, Table V]. The proposition follows from this list and the classification of the compact Gelfand pairs (U, K) with U simple (cf. [72, Tabelle 1]).

Let (U, K) be a pair from the list (i)–(iii) in Proposition 6.4.7, and let $(\mathfrak{u}, \mathfrak{k})$ be the associated pair of Lie algebras. Then $\mathfrak{k} = \mathfrak{k}_S$ for some subset $S \subset \Delta$ with $\#S^c = 1$. We call the simple root $\alpha \in S^c$ the Gelfand node associated with (U, K).

A dominant weight $\lambda \in P_+$ is called spherical if the the subspace of K-fixed vectors in $V(\lambda)$ is one dimensional. The corresponding representation $V(\lambda)$ is then also called spherical. We write $P_+^K \subset P_+$ for the subset of dominant spherical weights.

PROPOSITION 6.4.8. Let (U, K) be a pair from the list (i)–(iii) in Proposition 6.4.7, and let $\alpha \in \Delta$ be the associated Gelfand node with corresponding fundamental weight $\varpi := \varpi_{\alpha}$. Then we have a multiplicity free irreducible decomposition of $U_q(\mathfrak{g})$ -modules of the form

$$V(\varpi)^* \otimes V(\varpi) \simeq \bigoplus_{i=0}^l V(\mu_i)$$

for certain $l \in \mathbb{N}$, where $\mu_0 := 0 \in P_+$ and $\{\mu_i\}_{i=1}^l$ is a subset of the dominant spherical weights P_+^K . Furthermore, every $\lambda \in P_+^K$ can be uniquely written as a \mathbb{Z}_+ -linear combination of the μ_i 's $(i \in [1, l])$.

DEFINITION 6.4.9. The spherical weights μ_i $(i \in [1, l])$ are called the fundamental spherical weights associated with (U, K).

PROOF. It is well known that the trivial representation V(0) occurs with multiplicity one in the tensor product decomposition of $V(\varpi)^* \otimes V(\varpi)$. Furthermore, observe that

(6.4.7)
$$V(\varpi)^* \otimes V(\varpi) \simeq (B_{\varpi})^* B_{\varpi} \subset A_{\{\alpha\}^c} \subset \mathbb{C}_q[U/K]$$

as right $U_q(\mathfrak{g})$ -modules. By Proposition 6.4.2 we have the multiplicity free decomposition as right $U_q(\mathfrak{g})$ -modules

(6.4.8)
$$\mathbb{C}_q[U/K] \simeq \bigoplus_{\lambda \in P_+^K} V(\lambda),$$

from which it follows that the decomposition of $V(\varpi)^* \otimes V(\varpi)$ is multiplicity free, and that its irreducible constituents are all spherical.

Krämer [72, Tabelle 1] presented for each pair (U, K) from the list (i)–(iii) in Proposition 6.4.7 a set of dominant spherical weights $\{\mu_i\}_{i=1}^l$ satisfying the property that every $\lambda \in P_+^K$ can be uniquely written as a \mathbb{Z}_+ -linear combination of the μ_i 's $(i \in [1, l])$. The μ_i 's are explicitly given as \mathbb{Z}_+ -linear combination of the fundamental dominant weights ϖ_j $(j \in [1, r])$. In case of the Hermitean symmetric spaces U/K, there is an elegant procedure to recover the μ_i 's as linear combinations of the fundamental dominant weights from the corresponding Satake diagrams [125]. We show now that all spherical representations $V(\mu_i)$ $(i \in [1, l])$ are constituents of $V(\varpi)^* \otimes V(\varpi)$ by using the PRV conjecture, which states the following. Let $\lambda, \mu \in P_+$ and $w \in W$. Let $[\lambda + w\mu]$ be the unique element in P_+ which lies in the W-orbit of $\lambda + w\mu$. Then $V([\lambda + w\mu])$ occurs with multiplicity at least one in $V(\lambda) \otimes V(\mu)$. For each pair (U, K) from the list **(i)–(iii)** of Proposition 6.4.7, it is now possible to find explicit Weyl group elements $w_i \in W$ such that

$$[\varpi - w_i \varpi] = \mu_i, \quad (i = [1, l])$$

Combined with the PRV conjecture and the fact that $V(\varpi)^* \simeq V(-\sigma_0 \varpi)$, this implies that $V(\mu_i)$ is a consistent of $V(\varpi)^* \otimes V(\varpi)$ for all $i \in [1, l]$.

As an example, we follow the procedure for the compact Hermitian symmetric pair (U, K) = (SO(2l), U(l)) $(l \ge 2)$. We use the standard realization of the root system R of type D_l in the *l*-dimensional vector space $V = \sum_{i=1}^{l} \mathbb{R}\varepsilon_i$, with basis given by $\alpha_i = \varepsilon_i - \varepsilon_{i+1}$ (i = [1, l-1]) and $\alpha_l = \varepsilon_{l-1} + \varepsilon_l$. The fundamental weights are given by

$$arpi_i = arepsilon_1 + arepsilon_2 + \ldots + arepsilon_i, \quad (i < l - 1), \ arpi_{l-1} = (arepsilon_1 + arepsilon_2 + \ldots + arepsilon_{l-1} - arepsilon_l)/2, \ arpi_l = (arepsilon_1 + arepsilon_2 + \ldots + arepsilon_{l-1} + arepsilon_l)/2.$$

Set $\varpi = \varpi_l$ (i.e. $S^c = \{\alpha_l\}$). Let σ_i be the linear map defined by $\varepsilon_j \mapsto -\varepsilon_j$ (j = i, i+1)and $\varepsilon_j \mapsto \varepsilon_j$ otherwise. Then $\sigma_i \in W$ (i = [1, l-1]). If l = 2l' + 1, then

(6.4.9)
$$\begin{aligned} \varpi - \sigma_1 \sigma_3 \dots \sigma_{2i-1} \varpi = \varpi_{2i}, \quad (i = [1, l' - 1]), \\ \varpi - \sigma_1 \sigma_3 \dots \sigma_{2l'-1} \varpi = \varpi_{l-1} + \varpi_l. \end{aligned}$$

If l = 2l' then we have

(6.4.10)
$$\begin{aligned} \overline{\omega} &-\sigma_1 \sigma_3 \dots \sigma_{2i-1} \overline{\omega} = \overline{\omega}_{2i}, \qquad (i = [1, l'-1]), \\ \overline{\omega} &-\sigma_1 \sigma_3 \dots \sigma_{2l'-1} \overline{\omega} = 2\overline{\omega}_l. \end{aligned}$$

By comparison with [72, Tabelle 1] it follows from (6.4.9) (respectively (6.4.10)) that all the fundamental spherical weights of the pair (U, K) = (SO(2l), U(l)) have been obtained. The other cases are checked in a similar manner.

To complete the proof, we have to show that the $V(\mu_i)$ $(i \in [0, l])$ are the only irreducible constituents which can occur in the tensor product decomposition of $V(\varpi)^* \otimes V(\varpi)$. This is also proved case by case. The cases corresponding to the exceptional groups can be directly verified using for instance the maple-package "qtensor" of Stembridge [115]¹. The special case $(U, K) = (SU(p+l), S(U(p) \times U(l)))$ of this proposition was proven in the previous chapter, see (4.5.14). The remaining cases can be checked by showing that for $\lambda \in P_+^K \setminus {\mu_i}_{i=0}^l$, we have $\lambda \not\leq \varpi - \sigma_0 \varpi$, which implies that $V(\lambda)$ cannot occur as constituent of $V(\varpi)^* \otimes V(\varpi)$.

We are now in the position to prove the main result of this section.

¹http://www.math.lsa.umich.edu/~jrs/maple.html
THEOREM 6.4.10. The factorized *-subalgebra A_S is equal to $\mathbb{C}_q[U/K_S]$ if (i) $S = \emptyset$, i.e. $U/K_S = U/T$ is the full flag manifold; (ii) $\#S^c = 1$ and the simple root $\alpha \in S^c$ is a Gelfand node.

PROOF. To prove (i) we look at the simultaneous eigenspace decomposition of $\mathbb{C}_q[U]$ with respect to the left $U_q(\mathfrak{h})$ -action on $\mathbb{C}_q[U]$. The simultaneous eigenspace corresponding to the character ε on $U_q(\mathfrak{h})$ is exactly $\mathbb{C}_q[U/T]$. Using Soibelman's factorization of $\mathbb{C}_q[U]$ (cf. Theorem 6.3.1) and Lemma 6.4.5, it is then easily checked that $\mathbb{C}_q[U/T] = A_{\emptyset}$. To prove (ii) we note that

$$\bigoplus_{i=0}^{i} V(\mu_i) \simeq (B_{\varpi})^* B_{\varpi} \subset A_S$$

as right $U_q(\mathfrak{g})$ -modules by Proposition 6.4.8 and (6.4.7) (here we use the notations as introduced in Proposition 6.4.8). Now $\mathbb{C}_q[U]$ is an integral domain (cf. [49, Lemma 9.1.9 (i)]), hence $v_\lambda v_\mu \in A_S$ is a highest weight vector of highest weight $\lambda + \mu$ if $v_\lambda, v_\mu \in A_S$ are highest weight vectors of highest weight λ respectively μ . It follows that

$$\bigoplus_{\lambda \in P_+^K} V(\lambda) \hookrightarrow A_{\{\alpha\}^c}$$

as right $U_q(\mathfrak{g})$ -modules. Combining with (6.4.8), it follows that $A_{\{\alpha\}^c} = \mathbb{C}_q[U/K_{\{\alpha\}^c}]$, as requisted.

Observe that Theorem 6.4.10 for quantum Grassmannians was already proved in the previous chapter. In fact, for quantum complex Grassmannians we have shown that $A_S^0 = \mathbb{C}_q[U/K_S^0]$ in Theorem 5.4.1, so in particular $A_S = \mathbb{C}_q[U/K_S]$.

REMARK 6.4.11. By yet unpublished results of the author, it turns out that A_S equals $\mathbb{C}_q[U/K_S]$ for any generalized flag manifold U/K_S .

In the remainder of the chapter we study the irreducible *-representations of the *algebras A_S and $\mathbb{C}_q[U/K_S]$. In the next section the restriction of the irreducible *-representations of $\mathbb{C}_q[U]$ to the *-algebras A_S and $\mathbb{C}_q[U/K_S]$ is considered first.

6.5. Restriction of irreducible *-representations to $\mathbb{C}_q[U/K]$

We first recall some results from Soibelman [110] concerning the irreducible *-representations of $\mathbb{C}_q[U]$. Let $\{e_i\}_{i\in\mathbb{Z}_+}$ be the standard orthonormal basis of $l_2(\mathbb{Z}_+)$. Write $\mathcal{B}(l_2(\mathbb{Z}_+))$ for the algebra of bounded linear operators on $l_2(\mathbb{Z}_+)$. Then the formulas

(6.5.1)
$$\begin{aligned} \pi_q(t_{11})e_j &= \sqrt{(1-q^{2j})}e_{j-1}, \quad \pi_q(t_{12})e_j &= -q^{j+1}e_j, \\ \pi_q(t_{21})e_j &= q^j e_j, \quad \pi_q(t_{22})e_j &= \sqrt{(1-q^{2(j+1)})}e_{j+1}, \end{aligned}$$

with $\pi_q(t_{11})e_0 = 0$, uniquely determine an irreducible *-representation

$$\pi_q: \mathbb{C}_q[SU(2)] \to \mathcal{B}(l_2(\mathbb{Z}_+)).$$

Now the dual of the injective Hopf *-algebra morphism $\phi_i : U_{q_i}(\mathfrak{sl}(2;\mathbb{C})) \hookrightarrow U_q(\mathfrak{g})$ corresponding to the *i*th node of the Dynkin diagram $(i \in [1, r])$ is a surjective Hopf *-algebra morphism $\phi_i^* : \mathbb{C}_q[U] \twoheadrightarrow \mathbb{C}_{q_i}[SU(2)]$. In particular, we obtain irreducible *-representations $\pi_i := \pi_{q_i} \circ \phi_i^* : \mathbb{C}_q[U] \to \mathcal{B}(l_2(\mathbb{Z}_+))$.

On the other hand, there is a family of one dimensional *-representations τ_t of $\mathbb{C}_q[U]$ parametrized by the maximal torus $t \in T \simeq \mathbb{T}^r$ ($\mathbb{T} \subset \mathbb{C}$ denoting the unit circle in the complex plane). More explicitly, let $\iota_T : U_q(\mathfrak{h}) \hookrightarrow U_q(\mathfrak{g})$ be the natural Hopf *algebra embedding, and set $\mathbb{C}_q[T] := \operatorname{span} \{\phi_\mu\}_{\mu \in P} \subset U_q(\mathfrak{h})^*$, where $\phi_\mu(K^\sigma) := q^{(\mu,\sigma)}$ for $\sigma \in Q$. The formulas (6.3.5) define a Hopf *-algebra structure on $\mathbb{C}_q[T]$. Then $\iota_T^* : \mathbb{C}_q[U] \to \mathbb{C}_q[T], \iota_T^*(\phi) := \phi \circ \iota_T$ is a surjective Hopf *-algebra morphism. Any irreducible *-representation of $\mathbb{C}_q[T]$ is one dimensional and can be written as $\tilde{\tau}_t(\phi_\mu) :=$ t^{μ} for a unique $t \in T \simeq \mathbb{T}^r$. Here $t^{\mu} := t_1^{m_1} \dots t_r^{m_r}$ for $\mu = \sum_{i=1}^r m_i \varpi_i$. So we obtain a one dimensional *-representation $\tau_t := \tilde{\tau}_t \circ \iota_T^*$ of $\mathbb{C}_q[U]$, which is given explicitly on matrix elements $C_{\mu,i;\nu,j}^{\lambda}$ by the formula

(6.5.2)
$$\tau_t(C^{\lambda}_{\mu,i;\nu,j}) = \delta_{\mu,\nu}\delta_{i,j}t^{\mu}.$$

The following theorem completely describes the irreducible *-representations of $\mathbb{C}_q[U]$.

THEOREM 6.5.1 (Soibelman [110]). Let $\sigma \in W$, and fix a reduced expression $\sigma = s_{i_1}s_{i_1}\cdots s_{i_l}$. The *-representation

(6.5.3)
$$\pi_{\sigma} := \pi_{i_1} \otimes \pi_{i_2} \otimes \cdots \otimes \pi_{i_k}$$

does not depend on the choice of reduced expression (up to equivalence). The set

$$[\pi_{\sigma} \otimes \tau_t \mid t \in T, \sigma \in W]$$

is a complete set of mutually inequivalent irreducible *-representations of $\mathbb{C}_{q}[U]$.

Here tensor products of *-representations are defined in the usual way by means of the coalgebra structure on $\mathbb{C}_q[U]$. The irreducible representation π_e with respect to the unit element $e \in W$ is the one dimensional *-representation associated with the counit ε on $\mathbb{C}_q[U]$. In Soibelman's terminology, the representations $\pi_\sigma \otimes \tau_t$ are said to be associated with the Schubert cell X_σ of U/T (cf. Section 6.2).

We also mention here an important property of the kernel of π_{σ} , which we will repeatedly need later on. Let $U_q(\mathfrak{b}_+)$ be the subalgebra of $U_q(\mathfrak{g})$ generated by the $K_i^{\pm 1}$ and the X_i^+ ($i \in [1, r]$). For any $\lambda \in P_+$, the *-representation π_{σ} satisfies

(6.5.4)
$$\pi_{\sigma}(C_{v;v_{\lambda}}^{\lambda}) = 0 \quad \left(v \notin U_q(\mathfrak{b}_+)v_{\sigma\lambda}\right), \quad \pi_{\sigma}(C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda}) \neq 0$$

(cf. [110, Theorem 5.7]). Formula (6.5.4) combined with [9, Lemma 2.12] shows that the classical limit of the kernel of π_{σ} formally tends to the ideal of functions vanishing on X_{σ} .

Fix now a subset $S \subsetneq \Delta$. We freely use the notations introduced earlier in this chapter. Our next goal is to describe how the *-representations π_{σ} decompose under restriction to the subalgebra $\mathbb{C}_q[U/K_S]$. Consider the selfadjoint operators

(6.5.5)
$$L_{\sigma\lambda;\lambda} := \pi_{\sigma}((C^{\lambda}_{\sigma\lambda;\lambda})^* C^{\lambda}_{\sigma\lambda;\lambda})$$

for $\lambda \in P_+(S^c)$. Let $\sigma = s_{i_1} \cdots s_{i_l}$ be a reduced expression for σ , and set $\pi_{\sigma} = \pi_{i_1} \otimes \pi_{i_2} \otimes \cdots \otimes \pi_{i_l}$. Then it follows from [110, Proof of Proposition 5.2] (see also [110, Proof of Proposition 5.8]) that

(6.5.6)
$$\pi_{\sigma}(C^{\lambda}_{\sigma\lambda;\lambda}) = c \,\pi_{q_{i_1}}(t_{21})^{(\lambda,\gamma_1^{\vee})} \otimes \pi_{q_{i_2}}(t_{21})^{(\lambda,\gamma_2^{\vee})} \otimes \cdots \otimes \pi_{q_{i_l}}(t_{21})^{(\lambda,\gamma_l^{\vee})}$$

where the scalar $c \in \mathbb{T}$ depends on the particular choices of bases for the irreducible representations $V(\mu)$ ($\mu \in P_+$), and with

(6.5.7)
$$\gamma_k := s_{i_l} s_{i_{l-1}} \cdots s_{i_{k+1}} (\alpha_{i_k}) \quad (1 \le k \le l-1), \quad \gamma_l := \alpha_{i_l}$$

The proof of (6.5.6), which was given in [110] under the assumption that $\lambda \in P_{++}$, is in fact valid for all dominant weights $\lambda \in P_+$. It follows from (6.5.1), (6.5.5) and (6.5.6) that $l_2(\mathbb{Z}_+)^{\otimes l(\sigma)}$ decomposes as an orthogonal direct sum of eigenspaces for $L_{\sigma\lambda;\lambda}$,

(6.5.8)
$$l_2(\mathbb{Z}_+)^{\otimes l(\sigma)} = \bigoplus_{\gamma \in I(\lambda)} H_\gamma(\lambda),$$

where $I(\lambda) \subset (0,1]$ denotes the set of eigenvalues of $L_{\sigma\lambda;\lambda}$, and $H_{\gamma}(\lambda)$ denotes the eigenspace of $L_{\sigma\lambda;\lambda}$ corresponding to the eigenvalue $\gamma \in I(\lambda)$ (we suppress the dependance on σ if there is no confusion possible). Observe that $1 \in I(\lambda)$ and that $L_{\sigma\lambda;\lambda}$ is injective.

Recall the definition of the set W^S of minimal coset representatives (cf. (6.2.10)). An alternative characterization of W^S is given by

(6.5.9)
$$W^S = \{ \sigma \in W \mid \sigma(R_S^+) \subset R^+ \},\$$

where $R_S^+ := R^+ \cap \text{span}\{S\}$ (cf. [9, Proposition 5.1 (iii)]). Using this alternative description of W^S we obtain the following properties of $L_{\sigma\lambda;\lambda}$ for $\lambda \in P_{++}(S^c)$.

PROPOSITION 6.5.2. Suppose that $\sigma \in W^S$ and $\lambda \in P_{++}(S^c)$. Then (i) $L_{\sigma\lambda;\lambda}$ is a compact operator;

(ii) The eigenspace $H_1(\lambda)$ of $L_{\sigma\lambda;\lambda}$ corresponding to the eigenvalue 1 is spanned by the vector $e_0^{\otimes l(\sigma)}$.

PROOF. Fix a $\lambda \in P_{++}(S^c)$, and let $\sigma = s_{i_1}s_{i_2}\cdots s_{i_l}$ be a reduced expression of a minimal coset representative $\sigma \in W^S$. It is well-known that

(6.5.10)
$$R^+ \cap \sigma^{-1}(R^-) = \{\gamma_k\}_{k=1}^l,$$

where the γ_k are defined by (6.5.7). We have $\gamma_k \in R^+ \setminus R_S^+$ by (6.5.9). It follows that $(\lambda, \gamma_k^{\vee}) > 0$ for all k, since $\lambda \in P_{++}(S^c)$. By (6.5.1) and (6.5.6) it follows that $H_1(\lambda) = \text{span}\{e_0^{\otimes l(\sigma)}\}$ and that $H_{\gamma}(\lambda)$ is finite dimensional for all $\gamma \in I(\lambda)$. Since the spectrum of $L_{\sigma\lambda;\lambda}$ (which is equal to $I(\lambda) \cup \{0\}$) does not have a limit point except 0, we conclude that $L_{\sigma\lambda;\lambda}$ is a compact operator (cf. [105, Theorem 12.30]).

We recall now the following well-known inequalities for weights of finite dimensional irreducible representations of \mathfrak{g} (or, equivalently, $U_q(\mathfrak{g})$).

PROPOSITION 6.5.3. Let $\lambda \in P_+$ and $\mu, \nu \in P(\lambda)$. Then $(\lambda, \lambda) \ge (\mu, \nu)$, and equality holds if and only if $\mu = \nu \in W\lambda$.

For a proof of the proposition, see for instance [53, Proposition 11.4]. The proof is based on the following lemma, which we will also need later on. The lemma is a slightly weaker version of [53, Lemma 11.2].

LEMMA 6.5.4. Let $\lambda \in P_+$ and $\mu \in P(\lambda) \setminus \{\lambda\}$, and let $m_i \in \mathbb{Z}_+$ $(i \in [1, r])$ be the expansion coefficients defined by $\lambda - \mu = \sum_i m_i \alpha_i$. Then there is an $1 \le i \le r$ with $m_i > 0$ and $\lambda(H_i) \ne 0$.

We now have the following proposition, which can be regarded as a quantum analogue of the "if" part of Proposition 6.2.2.

PROPOSITION 6.5.5. Let $\sigma \in W^S$. Then π_{σ} restricts to an irreducible *-representation of the factorized *-algebra A_S . In particular, π_{σ} restricts to an irreducible *-representation of $\mathbb{C}_q[U/K_S]$.

PROOF. Let $\lambda \in P_{++}(S^c)$ and $\sigma \in W^S$. Suppose $H \subset l_2(\mathbb{Z}_+)^{\otimes l(\sigma)}$ is a non-zero closed subspace invariant under $\pi_{\sigma|A_S}$. Set $\gamma := ||L_{\sigma\lambda;\lambda|H}||$. Then $\gamma > 0$ since $L_{\sigma\lambda;\lambda}$ is injective, and γ is an eigenvalue of $L_{\sigma\lambda;\lambda|H}$ by Proposition 6.5.2(i). Let H_{γ} be the corresponding eigenspace. We claim that

(6.5.11)
$$\pi_{\sigma}((C_{\mu,i;\lambda}^{\lambda})^*C_{\mu,i;\lambda}^{\lambda})H_{\gamma} = 0, \quad \mu \neq \sigma \lambda$$

Suppose for the moment that the claim is correct. Then (6.3.10) and (6.5.11) imply $\gamma = 1$, hence $H_{\gamma} = \text{span}\{e_0^{\otimes l(\sigma)}\}$ by Proposition 6.5.2(ii). So every non-zero closed invariant subspace contains the vector $e_0^{\otimes l(\sigma)}$. Since H^{\perp} is also a closed invariant subspace, we must have $H^{\perp} = \{0\}$, i.e. $H = l_2(\mathbb{Z}_+)^{\otimes l(\sigma)}$. Remains therefore to prove the claim (6.5.11). By (6.5.4) we have $\pi_{\sigma}(C_{\mu,i;\lambda}^{\lambda}) = 0$ if $\mu < \sigma\lambda$. Hence

$$\begin{split} L_{\sigma\lambda;\lambda}\pi_{\sigma}((C^{\lambda}_{\mu,i;\lambda})^{*}C^{\lambda}_{\mu,i;\lambda}) &= q^{(\lambda,\lambda)-(\mu,\sigma\lambda)}\pi_{\sigma}((C^{\lambda}_{\mu,i;\lambda}C^{\lambda}_{\sigma\lambda;\lambda})^{*}C^{\lambda}_{\sigma\lambda;\lambda}C^{\lambda}_{\mu,i;\lambda}) \\ &= q^{2(\lambda,\lambda)-2(\mu,\sigma\lambda)}\pi_{\sigma}((C^{\lambda}_{\mu,i;\lambda})^{*}(C^{\lambda}_{\sigma\lambda;\lambda})^{*}C^{\lambda}_{\sigma\lambda;\lambda}C^{\lambda}_{\mu,i;\lambda}) \\ &= q^{2(\lambda,\lambda)-2(\mu,\sigma\lambda)}\pi_{\sigma}((C^{\lambda}_{\mu,i;\lambda})^{*}C^{\lambda}_{\sigma\lambda;\lambda})\pi_{\sigma}((C^{\lambda}_{\sigma\lambda;\lambda})^{*}C^{\lambda}_{\mu,i;\lambda}), \end{split}$$

where we used Proposition 6.3.2(i) in the second equality and Proposition 6.3.2(ii) in the first and third equality. So (6.5.11) will then follow from

(6.5.12)
$$\pi_{\sigma}((C^{\lambda}_{\sigma\lambda;\lambda})^*C^{\lambda}_{\mu,i;\lambda})H_{\gamma} = 0, \quad \mu \neq \sigma\lambda$$

in view of the injectivity of $L_{\sigma\lambda;\lambda}$. Fix $h \in H_{\gamma}$ and $\mu \in P(\lambda)$ with $\mu \neq \sigma\lambda$. By Lemma 6.4.5 we have $(C^{\lambda}_{\sigma\lambda;\lambda})^* C^{\lambda}_{\mu;i;\lambda} \in A_S \subset \mathbb{C}_q[U/K_S]$, hence the vector

(6.5.13)
$$\tilde{h} := \pi_{\sigma}((C^{\lambda}_{\sigma\lambda;\lambda})^* C^{\lambda}_{\mu;i;\lambda})h$$

lies in the invariant subspace H. Again using the commutation relations given in Proposition 6.3.2 and Corollary 6.3.3, we see that \tilde{h} is an eigenvector of $L_{\sigma\lambda;\lambda}$ with eigenvalue $\tilde{\gamma} := q^{2(\lambda,\sigma^{-1}(\mu)-\lambda)}\gamma$. We have $\tilde{\gamma} > \gamma$ by Proposition 6.5.3. By the maximality of γ , we conclude that $\tilde{h} = 0$. This proves (6.5.12), hence also the claim (6.5.11).

DEFINITION 6.5.6. The irreducible *-representation π_{σ} ($\sigma \in W^S$) of $\mathbb{C}_q[U/K_S]$ is said to be associated with the Schubert cell $X_{\overline{\sigma}} \subset U/K_S$.

The following proposition can be regarded as a quantum analogue of Proposition 6.2.3 as well as of the "only if" part of Proposition 6.2.2.

PROPOSITION 6.5.7. Let $\sigma \in W$, and let $\sigma = uv$ be the unique decomposition of σ with $u \in W^S$ and $v \in W_S$. For $\pi_{\sigma} = \pi_u \otimes \pi_v$ (cf. (6.2.11)) and $t \in T$, we have

$$(\pi_{\sigma} \otimes \tau_t)(a) = \pi_u(a) \otimes id^{\otimes l(v)}, \quad a \in \mathbb{C}_q[U/K_S].$$

PROOF. Recall that the one dimensional *-representation τ_t factorizes through ι_T^* : $\mathbb{C}_q[U] \to \mathbb{C}_q[T]$ and that π_i factorizes through $\phi_i^* : \mathbb{C}_q[U] \to \mathbb{C}_{q_i}[SU(2)]$. The maps ι_T^* and ϕ_i^* ($i \in S$) factorize through $\iota_S^* : \mathbb{C}_q[U] \to \mathbb{C}_q[K_S]$ since the ranges of ι_T and ϕ_i ($i \in S$) lie in the Hopf-subalgebra $U_q(\mathfrak{l}_S)$. Hence $\pi_v \otimes \tau_t$ ($v \in W_S, t \in T$) factorizes through ι_S^* , say $\pi_v \otimes \tau_t = \pi_{v,t} \circ \iota_S^*$. Then we have for $a \in \mathbb{C}_q[U/K_S]$,

$$\begin{aligned} \pi_{\sigma} \otimes \tau_t)(a) &= (\pi_u \otimes \pi_v \otimes \tau_t) \circ \Delta(a) \\ &= (\pi_u \otimes \pi_{v,t}) \circ (\mathrm{id} \otimes \iota_S^*) \Delta(a) \\ &= \pi_u(a) \otimes \pi_{v,t}(1) = \pi_u(a) \otimes \mathrm{id}^{\otimes l(v)}, \end{aligned}$$

which completes the proof of the proposition.

(

LEMMA 6.5.8. The *-representations $\{\pi_{\sigma}\}_{\sigma \in W^S}$, considered as *-representations of A_S respectively $\mathbb{C}_q[U/K_S]$, are mutually inequivalent.

PROOF. Let $\sigma, \sigma' \in W^S$ with $\sigma \neq \sigma'$ and $\lambda \in P_{++}(S^c)$. Then $\sigma\lambda \neq \sigma'\lambda$, since the isotropy subgroup $\{\sigma \in W | \sigma\lambda = \lambda\}$ is equal to W_S by Chevalley's Lemma (cf. [58, Proposition 2.72]). Without loss of generality we may assume that $\sigma\lambda \not\geq \sigma'\lambda$. Then we have $\pi_{\sigma'}((C^{\lambda}_{\sigma\lambda,\lambda})^*C^{\lambda}_{\sigma\lambda,\lambda}) = 0$ by (6.5.4). On the other hand, $L_{\sigma\lambda;\lambda}$ is injective. It follows that $\pi_{\sigma} \neq \pi_{\sigma'}$ as *-representations of A_S .

Let now $\|.\|_u$ be the universal C^* -norm on $\mathbb{C}_q[U]$ (cf. [23, Section 4]), so

(6.5.14)
$$\|a\|_u := \sup_{\sigma \in W, t \in T} \|(\pi_\sigma \otimes \tau_t)(a)\|, \quad a \in \mathbb{C}_q[U].$$

Let $C_q(U)$ (respectively $C_q(U/K_S)$) be the completion of the *-algebra $\mathbb{C}_q[U]$ (respectively of the *-algebra $\mathbb{C}_q[U/K_S]$) with respect to $||.||_u$. All *-representations $\pi_\sigma \otimes \tau_t$ of $\mathbb{C}_q[U]$ extend to *-representations of the C^* -algebra $C_q(U)$ by continuity. The results of this section can now be summarized as follows.

THEOREM 6.5.9. Let $S \subsetneq \Delta$. Then $\{\pi_{\sigma}\}_{\sigma \in W^S}$ is a complete set of mutually inequivalent irreducible *-representations of $C_q(U/K_S)$.

PROOF. This follows from the previous results, since every irreducible *-representation of $C_q(U/K_S)$ appears as an irreducible component of $\sigma_{|C_q(U/K_S)}$ for some irreducible *-representation σ of $C_q(U)$ (cf. [26, Proposition 2.10.2]).

Theorem 6.5.9 does not imply that $\{\pi_{\sigma}\}_{\sigma \in W^S}$ is a complete set of irreducible *-representations of the *-algebra $\mathbb{C}_q[U/K_S]$ itself. Indeed, it is not clear that any irreducible *-representation of $\mathbb{C}_q[U/K_S]$ can be continuously extended to a *-representation of $C_q(U/K_S)$. In the remainder of this chapter we will deal with the classification of the irreducible *-representations of A_S . In particular, this will yield a complete classification of the irreducible *-representations of $\mathbb{C}_q[U/K_S]$ for the generalized flag manifolds U/K_S for which the PBW factorization is valid (cf. Theorem 6.4.10 and Remark 6.4.11).

6.6. Irreducible *-representations of A_S

Let $S \subsetneq \Delta$ be any subset. In this section it is shown that $\{\pi_{\sigma}\}_{\sigma \in W^S}$ exhausts the set of irreducible *-representations of A_S (up to equivalence). We fix therefore an arbitrary irreducible *-representation $\tau : A_S \to \mathcal{B}(H)$ and we will show that $\tau \simeq \pi_{\sigma}$ for a (unique) $\sigma \in W^S$. In order to associate the proper minimal coset representative $\sigma \in W^S$ with τ , we need to study the range $\tau(A_S) \subset \mathcal{B}(H)$ of τ in more detail. For $\lambda \in P_+(S^c)$ and $\mu, \nu \in P(\lambda)$, let $\tau^{\lambda}(\mu; \nu), \tau^{\lambda}(\nu) \subset \mathcal{B}(H)$ be the linear subspaces

(6.6.1)
$$\begin{aligned} \tau^{\lambda}(\mu;\nu) &:= \{\tau((C^{\lambda}_{v;v_{\lambda}})^*C^{\lambda}_{w;v_{\lambda}}) \,|\, v \in V(\lambda)_{\mu}, \, w \in V(\lambda)_{\nu}\}, \\ \tau^{\lambda}(\nu) &:= \{\tau((C^{\lambda}_{v;v_{\lambda}})^*C^{\lambda}_{w;v_{\lambda}}) \,|\, v \in V(\lambda), \, w \in V(\lambda)_{\nu}\}. \end{aligned}$$

For $\lambda \in P_+(S^c)$, set

(6.6.2)
$$D(\lambda) := \{\nu \in P(\lambda) \mid \tau^{\lambda}(\nu) \neq \{0\}\}$$

and let $D_m(\lambda)$ be the set of weights $\nu \in D(\lambda)$ such that $\nu' \notin D(\lambda)$ for all $\nu' < \nu$. By (6.3.10), we have $D(\lambda) \neq \emptyset$, hence also $D_m(\lambda) \neq \emptyset$. We start with a lemma which is useful for the computation of commutation relations in $\tau(A_S) \subset \mathcal{B}(H)$.

LEMMA 6.6.1. Let $\lambda, \Lambda \in P_+(S^c)$ and $\nu \in D_m(\lambda)$. Let $v \in V(\lambda), v' \in V(\lambda)_{\nu'}$ with $\nu' < \nu$ and $w, w' \in V(\Lambda)$. Then the product of the four matrix elements $(C_{v;v_{\lambda}}^{\lambda})^*$, $C_{v';v_{\lambda}}^{\lambda}, (C_{w;v_{\lambda}}^{\Lambda})^*$, and $C_{w';v_{\Lambda}}^{\Lambda}$, taken in an arbitrary order, is contained in $Ker(\tau)$.

PROOF. Since $\text{Ker}(\tau)$ is a two-sided *-ideal in A_S , it follows from the definitions that

$$(C^{\Lambda}_{w;v_{\Lambda}})^* C^{\Lambda}_{w';v_{\Lambda}} (C^{\lambda}_{v;v_{\lambda}})^* C^{\lambda}_{v';v_{\lambda}} \in \operatorname{Ker}(\tau)$$

If the product of the four matrix coefficients is taken in a different order, then we can rewrite it by Proposition 6.3.2 and by Corollary 6.3.3 as a linear combination of products of matrix elements

$$(C^{\Lambda}_{u;v_{\Lambda}})^* C^{\Lambda}_{u';v_{\Lambda}} (C^{\lambda}_{x;v_{\lambda}})^* C^{\lambda}_{x';v_{\lambda}}$$

with $x' \in V(\lambda)_{\nu''}$ and $\nu'' \leq \nu' < \nu$. These are all contained in $\text{Ker}(\tau)$, since $\nu \in D_m(\lambda)$.

LEMMA 6.6.2. Let
$$\lambda \in P_+(S^c)$$
 and $\nu \in D_m(\lambda)$. Then
(i) $\tau^{\lambda}(\nu;\nu) \neq \{0\}$;
(ii) $\nu = \sigma \lambda$ for some $\sigma \in W^S$.

PROOF. Let $\lambda \in P_+(S^c)$ and $\nu \in D_m(\lambda)$. Fix weight vectors $v \in V(\lambda)_{\mu}$, $w \in V(\lambda)_{\nu}$ such that $T_{v;w} := \tau((C_{v;v_{\lambda}}^{\lambda})^* C_{w;v_{\lambda}}^{\lambda}) \neq 0$. By Lemma 6.6.1, we compute

(6.6.3)
$$(T_{v;w})^* T_{v;w} = q^{(\mu,\nu)-(\lambda,\lambda)} \tau(C^{\lambda}_{v;v_{\lambda}}(C^{\lambda}_{v;v_{\lambda}}C^{\lambda}_{w;v_{\lambda}})^* C^{\lambda}_{w;v_{\lambda}})$$
$$= \tau(C^{\lambda}_{v;v_{\lambda}}(C^{\lambda}_{v;v_{\lambda}})^*) T_{w;w},$$

where Proposition 6.3.2(ii) is used in the first equality and Proposition 6.3.2(i) is used in the second equality. On the other hand, $(T_{v;w})^*T_{v;w} \neq 0$ since $\mathcal{B}(H)$ is a C^* -algebra, so it follows that $T_{w;w} \neq 0$. In particular, $\tau^{\lambda}(\nu, \nu) \neq \{0\}$. Formula (6.6.3) for v = w gives

$$0 \neq (T_{w;w})^* T_{w;w} = \tau (C_{w;v_{\lambda}}^{\lambda} (C_{w;v_{\lambda}}^{\lambda})^*) T_{w;w} = q^{(\lambda,\lambda) - (\nu,\nu)} T_{w;w} T_{w;w},$$

where Proposition 6.3.2(ii) is used in the last equality. It follows that $(\lambda, \lambda) = (\nu, \nu)$, since $T_{w;w}$ is selfadjoint. By Proposition 6.5.3 we obtain $\nu = \sigma \lambda$ for some $\sigma \in W^S$. \Box

For
$$\lambda \in P_+(S^c)$$
 and $\nu \in D_m(\lambda)$, we set

(6.6.4)
$$L_{\nu;\lambda} := \tau((C_{\nu;\lambda}^{\lambda})^* C_{\nu;\lambda}^{\lambda})$$

This definition makes sense since dim $(V(\lambda)_{\nu}) = 1$ by Lemma 6.6.2(ii). Furthermore, $L_{\nu;\lambda}$ is a non-zero selfadjoint operator which commutes with the elements of $\tau(A_S)$ in the following way.

LEMMA 6.6.3. Let $\lambda, \Lambda \in P_+(S^c)$ and $\nu \in D_m(\lambda)$. For $v \in V(\Lambda)_{\mu}$, $w \in V(\Lambda)_{\mu'}$ we have

$$L_{\nu;\lambda}\tau((C^{\Lambda}_{v;v_{\Lambda}})^{*}C^{\Lambda}_{w;v_{\Lambda}}) = q^{2(\nu,\mu'-\mu)}\tau((C^{\Lambda}_{v;v_{\Lambda}})^{*}C^{\Lambda}_{w;v_{\Lambda}})L_{\nu;\lambda}$$

PROOF. By Lemma 6.6.1 and the commutation relations in Section 6.3 we compute

$$\begin{split} L_{\nu;\lambda}\tau((C^{\Lambda}_{v;v_{\Lambda}})^{*}C^{\Lambda}_{w;v_{\Lambda}}) &= q^{(\lambda,\Lambda)-(\nu,\mu)}\tau((C^{\Lambda}_{v;v_{\Lambda}}C^{\lambda}_{v;v_{\lambda}})^{*}C^{\lambda}_{v_{\nu};v_{\lambda}}C^{\Lambda}_{w;v_{\Lambda}}) \\ &= q^{2(\lambda,\Lambda)-2(\nu,\mu)}\tau((C^{\Lambda}_{v;v_{\Lambda}})^{*}(C^{\lambda}_{v_{\nu};v_{\lambda}})^{*}C^{\lambda}_{w;v_{\Lambda}}C^{\Lambda}_{w;v_{\Lambda}}) \\ &= q^{(\nu,\mu')+(\lambda,\Lambda)-2(\nu,\mu)}\tau((C^{\Lambda}_{v;v_{\Lambda}})^{*}(C^{\lambda}_{v_{\nu};v_{\lambda}})^{*}C^{\Lambda}_{w;v_{\Lambda}}C^{\lambda}_{v_{\nu};v_{\lambda}}) \\ &= q^{2(\nu,\mu'-\mu)}\tau((C^{\Lambda}_{v;v_{\Lambda}})^{*}C^{\Lambda}_{w;v_{\Lambda}})L_{\nu;\lambda}, \end{split}$$

where Proposition 6.3.2(ii) is used for the first and fourth equality, Proposition 6.3.2(i) is used for the second equality, and Corollary 6.3.3 is used for the third equality. \Box

It follows from Lemma 6.6.3 that $\text{Ker}(L_{\nu;\lambda}) \subsetneq H$ is a closed invariant subspace. By the irreducibility of τ , we thus obtain the following corollary.

COROLLARY 6.6.4. Let $\lambda \in P_+(S^c)$ and $\nu \in D_m(\lambda)$. Then $L_{\nu;\lambda}$ is injective.

The minimal coset representative σ of Lemma 6.6.2(ii) is unique and independent of $\lambda \in P_+(S^c)$ in the following sense.

LEMMA 6.6.5. There exists a unique $\sigma \in W^S$ such that $D_m(\lambda) = \{\sigma\lambda\}$ for all $\lambda \in P_+(S^c)$.

PROOF. Let $\Lambda \in P_{++}(S^c)$ and $\nu \in D_m(\Lambda)$. Then there exists a unique $\sigma \in W^S$ such that $\nu = \sigma \Lambda$ by Lemma 6.6.2(ii) and by Chevalley's Lemma (cf. [58, Proposition 2.27]). Fix furthermore arbitrary $\lambda \in P_+(S^c)$ and $\nu' \in D_m(\lambda)$. Choose a $\sigma' \in W$ such that $\nu' = \sigma' \lambda$. By Lemma 6.6.1 and the commutation relations of Section 6.3, we compute

$$\begin{split} L_{\nu;\Lambda}L_{\nu';\lambda} &= q^{(\Lambda,\lambda)-(\nu,\nu')}\tau((C^{\lambda}_{\nu';\lambda}C^{\Lambda}_{\nu;\Lambda})^*C^{\Lambda}_{\nu;\Lambda}C^{\lambda}_{\nu';\lambda}) \\ &= q^{3(\Lambda,\lambda)-3(\nu,\nu')}\tau((C^{\lambda}_{\nu';\lambda})^*(C^{\Lambda}_{\nu;\Lambda})^*C^{\lambda}_{\nu';\lambda}C^{\Lambda}_{\nu;\Lambda}) \\ &= q^{2(\Lambda,\lambda)-2(\nu,\nu')}L_{\nu';\lambda}L_{\nu;\Lambda}, \end{split}$$

where Proposition 6.3.2(ii) is used in the first and third equality and Proposition 6.3.2(i) is used twice in the second equality. If we repeat the same computation, but now using Corollary 6.3.3 twice in the second equality, then we obtain

$$L_{\nu;\Lambda}L_{\nu';\lambda} = q^{2(\nu,\nu')-2(\Lambda,\lambda)}L_{\nu';\lambda}L_{\nu;\Lambda},$$

hence

$$\left(q^{2(\Lambda,\lambda)-2(\nu,\nu')}-q^{2(\nu,\nu')-2(\Lambda,\lambda)}\right)L_{\nu';\lambda}L_{\nu;\Lambda}=0.$$

By Corollary 6.6.4 we have $L_{\nu';\lambda}L_{\nu;\Lambda} \neq 0$, so we conclude that

$$(\Lambda, \lambda) - (\nu, \nu') = (\Lambda, \lambda - \sigma^{-1} \sigma' \lambda) = 0.$$

Since $\Lambda \in P_{++}(S^c)$ and $\lambda \in P_+(S^c)$, it follows from Lemma 6.5.4 that $\lambda = \sigma^{-1}\sigma'\lambda$, i.e. $\nu' = \sigma\lambda$. Hence, $D_m(\lambda) = \{\sigma\lambda\}$ for all $\lambda \in P_+(S^c)$.

In the remainder of this section we write σ for the unique minimal coset representative such that $D_m(\lambda) = \{\sigma\lambda\}$ for all $\lambda \in P_+(S^c)$. We are going to prove that $\tau \simeq \pi_{\sigma}$. First we look for the analogue of the distinguished vector $e_0^{\otimes l(\sigma)}$ (cf. Proposition 6.5.2(ii)) in the representation space H of τ .

The spectrum $I(\lambda)$ of $L_{\sigma\lambda;\lambda}$ is contained in $[0, \infty)$, since $L_{\sigma\lambda;\lambda}$ is a positive operator. By considering the spectral decomposition of $L_{\sigma\lambda;\lambda}$, the following corollary of Lemma 6.6.3 and [60, Lemma 4.3] is obtained.

COROLLARY 6.6.6. Let $\lambda \in P_+(S^c)$. Then $I(\lambda) \subset [0,\infty)$ is a countable set with no limit points, except possibly 0.

The proof of Corollary 6.6.6 is similar to the proof of [110, Proposition 3.9] and of [60, Proposition 4.2].

By Corollary 6.6.6 we have an orthogonal direct sum decomposition

(6.6.5)
$$H = \bigoplus_{\gamma \in I(\lambda) \cap \mathbb{R}_{>0}} H_{\gamma}(\lambda)$$

into eigenspaces of $L_{\sigma\lambda;\lambda}$, where $H_{\gamma}(\lambda)$ is the eigenspace of $L_{\sigma\lambda;\lambda}$ corresponding to the eigenvalue γ . Let $\gamma_0(\lambda) > 0$ be the largest eigenvalue of $L_{\sigma\lambda;\lambda}$.

LEMMA 6.6.7. Let $\lambda \in P_+(S^c)$, $v \in V(\lambda)$, $w \in V(\lambda)_{\nu}$ and assume that $\nu \neq \sigma \lambda$. Then $\tau((C_{v;v_{\lambda}}^{\lambda})^* C_{w;v_{\lambda}}^{\lambda})(H_{\gamma_0(\lambda)}(\lambda)) = \{0\}.$

PROOF. Let $\lambda \in P_+(S^c)$, $v \in V(\lambda)_{\mu}$ and $w \in V(\lambda)_{\nu}$. By Lemma 6.6.1 and the commutation relations in Section 6.3, we compute

$$\begin{split} L_{\sigma\lambda;\lambda}\tau((C_{v;v_{\lambda}}^{\lambda})^{*}C_{w;v_{\lambda}}^{\lambda}) &= \tau(C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda}(C_{v;v_{\lambda}}^{\lambda}C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda})^{*}C_{w;v_{\lambda}}^{\lambda}) \\ &= q^{(\lambda,\lambda)-(\mu,\sigma\lambda)}\tau(C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda}(C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda})^{*}(C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda})^{*}C_{w;v_{\lambda}}^{\lambda}) \\ &= q^{2(\lambda,\lambda)-2(\mu,\sigma\lambda)}\tau((C_{v;v_{\lambda}}^{\lambda})^{*}C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda})\tau((C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda})^{*}C_{w;v_{\lambda}}^{\lambda}), \end{split}$$

where Proposition 6.3.2(i) is used in the second equality and Proposition 6.3.2(ii) is used in the first and third equality. This computation, together with the injectivity of $L_{\sigma\lambda;\lambda}$, shows that it suffices to give a proof of the lemma for the special case that $v = v_{\sigma\lambda}$. So we fix $h \in H_{\gamma_0(\lambda)}(\lambda)$ and $w \in V(\lambda)_{\nu}$ with $\nu \in P(\lambda)$ and $\nu \neq \sigma\lambda$. It follows from Lemma 6.6.3 that $\tilde{h} := \tau((C^{\lambda}_{v_{\sigma\lambda};v_{\lambda}})^*C^{\lambda}_{w;v_{\lambda}})h$ is an eigenvector of $L_{\sigma\lambda;\lambda}$ with eigenvalue $\tilde{\gamma}_0(\lambda) = q^{2(\lambda,\sigma^{-1}(\nu)-\lambda)}\gamma_0(\lambda)$. By Proposition 6.5.3 we have $\tilde{\gamma}_0(\lambda) > \gamma_0(\lambda)$, hence $\tilde{h} = 0$ by the maximality of the eigenvalue $\gamma_0(\lambda)$.

COROLLARY 6.6.8. $\gamma_0(\lambda) = 1$ for all $\lambda \in P_+(S^c)$.

PROOF. Follows from (6.3.10) and Lemma 6.6.7.

The linear subspace of $\mathbb{C}_q[U]$ spanned by the matrix elements $\{C^{\mu}_{\sigma\mu;\mu}\}_{\mu\in P_+}$ is a subalgebra of $\mathbb{C}_q[U]$ with algebraic generators $C^{\varpi_i}_{\sigma\varpi_i;\varpi_i}$ $(i \in [1, r])$. This follows from the fact that $C^{\mu}_{\sigma\mu;\mu}C^{\nu}_{\sigma\nu;\nu} = \lambda_{\mu,\nu}C^{\mu+\nu}_{\sigma(\mu+\nu);\mu+\nu}$, where the scalar $\lambda_{\mu,\nu} \in \mathbb{T}$ depends on the particular choices of orthonormal bases for the finite dimensional irreducible representations $V(\mu)$ and $V(\nu)$ (cf. [110, Proof of Proposition 3.12]). Then it follows from Proposition 6.3.2 and Lemma 6.6.1 that

$$(6.6.6) L_{\sigma(\mu+\nu);\mu+\nu} = L_{\sigma\mu;\mu}L_{\sigma\nu;\nu}$$

for all $\mu, \nu \in P_+(S^c)$, hence span $\{L_{\sigma\lambda;\lambda}\}_{\lambda \in P_+(S^c)}$ is a commutative subalgebra of $\mathcal{B}(H)$. Set

then $H_1 \subset H_1(\lambda)$ for all $\lambda \in P_+(S^c)$ by (6.6.6).

LEMMA 6.6.9. $H_1 = H_1(\lambda)$ for all $\lambda \in P_{++}(S^c)$. In particular, $H_1 \neq \{0\}$.

PROOF. For $\mu \in P_+(S^c)$ we have $||L_{\sigma\mu;\mu}|| = 1$. Moreover, for any $h \in H$,

$$(6.6.8) h \in H_1(\mu) \quad \Leftrightarrow \quad ||L_{\sigma\mu;\mu}h|| = ||h||.$$

This follows from the eigenspace decomposition (6.6.5) for $L_{\sigma\mu;\mu}$ and the fact that 1 is the largest eigenvalue of $L_{\sigma\mu;\mu}$. Let $\lambda \in P_{++}(S^c)$ and choose arbitrary $i \in S^c$. Then $\lambda = \mu + \varpi_i$ for certain $\mu \in P_+(S^c)$. By (6.6.6), we obtain for $h \in H_1(\lambda)$,

$$||h|| = ||L_{\sigma\lambda;\lambda}h|| = ||L_{\sigma\mu;\mu}L_{\sigma\varpi_i;\varpi_i}h||$$

$$\leq ||L_{\sigma\varpi_i;\varpi_i}h|| \leq ||h||,$$

hence we have equality everywhere. By (6.6.8), it follows that $h \in H_1(\varpi_i)$. Since $i \in S^c$ was arbitrary, we conclude that $h \in H_1$.

LEMMA 6.6.10. Let $\lambda \in P_+(S^c)$. For all $v \in V(\lambda)_{\mu}$ with $\mu \neq \sigma \lambda$ we have

$$\tau\big((C_{v;v_{\lambda}}^{\lambda})^*C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda}\big)(H_1) \subset H_1^{\perp}.$$

PROOF. Let $\Lambda \in P_{++}(S^c)$, $\lambda \in P_+(S^c)$, and $v \in V(\lambda)_{\mu}$ with $\mu \neq \sigma \lambda$ and $\mu \in P(\lambda)$. Then

(6.6.9)
$$L_{\sigma\Lambda;\Lambda}\tau((C_{v;v_{\lambda}}^{\lambda})^{*}C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda}) = q^{2(\Lambda,\lambda-\sigma^{-1}(\mu))}\tau((C_{v;v_{\lambda}}^{\lambda})^{*}C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda})L_{\sigma\Lambda;\Lambda}$$
by Lemma 6.6.3. By Lemma 6.5.4 we have $(\Lambda,\lambda-\sigma^{-1}(\mu)) > 0$. Hence,

$$\tau((C_{v;v_{\lambda}}^{\lambda})^{*}C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda})(H_{1}) = \tau((C_{v;v_{\lambda}}^{\lambda})^{*}C_{v_{\sigma\lambda};v_{\lambda}}^{\lambda})(H_{1}(\Lambda))$$
$$\subset \bigoplus_{\gamma < 1} H_{\gamma}(\Lambda) = H_{1}(\Lambda)^{\perp} = H_{1}^{\perp},$$

which completes the proof of the lemma.

COROLLARY 6.6.11. $\dim(H_1) = 1$.

PROOF. By Lemma 6.6.7 and Lemma 6.6.10 we obtain for any $0 \neq h \in H_1$,

$$\overline{\tau(A_S)h} \subset \operatorname{span}\{h\} \oplus H_1^{\perp},$$

where the overbar means closure. Since τ is irreducible, it follows that span $\{h\} = H_1$.

Any vector $h \in H_1$ with ||h|| = 1 can serve now as the analogue in the representation space H of the distinguished vector $e_0^{\otimes l(\sigma)}$ in the representation space of π_{σ} . By comparing the Gelfand-Naimark-Segal states of τ and π_{σ} taken with respect to the cyclic vector $h \in H_1$ (||h|| = 1) respectively $e_0^{\otimes l(\sigma)}$, we obtain the following lemma.

LEMMA 6.6.12. We have $\tau \simeq \pi_{\sigma}$ as irreducible *-representations of A_S .

PROOF. Fix an $h \in H_1$ with ||h|| = 1, and define the Gelfand-Naimark-Segal states $\phi_{\tau}, \phi_{\pi_{\sigma}} : A_S \to \mathbb{C}$ by

(6.6.10)
$$\phi_{\tau}(a) := (\tau(a)h, h), \quad \phi_{\pi_{\sigma}}(a) := (\pi_{\sigma}(a)e_{0}^{\otimes l(\sigma)}, e_{0}^{\otimes l(\sigma)}).$$

Then we have for $\phi = \phi_{\tau}$ (respectively $\phi = \phi_{\pi_{\sigma}}$),

(6.6.11)
$$\phi((C_{\mu,i;\lambda}^{\lambda})^* C_{\nu,j;\lambda}^{\lambda}) = \delta_{\mu,\sigma\lambda} \delta_{\nu,\sigma\lambda}$$

for $\lambda \in P_+(S^c)$, $\mu, \nu \in P(\lambda)$, $i \in [1, \dim(V(\lambda)_{\mu})]$, and $j \in [1, \dim(V(\lambda)_{\nu})]$. Indeed, (6.6.11) for $\phi = \phi_{\tau}$ follows from Lemma 6.6.7 and Lemma 6.6.10. For $\phi = \phi_{\pi_{\sigma}}$, recall that π_{σ} is an irreducible *-representation of A_S (Proposition 6.5.5). We have seen in the previous section that $L_{\sigma\lambda;\lambda} = \pi_{\sigma}((C^{\lambda}_{\sigma\lambda;\lambda})^*C^{\lambda}_{\sigma\lambda;\lambda})$ is injective for all $\lambda \in P_+(S^c)$, hence $\sigma\lambda \in D(\lambda)$ (cf. (6.6.2)) for all $\lambda \in P_+(S^c)$. By (6.5.4), we actually have $\sigma\lambda \in D_m(\lambda)$ for all $\lambda \in P_+(S^c)$. Hence the labeling $\sigma \in W^S$ of π_{σ} coincides with its (unique) minimal coset representative defined in Lemma 6.6.5. Furthermore, the one dimensional

182

subspace H_1 for π_{σ} is equal to span $\{e_0^{\otimes l(\sigma)}\}$ (cf. Proposition 6.5.2(ii), Lemma 6.6.11). So (6.6.11) for $\phi = \phi_{\pi_{\sigma}}$ follows again from Lemma 6.6.7 and Lemma 6.6.10.

By linearity it follows from (6.6.11) that $\phi_{\tau} = \phi_{\pi_{\sigma}}$, hence τ and π_{σ} are unitarily equivalent *-representations (cf. [26, Proposition 2.4.1]).

The results of this section may be summarized as follows.

THEOREM 6.6.13. For all $S \subsetneq \Delta$, $\{\pi_{\sigma}\}_{\sigma \in W^S}$ is a complete set of mutually inequivalent, irreducible *-representations of the factorized *-subalgebra A_S .

Combining Proposition 6.5.7, Theorem 6.4.10 and Theorem 6.6.13 we obtain the following theorem.

THEOREM 6.6.14. $\{\pi_{\sigma}\}_{\sigma \in W^S}$ is a complete set of mutually inequivalent, irreducible *-representations of $\mathbb{C}_q[U/K_S]$ in the the following cases: (i) $S = \emptyset$, i.e. $U/K_S = U/T$ is the full flag manifold; (ii) $\#S^c = 1$ and the simple root $\alpha \in S^c$ is a Gelfand node.

For these cases the restriction to $\mathbb{C}_q[U/K_S]$ of the universal C^* -norm on $\mathbb{C}_q[U]$ coincides with the universal C^* -norm on $\mathbb{C}_q[U/K_S]$.

In fact, Theorem 6.6.14 is true for all generalized flag manifolds U/K_S in view of Remark 6.4.11.

. .

Bibliography

- G.E. Andrews, R. Askey, *Enumeration of partitions: the role of Eulerian series and q-orthogonal poly*nomials, in "Higher Combinatorics", M. Aigner, ed., Reidel, Boston, MA (1977), pp. 3–26.
- [2] G.E. Andrews, R. Askey, Another q-extension of the beta function, Proc. Amer. Math. Soc. 81 (1981), pp. 97–100.
- [3] G.E. Andrews, R. Askey, *Classical orthogonal polynomials*, in "Polynômes Orthogonaux et Applications", C. Brezinski, A. Draux, A.P. Magnus, P. Maroni and A. Ronveaux, eds., Lecture Notes in Math. **1171**, Springer, New York (1985), pp. 36–62.
- K. Aomoto, On Elliptic Product Formulas for Jackson Integrals associated with Reduced Root Systems, J. Algebraic Combin. 8 (1998), pp. 115–126.
- [5] R. Askey, Some basic hypergeometric extensions of integrals of Selberg and Andrews, SIAM J. Math. Anal. 11 (1980), pp. 938–951.
- [6] R. Askey, J. Wilson, A set of orthogonal polynomials that generalize the Racah coefficients or 6-j symbols, SIAM J. Math. Anal. 10 (1979), pp. 1008–1016.
- [7] R. Askey, J. Wilson, Some basic hypergeometric orthogonal polynomials that generalize Jacobi polynomials, Mem. Amer. Math. Soc. 54 (1985), no. 319.
- [8] T.H. Baker, P.J. Forrester, Multivariable Al-Salam & Carlitz polynomials associated with the type A q-Dunkl kernel, Math. Nachr. 212 (2000), pp. 5–35..
- [9] I.N. Bernstein, I.M. Gelfand, S.I. Gelfand, Schubert cells and cohomology of the spaces G/P, Uspekhi Mat. Nauk 28 (1973), no. 3 (171), pp. 3–26; English translation in Russian Math. surveys 28 (1973).
- [10] N. Bourbaki, Groupes et Algèbres de Lie, Ch. 4,5,6, Masson, Paris, 1981.
- [11] V. Chari, A. Pressley, A Guide to Quantum Groups, Cambridge University Press, Cambridge (UK), 1994.
- [12] I. Cherednik, *Double affine Hecke algebras and Macdonald's conjectures*, Ann. of Math. **141** (1995), pp. 191–216.
- [13] I. Cherednik, Macdonald's evaluation conjectures and difference Fourier transform, Invent. Math. 122 (1995), pp. 119–145.
- [14] I. Cherednik, Nonsymmetric Macdonald polynomials, IMRN 10 (1995), pp. 483-515.
- [15] I. Cherednik, Intertwining operators of double affine Hecke algebras, Selecta Math. (N.S.) 3 (1997), pp. 459–495.
- [16] J.-F. van Diejen, Commuting difference operators with polynomial eigenfunctions, Compositio Math. 95 (1995), pp. 183–233.
- [17] J.-F. van Diejen, Self-dual Koornwinder-Macdonald polynomials, Invent. Math. 126 (1996), pp. 319–339.
- [18] J.-F. van Diejen, On certain multiple Bailey, Rogers and Dougall type summation formulas, Publ. Res. Inst. Math. Sci. 33 (1997), pp. 483–508.
- [19] J.-F. van Diejen, Properties of some families of hypergeometric orthogonal polynomials in several variables, Trans Amer. Math. Soc. 351 (1999), pp. 233–270.
- [20] J.-F. van Diejen, J.V. Stokman, Multivariable q-Racah polynomials, Duke Math. J. 91 (1998), pp. 89–136.
- [21] M.S. Dijkhuizen, Covariant Poisson brackets, quantized complex Grassmannians, and BC type qorthogonal polynomials, in "Quantum Group Symposium at Group21", (Proceedings of the Quantum Group Symposium at the XXI International Colloquium on Group Theoretical Methods in Physics, Goslar, 1996), eds. H.-D. Doebner, V.K. Dobrev, pp. 345–353, Heron Press, Sofia, 1997.

- [22] M.S. Dijkhuizen, T.H. Koornwinder, Quantum homogeneous spaces, duality and quantum 2-spheres, Geom. Dedicata 52 (1994), pp. 291–315.
- [23] M.S. Dijkhuizen, T.H. Koornwinder, CQG algebras: a direct algebraic approach to compact quantum groups, Lett. Math. Phys. 32 (1994), pp. 315–330.
- [24] M.S. Dijkhuizen, M. Noumi, A family of quantum projective spaces and related q-hypergeometric orthogonal polynomials, Trans. Amer. Math. Soc. 350 (1998), pp. 3269–3296.
- [25] M.S. Dijkhuizen, J.V. Stokman, Some limit transitions between BC type orthogonal polynomials interpreted on quantum complex Grassmannians, Publ. Res. Inst. Math. Sci. 35 (1999), pp. 451–500.
- [26] J. Dixmier, C*-algebras, North-Holland mathematical library; v.15 (1977)
- [27] V.G. Drinfeld, *Quantum groups*, in the Proceedings International Conference of Mathematicians, Berkeley, 1986, ed. A.M. Gleason, Amer. Math. Soc., Providence (RI), 1987, pp. 798–820.
- [28] A. Erdélyi, W. Magnus, F. Oberhettinger, F.G. Tricomi, *Higher transcendental functions Vol. 2*, McGraw-Hill (1953).
- [29] R.J. Evans, Multidimensional beta and gamma integrals, Contemp. Math. 166 (1994), pp. 341–357.
- [30] G. Gasper, M. Rahman, *Basic Hypergeometric Series*, Cambridge Univ. Press, Cambridge (1990).
- [31] R.A. Gustafson, A generalization of Selberg's beta integral, Bull. Amer. Math. Soc. (N.S.) 22 (1990), pp. 97–105.
- [32] R.A. Gustafson, Some q-beta and Mellin-Barnes integrals on compact Lie groups and Lie algebras, Trans. Am. Math. Soc. 341 (1994), pp. 69–119.
- [33] L. Habsieger, Une q-intégrale de Selberg et Askey, SIAM J. Math. Anal. 19 (1988), pp. 1475-1489.
- [34] W. Hahn, Über Orthogonalpolynome, die q-Differenzengleichungen genügen, Math. Nachr. 2 (1949), pp. 4–34.
- [35] M. Hashimoto, T. Hayashi, Quantum multilinear algebra, Tôhoku Math. J. 44 (1992), pp. 471-521.
- [36] G.J. Heckman, Root systems and hypergeometric functions II, Compositio Math. 64 (1987), pp. 353–373.
- [37] G.J. Heckman, An elementary approach to the hypergeometric shift operators of Opdam, Invent. Math. **103** (1991), pp. 341–350.
- [38] G.J. Heckman and E.M. Opdam, *Root systems and hypergeometric functions I*, Compositio Math. 64 (1987), pp. 329–352.
- [39] G.J. Heckman, H. Schlichtkrull, Harmonic Analysis and Special Functions on Symmetric Spaces, Perspectives in Math. 16, Academic Press, New York, 1995.
- [40] S. Helgason, *Differential geometry, Lie groups, and symmetric spaces*, Academic Press, New York (1978).
- [41] S. Helgason, Groups and Geometric Analysis, Academic Press, New York (1984).
- [42] J.E. Humphreys, *Introduction to Lie Algebras and Representation Theory*, Graduate Texts in Mathematics **9**, Springer-Verlag (1980).
- [43] M.E.H. Ismail, D. Stanton, On the Askey-Wilson and Rogers polynomials, Canad. J. Math. 40 (1988), pp. 1025–1045.
- [44] M. Ito, On a theta product formula for Jackson integrals associated with root systems of rank two, J. Math. Anal. Appl. 216 (1997), pp. 122–163.
- [45] A.T. James, A.G. Constantine, Generalized Jacobi polynomials as spherical functions of the Grassmann manifold, Proc. London Math. Soc. 29 (1974), pp. 174–192.
- [46] M. Jimbo, A q-difference analogue of $U(\mathfrak{g})$ and the Yang-Baxter equation, Lett. Math. Phys. 10 (1985), pp. 63–69.
- [47] M. Jimbo, A q-analogue of U(gl(n)), Hecke algebra and the Yang-Baxter equation, Lett. Math. Phys. 11 (1986), pp. 247–252.
- [48] A. Joseph, On the prime and primitive spectra of the algebra of functions on a quantum group, J. Algebra 169 (1994), pp. 441–511.
- [49] A. Joseph, Quantum groups and their primitive ideals, Ergebnisse der Mathematik und ihrer Grenzgebiete; 3, Folge, Bd. 29 (1995).
- [50] B. Jurco, P. Stovicek, *Quantum dressing orbits on compact groups*, Comm. Math. Phys. 152 (1993), pp. 96–126.

- [51] B. Jurco, P. Stovicek, *Quantum Grassmann Manifolds*, Comm. Math. Phys. 158 (1993), pp. 135–153.
- [52] B. Jurco, P. Stovicek, *Coherent states for quantum compact groups*, Comm. Math. Phys. **182** (1996), pp. 221–251.
- [53] V.G. Kac, Infinite dimensional Lie algebras, Cambridge University Press, third edition, (1990).
- [54] K.W.J. Kadell, A proof of Askey's conjectured q-analogue of Selberg's integral and a conjecture of Morris, SIAM J. Math. Anal. 19 (1988), pp. 969–986.
- [55] J. Kaneko, q-Selberg integrals and Macdonald polynomials, Ann. scient. Éc. Norm. Sup., 4^e série, t. 29 (1996), pp. 583–637.
- [56] S. Khoroshkin, A. Radul, V. Rubtsov, A family of Poisson structures on Hermitian symmetric spaces, Comm. Math. Phys. 152 (1993), pp. 299–315.
- [57] A. Klimyk, K. Schmüdgen, Quantum Groups and Their Representations, Texts and monographs in physics, Springer (1997).
- [58] A.W. Knapp, *Lie Groups Beyond an Introduction*, Progress in Math. 140, Birkhäuser Verlag, Boston (1996).
- [59] R. Koekoek, R.F. Swarttouw, The Askey-scheme of hypergeometric orthogonal polynomials and its qanalogue, Report 94-05 (Delft University of Technology, 1994).
- [60] H.T. Koelink, On *-representations of the Hopf *-algebra associated with the quantum group $U_q(n)$, Compositio Math. 77 (1991), pp. 199–231.
- [61] T.H. Koornwinder, Orthogonal polynomials in two variables which are eigenfunctions of two independent partial differential operators I–IV, Indag. Math. **36** (1974), pp. 48–66 and pp. 357–381.
- [62] T.H. Koornwinder, *Two-variable analogues of the classical orthogonal polynomials*, in "Theory and applications of special functions", R. Askey, ed., Academic Press, New York (1975), pp. 435–495.
- [63] T.H. Koornwinder, Representations of the twisted SU(2) quantum group and some q-hypergeometric polynomials, Proc. Kon. Ned. Akad. Wet. Series A 92 (1989), pp. 97–117.
- [64] T.H. Koornwinder, Orthogonal polynomials in connection with quantum groups in "Orthogonal polynomials: Theory and Practice", NATO ASI series C, vol. 294, Kluwer, Dordrecht (1990), pp. 257–292.
- [65] T.H. Koornwinder, Askey-Wilson polynomials for root systems of type BC, Contemp. Math. 138 (1992), pp. 189–204.
- [66] T.H. Koornwinder, Askey-Wilson polynomials as zonal spherical functions on the SU(2) quantum group, SIAM J. Math. Anal. 24 (1993), pp. 795–813.
- [67] T.H. Koornwinder, Compact quantum groups and q-special functions in "Representations of Lie groups and quantum groups", V. Baldoni and M.A. Picardello (eds.), Pitman Research Notes in Mathematics Series 311, Longman Scientific and Technical (1994).
- [68] T.H. Koornwinder, Dynkin diagrams of rank n with a subdiagram of rank n 1, informal note (1994).
- [69] T.H. Koornwinder, I.G. Sprinkhuizen-Kuyper, Generalized power series expansions for a class of orthogonal polynomials in two variables, SIAM J. Math. Anal. 9 (1978), pp. 457–483.
- [70] L.I. Korogodsky, Complimentary series representations and quantum orbit methods, preprint (1997).
- [71] B. Kostant, A formula for the multiplicity of a weight, Trans. Amer. Math. Soc. 93 (1959), pp. 53–73.
- [72] M. Krämer, *Sphärische untergruppen in kompakten zusammenhängenden Liegruppen*, Compositio Math. **38** (1979), pp. 129–153.
- [73] S. Kumar, *Proof of the Parthasarathy-Ranga Rao-Varadarajan conjecture*, Invent. Math. **93** (1988), pp. 117–130.
- [74] V. Lakshmibai, N. Yu. Reshetikhin, Quantum deformations of flag and Schubert schemes, C.R. Acad. Sci. Paris Sér. I 313 (1991), pp. 121-126.
- [75] V. Lakshmibai, N. Yu. Reshetikhin, *Quantum flag and Schubert schemes*, in "Deformation Theory and Quantum Groups with Applications to Mathematical Physics", M. Gerstenhaber, J. Stasheff (eds), Contemp. Math. **134** (1991), pp. 145–181.
- [76] S. Levendorskii, Ya. S. Soibelman, Algebras of Functions on Compact Quantum Groups, Schubert Cells and Quantum Tori, Comm. Math. Phys. 139 (1991), pp. 141–170.
- [77] P. Littelmann, A Littlewood-Richardson rule for symmetrizable Kac-Moody algebras, Invent. Math. 116 (1994), pp. 329–346.

- [78] P. Littelmann, On spherical double cones, J. Algebra 166 (1994), pp. 142–157.
- [79] J.-H. Lu, Coordinates on Schubert cells, Kostant's harmonic forms, and the Bruhat-Poisson structure on G/B, Transform Groups 4 (1999), pp. 355–374.
- [80] J.-H. Lu, A. Weinstein, Poisson-Lie groups, dressing transformations, and Bruhat decompositions, J. Diff. Geom. 31 (1990), pp. 501–526.
- [81] J.-H. Lu, A. Weinstein, Classification of SU(2)-covariant Poisson structures on S², appendix to: A.J.-L. Sheu, Quantization of the Poisson SU(2) and its Poisson homogeneous space the 2-sphere, Comm. Math. Phys. 135 (1991), pp. 229–231.
- [82] I.G. Macdonald, Orthogonal polynomials associated with root systems, unpublished manuscript (1987), math.QA/0011046.
- [83] I.G. Macdonald, Some conjectures for Koornwinder's orthogonal polynomials, unpublished manuscript (1988).
- [84] I.G. Macdonald, Some conjectures for root systems, SIAM J. Math. Anal. 13 (1982), pp. 988–1007.
- [85] I.G. Macdonald, Symmetric functions and Hall polynomials, Oxford University Press, Oxford, 2nd edition (1995).
- [86] I.G. Macdonald, Affine Hecke algebras and orthogonal polynomials, Séminaire Bourbaki, Vol. 1994/95. Astérisque No. 237 (1996), Exp. No. 797, 4, pp. 189–207.
- [87] T. Masuda, K. Mimachi, Y. Nakagami, M. Noumi, K. Ueno, *Representations of quantum groups and a q-analogue of orthogonal polynomials*, C. R. Acad. Sci. Paris Sér. I Math. **307** (1988), pp. 559–564.
- [88] O. Mathieu, *Construction d'un groupe de Kac-Moody et applications*, Compositio Math. **69** (1989), pp. 37–60.
- [89] M. Noumi, Quantum Grassmannians and q-hypergeometric series, CWI Quaterly 5, Nr. 4, Amsterdam (1992), pp. 293–307.
- [90] M. Noumi, Macdonald-Koornwinder polynomials and affine Hecke algebras, RIMS Kokyuroku 919 (1995), pp. 44–55 (in Japanese).
- [91] M. Noumi, Macdonald's symmetric polynomials as zonal spherical functions on some quantum homogeneous spaces, Adv. Math. 123 (1996), pp. 16–77.
- [92] M. Noumi, M.S. Dijkhuizen, T. Sugitani, *Multivariable Askey-Wilson polynomials and quantum complex Grassmannians*, in: "Special Functions, q-Series, and Related Topics", eds. M.E.H. Ismail e.a., pp. 167–177, Fields Inst. Comm. 14, Amer. Math. Soc., Providence (RI), 1997.
- [93] M. Noumi, K. Mimachi, Quantum 2-spheres and big q-Jacobi polynomials, Comm. Math. Phys. 128 (1990), pp. 521–531.
- [94] M. Noumi, T. Sugitani, Quantum symmetric spaces and related q-orthogonal polynomials, in "Group Theoretical Methods in Physics", Proceedings XX ICGTMP, Toyonaka (Japan), July 1994, eds. A. Arima et al., pp. 28–40, World Scientific, Singapore, 1995.
- [95] M. Noumi, T. Sugitani, *Quantum symmetric spaces and multivariable q-orthogonal polynomials*, preprint.
- [96] M. Noumi, H. Yamada, K. Mimachi, Finite-dimensional representations of the quantum group $GL_q(n, \mathbb{C})$ and the zonal spherical functions on $U_q(n-1)\setminus U_q(n)$, Japanese J. Math. 19 (1993), pp. 31–80.
- [97] D.I.Panyushev, Complexity and rank of double cones and tensor product decompositions, Comment. Math. Helv. 68 (1993), pp. 455–468.
- [98] D.I.Panyushev, Reductive group actions on affine varieties and their doubling, Ann. Inst. Fourier (Grenoble) 45 (1995), pp. 929–950.
- [99] K.R. Parthasarathy, R. Ranga Rao, V.S. Varadarajan, Representations of complex semisimple Lie groups and Lie algebras, Ann. of Math. 85 (1967), pp. 383–429.
- [100] A.M. Perelomov, *Generalized coherent states and their applications*, Springer, Berlin, Heidelberg, New York (1986).
- [101] P. Podles, Quantum spheres, Lett. Math. Phys. 14 (1987), pp. 193-202.
- [102] G.B. Podkolzin, L.I. Vainerman, Quantum Stiefel manifold and double cosets of quantum unitary group, Pacific J. Math. 188 (1999), pp. 179–199.

- [103] M. Rahman, A simple evaluation of Askey and Wilson's q-beta integral, Proc. Amer. Math. Soc. 92 (1984), pp. 413–417.
- [104] N. Reshetikhin, L.A. Takhtadjan, L.D. Faddeev, *Quantization of Lie groups and Lie algebras*, Leningrad Math. J. 1 (1990), pp. 193–225.
- [105] W. Rudin, Functional analysis, Tata McGraw-Hill (1973).
- [106] S. Sahi, Nonsymmetric Koornwinder polynomials and duality, Ann. of Math. 150 (1999), pp. 267-282.
- [107] A. Selberg, *Bemerkninger om et multipelt integral*, Norsk Mat. Tidsskr. **26** (1944), pp. 71–78; (Coll. Papers, Vol. 1, pp. 204–213).
- [108] M.A. Semenov-Tian-Shansky, Dressing transformations and Poisson group actions, Publ. Res. Inst. Math. Sci. 21 (1985), pp. 1237–1260.
- [109] Ya. S. Soibelman, Irreducible representations of the function algebra on the quantum group SU(n), and Schubert cells, Soviet Math. Dokl. 40 (1990), pp. 34–38.
- [110] Ya. S. Soibelman, The algebra of functions on a compact quantum group, and its representations, Leningrad Math. J. 2 (1991), pp. 161–178.
- [111] Ya. S. Soibelman, On quantum flag manifolds, Funct. Anal. Appl. 26(1992), pp. 225-227.
- [112] T. A. Springer, *Linear algebraic groups*, Progress in Math. Vol. 9 (eds. J. Coates, S. Helgason), Birkhäuser, Boston, Basel, Stuttgart (1981).
- [113] I.G. Sprinkhuizen-Kuyper, Orthogonal polynomials in two variables. A further analysis of the polynomials orthogonal on a region bounded by two lines and a parabola, SIAM J. Math. Anal. 7 (1976), pp. 501–518.
- [114] R. Steinberg, Lectures on Chevalley groups, Lecture Notes, Yale Univ., New Haven (1968).
- [115] J.R. Stembridge, A maple package for root systems and finite Coxeter groups, manuscript (1997).
- [116] J.V. Stokman, Multivariable big and little q-Jacobi polynomials, SIAM J. Math. Anal. 28 (1997), pp. 452–480.
- [117] J.V. Stokman, *Two limit transitions involving multivariable BC type Askey-Wilson polynomials*, in "Quantum Groups and Quantum Spaces" (eds. R. Budzynski, W. Pusz and S. Zakrzewski), Banach Center Publ. 40 (1997), pp. 415–428.
- [118] J.V. Stokman, Askey-Wilson polynomials with partly discrete orthogonality measure, Ramanujan J. 1 (1997), pp. 275–297.
- [119] J.V. Stokman, On BC type basic hypergeometric orthogonal polynomials, Trans. Amer. Math. Soc. 352 (2000), pp. 1527–1579.
- [120] J.V. Stokman, Koornwinder polynomials and affine Hecke algebras, Internat. Math. Res. Notices 2000, no. 19, pp. 1005–1042.
- [121] J.V. Stokman, T.H. Koornwinder, Limit transitions for BC type multivariable orthogonal polynomials, Canad. J. Math. 49 (2) (1997), pp. 373–404.
- [122] J.V. Stokman, T.H. Koornwinder, On some limit cases of Askey-Wilson polynomials, J. Approx. Th. 95 (1998), pp. 310–330.
- [123] J.V. Stokman, M.S. Dijkhuizen, Quantized flag manifolds and irreducible *-representations, Comm. Math. Phys. 203 (1999), pp. 297–324.
- [124] T. Sugitani, Zonal spherical functions on quantum Grassmann manifolds, J. Math. Sci. Univ. Tokyo 6 (1999), pp. 335–369.
- [125] M. Sugiura, Representations of compact groups realized by spherical functions on symmetric spaces, Proc. Japan Acad. 38 (1962), pp. 111–113.
- [126] E. Taft, J. Towber, Quantum deformations of flag schemes and Grassmann schemes. I. A q-deformation of the shape-algebra for GL(n), J. Algebra 142 (1991), pp. 1–36.
- [127] V. Tarasov, A. Varchenko, Geometry of q-hypergeometric functions, quantum affine algebras and elliptic quantum groups, Astèrisque 246 (1997).
- [128] J. Towber, Young symmetry, the flag manifold, and representations of GL(n), J. Algebra 61 (1979), pp. 414–462.
- [129] L.L. Vaksman, Ya. S. Soibelman, algebra of functions on the quantum group SU(2), Funct. Anal. Appl. 22 (1988), pp. 170–181.

- [130] L.L. Vaksman, Ya. S. Soibelman, The algebra of functions on the quantum group SU(n + 1), and odddimensional quantum spheres, Leningrad Math. J. 2 (1991), no. 5, pp. 1023–1042.
- [131] L. Vretare, Elementary spherical functions on symmetric spaces, Math. Scand. 39 (1976), pp. 343–358.
- [132] L. Vretare, Formulas for elementary spherical functions and generalized Jacobi polynomials, SIAM J. Math. Anal. 15 (1984), pp. 805–833.
- [133] G. Warner, Harmonic Analysis on Semi-Simple Lie Groups I, Springer-Verlag, New York (1972).

CWI TRACTS

- 1 D.H.J. Epema. Surfaces with canonical hyperplane sections. 1984.
- 2 J.J. Dijkstra. Fake topological Hilbert spaces and characterizations of dimension in terms of negligibility. 1984.
- 3 A.J. van der Schaft. System theoretic descriptions of physical systems. 1984.
- 4 J. Koene. Minimal cost flow in processing networks, a primal approach. 1984.
- 5 B. Hoogenboom. Intertwining functions on compact Lie groups. 1984.
- 6 A.P.W. Böhm. Dataflow computation. 1984.
- 7 A. Blokhuis. Few-distance sets. 1984.
- 8 M.H. van Hoorn. Algorithms and approximations for queueing systems. 1984.
- 9 C.P.J. Koymans. *Models of the lambda calculus*. 1984.
- 10 C.G. van der Laan, N.M. Temme. Calculation of special functions: the gamma function, the exponential integrals and error-like functions. 1984.
- 11 N.M. van Dijk. Controlled Markov processes; timediscretization. 1984.
- 12 W.H. Hundsdorfer. The numerical solution of nonlinear stiff initial value problems: an analysis of one step methods. 1985.
- 13 D. Grune. On the design of ALEPH. 1985.
- 14 J.G.F. Thiemann. Analytic spaces and dynamic programming: a measure theoretic approach. 1985.
- 15 F.J. van der Linden. Euclidean rings with two infinite primes. 1985.
- 16 R.J.P. Groothuizen. Mixed elliptic-hyperbolic partial differential operators: a case-study in Fourier integral operators. 1985.
- 17 H.M.M. ten Eikelder. Symmetries for dynamical and Hamiltonian systems. 1985.
- 18 A.D.M. Kester. Some large deviation results in statistics. 1985.
- 19 T.M.V. Janssen. Foundations and applications of Montague grammar, part 1: Philosophy, framework, computer science. 1986.
- 20 B.F. Schriever. Order dependence. 1986.
- 21 D.P. van der Vecht. Inequalities for stopped Brownian motion. 1986.
- 22 J.C.S.P. van der Woude. *Topological dynamix.* 1986.
- 23 A.F. Monna. Methods, concepts and ideas in mathematics: aspects of an evolution. 1986.
- 24 J.C.M. Baeten. Filters and ultrafilters over definable subsets of admissible ordinals. 1986.
- 25 A.W.J. Kolen. Tree network and planar rectilinear location theory. 1986.
- 26 A.H. Veen. The misconstrued semicolon: Reconciling imperative languages and dataflow machines. 1986.
- 27 A.J.M. van Engelen. Homogeneous zerodimensional absolute Borel sets. 1986.
- 28 T.M.V. Janssen. Foundations and applications of Montague grammar, part 2: Applications to natural language. 1986.
- 29 H.L. Trentelman. Almost invariant subspaces and high gain feedback. 1986.
- 30 A.G. de Kok. Production-inventory control models: approximations and algorithms. 1987.

- 31 E.E.M. van Berkum. Optimal paired comparison designs for factorial experiments. 1987.
- 32 J.H.J. Einmahl. Multivariate empirical processes. 1987.
- 33 O.J. Vrieze. Stochastic games with finite state and action spaces. 1987.
- 34 P.H.M. Kersten. Infinitesimal symmetries: a computational approach. 1987.
- 35 M.L. Eaton. Lectures on topics in probability inequalities. 1987.
- 36 A.H.P. van der Burgh, R.M.M. Mattheij (editors). Proceedings of the first international conference on industrial and applied mathematics (ICIAM 87). 1987.
- 37 L. Stougie. Design and analysis of algorithms for stochastic integer programming. 1987.
- 38 J.B.G. Frenk. On Banach algebras, renewal measures and regenerative processes. 1987.
- H.J.M. Peters, O.J. Vrieze (eds.). Surveys in game theory and related topics. 1987.
 J.L. Geluk, L. de Haan. Regular variation, exten-
- sions and Tauberian theorems. 1987.
- 41 Sape J. Mullender (ed.). The Amoeba distributed operating system: Selected papers 1984-1987. 1987.
- 42 P.R.J. Asveld, A. Nijholt (eds.). Essays on concepts, formalisms, and tools. 1987.
- 43 H.L. Bodlaender. Distributed computing: structure and complexity. 1987.
 44 A.W. van der Vaart. Statistical estimation in large
- 44 A.W. Van der Vaart. Statistical estimation in large parameter spaces. 1988.
 45 S.A. van de Geer. Regression analysis and empirical
- processes. 1988. 46 S.P. Spekreijse. Multigrid solution of the steady
- Euler equations. 1988. 47 J.B. Dijkstra. Analysis of means in some non-
- standard situations. 1988. 48 F.C. Drost. Asymptotics for generalized chi-square
- goodness-of-fit tests. 1988. 49 F.W. Wubs. Numerical solution of the shallowwater equations. 1988.
- 50 F. de Kerf. Asymptotic analysis of a class of perturbed Korteweg-de Vries initial value problems. 1988.
- 51 P.J.M. van Laarhoven. Theoretical and computational aspects of simulated annealing. 1988.
- 52 P.M. van Loon. Continuous decoupling transformations for linear boundary value problems. 1988.
- 53 K.C.P. Machielsen. Numerical solution of optimal control problems with state constraints by sequential quadratic programming in function space. 1988.
- 54 L.C.R.J. Willenborg. Computational aspects of survey data processing. 1988.
- 55 G.J. van der Steen. A program generator for recognition, parsing and transduction with syntactic patterns. 1988.
- 56 J.C. Ebergen. Translating programs into delayinsensitive circuits. 1989.
- 57 S.M. Verduyn Lunel. Exponential type calculus for linear delay equations. 1989.
- 58 M.C.M. de Gunst. A random model for plant cell population growth. 1989.
- 59 D. van Dulst. Characterizations of Banach spaces not containing l¹. 1989.

- 60 H.E. de Swart. Vacillation and predictability properties of low-order atmospheric spectral models. 1989.
- 61 P. de Jong. Central limit theorems for generalized multilinear forms. 1989.
- 62 V.J. de Jong. A specification system for statistical software. 1989.
- 63 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part I. 1989.
- 64 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part II. 1989.
- 65 B.M.M. de Weger. Algorithms for diophantine equations. 1989.
- 66 A. Jung. Cartesian closed categories of domains. 1989.
- 67 J.W. Polderman. Adaptive control & identification: Conflict or conflux?. 1989.
- 68 H.J. Woerdeman. *Matrix and operator extensions*. 1989.
- 69 B.G. Hansen. Monotonicity properties of infinitely divisible distributions. 1989.
- 70 J.K. Lenstra, H.C. Tijms, A. Volgenant (eds.). Twenty-five years of operations research in the Netherlands: Papers dedicated to Gijs de Leve. 1990.
- 71 P.J.C. Spreij. Counting process systems. Identification and stochastic realization. 1990.
- 72 J.F. Kaashoek. Modeling one dimensional pattern formation by anti-diffusion. 1990.
- 73 A.M.H. Gerards. Graphs and polyhedra. Binary spaces and cutting planes. 1990.
- 74 B. Koren. Multigrid and defect correction for the steady Navier-Stokes equations. Application to aerodynamics. 1991.
- 75 M.W.P. Savelsbergh. Computer aided routing. 1992.
- 76 O.E. Flippo. Stability, duality and decomposition in general mathematical programming. 1991.
- 77 A.J. van Es. Aspects of nonparametric density estimation. 1991.
- 78 G.A.P. Kindervater. Exercises in parallel combinatorial computing. 1992.
- 79 J.J. Lodder. Towards a symmetrical theory of generalized functions. 1991.
- 80 S.A. Smulders. Control of freeway traffic flow. 1996.
- 81 P.H.M. America, J.J.M.M. Rutten. A parallel object-oriented language: design and semantic foundations. 1992.
- 82 F. Thuijsman. Optimality and equilibria in stochastic games. 1992.
- 83 R.J. Kooman. *Convergence properties of recurrence sequences.* 1992.
- 84 A.M. Cohen (ed.). Computational aspects of Lie group representations and related topics. Proceedings of the 1990 Computational Algebra Seminar at CWI, Amsterdam. 1991.
- 85 V. de Valk. One-dependent processes. 1994.
- 86 J.A. Baars, J.A.M. de Groot. On topological and
- linear equivalence of certain function spaces. 1992. 87 A.F. Monna. The way of mathematics and mathematicians. 1992.
- 88 E.D. de Goede. Numerical methods for the threedimensional shallow water equations. 1993.
- 89 M. Zwaan. Moment problems in Hilbert space with applications to magnetic resonance imaging. 1993.

- 90 C. Vuik. The solution of a one-dimensional Stefan problem. 1993.
- 91 E.R. Verheul. Multimedians in metric and normed spaces. 1993.
 92 J.L.M. Maubach. Iterative methods for non-linear
- partial differential equations. 1994. 93 A.W. Ambergen. Statistical uncertainties in poste-
- rior probabilities. 1993. 94 P.A. Zegeling. Moving-grid methods for time-
- dependent partial differential equations. 1993. 95 M.J.C. van Pul. Statistical analysis of software re-
- liability models. 1993. 96 J.K. Scholma. A Lie algebraic study of some inte-
- grable systems associated with root systems. 1993. 97 J.L. van den Berg. Sojourn times in feedback and
- processor sharing queues. 1993. 98 A.J. Koning. Stochastic integrals and goodness-of-
- fit tests. 1993. 99 B.P. Sommeijer. Parallelism in the numerical inte-
- gration of initial value problems. 1993. 100 J. Molenaar. Multigrid methods for semiconductor
- device simulation. 1993. 101 H.J.C. Huijberts. Dynamic feedback in nonlinear synthesis problems. 1994.
- 102 J.A.M. van der Weide. Stochastic processes and point processes of excursions. 1994.
- 103 P.W. Hemker, P. Wesseling (eds.). Contributions to multigrid. 1994.
- 104 I.J.B.F. Adan. A compensation approach for queueing problems. 1994.
- 105 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems - solution methods. Part 1. 1994.
- 106 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems - solution methods. Part 2. 1994.
- 107 R.A. Trompert. Local uniform grid refinement for time-dependent partial differential equations. 1995.
- 108 M.N.M. van Lieshout. Stochastic geometry models in image analysis and spatial statistics. 1995.
- 109 R.J. van Glabbeek. Comparative concurrency semantics and refinement of actions. 1996.
- 110 W. Vervaat, H. Holwerda (ed.). Probability and lattices. 1997.
- 111 I. Helsloot. Covariant formal group theory and some applications. 1995.
- 112 R.N. Bol. Loop checking in logic programming. 1995.
- 113 G.J.M. Koole. Stochastic scheduling and dynamic programming. 1995.
- 114 M.J. van der Laan. Efficient and inefficient estimation in semiparametric models. 1995.
- 115 S.C. Borst. Polling models. 1996.
- 116 G.D. Otten. Statistical test limits in quality control. 1996.
- 117 K.G. Langendoen. Graph reduction on sharedmemory multiprocessors. 1996.
- 118 W.C.A. Maas. Nonlinear \mathcal{H}_∞ control: the singular case. 1996.
- 119 A. Di Bucchianico. Probabilistic and analytical aspects of the umbral calculus. 1997.
- 120 M. van Loon. Numerical methods in smog prediction. 1997.
- 121 B.J. Wijers. Nonparametric estimation for a windowed line-segment process. 1997.

- 122 W.K. Klein Haneveld, O.J. Vrieze, L.C.M. Kallenberg (editors). Ten years LNMB – Ph.D. research and graduate courses of the Dutch Network of Operations Research. 1997.
- 123 R.W. van der Hofstad. One-dimensional random polymers. 1998.
- 124 W.J.H. Stortelder. Parameter estimation in nonlinear dynamical systems. 1998.
- 125 M.H. Wegkamp. Entropy methods in statistical estimation. 1998.
- 126 K. Aardal, J.K. Lenstra, F. Maffioli, D.B. Shmoys (eds.) Selected publications of Eugene L. Lawler. 1999.
- 127 E. Belitser. Minimax estimation in regression and random censorship models. 2000.

- 128 Y. Nishiyama. Entropy methods for martingales. 2000.
- 129 J.A. van Hamel. Algebraic cycles and topology of real algebraic varieties. 2000.
- 130 P.J. Oonincx. Mathematical signal analysis: wavelets, Wigner distribution and a seismic application. 2000.
- 131 M. Ruzhansky. Regularity theory of Fourier integral operators with complex phases and singularities of affine fibrations. 2001.
- 132 J.V. Stokman. Multivariable orthogonal polynomials and quantum Grassmannians. 2001.
- 133 N.R. Bruin. Chabauty methods and covering techniques applied to generalised Fermat equations. 2001.

MATHEMATICAL CENTRE TRACTS

1 T. van der Walt. Fixed and almost fixed points, 1963.

2 A.R. Bloemena. Sampling from a graph. 1964. 3 G. de Leve. Generalized Markovian decision processes, part 1: model and method. 1964.

4 G. de Leve. Generalized Markovian decision processes, part II: probabilistic background. 1964.

5 G. de Leve, H.C. Tijms, P.J. Weeda. Generalized Markovian decision processes, applications. 1970.

6 M.A. Maurice. Compact ordered spaces. 1964.

7 W.R. van Zwet. Convex transformations of random variables. 1964. 8 J.A. Zonneveld. Automatic numerical integration. 1964.

9 P.C. Baayen. Universal morphisms. 1964.

10 E.M. de Jager. Applications of distributions in mathematical physics. 1964.

11 A.B. Paalman-de Miranda. Topological semigroups. 1964. 12 J.A.Th.M. van Berckel, H. Brandt Corstius, R.J. Mokken, A. van Wijngaarden. Formal properties of newspaper Dutch. 1965. 1965

13 H.A. Lauwerier. Asymptotic expansions. 1966, out of print: replaced by MCT 54.

14 H.A. Lauwerier. Calculus of variations in mathematical physics. 1966.

15 R. Doornbos. Slippage tests. 1966.

L. Soonnoos, suppage tests. 1966.
 I.M. de Bakker, Formal definition of programming languages with an application to the definition of ALGOL 60. 1967.

17 R.P. van de Riet. Formula manipulation in ALGOL 60, part 1. 1968.

18 R.P. van de Riet. Formula manipulation in ALGOL 60, part 2. 1968.

19 J. van der Slot. Some properties related to compactness. 1968.

20 P.J. van der Houwen. Finite difference methods for solving partial differential equations. 1968.

21 E. Wattel. The compactness operator in set theory and topology. 1968.

22 T.J. Dekker. ALGOL 60 procedures in numerical algebra, part 1. 1968.

23 T.J. Dekker, W. Hoffmann. ALGOL 60 procedures in numerical algebra, part 2. 1968.

24 J.W. de Bakker. Recursive procedures. 1971.

25 E.R. Paërl. Representations of the Lorentz group and projec-tive geometry. 1969.

26 European Meeting 1968. Selected statistical papers, part I. 1968

27 European Meeting 1968. Selected statistical papers, part II. 1968.

28 J. Oosterhoff. Combination of one-sided statistical tests. 1969.

29 J. Verhoeff. Error detecting decimal codes. 1969.

30 H. Brandt Corstius. Exercises in computational linguistics. 1970

33 F.W. Steutel. Preservations of infinite divisibility under mixing and related topics. 1970.
34 I. Juhász, A. Verbeek, N.S. Kroonenberg. Cardinal functions in topology. 1971.

35 M.H. van Emden. An analysis of complexity. 1971. 36 J. Grasman. On the birth of boundary layers. 1971.

37 J.W. de Bakker, G.A. Blaauw, A.J.W. Duijvess, 1971. Dijkstra, P.J. van der Houwen, G.A.M. Kamsteeg-Kemper, F.E.J. Kruseman Aretz, W.L. van der Poel, J.P. Schaap-Kruseman, M.V. Wilkes, G. Zoutendijk. MC-25 Informatica Symposium. 1971.

38 W.A. Verloren van Themaat. Automatic analysis of Dutch compound words. 1972.

39 H. Bavinck. Jacobi series and approximation. 1972.

40 H.C. Tijms. Analysis of (s,S) inventory models. 1972.

41 A. Verbeek. Superextensions of topological spaces. 1972. 42 W. Vervaat. Success epochs in Bernoulli trials (with applica-tions in number theory). 1972.

43 F.H. Ruymgaart. Asymptotic theory of rank tests for independence. 1973.

44 H. Bart. Meromorphic operator valued functions. 1973.

45 A.A. Balkema. Monotone transformations and limit laws. 1973.

46 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 1: the language. 1973.

47 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 2: the compiler. 1973.

48 F.E.J. Kruseman Aretz, P.J.W. ten Hagen, H.L. Oudshoorn. An ALGOL 60 compiler in ALGOL 60, text of the MC-compiler for the EL-X8. 1973.

49 H. Kok. Connected orderable spaces. 1974.

50 A. van Wijngaarden, B.J. Mailloux, J.E.L. Peck, C.H.A. Koster, M. Sintzoff, C.H. Lindsey, L.G.L.T. Meertens, R.G. Fisker (eds.). *Revised report on the algorithmic language ALGOL 68*, 1976.

51 A. Hordijk. Dynamic programming and Markov potential theory. 1974.

52 P.C. Baayen (ed.). Topological structures. 1974

53 M.J. Faber. Metrizability in generalized ordered spaces. 1974.

54 H.A. Lauwerier. Asymptotic analysis, part 1. 1974.
55 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 1: theory of designs, finite geometry and coding theory. 1974. 56 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 2: graph theory, foundations, partitions and combinatorial geometry. 1974.

57 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 3: combinatorial group theory. 1974.

58 W. Albers. Asymptotic expansions and the deficiency concept in statistics. 1975.

59 J.L. Mijnheer. Sample path properties of stable processes. 1975.

60 F. Göbel. Queueing models involving buffers. 1975. 63 J.W. de Bakker (ed.). Foundations of computer science. 1975

64 W.J. de Schipper. Symmetric closed categories. 1975.

65 J. de Vries. Topological transformation groups, 1: a categor-ical approach. 1975.

66 H.G.J. Pijls. Logically convex algebras in spectral theory and eigenfunction expansions. 1976.

68 P.P.N. de Groen. Singularly perturbed differential operators of second order. 1976.

69 J.K. Lenstra. Sequencing by enumerative methods. 1977. 70 W.P. de Roever, Jr. Recursive program schemes: semantics and proof theory. 1976.

71 J.A.E.E. van Nunen. Contracting Markov decision processes. 1976.

72 J.K.M. Jansen. Simple periodic and non-periodic Lamé functions and their applications in the theory of conical waveguides. 1977.

73 D.M.R. Leivant. Absoluteness of intuitionistic logic. 1979.

74 H.J.J. te Riele. A theoretical and computational study of generalized aliquot sequences. 1976. 75 A.E. Brouwer. Treelike spaces and related connected topological spaces. 1977.

76 M. Rem. Associons and the closure statements. 1976.

77 W.C.M. Kallenberg. Asymptotic optimality of likelihood ratio tests in exponential families. 1978.

78 E. de Jonge, A.C.M. van Rooij. Introduction to Riesz spaces. 1977.

79 M.C.A. van Zuijlen. Empirical distributions and rank statistics. 1977.

80 P.W. Hemker. A numerical study of stiff two-point boundary problems. 1977.

81 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 1. 1976. 82 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 2. 1976.

83 L.S. van Benthem Jutting. Checking Landau's "Grundlagen" in the AUTOMATH system. 1979.

4 H.L. Busard. The translation of the elements of Euclid from the Arabic into Latin by Hermann of Carinthia (?), books vii-xii. 1977.

85 J. van Mill. Supercompactness and Wallmann spaces. 1977. 86 S.G. van der Meulen, M. Veldhorst. Torrix I, a program ming system for operations on vectors and matrices over arbi-trary fields and of variable size. 1978.

88 A. Schrijver. Matroids and linking systems. 1977.

89 J.W. de Roever. Complex Fourier transformation and ana-lytic functionals with unbounded carriers. 1978.

90 L.P.J. Groenewegen. Characterization of optimal strategies in dynamic games. 1981.

91 J.M. Geysel. Transcendence in fields of positive characteristic 1979

92 P.J. Weeda. Finite generalized Markov programming. 1979. 93 H.C. Tijms, J. Wessels (eds.). Markov decision theory. 1977.

94 A. Bijlsma. Simultaneous approximations in transcendental number theory. 1978.

95 K.M. van Hee. Bayesian control of Markov chains. 1978. 96 P.M.B. Vitányi. Lindenmayer systems: structure, languages, and growth functions. 1980.

97 A. Federgruen. Markovian control problems; functional equations and algorithms. 1984.

 98 R. Geel. Singular perturbations of hyperbolic type. 1978.
 99 J.K. Lenstra, A.H.G. Rinnooy Kan, P. van Emde Boas (eds.). Interfaces between computer science and operations surgests. 1079. research. 1978.

100 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part ings bice 1. 1979.

101 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part ings bic 2. 1979.

102 D. van Dulst. Reflexive and superreflexive Banach spaces. 1978

103 K. van Harn. Classifying infinitely divisible distributions by functional equations. 1978. 104 J.M. van Wouwe. GO-spaces and generalizations of metri-zability. 1979.

105 R. Helmers. Edgeworth expansions for linear combinations of order statistics. 1982. 106 A. Schrijver (ed.). Packing and covering in combinatorics.

1979

107 C. den Heijer. The numerical solution of nonlinear opera-tor equations by imbedding methods. 1979. 108 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 1. 1979.

109 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 2. 1979.

110 J.C. van Vliet. ALGOL 68 transput, part I: historical review and discussion of the implementation model. 1979.

111 J.C. van Vliet. ALGOL 68 transput, part II: an implemen-tation model. 1979.

112 H.C.P. Berbee. Random walks with stationary increments newai theory. 1979.

113 T.A.B. Snijders. Asymptotic optimality theory for testing problems with restricted alternatives. 1979.

114 A.J.E.M. Janssen. Application of the Wigner distribution to harmonic analysis of generalized stochastic processes. 1979. 115 P.C. Baayen, J. van Mill (eds.). Topological structures II,

part 1. 1979. 116 P.C. Baayen, J. van Mill (eds.). Topological structures II, part 2. 1979.

117 P.J.M. Kallenberg. Branching processes with continuous state space. 1979.

118 P. Groeneboom. Large deviations and asymptotic efficiencies. 1980.

119 F.J. Peters. Sparse matrices and substructures, with a novel implementation of finite element algorithms. 1980.

120 W.P.M. de Ruyter. On the asymptotic analysis of large-scale ocean circulation. 1980.

121 W.H. Haemers. Eigenvalue techniques in design and graph 1980

122 J.C.P. Bus. Numerical solution of systems of nonlinear equations. 1980. 123 I. Yuhász. Cardinal functions in topology - ten years later.

124 R.D. Gill. Censoring and stochastic integrals. 1980. 125 R. Eising. 2-D systems, an algebraic approach. 1980.

126 G. van der Hoek. Reduction methods in nonlinear pro-gramming. 1980.

127 J.W. Klop. Combinatory reduction systems. 1980. 128 A.J.J. Talman. Variable dimension fixed point algorithms and triangulations. 1980.

129 G. van der Laan. Simplicial fixed point algorithms. 1980. 130 P.J.W. ten Hagen, T. Hagen, P. Klint, H. Noot, H.J. Sint, A.H. Veen. ILP: intermediate language for pictures. 1980

131 R.J.R. Back. Correctness preserving program refinements: proof theory and applications. 1980.

132 H.M. Mulder. The interval function of a graph. 1980.

133 C.A.J. Klaassen. Statistical performance of location esti-133 C.A.J. Niadoscu, Guinandi, J. mators. 1981. 134 J.C. van Vliet, H. Wupper (eds.). Proceedings interna-tional conference on ALGOL 68, 1981.

tional conference on ALGOL 00. 1501. 135 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.). Formal methods in the study of language, part I. 1981. 136 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.). Formal methods in the study of language, part II. 1981.

137 J. Telgen. Redundancy and linear programs. 1981.

138 H.A. Lauwerier. Mathematical models of epidemics. 1981.

139 J. van der Wal. Stochastic dynamic programming, succes-sive approximations and nearly optimal strategies for Markov decision processes and Markov games. 1981.

140 J.H. van Geldrop. A mathematical theory of pure exchange economies without the no-critical-point hypothesis. excna 1981.

141 G.E. Welters. Abel-Jacobi isogenies for certain types of Fano threefolds. 1981.

142 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 1. 1981.

143 J.M. Schumacher. Dynamic feedback in finite- and infinite-dimensional linear systems. 1981.

144 P. Eijgenraam. The solution of initial value problems using interval arithmetic; formulation and analysis of an algorithm. *interv* 1981.

145 A.J. Brentjes. Multi-dimensional continued fraction algo-rithms. 1981.

146 C.V.M. van der Mee. Semigroup and factorization methods in transport theory. 1981.

147 H.H. Tigelaar. Identification and informative sample size. 1982.

148 L.C.M. Kallenberg. Linear programming and finite Mar-kovian control problems. 1983.

149 C.B. Huijsmans, M.A. Kaashoek, W.A.J. Luxemburg, W.K. Vietsch (eds.). From A to Z, proceedings of a sympo-sium in honour of A.C. Zaanen. 1982.

150 M. Veldhorst. An analysis of sparse matrix storage schemes. 1982.

151 R.J.M.M. Does. Higher order asymptotics for simple linear rank statistics. 1982.

152 G.F. van der Hoeven. Projections of lawless sequencies. 1982.

153 J.P.C. Blanc. Application of the theory of boundary value problems in the analysis of a queueing model with paired ser-vices. 1982.

Vices. 1962.
154 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part I. 1982.
155 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part II. 1982.

156 P.M.G. Apers. Query processing and data allocation in distributed database systems. 1983.

157 H.A.W.M. Kneppers. The covariant classification of two-dimensional smooth commutative formal groups over an alge-braically closed field of positive characteristic. 1983.

158 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 1. 1983.

159 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 2. 1983. 160 A. Rezus. Abstract AUTOMATH. 1983.

161 G.F. Helminck. Eisenstein series on the metaplectic group, an algebraic approach. 1983.

162 J.J. Dik. Tests for preference. 1983. 163 H. Schippers. Multiple grid methods for equations of the second kind with applications in fluid mechanics. 1983.

164 F.A. van der Duyn Schouten. Markov decision processes with continuous time parameter. 1983.

165 P.C.T. van der Hoeven. On point processes. 1983.

166 H.B.M. Jonkers. Abstraction, specification and implemen-tation techniques, with an application to garbage collection. 1983.

167 W.H.M. Zijm. Nonnegative matrices in dynamic program-ming. 1983.

168 J.H. Evertse. Upper bounds for the numbers of solutions of diophantine equations. 1983.

169 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 2. 1983.